# T-cell receptor repertoire sequencing in health and disease

James Malcolm Heather Division of Infection and Immunity University College London

A thesis submitted for the degree of Doctor of Philosophy

2015

I, James Heather, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Signature:

Date:

"In a varying environment, changes in the character of the population take place. In every instance we are concerned to understand (a) whence the change in character of the individual components arises and (b) the nature of the process by which the constitution of the population is modified to conform to the requirements of the changing environment."

> Frank Macfarlane Burnet From 'The Clonal Selection Theory of Acquired Immunity'. 1958

"... [A] knowledge of sequences could contribute much to our understanding of living matter."

Frederick Sanger From the autobiography given to the Nobel Foundation upon receiving his second Prize. 1980

# Abstract

The adaptive immune systems of jawed vertebrates are based upon lymphocytes bearing a huge variety of antigen receptors. Produced by somatic DNA recombination, these receptors are clonally expressed on T- and B-lymphocytes, where they are used to help detect and control infections and help maintain regular bodily function. Full understanding of various aspects of the immune system relies upon accurate measurement of the individual receptors that make up these repertoires. In order to obtain such data, protocols were developed to permit unbiased amplification, high-throughput deepsequencing, and error-correcting bioinformatic analysis of T-cell receptor sequences.

These techniques have been applied to peripheral blood samples to further characterise aspects of the TCR repertoire of healthy individuals, such as V(D)J TCR gene usage and pairing distributions. A large number of sequences are also found to be shared across multiple individuals, including sequences matching receptors belonging to known and proposed T-cell subsets making use of invariant rearrangements. The resolution provided also permitted detection of low-frequency recombination events that use unexpected gene segments, or contained alternative splicing events.

Deep-sequencing was further used to study the effect of HIV infection, and subsequent antiretroviral therapy, upon the TCR repertoire. HIV-patient repertoires are typified by marked clonal inequality and perturbed population structures, relative to healthy controls. The data presented support a model in which HIV infection drives expansion of an subset of  $CD8^+$  clones, which – in combination with the virally-mediated loss of  $CD4^+$  cells – is responsible for driving repertoires towards an idiosyncratic population with low diversity. Moreover these altered repertoire features do not significantly recover after three months of therapy.

Deep-sequencing therefore presents opportunities to investigate the properties of TCR repertoires both in health and disease, which could be useful when analysing a wide variety of immune phenomena.

# Acknowledgements

Many people have helped me throughout the course of my PhD in one form or another, for which I am truly thankful.

First and foremost among those is Professor Benny Chain, who in his supervision of me through this project has displayed all of the qualities that make up not just great scientists but great supervisors, qualities that I hope to emulate in my own career. A clerical error meant that I was not assigned a secondary supervisor. Benny meant that I did not need one.

Similar thanks also go to Dr Maddy Noursadeghi, my *de facto* secondary supervisor. The group environment fostered between the collaboration of Chain and Noursadeghi has proven immensely fruitful to my academic development, and I am grateful to have the opportunity to work in it for a little longer.

While I am indebted to many current and past lab members for help, there are two that merit particular mention. Dr Theres Oakes and Katharine Best have made immeasurable contributions throughout my PhD, as sounding boards and collaborators for the development of the wet lab and bioinformatic protocols respectively.

Very large thanks also go to Dr Dan Depledge and Tony Brooks, both of whom very patiently answered my silly questions and helped me find my feet in the sequencing work. Thanks also go to Professor Paul Kellam and his group for sequencing several of the early experiments in this thesis.

My innumerable discussions and frantic schematic-drawing-sessions with Samir Aoudjane must also be noted, for helping me actually realise that I might know what I was talking about. I would also like to thank Professor Av Mitchison for his encouraging words before and after graciously looking over this thesis.

Finally, I must thank my parents, my partner Christy, and all the rest of my friends and family for supporting me throughout. You'll all be very pleased to know that it's done now (well, almost), so I can finally come out and play. Just after the next paper is written.

# Contents

A	bstra	nct		4
A	ckno	wledge	ements	5
Ta	able	of cont	ents	9
$\mathbf{Li}$	ist of	figure	s	10
Li	ist of	tables	5	13
G	lossa	ry		14
1	Intr	roduct	ion	16
	1.1	Adapt	vive immunity	16
	1.2	V(D).	$\label{eq:combination} I \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ $	17
	1.3	TCR	structure, function and selection	20
	1.4	TCR	discovery	22
	1.5	Histor	ical methods of investigating TCR repertoires	24
	1.6	TCR	repertoire analysis by deep-sequencing	25
	1.7	Aims	of this thesis	27
<b>2</b>	Ma	terials	and methods	28
	2.1	The in	nternational ImMunoGeneTics (IMGT)	
		inform	nation system and nomenclature	28
	2.2	Gener	al considerations and RNA extraction	29
	2.3	Ampli	fication and deep-sequencing protocols	29
		2.3.1	Reverse transcription	29
		2.3.2	Simple V amplification	30
			2.3.2.1 Magnetic assisted cell sorting and flow cytometry	30
		2.3.3	Poly-T RACE amplification	31
			2.3.3.1 Poly-T RACE MiSeq library preparation	32
			2.3.3.2 Cloning	33
		2.3.4	Randomly barcoded V (BV) amplification	34
		2.3.5	Barcoded ligation RACE (BLR) amplification	35
			2.3.5.1 Flow assisted cell sorting	36
			2.3.5.2 HIV patient criteria	36

# CONTENTS

		2.3.6	MiSeq operation	6
		2.3.7	Primers and indexes	7
			2.3.7.1 Primer sequences $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 3'$	7
			2.3.7.2 Indexing sequences $\ldots \ldots \ldots \ldots \ldots \ldots \ldots 38$	8
	2.4	Bioinf	ormatic analysis	8
		2.4.1	General considerations	8
		2.4.2	Demultiplexing BLR MiSeq data 39	9
		2.4.3	TCR assignment and error-correction 40	0
		2.4.4	TCR productivity and CDR3 amino acid determination 4	1
		2.4.5	Sorting of poly-T homopolymer containing reads	2
		2.4.6	Splicing analysis	3
			2.4.6.1 Mapping of reads to TCR germline loci	3
			2.4.6.2 Quantification of intra-V gene splicing events 44	3
		2.4.7	Calculation of diversity and sharing values	4
			2.4.7.1 Gini index	4
			2.4.7.2 Shannon entropy	5
			2.4.7.3 Jaccard index	6
			2.4.7.4 Jensen-Shannon divergence	6
	2.5	Comm	itment to open science 40	6
		2.5.1	Data deposition	7
		2.5.2	Source code	7
2	Fvn	loring	healthy T coll recontor reporteires	n
J	2 1	The gl	abal TCP reportains at standy state	9 0
	3.1	2 1 1 e gi	Clopel distribution of the healthy TCP reportaine	9 0
		0.1.1 9 1 0	V and L gaps usage	9 9
		0.1.2	TPPD detection and usage	2 7
		0.1.0 9.1.4	TCP sharing and publicness	1
		0.1.4 9.1.5	Invariant TCP detection	1
		3.1.0 9.1.6	Section discussion	4 0
	2.9	5.1.0	section discussion	ວ ດ
	0.2	Subset	-specific FOR repetitione infangs	2
		3.2.1	CD4 and $CD8$ TCR repertores	2
		3.2.2	Clonality distribution differences using targeted V amplification	-
			$Protocols \dots Protocols 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1$	о г
			3.2.2.1 UD4' UD45KA' versus UD45KO' clonal distributions 78	Э
				-
		0.0.0	3.2.2.2 Cord blood versus adult whole blood clonal distributions 7'	7

# CONTENTS

	3.3	Intra-gene splicing of TCR transcripts		
		3.3.1	Section discussion	87
	3.4	Chapt	er discussion	89
4	Per	turbat	ions to the T-cell receptor repertoire during HIV infection	93
	4.1	Introd	$uction \ldots \ldots$	93
		4.1.1	HIV and AIDS	93
		4.1.2	HIV TCR repertoire analysis	95
	4.2	HIV in	nfection associates with a dramatic loss of TCR diversity $\ldots$ .	96
	4.3	TCR 1	repertoire population structures are perturbed in HIV patients	99
	4.4	Effect	of antire troviral therapy on TCR repertoires of $\mathrm{HIV}^+$ individuals -	101
	4.5	Tracki	ing sequence-specific CDR3s on ART	103
	4.6	Discus	ssion	106
<b>5</b>	Pro	tocol a	development 1	110
	5.1	Introd	$uction \dots \dots$	110
		5.1.1	DNA sequencing technologies	110
			5.1.1.1 First-generation DNA sequencing	110
			5.1.1.2 Second-generation DNA sequencing	112
			5.1.1.3 Third-generation DNA sequencing	116
		5.1.2	Considerations for repertoire sequencing	117
		5.1.3	Published TCR methodologies	120
			5.1.3.1 Our protocol; initial decisions	121
		5.1.4	A note on FASTQ quality scores	121
	5.2	Seque	ncing development	122
		5.2.1	Comparison results from published datasets	122
		5.2.2	Simple V amplification	123
			5.2.2.1 Protocol	123
			5.2.2.2 Sequencing results	124
			5.2.2.3 Verdict	128
		5.2.3	Poly-T RACE	128
			5.2.3.1 Protocol	128
			5.2.3.1.1 Poly-T RACE cloning	129
			5.2.3.2 Sequencing results	130
			5.2.3.2.1 454	130
			5.2.3.2.2 $MiSeq - adapter-ligation \dots \dots \dots$	132
			5.2.3.2.3 MiSeq – PCR library preparation	133

			5.2.3.2.4 Effect of native TCR homopolymers	137
			5.2.3.3 Verdict	137
		5.2.4	Ligation RACE	140
			5.2.4.1 Protocol	140
			5.2.4.2 Sequencing results	141
			5.2.4.3 Verdict	141
		5.2.5	Randomly-barcoded V amplification (BV)	143
			5.2.5.1 Protocol	143
			5.2.5.2 Sequencing results	145
			5.2.5.3 Verdict	145
		5.2.6	Barcoded Ligation RACE (BLR)	147
			5.2.6.1 Protocol	147
			5.2.6.2 Sequencing results	147
			5.2.6.2.1 Assess bias	150
			5.2.6.3 Verdict	153
	5.3	Bioinfo	prmatic development	154
		5.3.1	$\mathrm{TCR}$ repertoire analysis with <code>Decombinator</code>	154
			5.3.1.1 $$ Generation of new Decombinator tag sequences $$	156
			5.3.1.2 $$ Limitations of standard Decombinator analysis $$	159
		5.3.2	'Collapsing' TCRs: error-correction and improved frequency es-	
			timation	160
			5.3.2.1 vDCR/Collapsing verdict $\ldots \ldots \ldots \ldots \ldots \ldots$	167
	5.4	Protoc	ol development discussion $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	169
6	Con	alusior		179
U	6 1	Futuro	work	173
	0.1	ruture	WOIK	110
Po	ostsci	$\operatorname{ript}$		175
Δ	Anr	endix	$\mathbf{A}$ - exploring the TCR repertoire of healthy individuals	176
11	1106		i exploring the rent repertone of neutriny individuals	110
в	App	oendix	B - the effect of HIV infection upon the TCR repertoire	187
$\mathbf{C}$	App	oendix	C - developing a TCR sequencing pipeline	197
	C.1	Previo	usly published datasets	197
	C.2	Protoc	ol development supplementary figures	198
Re	References 211			

# List of Figures

1.1	$V(D)J$ recombination in the human TRB locus $\ . \ . \ . \ . \ . \ .$	19
1.2	The human alpha-delta germline locus	20
1.3	TCR binding to peptide-MHC (pMHC)	21
2.1	Gini index calculation from the Lorenz curve	45
3.1	Clonal composition of the healthy human $\alpha\beta$ TCR repertoire	51
3.2	V and J gene usage in healthy human $\alpha\beta$ TCR rearrangements	54
3.3	V and J gene pairing in healthy human $\alpha$ TCR rearrangements	55
3.4	V and J gene pairing in healthy human $\beta$ TCR rearrangements	56
3.5	TRBD detection and usage in healthy human TCR $\beta$ rearrangements	59
3.6	TCR sharing and publicness of productive rearrangements	61
3.7	Non-productive TCR sharing and publicness.	64
3.8	Intra-donor TCR sharing.	65
3.9	Detection of known and putative invariant $\alpha$ chain TCRs	66
3.10	Clonal composition of the sorted CD4 and CD8 TCR repertoires	74
3.11	Clonal composition of CD4 <sup>+</sup> CD45RA <sup>+</sup> or CD45RO <sup>+</sup> TRAV8-4 <sup>+</sup> and	
	TRBV12-3 <sup>+</sup> repertoires	76
3.12	Clonal distributions and diversity indices of adult and cord blood V-	
	specific repertoires.	78
3.13	Adult and cord blood $\alpha$ chain TCR length distributions	80
3.14	TCR publicness and sharing in the BV data.	81
3.15	Examples of intra-V gene splicing.	84
3.16	Extent of intra-V gene splicing.	86
4.1	TCR-repertoires of HIV-infected patients are typified by lower clonal	
	diversity and higher inequality than healthy donors	97
4.2	Simulating TCR repertoires with different CD4:CD8 ratios	98
4.3	HIV <sup>+</sup> TCR repertories are less shared, public, and translationally con-	
	vergent than healthy controls, and are highly divergent in their V and J	
	gene use	100
4.4	ART partially reduces repertoire inequality, but general diversity and	
	idiosyncratic gene usage do not recover.	102

4.5	Dynamics of viral-associated and invariant MAIT cell CDR3s	104
5.1	How the choice of nucleic acid impacts on choice of TCR amplification	
	protocol	119
5.2	Example V-J usage plots for the previously published datasets	122
5.3	Schematic detailing the amplification and library preparation steps in-	
	volved in our initial simple V amplification experiment.	123
5.4	Results of simple V amplification and HiSeq sequencing experiment	126
5.5	Poly-T RACE strategy schematic.	128
5.6	Agarose gel scans showing amplicons produced using the preliminary	
	poly-T RACE protocol.	129
5.7	Summary data for the poly-T RACE amplification products as sequenced	
	on the 454 GS FLX.	131
5.8	Results for the first poly-T RACE amplicons being sequenced on the	
	MiSeq, having undergone NEBNext adapter-ligation library preparation.	134
5.9	Clustering on the MiSeq flowcells	135
5.10	The extent and effect of 'native' homopolymers (HP) in TCR repertoire	
	sequence data.	138
5.11	Ligation RACE strategies.	140
5.12	Performance of the basic RNA and DNA ligation RACE protocols	142
5.13	Schematic of the barcoded V (BV) amplification protocol. $\ldots$ .	144
5.14	Overview of barcoded simple V amplication protocol results	146
5.15	Schematic of the barcoded ligation RACE (BLR) protocol.	148
5.16	Quality score boxplots and VJ gene usage heatmaps for a representative	
	whole blood healthy donor sample (HV07), produced using the BLR	
	strategy.	149
5.17	Assessment of bias in the BLR protocol	152
5.18	Schematic detailing an Aho-Corasick string matching automaton	155
5.19	Comparison of original and extended Decombinator tags	159
5.20	Schematic detailing the need for error-correction	160
5.21	The global effect of collapsing TCR repertoires	163
5.22	Correlation between raw and collapsed TCR frequencies	164
5.23	Use of unique molecular identifiers allows measurement of duplication	
	rates of different TCR rearrangements	166
A.1	Detection of TRDV usage in $\alpha$ chain TCR data	176
A.2	Purity of CD4 and CD8 sorted T-cells.	177

## LIST OF FIGURES

A.3	Alpha chain VJ pairing frequencies for HV01	178
A.4	Alpha chain VJ pairing frequencies for HVD4	179
A.5	Beta chain VJ pairing frequencies for HV01	180
A.6	Beta chain VJ pairing frequencies for HVD4	181
A.7	Adult and cord blood $\beta$ chain TCR length distributions	182
B.1	Clinical score-diversity correaltions	187
B.2	HIV infection is associated with a loss of publicness and translational	
	convergence in $\beta$ chain CDR3s	189
B.3	Healthy donor and HIV-patient gene usage divergence	190
B.4	Raw TCR proportional gene usage in healthy and HIV-infected samples.	191
B.5	Clinical features of the HIV patient samples.	192
B.6	Relationship between years before treatment and CD4:CD8 ratio	193
B.7	Change in HIV patients' gene usage divergence in the first three months	
	of ART	194
B.8	HIV patients' change in diversity indexes on ART	195
B.9	HIV-associated CDR3s decrease in frequency on ART, while CMV- and	
	EBV-associated CDR3s increase.	196
C.1	Example amplicons cloned from the DNA amplified from two healthy	
	donors' PBMC using our poly-T RACE PCR	200
C.2	Sanger sequenced clones produced using the stepwise PCR protocol	202
C.3	Flowchart overview of how the error- and frequency-correcting code of	
	CollapseTCR works.	204
C.4	$TRAJ \ and \ TRBJ \ usage \ difference \ between \ original \ and \ extended \ {\tt Decombi}$	nator
	tag sets	205

# List of Tables

2.1	Primer sequences	37		
2.2	Index sequences	38		
A.1	Germline sequences of the human TRBD genes	176		
A.2	Sources of the TCRs of T-cell subpopulations bearing invariant			
	lpha chain and semi-invariant $eta$ chain rearrangements	183		
A.3	Table of putative invariant $\alpha$ chain TCRs from the BLR data	184		
A.4	Table of putative invariant $\alpha$ chain TCRs from the BV data	185		
A.5	List of intra-V gene splicing events	186		
B.1	Sources of published viral- or subset-specific CDR3s	188		
C.1	Overview read statistics for three sets of publicly available healthy hu-			
	man datasets that were available from papers published prior to comple-			
	tion of our own protocols. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	200		
C.2	Merging HiSeq reads	202		
C.3	Read and Decombinator output data for the Simple V amplification			
	protocol, as sequenced on the Illumina HiSeq	206		
C.4	Overall read statistics for the Simple V amplification protocol	207		
C.5	Overall read statistics for the poly-T RACE protocol, sequenced on the			
	Roche 454 GS FLX	207		
C.6	Overall read statistics for the poly-T RACE protocol, with NEBNext			
	adapter-ligation library preparation.	208		
C.7	Summary statistics for the ligation RACE protocols, sequenced on the			
	Illumina MiSeq	208		
C.8	Summary statistics for the randomly barcoded simple V amplification			
	protocol, sequenced on the Illumina MiSeq	209		
C.9	Summary statistics for the randomly barcoded ligation RACE (BLR)			
	amplification protocol, sequenced on the Illumina MiSeq	209		
C.10	C.10 Differential amplification observed between V genes during multiplex PCR.210 $$			

# Glossary

6N	A random hexamer (i.e. six random bases) in an oligonucleotide sequence			
$\mathbf{AC}$	Aho-Corasick			
Amplicon	The section of DNA amplified during an amplification reaction			
Antigen A substance to which the adapt immune system can respond				
APMI	As per manufacturer's instructions			
ART	Antiretroviral therapy			
$\mathbf{AS}$	Alternative splicing			
<b>Barcode</b> A random stretch of nucleotides in corporated before amplification to a low 'collapsing' of data back to orig nal cDNA molecules				
BCR	B-cell receptor			
BLR	Barcoded ligation RACE (amplifica- tion protocol)			
bp	Base pair			
BSA Bovine serum albumin				
<b>BV</b> Barcoded V (amplification protoc				
$\mathbf{C}\mathbf{D}$	Cluster of differentiation			
cDNA	Complementary deoxyribonucleic acid			
$\mathbf{CDR}$	Complementarity determining region			
DCR	Shorthand for the five-part TCR clas- sifier produced by Decombinator			
Decombina	<b>ator</b> The software used to perform basic TCR analysis in this thesis, in which each TCR chain sequence is classified using a five-field identifier from which the entire nucleotide se- quence can be recapitulated			
DN	Double negative			
DNA	Deoxyribonucleic acid			
dNTP	Deoxyribose nucleoside triphosphate			

DOI	<b>DOI</b> Digital Object Identifier			
DP	Double positive			
Epitope	The part of an antigen to which a TCR , BCR or antigen binds			
FACS	Flow assisted cell sorting			
<b>Functional</b> IMGT definition: A germline V gene predicted to be able to tribute to productive rearrange				
gDNA	Genomic DNA			
GEM T-o	cell Germline-Encoded Mycolyl- reactive T-cells			
HF	High fidelity buffer for Phusion polymerase			
HLA	Human leukocyte antigen			
HSCT Haematopoietic stem cell transplatation				
HTS	High-throughput sequencing			
Ig	Immunoglobulin			
IMGT	<b>IGT</b> ImMunoGeneTics; the organisation responsible for collating and annotat ing TCR and Ig information			
Index	One of a list of pre-determined nu- cleotide sequences that is incorpo- rated to all amplicons of a given sam- ple before sequencing, to allow multi- plexing of many samples on one run			
iNKT cel	l Invariant natural killer T-cell			
JSD	Jensen-Shannon divergence			
kb	Kilobase (1000 nucleotide base pairs)			
L-PART1 Library	/2 The two sections of the leader sequence, which lie immediately up- stream of each V gene and are spliced together in mature TCR mRNA Set of DNA molecules prepared to be			
	sequenced			
m7G	7-methylguanosine			
MACS	Magnetic assisted cell sorting			
MAIT cell Mucosal associated invariant T-cell				
Mbp	mega base pair (1,000,000 base pairs)			
MHC	Major histocompatibility antigen			
mRNA	Messenger ribonucleic acid			

Weight per volume

gene/segment A section of DNA that can be assembled through V(D)J recombination to produce a TCR

terminal deoxynucleotidyl transferase

T-cell receptor alpha chain J gene

T-cell receptor alpha chain V gene

T-cell receptor beta chain J gene

T-cell receptor beta chain V gene

Verbose Decombinator, the specific version of Decombinator developed during this thesis to permit error-correcting TCR analysis

Ν	N represents an ambiguous, unknown	$\mathbf{Rep}\text{-}\mathbf{Seq}$	Repertoire sequencing
	or random nucleotide	RNA	Ribonucleic acid
NEB	New England Biolabs	SBS	Sequence-by-synthesis
NGS	Next-generation sequencing	SMS	Single molecule sequencing
NKT cell	Natural killer T-cell	SP	Single positive
NP (Non-	productive) IMGT definition: A re-	SD1	Illumine's sequencing primer 1
	arrangement predicted to not be ca- pable of forming valid receptors		
ORF	IMGT definition: A germline V(D).	SP2	Illumina's sequencing primer 2
0101	gene predicted to be unlikely to be	$\mathbf{SRA}$	Sequence Read Archive
	able to contribute to productive re-	TCR	T-cell receptor
	arrangements	TCR gen	e/segment A section of DNA
P (Pseude	<b>ogene</b> ) IMGT definition: A germline $V(D)J$ gene predicted to not be able		can be assembled through V(I combination to produce a TCI
	to contribute to productive rearrange- ments	$\mathbf{Tdt}$	terminal deoxynucleotidyl tran
P5	One of two sequences required for Il-	TEMRA	T effector memory CD45RA+ $$
	lumina bridge amplification	TRA	T-cell receptor alpha
P7	One of two sequences required for Il- lumina bridge amplification	TRAJ	T-cell receptor alpha chain J g
PBMC	Peripheral blood mononuclear cells	TRAV	T-cell receptor alpha chain V
PCR	Polymerase chain reaction	TRB	T-cell receptor beta
PE	Paired end	TRBJ	T-cell receptor beta chain J ge
PEG	Polyethylene glycol	TRBV	T-cell receptor beta chain V g
nMHC	Pentide major histocompatability	TRD	T-cell receptor delta
piiiie	complex	TRG	T-cell receptor gamma
Productive IMGT definition: A rearrangement		UCL	University College London
	predicted to be capable of forming valid receptors	vDCR	Verbose Decombinator, the cific version of Decombinat
RACE	Rapid amplification of cDNA ends		veloped during this thesis to
Regex	Regular expression; a sequence of characters used to find strings that		error-correcting TCR analysis

match certain criteria

w/v

# Introduction

### 1.1 Adaptive immunity

Various mechanisms have evolved across the domains of life to protect organisms from the negative consequences of being infected or parasitised by other organisms. In vertebrates these defences are typically grouped into one of two functional distinctions. Germline-encoded mechanisms that provide generic barriers and rapid responses to conserved elements of infectious agents constitute innate immunity. Working in conjunction with these innate mechanisms is the adaptive immune system, that *de novo* generates novel protein-coding genes in certain lymphocyte cell types to provide defence from a potentially near-infinite array of different biological and chemical entities. While typically less rapid than innate mechanisms, adaptive responses develop immunological memory, leading to faster responses during subsequent exposures. Despite these differences, both innate and adaptive immune mechanisms involve the dynamic regulation of different tissues, cells and effector molecules.

One of the most important immunological questions is how the body distinguishes what is foreign – and potentially harmful – from its own component parts, many of which will be composed of the same basic chemical structures. This problem was famously addressed by Frank Macfarlane Burnet in 1949 with his theory of immunological tolerance (1). This theory states that during embryogenesis the developing immune system should essentially only be exposed to the 'self', i.e. the sum total of identifiable molecular entities expressed throughout the body. Developing animals become 'tolerant' to these self-features, and can therefore reserve immunological reactions to anything else that is 'non-self', thus targeting foreign material while limiting damage to themselves. While modern theories have moved on – not least due to the knowledge that the adaptive immune system does indeed mount responses towards self molecules – this concept of 'self' and 'non-self' remains central to our understanding of immune processes.

The mode of adaptive immunity that humans enjoy is shared by all gnathostomes (jawed-vertebrates), having evolved approximately 500 million years ago in our shared ancestors (2). In this system a series of DNA recombination events is used to generate

populations of cells bearing a wide-range of different receptors, which can detect a huge assortment of different molecules in order to trigger immune responses. These cells – B and T lymphocytes, or B-cells and T-cells – are derived from a common haematopoietic progenitor (3, 4), and express a variable recombined receptor based on their cell type. B-cells express B-cell receptors (BCRs) and secrete soluble variations as antibodies (or immunoglobulins), while T-cells express T-cell receptors (TCRs), which are the focus of this thesis.

## 1.2 V(D)J recombination

TCRs exist as surface-expressed glycoproteins composed of two polypeptide chains. In humans, most T-cells express a heterodimer consisting of an  $\alpha$  and a  $\beta$  chain (although a second lineage accounting for 5% of total T-cells makes use of an alternative TCR, composed of a  $\gamma$  chain paired with a  $\delta$  (5)). Unlike the majority of genes, TCRs (and BCRs) are not functionally encoded in the germline. Instead, the loci encoding each chain contain a number of non-contiguous sections of DNA that can be recombined together to form complete coding sequences. This somatic DNA recombination was first observed in B-cells (6), and was found to be due to a number of related gene segments (7, 8) bordered by specific recombination sequences (9, 10) being brought together.

These recombinable genes fall into one of three categories, or types: 'variable' (V), 'diversity' (D) or 'joining' (J). These can be physically joined together through the action of a recombinase complex which is recruited to conserved recognition signal sequences (RSS) at the gene boundaries (11). This complex is the product of two recombinase activating genes (RAG), RAG-1 and RAG-2, which work co-operatively to mediate recombination between RSS (12, 13). Each RSS consists of a conserved nonamer and heptamer nucleotide sequence, separated by either 12 or 23 nucleotides; the correct order of assembly is promoted by recombination only occurring efficiently between pairs of genes that are bordered by RSS with different length spacers (the '12/23 rule') (14). The complement of genes types are not equally distributed between chains, as only one per heterodimer contains a D gene in its rearrangement: in the TCR it is the  $\beta$  chain<sup>1</sup>. The distribution of RSS therefore directs recombination between D $\beta$  and J $\beta$ , and V $\beta$  with D $\beta$  (see Figure 1.1). There is only one type of recombination event in the  $\alpha$  chain, that between a V $\alpha$  and a J $\alpha$ . However recombination at this locus is complicated as the panel of V genes is shared between both  $\alpha$  and  $\delta$  rearrangements;

 $<sup>^1\</sup>mathrm{D}$  regions are also components of TCR  $\delta$  and BCR heavy chain rearrangements.



the  $\delta$  locus resides within the  $\alpha$  (see Figure 1.2) (15, 16). Following transcription, rearranged V(D)J sections are spliced onto a constant region, which is required for surface expression and intracellular transmission of TCR binding signals.

The consequence of using V(D)J recombination to produce variable antigen receptors is the breadth of molecules that it can produce. Recombinatorial diversity is generated by merit of there being multiple V, D and J genes, and from the final protein being a heterodimer of two recombined chains: without any other mechanism in place this system could produce over 2.8 million different TCRs<sup>2</sup>. However the majority of the potential diversity arises due to mechanisms that result in the non-templated addition and deletion of nucleotides at the recombination junction. The RAG enzymes catalyse formation of a DNA hairpin intermediate during recombination; action of the DNA repair enzymes recruited to this site can lead to exonucleolytic 'nibbling' away of nucleotides, and formation of 'P-nucleotides' (where non-symmetrical hairpin resolution causes insertion of a short palindromic repeat) at the recombining edges (22, 23).

However the greatest source of diversity is the non-templated addition of nucleotides ('N-nucleotides') (24) via the action of terminal deoxynucleotidyl transferase (Tdt) (25, 26, 27). Hypothetically, allowing up to ten non-templated additions and deletions at each recombination junction raises the potential diversity to more than  $10^{16}$  possible different TCR heterodimers. While these diversities will not even be approached in an individual – not least because this theoretical estimate is orders of magnitude greater than the total number of eukaryotic cells in a body – they serve to illustrate how a few megabases of DNA are able to produce a vast array of antigen receptors that can be used to detect and protect from a far greater number of immune challenges than a static genome could.

Figure 1.1 (preceding page): V(D)J recombination in the human TRB locus. A: Overview of the germline layout of the locus, with inset zoomed in on the DJC section. B: Whole locus view of the rearrangement found in the  $\beta$  chain of the Jurkat cell line (genes used marked with asterisks), with inset zoomed in on the rearrangement. C: Fully spliced mRNA of the Jurkat  $\beta$  rearrangement. 'A' represents the poly-A signal. Complementarity determining region 3 (CDR3) formed at the recombination site is marked.

 $<sup>^{2}(45</sup> V\alpha \times 50 J\alpha) \times (48 V\beta \times 2 D\beta \times 13 J\beta) = 2,808,000$ . Values taken from IMGT/GENE-DB (19), considering only the prototypical allele (thus ignoring the effect of polymorphism), for those V(D)J genes predicted to be functional. Note that this value is two orders of magnitude greater than the number of genes in the rest of the human genome, the estimated value of which varies between 19,000 and 31,000 (depending on the definition of 'gene') (20, 21).



Figure 1.2: The human alpha-delta germline locus. Schematic representation of the human TRA/TRD locus. Successful recombination of an  $\alpha$  chain prevents expression of a  $\delta$  chain by excision. Genes are coloured as per the IMGT standard system (described later in Section 2.1): V genes are green, D genes dark blue, J genes yellow and constant regions lighter blue. Black ' $\delta$ ' characters indicate a IMGT-designated TRA/DV gene, while red ' $\delta$ ' characters indicate those which are supposed to only contribute towards delta chain rearrangements. Note the furthest 3' V gene is underscored by a leftwards arrow: this signifies TRDV3, which appears downstream of the TRD cluster, forcing it to recombine by inversion (as opposed to standard deletional joining) (17, 18). Double grey lines indicate omitted genes.

## **1.3** TCR structure, function and selection

T-cells broadly fall into one of two cell types that perform separate immune functions. Cytotoxic T-lymphocytes (CTLs) survey cells of the body for indications of viral infection or tumour development, and kill such potentially dangerous cells. In constrast, T helper ( $T_{\rm H}$ ) cells are non-cytolytic and release cytokines to modulate the local immune response of other cells. These contrasting abilities are both mediated by their TCRs binding to specific ligands. These cell types can generally be distinguished by the mutually exclusive expression of a glycoprotein co-receptor: CTLs express CD8 while  $T_{\rm H}$  cells express CD4.

The exact nature of the TCR ligand was gradually revealed by a number of important experiments, starting with the discovery that CTLs from one animal could only lyse cells derived from an animal which shares a major histocompatibility complex (MHC) allele (29); TCRs are thus 'MHC-restricted'. Unlike antibodies, TCRs were not found to be able to bind 'native' antigens; instead their protein antigens must first be processed and presented by cells (30). In combination with the solved crystal structure of an MHC molecule<sup>3</sup> (31), the observation that addition of appropriate synthetic peptides to cells could cause T-cells to specifically activate against them (32, 33) lead to the realisation that TCR ligands consist of short peptides, bound in the membrane-distal groove of MHC molecules (34, 35) (Figure 1.3A). CD4 and CD8 contribute by binding specific families of MHC molecules: CD8 on CTLs binds to MHC class I, while CD4 binds to MHC class II, directing T-cell restriction (36). Further structural data revealed that the region of each TCR polypeptide chain encoded by the hypervariable DNA at

<sup>&</sup>lt;sup>3</sup>Human leukocyte antigen (HLA)-A2.



Figure 1.3: TCR binding to peptide-MHC (pMHC). Images of the A6 TCR recognising a self-peptide from neuronal protein HuD, bound in the context of the class I MHC HLA A2 (28). A: Whole structure profile revealing the peptide (yellow), bound in the groove of the MHC molecule (light grey,  $\beta$ 2-microglobulin in dark grey). Constant and variable domains of the TCR are indicated above.  $\alpha$  chain is coloured green,  $\beta$  chain is coloured purple. B and C: Demonstration of the proximity of the  $\alpha$  and  $\beta$  chain CDR3 regions (coloured red and aqua respectively) with the MHC-bound peptide (note that C is taken from the point of view of the MHC, not shown). Crystal structure obtained via the PDB accession 3PWP, images made using PyMo1.

the site of recombination – named complementarity determining region 3 (CDR3) – forms the primary contacts with the peptide bound in the MHC groove (Figure 1.3B and C), while other (germline encoded) variable sites (CDR1 and CDR2) usually make contacts elsewhere on the MHC (37).

T-cells are derived from lymphoid progenitor cells that originate in the bone marrow, which must travel to the thymus to develop<sup>4</sup>. Thymocytes begin their thymic development with unrearranged TCR gene loci, double negative (DN) with respect to the TCR co-receptors. As they navigate through various thymic compartments they undergo various genetic events and checkpoints including the rearrangement of their  $\beta$  chain, which must be confirmed by successfully pairing with an invariant pre-TCR to pass ' $\beta$ -selection', and expression of both CD4 and CD8 during a double positive stage (DP), followed by recombination of their  $\alpha$  chain (39). DP cells must then undergo two rounds of selection to ensure their TCR has the required properties, through interaction with thymic epithelial cells expressing a wide variety of self peptide-MHC (pMHC) antigens: cells must demonstrate the ability to bind to pMHC (positive selection), while not binding so strongly to self antigens so as to pose an autoimmune risk (negative selection) (40). DP cells that pass through thymic selection lose expression of one of the co-receptors, becoming CD4 or CD8 single positive (SP), and can leave the thymus to enter the periphery as a naïve T-cell.

This system of T-cell development therefore uses a somatic DNA recombination process to produce a diverse anticipatory array of different TCRs – the TCR repertoire – that stands to activate in response to antigens derived from a world of unknown and potentially pathogenic organisms.

### 1.4 TCR discovery

Study of the TCR began with its discovery in the early 1980s. By this time it was already known that different T lymphocyte clones (or 'clonotypes') could respond to specific antigens, producing the cellular immune response (41). However the quest to find the receptor responsible for this specificity proved difficult, perhaps hampered by initial expectations that such receptors would resemble the B-cell style immunoglobulins (42). In the space of one year, three groups had raised monoclonal antibodies against T-cell lines which recognised clonotype-specific surface-expressed disulphide-linked heterodimeric glycoproteins, in both mouse and man (43, 44, 45). These heterodimers

<sup>&</sup>lt;sup>4</sup>Jacques Miller's discovery of this immune function for the thymus not only finally answered the long standing mystery of what role the organ fulfilled, but also began the chain of events that lead to the identification of T-cells and B-cells as separate lymphocyte lineages (38).

associated with CD3<sup>5</sup>, which was already known to be a marker of T lymphocytes that promoted activation when targeted by antibodies (44). These clonotypic protein structures - that were of similar structure but comprised of different protein sequences - were found on T-cells, but not other cells such as B-cells, providing the first clue that these discoveries might correspond to the TCR (46, 47). These findings began the search for the transcripts and genes encoding these proteins. Two groups headed by Mark Davis and Tak Mak published back to back papers describing this achievement, using similar strategies working on the hypothesis that TCR genes would possess qualities of rearranged immunoglobulin genes (having both a variable, recombined region for antigen binding and a constant region for signalling) yet only be expressed in T-cells, not B-cells (48, 49). These assumptions informed experimental designs that allowed isolation of the first human and murine TCR sequences through a process of cDNA screening called subtraction<sup>6</sup>. Sequence analysis revealed homology to immunoglobulin genes (49, 50), as well as a comparable arrangement scheme consisting of a constant and a variable region, which was further divided into separate segments (51). The first  $TCR\beta$  sequences had been found, beginning the modern age of TCR analyses. Two additional chains were cloned soon afterwards using similar techniques and proposed as potential binding partners of TCR $\beta$  (52, 53). Partial peptide sequences of the TCR $\beta$ associated molecules revealed that one of these chains (the second to be discovered) was the partner of the  $\beta$  chain, and therefore the  $\alpha$  chain (54)<sup>7</sup>. The remaining chain would become known as the  $\gamma$  chain, whose contribution to TCR receptors would remain unknown until pulsed-field gel electrophoresis experiments revealed an additional recombining locus nestled within that encoding the  $\alpha$  chain (15), which was found to produce  $\delta$  chains, the binding partner of  $\gamma$ .

<sup>&</sup>lt;sup>5</sup>Then known as 'T3'.

<sup>&</sup>lt;sup>6</sup>Subtraction consists of incubating labelled cDNA from the cell of interest (T-cell clones) with unlabelled RNA from a related cell (B-cells), which removes the sequences that are common to both by separating single and double stranded sequences.

<sup>&</sup>lt;sup>7</sup>The chains had already been named arbitrarily upon initial discovery of the protein heterodimer, following electrophoresis under denaturing conditions: the larger chain was named  $\alpha$ , while the smaller chain was named  $\beta$  (55).

### 1.5 Historical methods of investigating TCR repertoires

Having identified the individual TCR chains, the challenge then became how to study the distribution of different rearrangements within a sample. However, the complexity produced by V(D)J recombination presents a number of technical challenges to study. Various technologies have been employed to monitor the TCR repertoire with varying throughput, resolution, and capabilities.

Since their role in the discovery of the TCR proteins, monoclonal antibodies directed against specific V families have been widely used to measure T-cells bearing TCRs using particular V genes via flow cytometry (56). Flow cytometry is able to screen millions of cells for surface expression of specific Vs, both quickly and at relatively low cost (57). Modern flow cytometers can detect up to 18 colours, theoretically giving simultaneous recording of large swathes of the repertoire in one experiment; the advent of mass cytometry – in which mass spectrometry is used to detect antibody-bound rare-earth elements, instead of fluorescence detection via CCDs – could even allow detection of all Vs of a given chain (58). Cytometric techniques can be used to further characterise Vspecific T-cells by staining for phenotypic and lineage cell markers. However, in practice even modern studies using flow cytometry tend to only monitor the  $\beta$  chain, using a panel of antibodies designed to cover the majority of TRBV genes (57, 59), and there are currently no published accounts of mass cytometry being employed for TCR repertoire studies. Furthermore cytometry can only provide information as to the frequency of cells using particular V genes in a population, providing no further information about the nature of the rearrangement. Therefore despite the large throughput, flow cytometry produces very low resolution data. The availability of antibodies covering the majority of V genes can also be limiting (47, 60), particularly in non-model organisms, and V gene polymorphisms may prevent binding.

The discovery of TCR genes was contemporaneous with the development of polymerase chain reaction (PCR) (61, 62); this technology was rapidly adopted by immunologists to quantitatively monitor the expression of V gene families (63, 64). PCR has the advantage of being able to selectively amplify nucleic acids bearing specific sequences, allowing one to target TCR chains at various levels. One can target and amplify individual clonotypes (to track leukaemic clones (65), for instance), individual V or J gene families (64) or even all TCRs within a chain. PCR has also facilitated the development of almost every other technique to measure TCR repertoires which uses genetic information.

One of the more widely used PCR-based techniques to study repertoires of variable immune receptors involves amplifying from RNA using specific V gene family and constant region primers, before visualising the profile of CDR3 lengths produced by electrophoresis  $(66, 67)^8$ . These 'spectratypes' give a great deal of information about the spread of clonalities within a V family; expansions of clones can be visualised as departures from the typical Gaussian-like distribution of CDR3 lengths. Spectratyping therefore provides a relative increase in resolution with respect to flow cytometry, providing information regarding clonal expansions within a V family. Variations upon spectratyping such as heteroduplex analysis (69) or single-strand conformation polymorphism analysis (70) exist that further separate amplicons based on their nucleotide conformational properties, which can offer resolution beyond CDR3 length distributions.

Having played a key role in the identification of TCR chains, sequencing of cloned TCR cDNA molecules remains an important means of assessing TCR identity and diversity (71). Sequencing produces the highest resolution data of all methods discussed thus far: nucleotide level information, often for the complete variable region. This information typically comes at the cost of throughput, as cloning and sequencing are relatively time, labour and expense intensive. However, cloning and sequencing from single cells allows extraction of paired  $\alpha$  and  $\beta$  chain sequences, giving the actual complete clonotype of each cell (72, 73).

### **1.6** TCR repertoire analysis by deep-sequencing

As with the emergent technologies described above, the field of immunology has begun to adopt high-throughput sequencing (HTS) (described in detail in Chapter 5) in order to study TCR repertoires. These technologies offer both high-throughput and highresolution data, as they can produce nucleotide sequence information for thousands to millions of clones in one experiment. Whilst still a technique in its infancy, repertoire sequencing (or 'RepSeq' (74)) has revealed a number of interesting features of T-cell biology.

The earliest TCR deep-sequencing papers set out to parameterise the TCR $\beta$  repertoires of healthy individuals, measuring clonotype frequencies, CDR3 length and V(D)J usage (75, 76). Early reports used such data to estimate the lower bound of number of distinct peripheral TCR $\beta$  rearrangements. Values ranged from one to four million sequences (76, 77), in agreement with the oft-cited estimate from Arstila *et al* (78), although at least one group claims it could be two order of magnitude greater (79). A

 $<sup>^{8}</sup>$ It is also possible to perform this procedure using gDNA as a template, but this requires using primers directed against the J genes instead of the constant region, due to the presence of the large J-C intron(68).

profound decrease in repertoire diversity has also been reported to occur with increased age, consistent with findings from previous technologies (79, 80). The repertoires of T-cell responses to Epstein-Barr Virus (EBV) and cytomegalovirus (CMV) have been found to be very stable, with the hierarchy of expanded sequences being maintained over several years (81).

TCR RepSeq has also shown its use in answering various other immunological research questions. For instance,  $\alpha\beta$  and  $\gamma\delta$  T-cells were sorted from healthy donors and both TCR $\beta$  and  $\gamma$  chains were sequenced; while only a small number of  $\beta$  sequences were present in the  $\gamma\delta$  cell libraries, recombined  $\gamma$  chain sequences were found at high frequency in  $\alpha\beta$  populations, supporting the hypothesis that TCR $\gamma$  rearrangement occurs prior to T-cell lineage commitment, while TCR $\beta$  rearranges afterwards (82). Similarly, the presence of some sequences in multiple different T-helper subsets of the same individual argues against the possibility that TCRs are deterministic with regards to T-helper fate (83). Extensive profiling of repertoires of multiple different tissues from organ donors reveals considerable compartmentalisation of different clonotypes to different tissues (84). Patients with various immunodeficiencies, including aplastic anaemia (85), Wiskott-Aldrich syndrome (86, 87), and Omenn syndrome (88) have been sequenced, invariably revealing restriction or skewing of the TCR repertoire relative to healthy controls.

One of the first major clinical applications of TCR RepSeq is in identifying and tracking the clonal sequences of T-cell derived leukaemias and lymphomas (89, 90, 91, 92), as the clonal rearrangement of the original cell acts a specific marker of neoplastic cells. Different groups have employed sequencing of  $\beta$  and/or  $\gamma$  chains, depending on the nature of progenitor cell's lineage <sup>9</sup>. Deep-sequencing has shown itself to be a particularly sensitive tool for detecting minimal residual disease (MRD), i.e. the continued presence of neoplastic cells during therapy, out-performing the existing techniques of flow-cytometry (89) allele-specific oligonucleotide PCR(90), and (in a model system of MRD) capillary sequencing (96). In a recent study by Logan *et al*, MRD detected by HTS was completely predictive of relapse to disease following haematopoietic stem cell transplantation (HSCT) (95). In addition to detecting MRD, TCR sequencing produces other data that could be of diagnostic or prognostic value in cancer patients, such as monitoring the dynamics of the repertoire fluctuation on treatment.

Measuring immune system reconstitution following HSCT is another important clinical application of repertoire sequencing, which has been explored beyond treatment of

 $<sup>^{9}\</sup>mathrm{Immunoglobulin}$  rearrangements have also been sequenced to track B-cell neoplasms (90, 91, 93, 94, 95).

cancer. One patient with ankylosing spondylitis who received HSCT has been extensively studied in order to evaluate immune recovery (97, 98). Not only did sequencing reveal the emergence of a skewed but stable repertoire, but a number of clones that were present at low frequency pre-transplantation were found to be both retained and expanded at later timepoints. A later study aimed to test whether whether TCR sequencing might stratify responses in patients receiving transplants from different stem cell sources (99). Six months after transfusion, cord blood stem cell recipients developed repertoires with diversities approaching that of healthy donors, while those who received stem cells derived from adult peripheral blood were far less diverse, potentially leaving them at greater risk of opportunistic infections and cancer relapse. Graft-versus-host disease (GVHD), a major risk factor for HSCT patients, has also been investigated with HTS. Steroid-refractory patients showed greater repertoire conservation between samples from multiple gastrointestinal (GI) sites (100). Moreover, clones identified in these GI samples were found to be expanded in longitudinal peripheral blood samples. Therefore TCR RepSeq potentially offers both a means of quantifying immune reconstitution in HSCT patients and stratifying those who might fail to respond to therapy.

## 1.7 Aims of this thesis

TCR repertoire sequencing represents a powerful tool that can be applied in the study of diverse immunological and clinical phenomena. However it is still a developing technology, and its full potential is yet to be fully realised. Therefore the first aim of this thesis was to develop protocols for the amplification, high-throughput sequencing and bioinformatic analysis of TCR repertoires from human blood samples. These protocols should be robust, repeatable, and incorporate as little bias as possible. Having established repertoire-sequencing techniques, the next aim was to apply them to blood samples taken from healthy volunteers. This should allow both testing of the capabilities and limitations of the method, as well as produce an understanding of the nature of the repertoire features in healthy adult individuals using such high-resolution data. The final aim of this thesis was to sequence the repertoires of patients suffering from a clinically relevant infection, to assess the merit of using such technologies to understand the effect of infection upon the immune system. As a pathogen of global clinical importance, that is known to have a large impact on the T-cell compartment, human immunodeficiency virus (HIV) was chosen as a suitable model. Patient samples were to be sequenced and compared to those of healthy, uninfected donors, to allow measurement of the differences between the two groups.

# Materials and methods

# 2.1 The international ImMunoGeneTics (IMGT) information system and nomenclature

The IMGT system was established in the Montpelier Laboratoire d'Immuno Génétique Moléculaire under the direction of Professor Marie-Paule Lefranc, to provide standardised references for immunogenetic and immunogenomic research (101). I have made frequent use of IMGT system throughout the generation of my data and the writing of this thesis. All annotated TCR gene sequences were obtained from the IMGT database IMGT/GENE-DB (19), and IMGT/LocusView was very useful in the production of a number of schematics in this thesis.

The IMGT-defined and Human Genome Organisation (HUGO)-approved TCR nomenclature has been used throughout this document (102, 103). In this system, TCR (and immunoglobulin) loci are first divided into gene types, i.e. V, D and J<sup>1</sup> Any genes within a given locus of a species that share 75% nucleotide sequence identity with each other are designated as subgroups. Specific genes can furthermore exist as one of several alleles. Genes are referred to with identifiers that incorporate all of this information. For example, TRAV8-4\*01: TR indicates a TCR gene, AV indicates that it's an  $\alpha$  chain V gene; 8-4 signifies that it's the fourth gene in the eighth subgroup in that locus, while \*01 indicates that it's the prototypical allele of that gene<sup>2</sup>.

Where possible, I have also adopted the IMGT colour conventions for use in schematic illustrations, as detailed in the IMGT Scientific Chart (104, 105). Accordingly, V genes have been coloured green (hexadecimal colour code #00EE00), D genes dark blue (#0000E4), J genes yellow (#FFE400), constant regions light blue (#0099FA), N nucleotides red (#FF0000) and where featured, leader sequences in light yellow (#FFFF00 and bordered in black to distinguish from J sequences).

<sup>&</sup>lt;sup>1</sup>The recombinable DNA segments are referred to as 'genes' under IMGT terminology, for both practical and biological reasons. Practically, as labelling them as genes was a requirement for them to be indexed in various databases, and biologically on the grounds that each can be transcribed independently as they have their own promoters.

<sup>&</sup>lt;sup>2</sup>Note that in this system variable antigen receptor gene names are not italicised, in contrast to the standard convention for human genes.

# 2.2 General considerations and RNA extraction

In order to minimise the risk of contamination or biohazards, all molecular or cell biology was undertaken with appropriate control measures. Handling of nucleic acids prior to amplification was undertaken in UV- and chemically-treated PCR hoods, and highlyconcentrated PCR or cloning products were purified in separate rooms, to minimise the potential for DNA contamination. Peripheral blood and cell lines were manipulated in laminar flow hoods. Bacterial cloning was carried out at the bench under asceptic conditions.

RNA samples were obtained from consenting adults in one of two ways. RNA for protocols that did not use molecular barcoding (i.e. simple V and poly-T amplification), was derived from peripheral blood mononuclear cells (PBMC) or sorted cell fractions, obtained by density centrifugation of up to 120 ml of peripheral blood using Lymphoprep (Axis-Shield). PBMC were then lysed in RLT buffer (Qiagen) and RNA extracted using the Qiagen RNeasy kits and eluted in 60  $\mu$ l of water. RNA for the protocols using random barcodes (BV and BLR amplification) was extracted by other researchers (see relevant sections) by drawing 2.5 ml of peripheral blood into Tempus tubes (Life Technologies) and eluting RNA in approximately 90  $\mu$ l of water. All cell lysates, RNA and Tempus tubes were stored at -80°C, while cDNA and amplified DNA products were stored at -20°C. RNeasy and Tempus extractions, and Lymphoprep separation were performed as per the manufacturer's instructions (APMI).

## 2.3 Amplification and deep-sequencing protocols

The development and application of the following protocols made up the major component of all practical molecular biology undertaken during this body of work; the rationale and details of this development is covered in detail in Chapter 5. Primer sequences for all protocols are detailed below in Section 2.3.7.1.

### 2.3.1 Reverse transcription

After extraction, residual genomic DNA (gDNA) was removed from RNA samples using commerically available DNase I (or derivative engineered forms). For all protocols except the BLR protocol, RNA was incubated with 0.1 U/ $\mu$ l of RQ1 DNase (Promega) in 1X reaction buffer for 30 min at 37°C. Samples processed using the BLR protocol had gDNA removed using the TURBO DNase kit, and globin mRNA was depleted using GLOBINclear<sup>3</sup> (both Life Technologies, APMI).

<sup>&</sup>lt;sup>3</sup>Globin message was removed so that this RNA could be used for additional microarray studies.

Reverse transcription was carried out using oligonucleotides named  $\alpha$ RC2 and  $\beta$ RC2, which are complementary with sequences towards the 5' end of the  $\alpha$  and  $\beta$  constant regions respectively.  $\beta$ RC2 targets a region that is conversed between the two human  $\beta$  constant regions, TRBC1 and TRBC2. RNA (typically 500 ng) was mixed with 1  $\mu$ l of each RC2 primer (10  $\mu$ M) and 1  $\mu$ l of a 10mM dNTP mix and heated at 65°C, 5 min and cooled rapidly on ice. 1  $\mu$ l of DTT (0.1  $\mu$ M), 1  $\mu$ l of RNasin (RNase inhibitor, 20-40U/ $\mu$ l, Promega), 1  $\mu$ l of SuperScript III reverse transcriptase (RT) (200 U/ $\mu$ l, Life Technologies) and 4  $\mu$ l of 5X FS buffer was added, before incubation at 55°C, 20 min, then 70°C, 15 min for heat inactivation. Complementary DNA (cDNA) from RT reactions was purified and eluted into either 10  $\mu$ l or 30  $\mu$ l of buffered Tris-Cl using MinElute or QIAquick columns respectively (Qiagen).

### 2.3.2 Simple V amplification

Primers were designed against TRAV8-4 and TRBV12-3 (named  $\alpha$ F1 and  $\beta$ F3) and the very 5'-most end of TRAC and TRBC ( $\alpha$ RC1 and  $\beta$ RC1). The two  $\beta$  chain constant region genes differ at two positions in this region, therefore  $\beta$ RC1 contains two degenerate bases. Also note that while  $\alpha$ RC1 ends at the most 5' nucleotide of the constant region, the beta primer actually overlaps with the last nucleotide of the J gene, as all TRBJ genes end with the same nucleotide.

1  $\mu$ l reverse transcribed cDNA from lysed PBMC or sorted cells underwent a multiplex PCR using all four primers ( $\alpha$ F1,  $\beta$ F3,  $\alpha$ RC1 and  $\beta$ RC1). Reagent final concentrations in 50  $\mu$ l reaction volumes were: 1X GoTaq Green buffer (Promega); 0.1 mM each dNTP; 0.2  $\mu$ M each primer; 0.5 mM additional MgCl<sub>2</sub> and 2.5 U GoTaq DNA Polymerase (Promega). PCR mixtures were subjected to 30 cycles of: 94°C, 30 sec; 60°C, 2 min; 72°C, 1 min, with a 1 min final extension. PCR products were visualised on 1% agarose gels, and gel purified before submission for 100 bp paired-end (PE) Illumina HiSeq sequencing. Sequencing was performed at the Wellcome Trust Sanger Institute, in collaboration with Professor Paul Kellam's group.

#### 2.3.2.1 Magnetic assisted cell sorting and flow cytometry

PBMC from two donors – D3 and D4 – were sorted into  $CD4^+CD45RA^+$  and  $CD4^+CD45RO^+$  populations (broadly representative of  $CD4^+$  naïve and memory cells) prior to lysis, using negative-selection magnetic assisted cell sorting (MACS). Negative-selection refers to the targets of the selection: antibodies are used to target beads to cells other than those of interest by binding surface antigens that they do not express, and collecting unbound enriched desired cells from the eluate.

After isolation, PBMC were divided in two and subjected to either the Naïve or Memory CD4<sup>+</sup> T cell Isolation Kit (Miltenyi Biotech). Cells were counted and suspended in 40  $\mu$ l of MACS buffer, before adding 10  $\mu$ l of relevant antibody cocktail and incubating at 4°C for 10 min, after which the protocols diverge.

- Naïve: cells were washed in 20 ml of MACS buffer, spun down and supernatants removed. Cells were then resuspended in 80 μl of buffer before adding 20 μl of microbeads, before mixing and incubation at 4°C for 15 min. Another wash with 20 ml of MACS buffer followed, before cells were spun down and supernatants removed. Sorted cells were resuspended in 500 μl of buffer.
- Memory: 30 μl MACS buffer was added, followed by 20 μl of microbeads, before mixing and incubation at 4°C for 15 min. Cells were then washed in 20 ml buffer, spun down and supernatants were discarded. Sorted cells were finally resuspended in 500 μl buffer.

Cells underwent two rounds of negative selection to achieve greater purification, which was assessed by flow cytometry. Staining was performed in PBS with 10% goat serum. The following antibodies were used: CD3-FITC (Miltenyi), CD4-RPE (Dako), CD45RO-PeCy7 (BD) and CD45RA-APC (BD). Cells were incubated with antibodies at 4°C for a minimum of 30 min, before being washed and resuspended in a suitable volume. Flow cytometry was performed on a FACS Calibur (BD Biosciences) and analysed in FlowJo.

### 2.3.3 Poly-T RACE amplification

Rapid amplification of cDNA ends (RACE) was performed by addition of an adenosine tail to cDNA followed by amplification using a poly-T containing primer. First, 5  $\mu$ l of cDNA was polyadenylated using terminal deoxytidyl transferase (Tdt, Promega). Final concentrations were: 1X Tdt reaction buffer; 0.4 mM ATP and 30 U Tdt (Promega). Reagents were mixed and incubated at 37°C for one hour, heat inactivated at 75°C for 15 min and purified using QIAquick PCR purification columns (Qiagen).

5  $\mu$ l of poly-adenylated cDNA was then amplified using a combination of two different poly-T containing oligonucleotides ( $\alpha$ L2 and  $\beta$ L2), and the constant region primers  $\alpha$ RC1 and  $\beta$ RC1. Unequal representation of TRBJ genes from the two TRBD clusters in the 454 data was observed, leading to the degenerate  $\beta$ RC1 oligonucleotide being replaced with an equimolar pool of two primers that specifically target each TRBC gene ( $\beta$ RC1.1 and  $\beta$ RC1.2). This allows for greater control of the concentration of each desired oligonucleotide sequence in the mixture, and removes the additional unwanted DNA species that would be produced in the original degenerate  $\beta$ RC1 formulation.

Final concentrations for the poly-T PCR were: 0.2 mM each dNTP; 0.5  $\mu$ M each primer; 1 U Phusion polymerase (New England Biolabs, or NEB) and 1X HF buffer. Due to the reported strong 3' $\rightarrow$ 5' endonuclease activity of Phusion, all reagents save enzyme and buffer were mixed separately first, followed by rapid addition of a mixture of enzyme in buffer on ice immediately before commencing thermal cycling, APMI. Thermal cycling conditions were: 98°C, 3 min; 54°C, 1 min; 72°C, 1 min, then 35 to 40 cycles of: 98°C, 15 sec; 70.7°C, 30 sec; 72°C, 20 sec, before final extension at 72°C, 5 min. Amplified TCRs were then selected and purified by gel extraction from 1-2% agarose gels. Amplicons for 454 sequencing were sent to the Wellcome Trust Sanger Institute for sequencing on the GS FLX instrument.

#### 2.3.3.1 Poly-T RACE MiSeq library preparation

A second PCR was developed to add the required Illumina sequences specifically to the ends of the poly-T amplicons to allow processing on the MiSeq. Sequencing Primer 1 and 2 (SP1 and SP2) – the binding sites which define the start of the two sequencing reads – must reside immediately adjacent to amplicons sequences. P5 and P7 sequences are required immediately adjacent to the sequencing primers, as these bind to the flowcell bound oligonucleotides to facilitate bridge amplification. Therefore I designed a set of primers to add each of these elements in one reaction by adding the innermost primers at limiting dilutions: as the reaction progresses these should be outcompeted by the higher concentration outermost primers, promoting the assembly of full length amplicons. SP1 and SP2 were added to the constant regions and poly-T sequences respectively with long primers SP1-N6- $\alpha$ RC1 or SP1-N6- $\beta$ RC1.1+2 (where N6 represents a random hexamer to augment nucleotide diversity), and SP2-pT. P5 and P7 elements were then added to these by extension with long primers P5-SP1 and P7-X-SP2, where X is a six-base index to allow sequencing of multiple samples. This index is read in a separate read from a primer targeting the reverse complement of the SP2 sequence. Finally, two primers at the highest concentration target the outermost sequences: a 3' shortened version of P5 (P5s, or short) and P7.  $\alpha$  and  $\beta$  reactions were carried out separately.

Final concentrations were: SP1-N6- $\alpha/\beta$ RC1 and SP2-pT, 0.05  $\mu$ M; P5-SP1 and P7-X-SP2 0.1  $\mu$ M; P5s and P7, 0.5  $\mu$ M; 0.2 mM each dNTP, 1 U Phusion and 1X HF buffer (enzyme in buffer added last). Thermal cycler steps were: 98°C, 3 min; 3 cycles of 98°C for 15 sec, 56°C for 45 sec, 72°C for 45 sec; then 17 cycles of: 98°C, 15 sec; 69°C,

30 sec; 72°C, 30 sec; final extension at 72°C, 5 min. The first three cycles with lower annealing temperatures allow annealing of homopolymer RACE sequences, which have low melting temperatures. Amplified TCRs were purified by gel extraction from 1-2% agarose gels and processed for MiSeq sequencing.

#### 2.3.3.2 Cloning

Initial clones produced to test the efficacy of the Tdt-RACE PCR were produced through blunt-ended cloning of PCR products into pUC19. Plasmid DNA was linearised using the restriction enzyme SmaI and dephosphorylated using calf intestinal alkaline phosphatase (both NEB, APMI) to prevent recircularisation, before gel excision and purification. Physical DNA polymerase produces blunt-ended products and therefore required no further treatment post-purification. Ligation reactions using 2 U T4 DNA ligase (Promega), 1X reaction buffer and varying ratios of purified insert and linearised vector DNA were incubated overnight at 4°C. 10  $\mu$ l of ligation mixtures were used to transform 50  $\mu$ l of  $\alpha$ -Select Silver competent *E. coli* (Bioline). Ligation mixes were added to bacteria (thawed on ice), before being briefly flick-mixed and incubated on ice for 30 min. Cells were then heat-treated (42°, 30 sec), immediately cooled on ice (2 min) and diluted in 1 ml antibiotic-free lysogeny broth (LB, Sigma-Aldrich) before shaking at 37°C, 220 rpm for 1 hour. Cells were then spread on AIX LB agar plates (0.1 g/ml ampicillin, 0.04 g/ml IPTG and 0.03 g/ml X-gal) and incubated overnight at 37°C. Potential recombinants were selected by blue/white screening and digested to check for the presence of inserts.

Clones generated to test for the correct incorporation of MiSeq elements were ligated into pGEM-T-Easy (Promega) to allow TA-cloning, facilitated by amplification using GoTaq polymerase (which produces amplicons with a single adenosine overhang). Purified amplicons produced from the Phusion reaction were re-amplified using GoTaq. PCR mixtures consisted of: 1X GoTaq Green buffer (Promega); 0.1 mM each dNTP; 0.2  $\mu$ M of P5s and P7 primers; 0.5 mM additional MgCl<sub>2</sub> and 2.5 U GoTaq DNA Polymerase (Promega). PCR mixtures were subjected to 35 cycles of: 95°C, 30 sec; 64°C, 30 sec; 72°C, 1 min, with a 5 min final extension. PCR products were gel-purified and ligation into pGEM-T-Easy in an equivalent protocol as used above.

All plasmids were Sanger sequenced by the Wolfson Institute for Biomedical Research DNA Sequencing Service at UCL<sup>4</sup>, using AB capillary sequencing technology. Sequencing traces were processed using FinchTV (Geospiza); poor quality bases were trimmed before analysis.

<sup>&</sup>lt;sup>4</sup>Now defunct.

### 2.3.4 Randomly barcoded V (BV) amplification

The BV stategy targets two more V genes than the simple V amplification protocol (described in Section 2.3.2), while improving upon the MiSeq library preparation protocol developed for the poly-T RACE (Section 2.3.3.1). Long primers were designed to add SP2 elements and random hexamers to four specific V genes: those targeted in the simple V amplification, TRAV8-4 and TRBV12-3 (those used by Jurkat), and TRAV21 and TRBV20-1 (which had both consistently appeared at high frequency in previous experiments). These primers were used to perform a second strand extension on our cDNA, making double stranded DNA. A further extension reaction step - which I refer to as a 'third strand' reaction - then uses long oligonucleotides to add SP1 and random hexamer elements immediately downstream of the constant region sequence, allowing bidirectional sequencing through two random sequences, which can be used as molecular barcodes. Two further PCRs are then performed, the first adding P5 and P7 elements, the second amplifying to required concentrations. AMPure XP bead purifications (Agencourt Bioscience) are performed in between each round of DNA polymerisation using 1:1 bead:solution ratios to remove primers (APMI), eluting into 30  $\mu$ l of water each time. These purifications make use of the propensity of high concentrations of polyethylene glycol (PEG) and salt to flocculate DNA (106) onto carbonyl-coated paramagnetic beads, allowing contaminants or low-molecular weight DNA to be removed by washing.

5  $\mu$ l of cDNA was carried into 25  $\mu$ l separate  $\alpha$  or  $\beta$  second strand reactions, using: 0.2 mM each dNTP; 0.5  $\mu$ M of each chain-specific primer pair (SP2-N6-TRAV21 and SP2-N6-TRAV8-4 or SP2-N6-TRBV12-3 and SP2-N6-TRBV20-1); 0.4 U Phusion, and 1X HF buffer. Reactions underwent the following program: initial denaturation at 95°C, 3 min, followed by a slow ramp to 80°C, 10 sec and another slow ramp to 58°C, 45 sec, and final extension for 5 min at 72°C. Slow ramping involves descending at 0.5°C/sec; this slow descent aims to encourage optimal annealing of the V region specific component of the long oligonucleotides to anneal at the highest possible temperature, minimising the chance of off-target priming. After purification, all 30  $\mu$ l of doublestranded DNA is carried into 50  $\mu$ l third strand synthesis reactions. These use the same reagent concentrations and cycling conditions as the second strand reaction, save that the SP2 primers are replaced by SP1-adding primers SP1-N6-I- $\alpha$ RC1 or SP1-N6-I- $\beta$ RC1.1+2, where 'I' represents a second indexing sequence, allowing greater flexibility when planning sample multiplexing.

All of the purified third strand DNA undergoes 'PCR1' to add the P5 and P7 elements: 0.2 mM each dNTP; 0.5  $\mu$ M of each primer (P5-SP1 and P7-X-SP2) and

1 U Phusion in 1X HF buffer. Mixtures were then thermally cycled: initial denaturation occurred at 95°C for 3 min, before ramping slowly to 69°C, 15 sec, and extended at 72°C, 1 min, before 3 cycles of 98°C, 10 sec to 72°C, 1 min. Final extension was at 72°C for 5 min. DNA was purified and all carried into the final reaction, 'PCR2', to amplify sufficient product for sequencing. Reaction conditions were the same as in PCR1, save that the primers used were just the outermost sequences, P5s and P7. PCR program began with initial denaturation at 95°C, 3 min, before 22 cycles of: 98°C, 10 sec; 69°C, 15 sec; 72°C, 40 sec, before final extension at 72°C for 5 min. Samples thus underwent a total of 26 cycles of PCR. Samples were then bead purified before dilution to equilibrate concentrations.

### 2.3.5 Barcoded ligation RACE (BLR) amplification

For each donor sequenced using the BLR protocol, 500 ng of RNA was reverse transcribed and purified using MinElute RNeasy columns (Qiagen). Half of the purified solution (5  $\mu$ l of the 10  $\mu$ l produced) was used in each experiment. The SP2 sequence, followed by a hexamer of random bases, was ligated to the V region end of the cDNA. T4 RNA ligase was used to ligate single stranded oligonucleotides to cDNA in a 30  $\mu$ l reaction mix containing: 1X T4 RNA ligase buffer; 20 U T4 RNA Ligase 1 (NEB), 0.33 mM dATP; 25% (w/v) PEG 8000; 1 mM hexamine chloride; 0.1 mg/ml acetylated-BSA, and 0.33  $\mu$ M SP2-6N ligation oligo. Master mixes were vortexed for several seconds to ensure mixing with the PEG, added to cDNA and mixed thoroughly. Ligations were incubated for 23 hours at 16°C, heat inactivated for 10 min at 65°C, diluted in 70  $\mu$ l water and bead purified.

All purified ligated DNA was subjected to second strand synthesis, using just the SP2 sequence as the priming target. Final concentrations in the 50  $\mu$ l reactions were: 0.2  $\mu$ M of each dNTP; 0.5  $\mu$ M SP2 primer and 1 U Phusion, in 1X HF buffer. The single strand extension program was the same as that described for the BV protocol in Section 2.3.4. Purified double-stranded samples were split in two and each half underwent either  $\alpha$  or  $\beta$  chain specific third strand synthesis. Reaction conditions were the same as the second strand reaction, except the SP2 primer was replaced with either SP1-6N-I- $\alpha$ RC1 or SP1-6N-I- $\beta$ RC1. PCR1 and PCR2 were then performed to add flowcell-binding elements and amplify as in Section 2.3.4, with the exception that PCR2 instead contained 23 instead of 22 cycles. BLR samples thus underwent a total of 27 PCR cycles. Samples were purified before normalisation.

Blood samples from HIV patients for BLR sequencing were obtained by the research nurse team based in the Mortimer Market Centre Clinic, Bloomsbury, and RNA extraction was performed by Dr. Eleanor R. Gray. Samples from healthy donors were obtained and had RNA extracted by Drs. Nandi Simpson and Jennifer Roe. RNA in both cases was extracted from approximately 2.5 ml of whole peripheral blood lysed in Tempus tubes (Life Technologies), APMI.

#### 2.3.5.1 Flow assisted cell sorting

PBMC from two healthy donors were sorted into  $CD4^+$  and  $CD8^+$  populations by FACS. Cells were isolated through density-gradient centrifugation using Ficoll-Paque PLUS (GE Healthcare Lifescience) and incubated in 10% goat serum at 4°C for 30 min to block Fc-receptors. Cells were subsequently stained with CD8-FITC, CD3-PE, CD4-PerCp-Cy5.5 (BD Biosciences) and LIVE/DEAD fixable far-red dead cell stain (Life Technologies) APMI. CD3<sup>+</sup>CD4<sup>+</sup>CD8<sup>-</sup> and CD3<sup>+</sup>CD8<sup>+</sup>CD4<sup>-</sup> T-cells were sorted on a BD FACS Aria machine (BD Biosciences).  $7.2 \times 10^6$  CD4<sup>+</sup>CD8<sup>-</sup> cells and  $6 \times 10^6$ CD8<sup>+</sup>CD4<sup>-</sup> T-cells were lysed in RLT buffer (Qiagen) prior to RNA extraction. Cell sorting was performed by Dr. Theres Oakes, at the UCL Flow Cytometry Core Facility.

#### 2.3.5.2 HIV patient criteria

Adult HIV+ HAART-naive patients due to start treatment imminently were recruited to the study. Exclusion criteria included: febrile or AIDS-related illness; tumours; coinfection with Hepatitis B or C; immunomodulatory therapy, or recent vaccination. Interruption or adverse reactions to ART were further grounds for exclusion.

### 2.3.6 MiSeq operation

Apart from initial libraries amplified using the poly-T protocol (which were sequenced commercially via UCL Genomics), all libraries were sequenced by myself. Purified amplified samples were first normalised to a suitable concentration. Concentrations were assessed using a Qubit fluorometer, using the dsDNA HS Kit (Life Technologies). Approximately 2 ng of DNA per sample was then inspected on a Bioanalyzer using the High Sensitivity DNA Analysis Kit (Agilent Technologies), allowing determination of amplification success and peak band size. Amplicon length and concentration were then used to normalise samples to 4 nM before pooling. Typically 10 to 12 amplified samples ( $\alpha$  and  $\beta$  samples from five to six donors) were pooled per run. MiSeq flowcells were prepared as per Illumina's instructions (107). Denatured libraries were loaded at 11-12 pM, using a 1% PhiX spike in. 2x250PE kits with version 2 chemistry were used.

Read quality was assessed using in-built MiSeq metric software in combination with FastQC (108).
# 2.3.7 Primers and indexes

# 2.3.7.1 Primer sequences

Primer	Sequence
lpha  m RC2	GAGTCTCTCAGCTGGTACACG
$\beta RC2$	ACACAGCGACCTCGGGTGGGAA
lpha F1	AGCTTGGCAGCCACGTCTCTG
$\beta { m F3}$	CCAATTTCAGGCCACAACTCC
$\alpha RC1$	ACGGCAGGGTCAGGGTTCTGGATAT
$\beta RC1$	GGTGGGAACACSTTKTTCAGGTCCTC
lpha L2	TCACACAGGAAACAGCTATGACTTTTTTTTTTTTTTTTT
$eta \mathrm{L2}$	GTCATAGCTGTTTCCTGTGTGATTTTTTTTTTTTTTTTT
$\beta RC1.1$	GGTGGGAACACCTTGTTCAGGTCCTC
$\beta RC1.2$	GGTGGGAACACGTTTTTCAGGTCCTC
SP1-N6- $\alpha$ RC1	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNACGGCAGGGTCAGGGTTCTGGATAT
$\operatorname{SP1-N6-}\beta\mathrm{RC1.1}$	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNGGTGGGAACACCTTGTTCAGGTCCTC
$\operatorname{SP1-N6-}\beta\mathrm{RC1.2}$	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNGGTGGGAACACGTTTTTCAGGTCCTC
SP2-pT	GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTTTTTTTT
P5-SP1	AATGATACGGCGACCACCGAGATCTACACTCTTTCCCTACACGACGCTCTTCCGATCT
P7-X-SP2	CAAGCAGAAGACGGCATACGAGATxxxxxGTGACTGGAGTTCAGACGTGTGCTCTTCCGATC
P5s	AATGATACGGCGACCACCGAGATC
P7	CAAGCAGAAGACGGCATACGAGAT
SP2-N6-TRAV8-4	GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTNNNNNTCATCGTCTGTTCCACCATATCTCTTCTG
SP2-N6-TRAV21	GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTNNNNNAGACTTAATGCCTCGCTGGA
SP2-N6-TRBV12-3	GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTNNNNNATGCCCGAGGATCGATTCTCAG
SP2-N6-TRBV20-1	GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTNNNNNTTTGGTATCGTCAGTTCCCG
${\rm SP1\text{-}N6\text{-}I\text{-}}\alpha{\rm RC1}$	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNXXXXXACGGCAGGGTCAGGGTTCTGGATAT
$\operatorname{SP1-N6-I-}\beta\mathrm{RC1.1}$	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNXXXXXXGGTGGGAACACCTTGTTCAGGTCCTC
$\operatorname{SP1-N6-I-}\beta\mathrm{RC1.2}$	ACACTCTTTCCCTACACGACGCTCTTCCGATCTNNNNNXXXXXXGGTGGGAACACGTTTTTCAGGTCCTC
SP2-6N (ligation)	[Phos]NNNNNAGATCGGAAGAGCACACGTCTGAACTCCAGTCAC[SpcC3]

Table 2.1: Primer sequences used in the protocols described above. IUPAC nucleotide codes used: 'N' = any base; 'V' = any base apart from T (A, C or G); 'S' = C or G; 'K' = G or T. Variable bases for indexes are represented by 'x' (see Section 2.2). [Phos] represents a 5' phosphate group, to permit ligation with the 3' hydroxyl group of the cDNA. [SpcC3] = Spacer C3, a blocking moiety to prevent polymerisation of ligation oligonucleotides.

Number	P7-X-SP2 indexes	SP1-N6-I- $\alpha/\beta {\rm RC1}$ indexes
1	ATCACG	ATCACG
2	CGATGT	CGATGT
3	TTAGGC	TTAGGC
4	TGACCA	TGACCA
5	ACAGTG	ACAGTG
6	GCCAAT	GCCAAT
7	CAGATC	
8	ACTTGA	
9	GATCAG	
10	TAGCTT	
11	GGCTAC	GGCTAC
12	CTTGTA	CTTGTA
13	AGTCAA	TAGACT
14	AGTTCC	ACACGG

#### 2.3.7.2 Indexing sequences

Table 2.2: Index sequences used to multiplex samples. Note that the hexamer index sequences ('X') in the P7-X-SP2 sequences are actually the reverse complement of the equivalent standard Illumina indexes of the same number. This makes pooling libraries with others easier (as it minimises the chance of sharing indexes) while maintaining the edit distances between the sequences.

# 2.4 Bioinformatic analysis

## 2.4.1 General considerations

The bioinformatic work described in this thesis was almost exclusively run on a standard work-range laptop computer, with a quad-core Intel i5 CPU ( $4 \times 2.67$ GHz), with 7.6 GiB of RAM, running 64-bit Ubuntu Linux. The majority of analyses were performed using Python (v2.7.6) or R (v3.0.2) and custom scripts, while general text file exploration and manipulation were performed on the command line using bash, awk or sed commands. All manipulation of FASTQ data was performed using Biopython v1.63 (109). Python was typically employed for any rearrangement- or sequence-specific analysis, while R was used for plotting numerical features of whole files. The Python library collections was regularly employed for storing TCR features, the difflib module was used to compare sequences, and statistical analysis was achieved through a combination of NumPy and SciPy libraries, with results plotted using Matplotlib. The

R package lattice was additionally used in the production of heatmaps. Additional sequence file manipulation was performed using the FASTX-toolkit (110). Schematics were made and figures collated using Inkscape.

Unless otherwise stated, analyses are described in this section as they were applied to the data produced using the BLR protocol, as this constitutes the majority of the results of this thesis.

#### 2.4.2 Demultiplexing BLR MiSeq data

High-throughput sequencing (HTS) machines produce data in FASTQ format. FASTQ files are text files, where every four lines corresponds to a different sequencing read, providing an identifier, the sequence, and quality data. Multiple samples are typically processed on one run of a machine – or 'multiplexed' – by introducing sample-specific index sequences prior to pooling and sequencing.

Data produced using the BLR protocol requires additional processing so that both random hexamer sequences are present at known positions in the first read, to allow identified TCRs to be linked to the barcode sequence when analysing a single read file. This pre-processing occurs at the same time as demultiplexing, in one of two ways. The earliest samples were first subjected to the standard Illumina demultiplexing procedure that occurs by default on the MiSeq during FASTQ generation, which deconvolutes samples based on their P7-X-SP2 index sequence. This produces two FASTQ files (one starting from each end of the amplicon) for every P7-X-SP2 index used. A custom script (named AddAl1R2Hex) then pastes the random hexamer sequence (and associatd quality scores) from the second read to the start of the first. This produces a read where the first twelve bases are made up of the two random hexamers, followed by the rest of the read, which could contain V(D)J information. These reads can then be further demultiplexed based on the presence of the relevant SP1-N6-I- $\alpha/\beta$ RC1 index immediately downstream of the barcode, which was done using the grep command in bash to pull out direct matches.

The second technique implemented (used on the majority of runs) involves simultaneously demultiplexing on both indexes, allowing complete control over index string matching criteria. This method involves first obtaining FASTQ data for all three reads, i.e. including the index read (which is not produced by default). This was achieved in one of two ways, depending on which of the two MiSeq machines I had access to was available. One of the machines had an altered configuration file which overrode various default settings, making the MiSeq output the index read simultaneously with the two sample reads. To obtain the index read for data from the other machine, the entire output run folder (containing the raw base-call data) was copied and reprocessed using the Illumina software CASAVA/bcl2fastq to produce all three FASTQ files. All three reads could then be simultaneously processed to enable the production of reads containing: both random hexamer sequences (to become the barcode), both hexamer indexing sequences (to allow demultiplexing) and the rest of the read one sequence (containing the V(D)J information). A single script (**DualIndexDemultiplexing**) scrolls through all three FASTQs simultaneously, moving the relevant sequences to known locations in one output read, which is then written to a specific file based on its sample-specific index sequence.

Demultiplexed reads can then be analysed for TCR rearrangements, while still retaining any barcode information without needing to refer to multiple files.

## 2.4.3 TCR assignment and error-correction

Our group has previously released software called Decombinator, which rapidly scans HTS data for recombined TCR sequences (111). 'Decombined' data stores TCR rearrangements as a five-part classifier, where the fields detail which V and J genes were used, how many deletions there were from each of these genes, and the nucleotide sequence (or 'insert') between the remaining ends of the two genes. As described in Chapter 5, a modified version of Decombinator called 'verbose Decombinator' (**vDCR**) was developed and employed in the work of this thesis. vDCR was built on Decombinator version 1.4, and incorporates additional error-filtering capabilities and outputs additional read information for downstream processes. Note that the most recent release of Decombinator (v2.0) incorporates a number of the general improvements made in vDCR, and thus can be considered broadly functionally equivalent (112). An extended set of V and J gene tags (relative to the standard Decombinator tag set) was used in this work.

Rearrangements detected by vDCR are output in 'n12' file format. If the libraries were produced using either the BV or BLR protocols – which introduce hexamers of random nucleotides to both ends of each cDNA molecule prior to amplification – then the additional information contained in the n12 file can be used for further data correction. By using the 12 random nucleotides as a barcode of unique cDNA molecules, comparison of TCR sequences sharing barcodes allows removal of those containing errors. Furthermore, counting the number of barcodes associated with an error-corrected TCR (after clustering to remove probable errors in the barcode) provides an estimate of cDNA frequency. Barcodes were clustered if they possessed an edit distance of three or fewer to all other sequences in that cluster. Error- and frequencycorrected data is said to be 'collapsed', therefore the script that performs this function is named **CollapseTCRs**. The development of this process is described in greater detail later, in Chapter 5.

By default the 'OmitN' option in the vDCR code is set to 'False' (unlike in standard Decombinator), thus allowing reads through even if they contain ambiguous base calls. Reads that contain Ns in their inter-tag region (or barcode sequence, where applicable) are instead filtered after Decombining by using bash/awk commands. Reads with erroneously long inter-tag distance greater than 130 nucleotides were removed at the same time. If the data was not generated using a barcoded protocol, then the following command was used, which only filters out reads with ambiguous base call in the inter tag region (the seventh field of the n12 files):

awk '{FS=`, `} {if (\$7~/N/){next} else{ if (length(\$7) < 130){print}}}' \
FILE.n12 > temp && mv temp FILE.n12

If the data had been generated using a barcoded protocol, then the command was altered to also filter out n12 entries containing an ambiguous base in their barcode field as well (the ninth field), as this would impact upon the clustering process:

awk '{FS=`, `} {if (\$7~/N/ || \$9~/N/) {next} else{ if (length(\$7) < 130) \
 {print}}' FILE.n12 > temp && mv temp FILE.n12

Note that in the BV analysis of the cord blood data, low-frequency rearrangements found to not be using the primed V regions (as a result of off-target priming or misassignment) were similarly filtered out prior to further analysis.

## 2.4.4 TCR productivity and CDR3 amino acid determination

TCR productivity – i.e. the likelihood of producing polypeptide chains that could functionally serve in a surface-expressed receptor complex – was determined for raw or collapsed Decombinator identifier files. This is achieved by inferring the complete nucleotide sequence, determining the reading frame, translating the sequence and searching for the presence of particular TCR motifs in the appropriate locations. These tasks are all performed by a single Python script (**CDR3ulator**), which makes use of functions adapted from equivalent scripts written for similar purposes by other group members Dr. Niclas Thomas and Katharine Best. In order to be considered productive a TCR classifier must produce a sequence that meets the required criteria:

• Sequences must be in frame.

- This is determined by the number of nucleotides from the first base of the V region to the last base of the J region minus one being a multiple of three (as the last nucleotide of the J region contributes to the first codon of the constant region).
- Sequence must contain no stop codons.
  - Using Biopython, this equates to discounting any translated sequences that contain the '\*' character.
- Sequences must contain the conserved CDR3 cysteine residue.
  - This is the second conserved cysteine residue that marks the start of the CDR3 sequence (113).
- Sequences must contain the conserved FGXG motif.
  - This is the motif that ends the CDR3 sequence (113). Note that in human  $\alpha\beta$  TCR chains a number of germline J genes predicted to be functional diverge from this motif by one residue; in order to be correctly assigned each gene must contain the appropriate motif.

Productive and non-productive reads are then written into separate files. In this thesis, I have used the standard CDR3 definition that runs inclusively from the second conserved cysteine to the phenylalanine in the conserved FGXG motif. Note that these statuses are predicted, and may not reflect the true productiveness of a given cDNA. It is possible that transcripts assigned as productive contain non-sequenced post-transcriptional modifications that prevent their function, or that some non-productive sequences are able to contribute to some biological function.

## 2.4.5 Sorting of poly-T homopolymer containing reads

Data produced using the poly-T RACE protocol (described above in Section 2.3.3) were sorted into reads that did or did not contain poly-T sequences in their 5' (i.e. those which were sequenced in the same orientation as transcription or not) through use of a custom Python script (**SortToPolyTREGEX**). This script makes use of the re package to search for various regular expressions (regex) that look for the presence of poly-T tracts. Regex analysis is required as polymerase slippage and homopolymer-induced error results in exact string matching of long homopolymers being unfeasible. Three regexes were employed: `TTTTTTT', `TT+[ACGN]T+' and `TT+[ACGN]T\*[ACGN]T+'. The search was configured to detect any poly-T tract of eight or more bases in the first 70 basecalls of a read, making allowances for up to two non-T bases to be included. Reads were sorted into one of two output files depending on the presence of poly-T homopolymers being detected in their 5'.

## 2.4.6 Splicing analysis

## 2.4.6.1 Mapping of reads to TCR germline loci

Alternative splice events were detected by mapping reads to the germline genomic TRA/D and TRB loci. The TRA/D germline sequence consisted of nucleotides 3080000-4030000 from the GRCh37.p5 primary assembly of the *Homo sapiens* chromosome 14. The TRB sequence used was the NG\_001333.2 version of the curated locus sequence from chromosome 7, which is 684973 bp long (accession number 114841177).

Reads were mapped against these references using the RNA sequence aligner STAR (114). Bam files produced by STAR (v2.3.0) were sorted and indexed using SAMtools (v0.1.19) (115), and visualised using the Broad Institute's Integrative Genomics Viewer (IGV, v2.3.32) (116). Mapped reads were visually compared against mapped germline sequences for L-PART1, L-PART2 as well as V and J genes, downloaded from IMGT/GENE-DB (19).

#### 2.4.6.2 Quantification of intra-V gene splicing events

Having identified intra-V splice sites through manual inspection of one donor's reads, these sequences were searched for in a number of donors. Paired-end reads were first merged using FLASH (117), which finds those which overlap in their 3' sequences, reverse complements one and combines them, producing reads with the longest possible coverage over the V region. This process typically merged around one third of all reads per file of data produced using the BLR protocol.

Merged reads were then analysed with a single Python script (**SpliceAnalysis**) that performed a number of tasks, including: collapsing, to reduce errors; Decombining and translation, to assess the presence and productive status of rearrangements, and checking for the presence of previously identified intra-V splice events. Collapsing of merged reads occurs slightly differently to collapsing barcoded Decombinator output, as it is the whole read that is being collapsed, not just the section that has been determined to cover the rearrangement. As before, the most common sequence within a barcode is taken as the likely-actual sequence, and any others but error-containing PCR-descendants. Only barcodes with a minimum of three separate reads are considered. If there is no sequence within a barcode that is more prevalent than the others, then a consensus sequence is created by taking the most common nucleotide at each

position of each equivalent read: if two or more nucleotides are equally present at a position then that barcode is ignored. All barcode sequences shared by a particular corrected sequence are then clustered as in CollapseTCRs to estimate the frequency of sequences. Corrected sequences were then inspected for rearrangements using the core functions of vDCR. The original, published V and J gene tag set was used, as these tend to lie closer to the recombining edge. Corrected sequences with rearranged TCRs are then translated by inference from the Decombinator classifier and predicted productivity is recorded. Amino acid sequences are then also translated from the complete actual sequence (not that inferred from the classifier). Correct reading frames were deduced by finding the frame that contains the amino acids of the constant region covered by the RC1 primer used during amplification: 'IQNPDPA' for TRAC<sup>5</sup>, 'DLKNVF' for TRBC1 and 'DLNKVF' for TRBC2. The productiveness of the transcript can therefore be inferred from the whole sequence itself, i.e. whether there is an in-frame start codon relative to the required constant region residues, and the absence of an interrupting stop codon. Sequences were also searched for the exact string matches of splice events as determined from the mapping described in Section 2.4.6.1. All findings are input to a dictionary, recording for a given corrected transcript sequence: its frequency; the presence of a rearranged TCR; the predicted CDR3 (or criteria for being deemed nonproductive); the actual open-reading frame of the complete sequence, and whether or not it contains any identified intra-V splice events.

## 2.4.7 Calculation of diversity and sharing values

#### 2.4.7.1 Gini index

The Gini index is a measure of inequality or dispersion of a distribution, often employed in economics to represent the distribution of wealth in a system. Gini indexes are calculated in reference to a Lorenz curve, where the proportion of resource allocation is plotted as a function of the proportion of the population (Figure 2.1):

$$\mathbf{G} = \frac{A}{A+B}$$

Gini index values can range from zero, representing total equality or maximum dispersion (e.g. all money is shared equally) to one, representing total inequality (e.g.

<sup>&</sup>lt;sup>5</sup>This sequence actually starts from the second amino acid of the constant region: the last nucleotide of the J gene starts the first codon of the constant region, and TRAJ genes do not all end in the same nucleotide, thus after splicing the first codon of the constant region can encode one of four different amino acids.



Figure 2.1: Gini index calculation from the Lorenz curve. The Gini index is defined as the ratio between the area under the Lorenz curve (LC) and the total area, under the line of equality (LE, or perfect distribution line).

one entity has all of the money). Gini indexes were calculated using the ineq library in  $\ensuremath{\mathtt{R}}$  .

## 2.4.7.2 Shannon entropy

Shannon entropy is a measure borrowed from information theory, where it is used to quantify the 'entropy' of a system. It can also be considered to be measuring either the uncertainty or information content, where higher entropies correspond to more uncertainty or information. Consider a comparison of a fair dice roll and a fair coin toss: there is more uncertainty as to the output of the die than the coin, as there are more options (more entropy). The greater array of possible values produced from an equal number of rolls compared to tosses similarly could contain more information ( $X^6$ as opposed to  $X^2$ ). When Pi is the probability of instance *i* occurring in a distribution (and *b* is the base of the logarithm), the Shannon entropy is defined as:

$$H = -\sum_{i} P_i \log_b P_i$$

Shannon entropies were calculated using the entropy library in R, using the base 2 logarithm to produce output in bits.

#### 2.4.7.3 Jaccard index

The Jaccard index is a normalised measure of sharing between two sets of items. It is defined as the size of the intersection of those two sets (i.e. how many unique items are found in both) divided by the size of the union (how many different unique items are found in total between the two sets):

$$J(A,B) = \frac{|A \cap B|}{|A \cup B|}$$

In this thesis, Jaccard indexes were calculated using a custom function in R, using the default functions intersect and unique.

#### 2.4.7.4 Jensen-Shannon divergence

The Jensen-Shannon divergence (JSD) value is a modification of the Kullback-Leibler divergence (KLD) – to make it symmetric and produce finite values – that relates to the amount by which two distributions (P and Q) differ:

$$I(P||Q) = \frac{1}{2} \left[ D\left(P||\frac{P+Q}{2}\right) + D\left(Q||\frac{P+Q}{2}\right) \right]$$

Here D represents the KLD using the base 2 logarithm, defined as:

$$D(P||Q) = \sum_{i} P(i) \log \frac{P(i)}{Q(i)}$$

The KLD is therefore the logged difference between distributions P and Q, using P as the expected distribution. By taking the average of KLDs calculated in either orientation, the measure becomes symmetric. In this work, JSD was calculated in Python using a function written by Jonathan Friedman (118).

# 2.5 Commitment to open science

I adhere to the principles of open science, under the expectation that verification, reproduction and proper dissemination of research can only be properly achieved through adequate sharing of the processes and results involved. Accordingly, the following components of the content of this thesis have been made publicly available.

## 2.5.1 Data deposition

The vast majority of the raw sequence data produced and analysed in this data has been uploaded to the NCBI Sequence Read Archive (SRA). The SRA accession numbers of these datasets are:

- $\bullet\,$  Simple V amplification HiSeq data:  ${\bf SRP055031}$
- Poly-T RACE 454 data: **SRP055040**
- Barcoded V MiSeq data:  ${\bf SRP055080}$
- Barcoded ligation RACE MiSeq data: SRP045430

## 2.5.2 Source code

The scripts required for the standard analysis of TCRs from high-throughput sequencing data obtained using these protocols can be downloaded from a freely accessible GitHub repository at https://github.com/JamieHeather/tcr-analysis (119). This repository contains vDCR, CDR3ulator, the extended tag sets as well as the associated demultiplexing and basic plotting scripts.

Another repository, available from https://github.com/JamieHeather/ThesisScripts was generated to contain a sample of the more specific analyses that were used during this thesis. In addition to the scripts listed in bold in preceding sections, this repository also contains scripts that:

- Randomly draw a given number of sequences (either DCR classifiers or CDR3s) from a larger file, for the purpose of size-matching data for fair comparison (SizeMatch and CDRSizeMatch)
- Construct FASTQ files from the inter-tag region data contained in n12 files, and assess the average quality score and converted error probabilities (FastqInterTag, PrintTotalAvgQuals and PrintErrorProbabilities)
- Take two DCR index files and output a comma-delimited table, giving the proportional frequency of each TCR sequence in both of the two files (**ProportionPrePlot**), allowing for convenient correlation measurements
- Quantify and compare the presence of published TCRs sequence in multiple samples (HIVPublished\_CDR3s and InvariantTCR)

- Measure TCR repertoire population level differences across multiple samples, such as publicness, translational convergence and similarity of TCR gene usage (Publicness, HIVTranslational\_Convergence and LigRaceJSDs)
- Quantify the presence of certain genes or subsequences (GeneUsage and FindDs)
- Mix TCR classifiers from multiple subsets to generate *in silico* simulated repertoires of mixes populations (HIVCD4\_CD8\_Mixing\_CDR3s)

Note that the sequences of invariant and antigen specific TCRs harvested from the literature are contained within the HIVPublished\_CDR3s code.

# Exploring healthy T-cell receptor repertoires

# 3.1 The global TCR repertoire at steady state

3

In this section I explore the data generated from sequencing ten healthy donors' whole peripheral blood repertoires using the barcoded ligation RACE strategy (or 'BLR')<sup>1</sup>. In these experiments a fraction of RNA extracted from a 2.5 ml sample of whole blood served as the template for reverse transcription, TCR amplification and sequencing. Sequences produced were analysed using Decombinator, the TCR analysis software developed in our group, and further error-corrected using CollapseTCR. The error-corrected TCR classifiers produced by Decombinator ('DCRs') – which are equivalent to unique nucleotide sequences – are the basis for the following analyses.

#### 3.1.1 Clonal distribution of the healthy TCR repertoire

The first aspect of the repertoire considered is its basic clonal composition. The distribution of TCR rearrangement frequencies is an intrinsically difficult property to visualise, as the vast majority of unique DNA recombinations – 50 to 80% – only occur once in a given sample, while a small fraction of sequences are found repeated tens, hundreds, or even thousands of times (Figure 3.1A). This produces a characteristic 'L' shaped distribution, which makes the data somewhat difficult to interpret. By plotting the logged rank (with the most common rearrangement as rank one) against the logged frequency (Figure 3.1B) we observe an approximately linear relationship between the two (at least for sequences with more than five copies), indicating that a given sequence will be more frequent than the next most common sequence by a fixed proportion. Thus there are a small number of highly frequent rearrangements, and a large number of sequences that only occur very rarely in these samples.

The frequency distribution of different TCR chains relates to another property of a repertoire, that of diversity. This is thought to be a very important repertoire characteristic, as the number of different types of antigen receptor in a system should

<sup>&</sup>lt;sup>1</sup>The development of this protocol, and all others, is covered in detail later in Chapter 5.



relate to how many antigens that system can respond to. TCR receptor diversity can be measured in many different ways (120); in this thesis I make use of two indexes of diversity borrowed from other fields. The Gini index is a measure of inequality (or unevenness) frequently used in economics, as it permits an overview of the distribution of a resource such as wealth. For example, a Gini index value of one represents total inequality, in which all of the wealth belongs to one entity, while a Gini index of zero represents total equality, in which all entities possess an equal share of wealth (121). With respect to TCRs, Gini index values of zero would correspond to all rearrangements being equally present in a sample, and as oligoclonality increases – i.e. as subsets of sequences began to become more common and dominate the repertoire – values would tend towards one. Figure 3.1C shows that the  $\alpha$  and  $\beta$  repertoires of healthy donors have Gini indexes towards the low end of the scale, indicative of a generally even distribution structure. The other measure employed is Shannon entropy, borrowed from information theory. Shannon entropy is a more complicated measure that incorporates both aspects of diversity – the number and frequency of different species – to produce a value that can be interpreted as the information content or complexity of a system (122). More species (or TCRs) being present increases the total information content, as does increasing the evenness. Thus the higher the Shannon entropy the more information content a system has, and the harder it is to predict to what species additional observations are likely to belong. Conversely, a distribution where all species are the same has no information

Figure 3.1 (preceding page): Clonal composition of the healthy human  $\alpha\beta$  TCR **repertoire.** A: Percentages of each of the ten donor's repertoires that are accounted for by rearrangements of different frequencies (i.e. how many times that rearrangements appears). B: Logged rearrangement rank versus logged frequency reveals a linear distribution for all but the lowest frequency sequences. C: Gini indexes of the distribution of unique rearrangements of healthy donor repertoires; dotted lines connect the  $\alpha$  and  $\beta$  values of the same donor. There is no significant difference between the paired  $\alpha$  and  $\beta$  Gini index values (p-value = 0.38, Wilcoxon signed rank test). D: Shannon entropies of the healthy donor samples. There is no significant change between the  $\alpha\beta$  paired Shannon entropy values (p-value = 0.08), although the majority of donors tend to have greater values in their  $\beta$  chains. E: The proportion of unique rearrangements (i.e. ignoring the frequency of each) that are predicted to be productive (P) or non-productive (NP).  $\mathbf{F}$ : The proportions of the criteria by which the NP rearrangements shown in E were determined to be so.  $OOF = out-of-frame; OOF^*$ = out-of-frame, also containing a stop codon;  $IF^* =$  in-frame but containing a stop codon; C = lacking the conserved cysteine residue at the start of the CDR3. Note that the translation software only generates sequence up until the end of the J region, thus OOF transcripts without stop codons in this analysis could contain stop codons further downstream in the sequence. All error bars indicate standard deviation around the mean value calculated using all ten donors.

per se, and thus would have a Shannon entropy of zero. Figure 3.1D shows the Shannon entropies of the healthy donor repertoires, showing that typical healthy repertoires (of these sample sizes) possess in excess of twelve bits of information.

V(D)J recombination will not always produce a recombined gene that is capable of being translated into a functional polypeptide. Due to the random nature of the recombination process, rearrangements may fail to be in the correct frame downstream of the junction, or contain premature stop codons, preventing translation of valid TCR chains. When developing the software to translate TCR nucleotide sequences and extract CDR3 regions, I counted TCR sequences that are expected to fail to produce a valid CDR3, and are thus considered non-productive. When considering unique rearrangements (i.e. ignoring how prevalent each rearrangement is) I observed that on average 16% of all  $\alpha$  chain rearrangements and 2.6% of  $\beta$  were non-productive (Figure 3.1E). Examination of these non-productive sequences reveals that the majority of them are out-of-frame and contain stop codons (Figure 3.1F). A small number are in-frame yet contain a stop codon, and a similar number lack the conserved cysteine residue in the V gene which marks the N-terminal end of the CDR3 region. Note that there was a similar requirement for detection of the FGXG-based motif in the J gene that marks the C-terminal CDR3 end, however no sequences were deemed non-productive due to lacking this sequence. This difference is likely due to the fact that the conserved cysteine in the V is much closer to the recombining edge than the FGXG is in the J, and thus is more likely to be lost through base deletion during recombination.

## 3.1.2 V and J gene usage

Another important feature that will define a given repertoire is the frequency with which different V and J genes are used. Figure 3.2A-D shows the mean proportional usage of all the different TRAV, TRAJ, TRBV and TRBJ genes in our ten healthy donor repertoires respectively. The distributions are non-uniform; some genes are used in a far higher proportion of rearrangements than others. Figure 3.2E shows that the more a given V or J gene is used, the more deviation there is between how much different individuals use it.

During this analysis I observed a large number of unexpected recombination events. The first class of unexpected rearrangements involved detection of supposedly TCR  $\delta$  chain specific V genes rearranged onto  $\alpha$  chains, using the  $\alpha$  constant region in combination with  $\alpha$  chain J genes. Despite  $\alpha$  and  $\delta$  V genes sharing a common pool of V genes (Figure 1.2) it is currently reported that not all genes can be used by either chain: of the eight V regions known to contribute to  $\delta$  chain receptors, five are also known



to participate in  $\alpha$  chain production (termed 'TRAV/DV' genes) (123). According to IMGT, the three remaining genes – TRDV1, TRDV2 and TRDV3 – are supposedly exclusively limited to  $\delta$  chain rearrangements. However as Figure 3.2 demonstrates, all three can be found in  $\alpha$  chain recombinations, even present at greater amounts than some of the officially-recognised TRAV genes. In order to confirm these findings, I also looked for such sequences in the only other publicly available high-throughput  $\alpha$  chain data set: Figure A.1 in Appendix A shows that rearrangements using all three genes can also be found among these independently generated sequences.

The second class of unexpected TCRs are those that do not use genes designated as functional. For various reasons, several of the genes in the germline receptor gene loci are predicted to either not participate in rearrangements, or not be able to encode working polypeptides if they do (124). These non-functional germline genes come in two varieties: open-reading frame genes (ORFs), or pseudogenes. ORFs are predicted to not participate in rearrangements due to alterations in regulatory sequences, such as splice sites or recombination signal sequences, or in conserved amino acid sequences (such as the cysteine and phenylalanine residues that bound the CDR3 region), while pseudogenes are those that contain stop codons or frameshift mutations. However examples of both V and J ORFs and pseudogenes can be found for both TCR chains in the healthy donor data, with the exception TRBJ (which only has one such gene, a pseudogene). Some of these genes account for a considerable proportion of the repertoire, such as TRAJ35 and TRAJ58 (Figure 3.2C) or TRBV21-1 and TRBV23-1 (Figure 3.2D), approximately occurring more frequently than one in every 500 sequences.

In addition to monitoring how frequently each V and J gene appears, I also measured how frequently each combination of V and J gets used across our healthy donors. The mean frequencies and standard deviations are shown for  $\alpha$  and  $\beta$  in Figures 3.3 and 3.4 respectively. In terms of average  $\alpha$  chain rearrangements (Figure 3.3A) most V-J combinations account for small, roughly equivalent proportions of the repertoire, with one particular V-J pair – TRAV1-2 - TRAJ33 – being far more prevalent than any other. However comparison with the standard deviation of those averages (Figure 3.3B) reveals an increased number of mid-level hot-spots, indicating that at least in some donors certain V-J combinations are used preferentially.  $\beta$  chain VJ pairing shows

Figure 3.2 (preceding page): V and J gene usage in healthy human  $\alpha\beta$  TCR rearrangements. Mean proportional usage of the different genes for TRAV (A), TRAJ (B), TRBV (C) and TRBJ (D) in the ten healthy donors. F, ORF and P stand for functional, open-reading frame and pseudogene genes respectively, as described above. Error bars indicate standard deviation.



3.1 The global TCR repertoire at steady state

Figure 3.3: V and J gene pairing in healthy human  $\alpha$  TCR rearrangements. A: Heat-map showing the mean frequency with which the different  $\alpha$  chain V and J genes appear together in rearrangements, averaged across our ten healthy donor samples (taking the frequency of each rearrangement in each donor into account). B: Heat-map showing the standard deviation of the averages in A.



Figure 3.4: V and J gene pairing in healthy human  $\beta$  TCR rearrangements. A: Heat-map showing the mean frequency with which the different  $\beta$  chain V and J genes appear together in rearrangements, averaged across our ten healthy donor samples (taking the frequency of each rearrangement in each donor into account). B: Heat-map showing the standard deviation of the averages in **A**.

a number of hotspots in usage, particularly at the convergence of specific V and J genes which appear to be highly expressed relative to the others (e.g. TRBV20-1 - TRBJ2-1), while standard deviations appear relatively lesser in comparison. The upper limit of the scale for the  $\alpha$  chain average proportions is more than twice as much as that in  $\beta$ , indicating that the TRAV1-2 - TRAJ33 rearrangements account for a greater proportion of the  $\alpha$  repertoire than the most frequent  $\beta$  V-J pairs do in theirs.

#### 3.1.3 TRBD detection and usage

The two human TRBD genes are both very short and highly similar (Table A.1, Appendix A) and subject to nucleolytic degradation from both ends, making their assignation by the tag-based gene identification system of  $Decombinator^2$  not feasible. I developed additional methods to ascertain what the measurable contribution of the different D genes to  $\beta$  chain recombinations was, by searching the Decombinator classifier insert string – the nucleotides sequence found between the end of the V gene and the start of the J gene – for residual TRBD sequences. Figure 3.5A shows what percentage of total rearrangements contain detectable sequences from either TRBD gene. This search was performed multiple times, each time with an increased minimum length of matching sequence required. By applying the same search to the inserts of  $\alpha$  chain rearrangements (which should not contain TRBD sequences except by chance) we can get an estimate of the false positive rate. For example, if we look in  $\beta$  chain data for matches of only five nucleotides or longer we observe that seemingly 70% of reads contain TRBD sequences; however by the same criteria, almost 8% of  $\alpha$  reads appear to as well. As the threshold increases the rate of detection falls – as you would expect, given that there will be fewer sequences of the required length due to nucleolytic 'nibbling' – along with the false positive  $\alpha$  assignation rate. This method represents an improvement in measuring TRBD usage over existing techniques, both in the inclusion of false positive detection as well as the separation of different length matching; existing published analyses often just use thresholds of five (125) or six (126, 127) nucleotides.

One previous study additionally reported the presence of low-frequency tandem TRBD gene usage, producing VDDJ transcripts, which were detected both in their own data (127) and a previously published dataset (77). Figure 3.5B shows the extent of such rearrangements in our data. We see at least an order of magnitude less fusion D-D usage than in the Liu *et al* report (an average 0.38% of rearrangements using a threshold of six nucleotides, as opposed to the lower bound of 1.73% stated in their

<sup>&</sup>lt;sup> $^{2}$ </sup>This system is discussed in detail later in Section 5.3.1.



publication). This is likely due to the current analysis requiring exact sequence matches, while that previously published allowed mismatches.

I also assessed the extent to which each TRBD gene associated with different TRBV and TRBJ genes. Figure 3.5C and D show that despite large deviations, there are both V and J genes which significantly pair more frequently with a particular  $\beta$  D gene. Of particular note is the separation of TRBD-TRBJ gene pairing between the two different TRBDJ gene clusters (Figure 3.5D). Despite pairing far less frequently with the TRBD1 cluster (TRBJ1-1 to TRBJ1-6, see Figure 1.1), TRBD2 does indeed pair with each of the TRBJ genes at some level. Extending this analysis, Figure 3.5E quantifies the contribution that each of the individual TRBD genes makes to the rearrangements that have detectable D regions, and how long the sequence match is. The majority of rearrangements with shorter D regions used TRBD1, while TRBD2 accounts for a greater proportion of the longer matches. This is not unexpected: the germline TRBD2 sequence is longer than TRBD1 (16 versus 12 nucleotides). Across all ten donors we note that TRBD1 is cumulatively used more frequently, accounting for 64% of all detectable TRBD usage. This is in contrast with a previous report of a more equitable distribution (75). This value is likely to be extremely cohort-specific, given that there is some evidence to suggest such patterns are influenced by both age and sex (128).

Figure 3.5 (preceding page): TRBD detection and usage in healthy human TCR  $\beta$  rearrangements. A: Mean percentages of total healthy volunteer  $\beta$  chain rearrangements that contain exact contiguous matches to any TRBD gene sequences (see Table A.1, Appendix A). TRBD search was performing using an increasing threshold for the shortest sequence match to be counted. Each search was repeated using the  $\alpha$  chain Decombinator rearrangements of the same donors as a specificity control, as these should only contain TRBD sequences as a result of non-templated nucleotide insertions and deletions. B: Mean percentages of total  $\beta$  chain rearrangements that contain tandem TRBD sequences, i.e. both a TRBD1 and TRBD2 sequence. C and D: Proportions of TRBV and TRBJ genes, respectively, that are found in combination with particular TRBD genes, for those rearrangements where one D gene was unambiguously assigned. E: The length distribution profiles for all of the TRBD matches given a particular match length threshold. C, D and E were all performed using a TRBD search with a minimum match length of six nucleotides. TRBD2 results include sequences that map to either allele (TRBD2\*01 and TRBD2\*02). All error bars show standard deviation. Asterisks indicate significance by Wilcoxon rank sum tests: \* for p-values  $\leq 0.05$ , \*\* for  $\leq 0.01$  and \*\*\* for  $\leq 0.001$ .



#### 3.1.4 TCR sharing and publicness

An emerging aspect of TCR repertoire research is the extent to which different TCR receptor sequences are shared between individuals. This phenomena is manifest at multiple levels, from any two individuals sharing a certain proportion of their sequences, up to so-called 'public' sequences, that are found across multiple individuals (129, 130, 131). I assessed the extent of such sequences in my healthy donor repertoire data using several techniques.

I first adopted a TCR-centric view, and measured how many productive sequences appear in what number of donors. This analysis was repeated at three levels of TCR sequence, considering: the nucleotide sequence level, represented by the Decombinator classifier (or 'DCR'); the CDR3 alone, and finally the CDR3 in combination with the V and J genes used in that rearrangement ('VJ-CDR3'). Figure 3.6A shows what we might expect; as we increase the number of donors we find fewer sequences that are shared across that many (for example, there are approximately  $1 \times 10^5 \beta$  chain DCR sequences that appear exclusively in one donor, but only one sequence that is shared by five donors). Therefore it may be more useful to consider 'publicness' as a scalar rather than a binary property. There is a general trend among the different types of sequence, where VJ-CDR3s are more public than DCR classifiers, and CDR3s are yet still more public. This may be explained by the possible convergence between these levels: multiple nucleotide sequences (DCRs) can encode the same VJ-CDR3, and the same CDR3 can be made in multiple V-J gene contexts.

I next took a donor-centric view, by performing pairwise comparisons of every combination of two donors and measuring by how much their repertoires overlap, using the Jaccard index (Figure 3.6B). This was performed both on complete repertoire

Figure 3.6 (preceding page): TCR sharing and publicness of productive rearrangements. A: Publicness of productive TCR sequences in healthy human donor repertoires, i.e. how many unique sequences were found only in one donor, or two donors, or three and so on. DCR = Decombinator classifier, which is representative of unique nucleotide sequences. VJ-CDR3 = a CDR3 amino acid sequence in combination with the V and J genes with which that CDR3 was found. CDR3 = the CDR3 region alone, from the conserved cysteine in the V region to the conserved phenylalanine in the J region. B: Sharing of unique sequences between pairwise comparisons of all donors, measured using the Jaccard index, for both complete and subsampled repertoires. Subsampling involves randomly selecting a fixed number of samples from each donor's repertoire before comparison.  $\alpha$  chain repertoires were subsampled to 6000 unique sequences,  $\beta$  chain repertoires to 9000 (reflecting the lower bounds of number of sequences found in these groups). C: Frequencies of the CDR3s summarised in A. Horizontal dashes indicate group means. samples and on subsamples, where a fixed and equal number of rearrangements were randomly selected from each donor before comparison, in order to control for differences in sequencing depth. We again see the sharing increase as we consider the different sequence types (CDR3 > VJ-CDR3 > DCR). As with Figure 3.6A, we also see that  $\alpha$  chains are also far more shared than  $\beta$  chains. This probably reflects both the relative decrease in  $\alpha$  chain diversity (Figure 3.1D) and the fact that there exist T-cell subpopulations that make use of so-called 'invariant' TCR  $\alpha$  chains (discussed in more detail later in Section 3.1.5).

Shared and public clones are also reported to be present in higher frequencies in the repertoire (129). I investigated this in my data by plotting the proportional frequency of each CDR3 that was found shared among each number of donors (Figure 3.6C). Average values remain low in both chains up until the highest possible number of donors, indicating that the more public clonotypes are indeed more likely to be higher frequency than those that are less shared. However by looking at the underlying distributions we do see that there are equally high private clones (found in one or two donors), but these are just outweighed by the relative surplus of private clones that only occur at very low frequencies.

This analysis also revealed that there are non-productive nucleotide rearrangements (those expected to not translate correctly) that are also shared among multiple individuals (Figure 3.7A). Despite only a fraction of unique rearrangements being non-productive, there were still  $\alpha$  chain sequences that were shared among all ten individuals, and  $\beta$  sequences that were found in four different donors. Pairwise comparisons revealed that between any two donors, non-productive sequences are actually significantly more shared than productive sequences, in either chain (Figure 3.7B). As with productive  $\alpha$  chains, non-productive  $\alpha$  sequences that were shared among more people showed larger mean frequencies than those shared among fewer (Figure 3.7C). Remarkably, for those rearrangements that were common to either nine or ten individuals, the non-productive recombinations averaged larger clone sizes than the productive.

Inspection of the overlap in repertoires between individuals (intra-donor) provides us information on TCR sharing and publicness. By taking multiple bleeds from one individual and comparing these (inter-donor), we are able to obtain information about the dynamics of the repertoire within a person. Four of our healthy donors were bled twice, three months apart, and processed. As before, I then calculated the Jaccard indexes of intra- and inter-donor pairwise repertoire comparisons (Figure 3.8A). As might be expected, two repertoires from the same donor share more TCR sequences



than repertoires from two different donors. I did not observe non-productive rearrangements being more shared than productive for intra-donor as we do for inter-individual comparisons.

By plotting the proportional frequencies of rearrangements in the first bleed against those in the second we can see how stable the clonotypic hierarchy is within individuals. The examples shown in Figure 3.8B demonstrate the donors that had the best and worst correlation by  $\mathbb{R}^2$  value. Even for the samples with a poorer fit by linear regression show a clear pattern; low-frequency sequences (below  $1 \times 10^{-3}$ , or one in every 1000 recombinations) display little correlation and are often simply not present in the other bleed, while higher-frequency sequences match exceedingly well. This might be explained by lower frequency clones being less likely to be captured by sequential sampling events.

## 3.1.5 Invariant TCR detection

There exist various subpopulations of phenotypically distinct  $\alpha\beta$  T-cells that make use of so-called 'invariant' TCRs, which are composed of near-identical  $\alpha$  chains and a small pool of  $\beta$  V genes. Currently, we know of three such cell types, which use these restricted diversity receptors to recognise non-canonical ligands bound to various MHC-like molecules. Discovered first, invariant natural killer T-cells (NKT) respond to glycolipid antigens bound to CD1d, using a TRAV10 - TRAJ18 containing TCR (132, 133, 134). Mucosal-associated invariant T-cells (MAIT) are MR-1 restricted and recognise microbial vitamin metabolites with TCRs using TRAV1-2 primarily in combination with TRAJ33, although variants using TRAJ12 and TRAJ20 have recently been described (135, 136, 137, 138). The most recently categorised population are the germline-encoded mycolyl lipid-reactive (GEM) T-cells, which use TCRs containing a TRAV1-2 - TRAJ9  $\alpha$  chain, and recognise mycobacterial lipids presented by CD1b (139, 140). Having noticed that two types of invariant TCR-bearing cells make use

Figure 3.7 (preceding page): Non-productive TCR sharing and publicness. A: Publicness of non-productive TCR rearrangements, inferred from Decombinator indexes (or DCRs) that are not expected to encode a productive TCR polypeptide. P = productive, NP = non-productive. B: Sharing of DCRs that describe either productive or non-productive rearrangements, for either whole or subsampled repertoires. Samples were subsampled down to the lower bound of NP samples, which were 1000 for  $\alpha$  chains and 260 for beta chains. Note that there are no subsampled P DCR results for beta, as all of these comparisons resulted in no shared clones. C: Frequencies of P and NP DCRs summarised in A. Horizontal dashes indicate group means.



Figure 3.8: Intra-donor TCR sharing. A: Overlap in DCR repertoires of the four healty donors from whom we took two peripheral blood RNA samples, three months apart, measured by Jaccard index. Both productive and non-productive (P and NP) DCR were compared, for both complete and subsampled repertoires. 'Intra' indicates comparisons of the two bleeds from one individual, while 'inter' indicates comparison of two bleeds from different individuals (but taken at the same time).  $\alpha$  repertoires were sampled down to 1,300 rearrangements, while  $\beta$  repertoires were subsampled to 300. B: Example plots demonstrating the intra-individuals concordance of rearrangement frequencies.  $\alpha$  and  $\beta$  repertoires are shown for the individuals with the highest and lowest degrees of correlation by R<sup>2</sup> value (HV01 and HV03; higher and lower two plots respectively). Note that the first and second sample bleeds were all taken at the same times of the year for each donor.



Figure 3.9: Detection of known and putative invariant  $\alpha$  chain TCRs. A: Number of published invariant TCR VJ-CDR3s that were found in our healthy donor peripheral blood repertoires. The sources of these sequences are listed in Table A.2. B: The size of the published CDR3s and VJ-CDR3s searched for in our donors, as proportions of the repertoires that they were found in. Note that the 'Putative' group shown in **A** does not include the putative sequences derived from the healthy donors presented later in this chapter.

of the same  $\alpha$  V gene (TRAV1-2), the group that discovered GEM cells have recently performed targeted deep-sequencing of  $\alpha$  chains that specifically use this gene (141). This process has generated a number of novel sequences that are shared among multiple individuals, which they claim represent potential invariant T-cell subpopulations. I harvested published sequences for these invariant TCRs and the putative invariant shared TRAV1-2 sequences (referenced in Table A.2), and looked for them in the healthy donor repertoires in the context of the correct V and J genes.

As Figure 3.9A shows, we are able to detect invariant  $\alpha$  chains belonging to each population at the VJ-CDR3 level to some extent. Most of these TCRs are only detectable in a small subset of donors. Only two of the five described GEM VJ-CDR3 sequences were found in this data, in different donors, with a slightly greater number of NKT and putative TRAV1-2 invariant sequences in more donors. This is likely due the prevalence of these cell types: human NKT cells account for approximately 0.2% of T-cells in blood (142), while GEM are two orders of magnitude less frequent (140). However, far more sequences derived from MAIT cells are found, with at least ten VJ-CDR3s being detected in each of the donors. While this search was repeated for  $\beta$  chains, this search yielded almost no results; only two MAIT  $\beta$  CDR3 sequences (lacking V and J gene context) were detectable, at low frequency.

Figure 3.9B shows how prevalent each detected invariant TCR sequence was in the repertoire file in which it was found. GEM sequences were only identified close to the limit of detection for this assay, with NKT and the putative sequences from Van Schaik *et al* being slightly more frequent. MAIT cell sequences are detected far more often and at higher frequency, with one sequence making up 3% of all sequences in the  $\alpha$  repertoire it was found in. MAIT cell sequences reach such high proportions in blood that the contribution of their invariant  $\alpha$  chain is readily apparent even by looking at V-J gene pairing frequencies, as was was shown earlier in Figure 3.3 (the most frequent pairing is TRAV1-2 - TRAJ33, the canonical MAIT combination). This is consistent with existing flow cytometry data showing that MAIT cells can be exceedingly common in blood, sometimes accounting for upwards of 10% of CD8<sup>+</sup> T-cells (143).

I also used my data to identity potential invariant TCRs based on degree of publicness. We have already observed that there are a number of  $\alpha$  chain VJ-CDR3 sequences shared among the majority of the healthy cohort (Figure 3.6A). I took the sequences that were found in seven or more donors, and subtracted any that matched a preexisting known or potential invariant population. This left 40 sequences (listed in Table A.3, Appendix A). One sequence was found in all ten donors, using TRAV13-1 and TRAJ6, bearing the CDR3 CAASKGGSYIPTF. This does not appear to be any kind of contamination, as this same sequence is generated using different nucleotide sequences in different donors. Searching for these sequences in the data produced a large number of hits, at greater frequencies than all other groups apart from MAIT (which might be expected, seeing as this list was defined by being common in this dataset).

## 3.1.6 Section discussion

TCR repertoires have a number of properties by which they may be characterised. Perhaps the simplest attribute to measure is the spread of frequencies of different rearrangements. The resultant distribution is characterised by an abundance of very low frequency sequences, and a small number of more common recombinations. Description of the frequency landscape of recombinations in healthy donors may function as a benchmark against which to assess samples with expected repertoire perturbations. When doing so, we must bear in mind that approaches that start with an RNA substrate (such as the one used in this thesis) do not allow one to infer a direct correlation between sequence frequency and cell counts, as different cells may express different amounts.

The presence of non-productive (NP) rearrangements detected in this RNA-derived data may initially seem surprising, given that mechanisms exist to downregulate such sequences. Various epigenetic systems mediate allelic exclusion of chromosomes that do not contain productively rearranged alleles (144, 145, 146, 147), and any transcripts that are made from such rearrangements are thought to be efficiently downregulated by nonsense-mediated decay (NMD) (148, 149). Therefore NP reads potentially just represent the 'leakiness' of the system, which seems fairly considerable in the  $\alpha$  chain data. Allelic exclusion has been reported to be stronger at the TRB locus, a fact which can be inferred from the relative ability to find cells transcribing two productively rearranged  $\alpha$  chains (72, 150). The majority of observed NP sequences from either chain contain premature stop codons, which are thought to be the strongest signal for downregulation by NMD (151). These two findings might seem incongruent, but likely reflect that the majority of NP reads would be expected to contain stop codons.

These data also demonstrate that some supposedly non-functional germline TCR genes contribute to relatively high frequency rearrangements. This does not necessarily correspond to their involvement in functioning TCR heterodimers: indeed some of the pseudogenes would be expected to not be able to produce any peptide at all. However I believe that there is a strong case for the official status of several of these genes to at least be reconsidered, if not reclassified. For instance, the frequently-found gene TRAJ58 is classified as an ORF as it contains an in-frame stop codon (152); however

as this resides towards the very 5' of the gene it can very easily be removed by nucleolytic nibbling during recombination. This allows it to produce perfectly theoretically functional rearrangements as appear in this data, as well as in several published reports that selected T-cells based on TCR binding (153, 154, 155). Another oft-detected ORF in our data was TRAJ35 (which is classified as such due to the presence of a cysteine in place of the phenylalanine in its FGXG motif) is found in valid rearrangements in several papers (156, 157, 158). TRDV1 particularly, which is shown to be consistently used at reasonably high frequencies in this data, has been found to be rearranged to a variety of TRAJ genes by a number of groups, some of whom detected surface expressed protein (127, 159, 160, 161, 162, 163, 164, 165, 166). The most recent paper to make note of this claims that TRDV1 is used with sufficient frequency and breadth of TRAJ genes that it merits a change of name to TRAV42/DV1 (127) (although this would not fit current IMGT numbering conventions). It even appears that IMGT is aware of TRDV1 recombining with at least one TRAJ gene (167), but does not count this as a dual TRAV/DV usage V gene (168), stating:

"Although the TRDV1 gene was also found rearranged to a TRAJ gene and therefore expressed with the TRAC gene ... it is considered as a TRDV gene."

Even recombinations using genes which would prevent translation still contain information, even if they arise due to some basal transcriptional noise of the system, and thus are worthy of inclusion at the primary stages of TCR analysis. For instance, to the best of my knowledge this thesis is the first report of recombination between TRDV2 and TRDV3 with any TRAJ genes in a healthy donor's repertoire<sup>3</sup>. I predict that comparably high-resolution investigation of  $\delta$  chain TCR repertoires would find V genes currently considered to be purely involved in  $\alpha$  chain recombination contributing to TRDJ- and TRDC-containing  $\delta$  chain rearrangements.

I have also demonstrated that  $\beta$  chain Decombinator classifier insert sequences can be searched to permit detection of TRBD genes. Due to the nature of these genes, they are often impossible to detect or assign reliably. One gains most sensitivity by using shorter match length thresholds, but this reduces the specificity of the search as shorter sequences are more readily produced by chance. This is evidenced by the presence of false positive TRBD hits in  $\alpha$  chain sequences. Even taking these limitations

 $<sup>^{3}</sup>$ The only papers I have found to describe TRDV2-TRAJ recombinations – which remarkably even included TRDD genes – did so in cells derived from leukaemia patients, and were predicted to not produce functional protein (169, 170).

into account, these results demonstrate that one is able to extract some information regarding the recombination of TRBD genes in data generated using the pipeline in this thesis. Doing so I have repeated some notable findings from other studies, such as the pairing of TRBD2 with TRBJ genes from the TRBD1 cluster, and preferential pairing between TRBV and TRBD genes. The presence of VDDJ rearrangements is particularly interesting, as among T-cells these are often thought to be the sole province of  $\delta$  chains, due to the configuration of the RSS around the D genes (171, 172). This raises the possibility of there also being direct VJ  $\beta$  chain rearrangements existing (173). Such rearrangements would however be hard to distinguish from genuine VDJ sequences that just had heavily deleted TRBD genes. While assignation of residual D gene nucleotides is not a requirement for the majority of analyses I describe in this thesis, such efforts would be of great importance in experiments inferring the mechanisms of recombination from sequence data.

There is a growing body of work describing sharing, 'publicness' and bias in TCR repertoires. Since the first descriptions of public TCRs recognising specific influenza, lysozyme and Epstein-Barr virus antigens (174, 175, 176), such responses have been discovered in many reactions across all species studied, prompting a great deal of discussion (177, 178, 179, 180). Turner *et al* have proposed a system of classification for the different types of bias observed in a response to an antigen (177). Type I bias describes responses that deploy multiple TCRs all using the same V gene, type II bias involves TCRs that share common amino acid motifs or features in their CDR3s, and type III bias denotes the sharing of actual nucleotide sequences. Miles *et al* later proposed an additional class, type IV bias, which describes those TCRs that fall between types II and III; such sequences share V gene identity and a significant proportion of matching CDR3 sequence, but differ in a small number of residues (179). While I am not presenting any antigen-specific data in this thesis, we can still infer the existence of each of these levels of bias in this global repertoire data by merit of the shared TCR features and sequences.

One surprising result from the publicness analyses described in this chapter is the extent to which non-productive (NP) rearrangements are shared between people. While infrequently detected (for reasons discussed in Section 3.1.1), on a per-TCR basis NP sequences were more likely to be shared between people than those that were productively rearranged. I propose that these data are consistent with the hypothesis that TCR sharing occurs as a function of generation probability, i.e. these TCRs are for some reason more likely to be made in multiple individuals (131). In this model, TCRs are produced at different efficiencies, therefore those that are generated most easily or most

often are those most likely to be shared. Productively rearranged receptors, regardless of potential publicness, will be subject to thymic selection, thus a portion of these will not be detected in the periphery. Non-productively rearranged receptors however do not contribute to ligand recognition and therefore cannot be selected against, therefore it is reasonable to assume they will exist in different individuals at some rate more proportional to their production. NP sequences are therefore more shared than productive as they have not had some potential public sequences removed. This higher level of sharing is only manifest between individuals; NP TCRs were not more likely to be found in two bleeds of the same person taken several months apart. This is consistent with the idea that we are observing NP TCRs at some rate proportional to their production; those we see at relatively high-frequencies being retained have likely been generated in thymocytes whose functional TCR chains have promoted their expansion.

These data do not however provide information as to the mechanism of unequal TCR production frequencies. One of the leading models for explaining public TCR responses is that of convergent recombination, which postulates that such sequences arise by merit of multiple recombination events producing the same nucleotide sequence (178, 181, 182). However one group has applied various statistical analyses to a selection of NP TCR sequences (obtained by sequencing from genomic DNA) and found that convergent recombination does not explain sharing, lending weight to the possibility that biases in the recombination machinery are instead responsible for production of public sequences (183, 184).

Related to measuring publicness is the detection of sequences known to exist in the non-pMHC-restricted T-cell subsets that express relatively invariant TCRs<sup>4</sup>. In this study, working from a fraction of RNA extracted from a 2-3ml sample of peripheral blood, the sequences we can most readily detect are those belonging to MAIT cells, which are highly prevalent cells in human blood<sup>5</sup>. However if other populations were a focus of research one could increase their detection through sequencing RNA either from larger volumes of blood, sorting cells, or by isolating RNA from other tissues in which these cells reside.

One exciting possibility offered by repertoire sequencing is the identification of potential unknown invariant populations. Van Schaik *et al* recently reported just such an exercise, looking for sequences that were highly shared in TRAV1-2<sup>+</sup> cells from three

<sup>&</sup>lt;sup>4</sup>However seeing as each so-called 'invariant' TCR is actually a pool of several similar but nonidentical sequences, the field may at some point wish to address this choice of word.

<sup>&</sup>lt;sup>5</sup>Note that NKT cells are more prevalent in mice than in humans (185), while MAIT cells are more prevalent in humans than in mice (135), so this finding is likely species-specific.

or more donors (141). They found a number of sequences, some of which were identifiable in this data, with more regularity than seen for iNKT or GEM cell sequences. An analogous search for highly shared sequences in my data, found a variety of highly shared sequences, one of which was present in all ten donors. Without any evidence of a phenotypically unique cell subset possessed of non-pMHC restriction we cannot differentiate these sequences from just highly shared clonotypes, perhaps representing public responses against common antigens. The line between invariance and type IV TCR bias will be a technologically difficult one to draw. Characterisation of cells using single-cell RNA sequencing data, permitting both identification of TCR chains and expression of lineage-specific markers, could represent a sequencing based solution to more confidently assert the existance of such cells. As for presentation, there are a great many MHC and MHC-like genes that could be responsible for displaying antigens to such hypothetical subsets. The original draft of the human MHC predicted 128 different protein-coding genes (186), while later work – which counted a larger portion of chromosome 6 – raised this estimate to 252, of which roughly 70 are predicted to have immune functions (187). A number of class I-like genes additionally exist outside of the MHC (188) which might serve in such a role. Indeed, all of the molecules that we currently know to display antigens to invariant T-cell populations – CD1 and MR1 - fall into this category, with their genes located on chromosome 1 (189, 190).

High-throughput sequencing is therefore able to extract and quantitate a number of diverse inter- and intra-individual TCR repertoire features, working from a small sample of blood.

# 3.2 Subset-specific TCR repertoire findings

In addition to investigating the global repertoire of peripheral blood, I also undertook a number of trial experiments to assess features of TCR repertoires of particular cell subsets. Herein I additionally explore data produced using different protocols, which only amplify rearrangements using a limited selection of V genes.

## 3.2.1 CD4 and CD8 TCR repertoires

T-cells can be divided into two major functional subsets – T helper cells or cytotoxic Tlymphocytes (CTLs) – based on the expression of either the CD8 or CD4 glycoprotein TCR co-receptor. As these molecules associate with T-cells that recognise peptides bound to different families of MHC molecules – class I or II – we hypothesised that their TCR repertoires would differ substantially, reflecting the significant structural


differences in ligands. In collaboration with Dr Theres Oakes, PBMC from two healthy donors were sorted into CD4<sup>+</sup> and CD8<sup>+</sup> populations and their TCR repertoires were sequenced. Cell purities are displayed in Appendix A in Figure A.2.

Overlap analysis revealed that there is very little sharing between the CD4 and CD8 compartments within a donor (Figure 3.10A). The CD4 and CD8 compartments of the two different individuals actually share more sequences with one another than do the separate populations within each individual. One of the healthy volunteers who donated blood for sorting was also one of the donors who provided two samples for previous analyses (HV01, see Figure 3.8). The blood to be sorted was drawn approximately two years after the two previous whole blood samples, yet still shares a large number of sequences with both CD4 and CD8 populations. The overlap between the two whole blood samples of this sample taken three months apart was included in Figure 3.10A for reference.

I also inspected the inequality of the sorted repertoires. CD8 repertoires display a far more unequal distribution than CD4, representative of a more oligoclonal distribution (Figure 3.10B). The whole blood samples from the same donor display intermediate equalities, being a combination of both populations. This difference in inequality is apparent even when viewing the VJ gene pairing frequencies of each population (Figure A.3 to A.6, Appendix A). CD4 populations in both chains clearly display a more even spread of gene usage compared to CD8, which is typified by a combination of hotspots and many infrequently used combinations.

Another feature that was apparent from viewing the VJ pairing of one of our sorted repertoires (HV01) was the segregation of large amounts of TRAJ1-2 - TRAJ33 rearrangements (the gene pair that contributes to canonical MAIT cell invariant TCRs) to the CD8 repertoire (Figure A.3). Donor HVD4's CD8 repertoire does not prominently display this pairing (Figure A.4, although there are indeed a number of MAIT

Figure 3.10 (preceding page): Clonal composition of the sorted CD4 and CD8 TCR repertoires. A: Heatmap showing the overlap between sorted CD4 and CD8 subsets of two donors, HV01 and HVD4 (D4), by Jaccard index. Sorted samples from HV01 were taken approximately two years after the two HV01 bleeds (v1 and v2) that were used in Section 3.1. Upper-left hand triangle shows comparisons of  $\alpha$  chain rearrangements, lower-right triangle shows beta. B: Clonal inequality of CD4 and CD8 compartments by Gini index. Unsorted whole blood samples from HV01 from two years before were included for reference. Plots A and B were both generated using subsampled repertoires to control for different sample sizes; 10,000 DCRs were used for alpha, while 20,000 were used for beta. C: Proportional frequencies of the VJ-CDR3 sequences of invariant  $\alpha$  chain TCRs (discussed in Section 3.1.5) as found in either file of CD4 and CD8 sorted TCR data.

cell sequences present). Instead this population is dominated by large expansions of clonotypes made using TRAV12-2 and TRAJ18, which account for 16.4% of all rearrangements. Within this VJ combination, 88.5% produce CDR3s that fit the pattern CAVXXDRGSTLGRLYF (where X represents any amino acid). This raised the possibility that we might be able to confirm and/or determine the CD4/CD8 expression patterns of different invariant TCR clonotypes. I searched for the presence of known and potential invariant  $\alpha$  chain sequences in the sorted repertoire data (Figure 3.10C). As with previous analyses, very few GEM and iNKT sequences were detected, however those that were found were predominantly in the CD4 compartment. MAIT sequences were present in both populations, but only reached the large populations sizes observed previously in the CD8 subset. The putative rearrangements from Van Schaik's and my own data are found distributed approximately evenly between both populations. Of particular note is the sequence that was earlier found to be present in all ten healthy donor whole blood samples (TRAV13-1 - TRAJ6, CAASKGGSYIPTF). Not only was it found exclusively within the  $CD4^+$  compartment of both sorted samples, but it was highly prevalent in each, being the most frequent non-MAIT invariant sequence found in the CD4 data (from HVD4). The CD8 VJ-CDR3 sequence with the highest frequency was TRAV12-2 - TRAJ5, CAVNNTGRRALTF (also in HVD4), one of the sequences that was earlier identified in eight of ten of our whole blood healthy samples, although this sequence was found at low frequency in the CD4 population of both donors. Of those 40 sequences, the majority were found in both CD4 and CD8 repertoires; ten were found exclusively in CD4, while only one was found exclusively in CD8 (Table A.3).

## 3.2.2 Clonality distribution differences using targeted V amplification protocols

In the process of developing the error-correcting pan-TCR amplification protocol used in the rest of this thesis (the process of which is covered in Chapter 5), I tested various intermediate protocols which specifically target a limited number of V genes. These intermediate protocols were used to compare the repertoires of two pairs of cell subsets: naïve versus memory, and cord blood versus adult peripheral blood.

### $3.2.2.1 \quad {\rm CD4^+ \ CD45RA^+ \ versus \ CD45RO^+ \ clonal \ distributions}$

The first such experiment targeted rearrangements using the  $\alpha$  and  $\beta$  V genes TRAV8-4 and TRBV12-3 (see Section 5.2.2 for details). Three healthy donors were bled and PBMC extracted. One donor's PBMC were processed and sequenced, while the other



Figure 3.11: Clonal composition of CD4<sup>+</sup> CD45RA<sup>+</sup> or CD45RO<sup>+</sup> TRAV8-4<sup>+</sup> and TRBV12-3<sup>+</sup> repertoires. A: Logged rearrangement rank versus logged frequency for the five samples sequenced using the simple V amplification protocol, targeting specific V genes. Thus  $\alpha$  results are exclusively those that are shown by Decombinator to be using TRAV8-2 or TRAV8-4 genes (which are highly similar in their 3' regions, and are thus indistinguishable).  $\beta$  samples shown use TRBV12-3. D2 samples were whole, unsorted PBMC. D3 and D4 were sorted into CD4<sup>+</sup>CD45RA<sup>+</sup> ('naïve' or 'n') and CD4<sup>+</sup>CD45RO<sup>+</sup> ('memory' or 'm'). B: Gini indexes for the repertoires shown in A.

two donors' cells were sorted into CD4<sup>+</sup> populations expressing either CD45RA or CD45RO, using magnetic assisted cell sorting (MACS), before sequencing. These populations have been used to enrich for naïve and memory populations respectively (191, 192). It must however be noted that using differential CD45 isoform expression does not provide true naïve-memory separation, as there is a subset of antigen-experienced memory cells that lose CD45RO and re-express CD45RA; such highly differentiated cells are called  $T_{\rm EMRA}$  (T effector memory RA<sup>+</sup>) (193), and will be included in our 'naïve' sorted population.

Figure 3.11A shows that the pattern of rearrangement frequencies by rank are broadly analogous between the groups. This implies that there are both expanded clones in the naïve population and a large number of memory clonotypes that are not expanded. Examination of the Gini indexes of these populations (Figure 3.11B) does however reveal that within a donor, the CD4<sup>+</sup>CD45RO<sup>+</sup> population does display greater inequality, indicative of a greater proportion of the memory repertoire being occupied by relatively expanded clones.

#### 3.2.2.2 Cord blood versus adult whole blood clonal distributions

The next subset experiment used a more refined experiment: while still priming from specific V genes in the PCR, the requisite sequences were introduced to allow errorand frequency-correction (described later in Sections 5.2.5 and 5.3.2), analogous to the approach used to correct the majority of the data in this thesis. This 'barcoded V' (BV) protocol targeted two additional V genes to the previous V-primed strategy – TRAV21 and TRBV20-1 – which had consistently appeared at high frequencies in our preliminary pan-V experiments. This protocol was used to explore the clonotypic structure of TCR rearrangements in cord blood relative to whole peripheral adult blood, focusing on this small number of genes.

Analysis of the specific-V data revealed that both adult and cord blood display the typical linear log-rank pattern, in which each rearrangement is a certain proportion bigger than the previous one (Figure 3.12A and B). Thus both populations contain a small number of relatively expanded and a large number of very-low frequency sequences; however sequences in adult blood reach greater frequencies than cord blood. This feature is recapitulated in their diversity indexes: cord blood repertoires display lower Gini indexes and greater Shannon entropies, indicative of more evenly distributed diverse repertoires (Figure 3.12C and D).



Figure 3.12: Clonal distributions and diversity indices of adult and cord blood V-specific repertoires. Logged rank-frequency distributions for whole peripheral adult blood (A) and neonatal cord blood (B). C: Gini indexes for adult and cord blood samples. D: Shannon entropies for adult and cord blood samples. Cord blood samples displayed significantly lower Gini indexes (p = 5.746e-05) and higher Shannon entropies (p = 0.03458), by one-sided Wilcoxon rank sum tests. All data in this section was produced using the BV protocol, and therefore only contains rearrangements using TRAV8-2/TRAV8-4, TRAV21, TRBV12-3 and TRBV20-1.



I investigated the distributions of the lengths of the inserts<sup>6</sup> and CDR3s of the two groups. Interestingly, despite adult donors' rearrangements tending to contain a greater number of non-templated nucleotides than cord blood (Figure 3.13A), CDR3s from both groups contain a median of 14 amino acids (Figure 3.13B). Despite the confounding presence of residual TRBD gene nucleotides this pattern was also observed in  $\beta$  chains, producing the same median length CDR3s (Figure A.7A and B). Pairwise comparisons of either length distribution between all donors reveals that inter-group pairs are more significantly different than intra-group pairs (Figure 3.13C), i.e. any given repertoire from one group is likely to have a length distribution more similar to another member of that group than the other. Within groups CDR3 length profiles also differ less than insert length profiles. We can infer two main findings from these data. First, that there exists a greater level of heterogeneity in how many inserted nucleotides one finds in TCR repertoires than in the lengths of the resultant CDR3s of those rearrangements. The second finding is that that both these length distributions differ significantly between neonatal cord and peripheral adult blood.

Sharing analyses that were first performed on the pan-V healthy data (presented earlier in Section 3.1.4) were applied to the BV data of this Section. While only considering a limited number of V genes, there are sequences from both chains that are present in the majority of donors (Figure 3.14A). These groups also share similar numbers of sequences with one another, although the variation is greater among adult samples (Figure 3.14B). This data thus provides another resource to look for highly public  $\alpha$  chain sequences. Table A.4 in Appendix A details those VJ-CDR3s that were present in seven or more of the BV-processed samples. One of the V genes targeted – TRAV21 – was also found to be used by some of the highly public sequences discovered

Figure 3.13 (preceding page): Adult and cord blood  $\alpha$  chain TCR length distributions. A: Box plot of  $\alpha$  chain rearrangement insert lengths, i.e. the number of nucleotides between the recognisable ends of the V and J genes, for adult (HV) and cord blood (CB) samples. B: Box plot of  $\alpha$  chain CDR3 lengths (running inclusively from the conserved cysteine contributed by the V gene to the conserved phenylalanine in the J gene). C: P-values obtained from running two-sided Wilcoxon rank sum tests on all the insert and CDR3 length distributions of pairs of donors shown in A, B and Figure A.7 in Appendix A. 'Intra' indicates comparisons within a group (e.g. cord blood versus cord blood) while 'Inter' indicates one distribution from each group. Dashed horizontal lines indicate levels of significance: 0.05 (\*), 0.01 (\*\*) and 0.001 (\*\*\*). Note that despite a similar nomenclature, the adult healthy volunteers were not the same as those used in the pan-V gene results in Section 3.1.

 $<sup>^{6}</sup>$ The fifth field of the Decombinator classifier that contains the nucleotide sequence between the recognisable ends of the V and J genes.



Figure 3.14: TCR publicness and sharing in the BV data. A: Publicness of nucleotide (DCR) and VJ-CDR3 sequences in the data produced using the barcoded V protocol, i.e. how many sequences were there found in a given number of donors. Both healthy adult and cord blood samples were pooled. Note that CDR3 sequences alone are not plotted as they exactly matched the VJ-CDR3 data, possibly as a result of targeting specific V genes in the amplification process. B: Sharing between the different groups amplified with the BV protocol, measured by Jaccard index.  $\alpha$  repertoires were subsampled to 400 sequences,  $\beta$  repertoires to 1,200.

previously in data produced with the BLR protocol (Table A.3); cross-referencing these results showed that a number of sequences are found in both analyses. Thus we find sequences that are shared among up to fifteen different donors in our data alone.

#### 3.2.3 Section discussion

In this section I report results from preliminary analyses made using combinations of specific cell populations and V gene repertoires. Most notable are the differences between the TCR repertoires of  $CD4^+$  and  $CD8^+$  T-cell populations. The most striking difference is the disparity of evenness of repertoires between the two groups, in agreement with previous findings (194), including deep-sequencing reports (99). This separation of polyclonality between  $CD4^+$  and  $CD8^+$  might be explained by differences in their proliferative programs; upon stimulation, CD8 cells undergo a greater amount of division, achieving larger burst sizes, that are more persistent than clonal exansions in CD4 cells (69, 195). Serendipitously, one of the donors bled for cell sorting was also one of the donors for whom there were data for the two whole blood samples (discussed in Section 3.1, which were taken approximately two years apart from the sorted samples), revealing a slightly higher long-term retention of sequences in the CD8 compartment.

The other notable finding is one replicating an observation from a previous TCR repertoire paper, that an individual's CD4<sup>+</sup> and CD8<sup>+</sup> populations do not show much overlap in TCR usage (83). In fact the CD4 and CD8 populations of the different donors shared more sequences than did the two populations within the same donor, highlighting the relationship between TCR identity and T-cell subtype. This makes intuitive sense, as CD4 and CD8 cells are recognising different pools of peptides (or 'peptidomes') in the context of different classes of MHC molecules, therefore one might expect a separation in their TCR repertoires. It was shown as early as the late 1980s that TCR sequences were in some way deterministic of the fate of that cell, as T-cells in mice expressing transgenic TCR sequences derived from a CD8 cell line were found to preferentially develop into  $CD8^+$  cells (196). According to a current model of lineage commitment, this repertoire property is likely enforced by the signal strength of the TCR-pMHC reaction during development: stronger, longer signals produce CD8 cells, and short, weaker signals lead to CD4, while any cells assigned the incorrect fate are later deleted (40). Note that this if effectively a combination of the 'instructive' and 'stochastic' models originally proposed (197).

Data produced from MACS sorted cells underwent the initial pilot TCR amplification and library preparation protocol featured in this thesis. As this protocol lacks the ability to apply barcode-based error correction, the data are somewhat rudimentary and are likely to be less robust quantitatively. Comparison of Figure 3.11 with similar data (e.g. Figure 3.1B and C) suggests the effect of PCR amplification and errors are skewing the results. The broad findings however – that low frequency clones occur in memory populations, and expanded clones could well exist in naïve populations – agree with those found in previous deep-sequencing efforts (83, 198).

A more refined protocol that also only targets specific V genes for amplification was applied to cord and adult whole blood samples. Cord blood is of particular medical interest due to the combination of large numbers of haematopoietic stem cells and relatively naïve T-cell populations possessed of lesser cytotoxic abilities, making it an attractive source of material for stem cell transplants (199). It also presents an interesting research opportunity to immunologists, as it allows inspection of T-cells arguably at their most naïve, having been shielded from microbial antigens whilst in *utero.* This likely explains the clonal distribution we observe in the cord blood, which is far more evenly distributed than adult blood, without so many clonal expansions, confirming previous spectratyping results (200). However there are some expanded sequences, which may reflect sequences that were produced frequently, perhaps as a result of convergent recombination or recombination bias as discussed above. I also demonstrate that despite producing the same median length CDR3s, adults' repertoires tend use TCRs with more inserted nucleotides (either N or P nucleotides) to encode them. Again, this supports previous published reports that describe foetal rearrangements tending towards shorter CDR3s than infants' (201), or adult blood tending towards longer CDR3s than cord blood (202). We know that in human embryos, Tdt – the enzyme responsible for the non-templated addition of nucleotides during recombination – is expressed before the  $22^{nd}$  week of gestation<sup>7</sup> (204). Therefore the difference between the degree of non-templated nucleotide addition between cord and adult blood cannot be explained by lacking the enzyme. Instead, I hypothesise this difference is a result of two features; the lack of expanded clones in cord blood and the probability distribution of producing TCRs with different numbers of insertions. Probabilistically speaking, the formation of TCRs with increasing numbers of nucleotides is less likely with each addition, therefore the longer the insert sequence the rarer that sequence can be expected to be produced by recombination. Adult repertoires have had both more chances to make such sequences and to increase their frequency by cell division

 $<sup>^{7}</sup>$ Note that is not the case in mice, as they do not express Tdt until three to five days after birth (203).

(as some cells bearing many insertions can be expected to expand in response to their cognate antigen), thus such sequences should be more frequent in adults.



## 3.3 Intra-gene splicing of TCR transcripts

Figure 3.15: Examples of intra-V gene splicing. Examples of genes that display noncanonical splice events that interrupt V gene sequences, for TRAV1-1 (A), TRBV7-8 (B), TRBV7-3 (C) and TRBV2 (D). Chromosomal germline (GL) sequences for each sequence are included for comparison. Schematics are to scale: vertical dashed lines indicate 100 bp intervals.

Detecting rearrangements using non-functional TCR genes (as was shown in Figure 3.2) required the design a new set of V and J gene tag sequences, which are used to bioinformatically determine which specific genes are used in a TCR rearrangement<sup>8</sup>.

 $<sup>^8\</sup>mathrm{Discussed}$  later in Section 5.3.1.1

During the course of testing this new 'extended' tag set, the frequency of one particular TRAV gene dropped significantly relative to its use in the original tag set. By inspecting the nucleotide sequences of reads that were assigned recombinations using the original tag set but not with the new, it was revealed that a number of transcripts using TRAV1-1 were subject to alternative splicing, to an splice acceptor site that lies within the new tag sequence.

In order to more systemically assess the extent of splicing within V genes I first needed to obtain nucleotide sequences that spanned the complete V gene. The sequencing protocol used in this thesis produces paired-end reads (i.e. one read from both ends of the template molecule), each 250 base pairs  $\log^9$ . By using the program FLASH (117) I was able to merge all of the reads that overlap in their 3's for one donor to obtain sequences which should contain whole transcript sequences. I then made use of the molecular barcodes introduced uniquely to each cDNA during the amplification process to error-correct our data. This differs to the error-correction employed in assigning the recombined TCR sequence: this splicing analysis requires accurate base calls across the entire length of the sequence, whereas the standard analyses are more tolerant of inaccuracies that would not affect calling the identity of a rearrangement. Accordingly more stringent filtering criteria were applied: in short, only those amplicons which were covered by more than three high-quality sequencing reads were considered, and then the most common base at each position was calculated to obtain a consensus for each transcript. These error-corrected transcripts were then processed using Decombinator, giving us rearrangement information. Recombined sequences up to the 3' of the V gene were then mapped onto the appropriate germline locus using the RNA alignment software STAR (114), and inspected for alternative splicing (AS).

A number of V genes in both chains were found to be subject to alternative splicing at some frequency: 30 distinct splice events were recorded divided across the 17 genes listed in Table A.5. Figure 3.15 shows scaled schematic representations of four example genes that between them demonstrate the breadth of different splice events taking place. These alternative spliceoforms often employ the standard splice event required to add the first exon (which contains the majority of the leader sequence, termed L-PART1 under IMGT nomenclature (19)) but then also splice out a downstream section of V sequence (e.g. Figure 3.15A i to iii). Some transcripts do however display other splicing profiles, such as splicing the L-PART1 sequence directly into the middle of the V gene (Figure 3.15A iv), making use of a splice donor upstream of L-PART1 (3.15C i), or failing to remove the leader sequence intron (Figure 3.15D i).

<sup>&</sup>lt;sup>9</sup>This process is detailed later in Section 5.2.6.



Figure 3.16: Extent of intra-V gene splicing. A: Number of merged, paired-end individual transcript consensus sequences that were successfully produced per healthy volunteer (HV). **B**: Proportions of transcripts that made it through each step of the analysis. DCR =those transcripts in which Decombinator successfully assigned a rearranged TCR, using the original tag set. DCR:P = of those assigned a rearrangement, how many appear to contain a valid productive rearrangement based solely on the Decombinator classifier, performed as in Figure 3.1. P/NP: AS = Of the reads determined to be productive (P) or non-productive (NP) based on their DCR, those that were shown to contain one of the alternative splicing (AS) events listed in Table A.5. C: Assessment of what proportion of reads predicted to be productive based on their Decombinator classifiers stand to theoretically translate based on the presence of start and stop codons in their complete transcript sequence. Categories are: those transcripts that contain only start codons (+ start - stop); those that contain both with the last start codon being downstream of the last stop codon (and thus might be translated, + start > + stop); those with both but stop codons after the last start codon (+ start < + stop), and those that contain no start codons at all (-start +/- stop). D: The proportions of different criteria found among the NP transcripts preventing their productivity based on Decombinator classifiers. OOF = out-of-frame but lacking a stop codon (at least upstream of where the constant region would start);  $OOF^* = out-of-frame$ , also containing a stop codon;  $IF^* = in$ -frame but containing a stop codon; FGXG = lacking the conservedmotif at the end of the CDR3; C = lacking the conserved cysteine residue at the start of theCDR3.  $\alpha$  chain results are shown in green,  $\beta$  chain in purple, with any transcripts found to contain splice events (as listed in Table A.5) hatched. All error bars indicate standard deviation.

I then quantified how frequently these splice events occurred in each of the ten healthy donors sequenced, after equivalent processing. Figure 3.16A shows how many overlapped and error-corrected transcripts were obtained using the high stringency pipeline. Of those transcripts produced, the majority are successfully assigned rearrangements and seem to encode productive polypeptides (Figure 3.16B). By looking for the splice sites identified above in either the productive (P) or non-productive (NP) we can see that splicing is proportionally enriched among the NP. In order to assess what effect AS might be having on seemingly-productively rearranged receptors, such transcripts were searched for the presence of start and stop codons (Figure 3.16C). In order to feasibly be productive, a transcript must contain a start codon that is followed by no stop codons (as these transcripts were reverse transcribed upstream of their proper stop codon). We can see that AS-transcripts show a large reduction in the proportion of transcripts that contain a start codon without a stop codon present. Instead they possess a far larger proportion of transcripts that contain stop codons but might still yet be productive as the last stop codon precedes the 3'-most start codon. Interestingly there is an increased tendency among  $\alpha$  chain transcripts to fall in this category relative to  $\beta$  chains, even in unspliced transcripts. There does not appear to be any considerable difference between unspliced and AS transcripts in terms of the profiles of transcripts that would not be expected to translate, i.e. those that lack a start codon or have one followed by a stop codon. Note that this analysis is limited in two respects. One, we can only obtain full-length sequences for those transcript that are short enough to be overlapped, and two, these sequences may not actually correspond to the actual 5' end of mRNA sequences, rather just where the reverse transcriptase molecule dissociated. I also looked to see whether there was a difference in the reasons why different unspliced and AS transcripts were classified as NP (Figure 3.16D). As there are so few sequences that are NP, of which only a fraction are spliced (with averages of 60 and 8 NP/AS transcripts per donor) this analysis is limited. However there does appear to be at the very least an increase in the frequency of in-frame sequences that contain stop codons (i.e. as a result of non-templated nucleotide deletion or addition) in AS-transcripts.

#### 3.3.1 Section discussion

There are a number of insights that can be drawn from this analysis. The first is that alternative splicing at the TCR loci appears to consist of a far more diverse series of splice events than has previously been recognised. In addition to describing a variety of novel cryptic splice donor sites, I believe this to be the first report of human TCR pre-mRNAs undergoing alternative splicing to cryptic acceptor sites within the sequence of the V gene itself. There is one report that I can find of an equivalent phenomenon, in which a non-productively rearranged  $\alpha$  chain in a clonal murine hybridoma was shown to have undergone inter-V gene splicing (205). This analysis finds over 10,000 transcripts that undergo this intra-V gene AS, allowing one to quantitatively and qualitatively probe the properties of such sequences. These splice events should theoretically be able to dramatically regulate a transcript's expression. That they are enriched among TCRs that should not be able to produce a valid polypeptide based on their rearrangement ('recombinatorially non-productive') suggests that it might be employed as a mechanism in these cases to further regulate message that will not be contributing towards surface receptors. The different profiles in criteria for non-productivity between spliced and unspliced transcripts might reflect AS preferentially being employed against particular subsets of transcripts. For instance it known that AS-mediated incorporation of stop codons can lead to degradation of message by nonsense-mediated decay (206). However the possibility remains that these transcripts just represent some low-level errors of the transcription and splicing machinery, and the tremendous depth of high-throughput sequencing has brought them above the level of detection; functional experiments would be required to settle the matter.

Perhaps more interesting is the small fraction of sequences which we would determine to be productive, based on their recombined hypervariable region, but which contain upstream intra-V splicing events. It is probable that this would either prevent translation entirely, or at least alter the protein structure so much that the product would no longer be able to contribute to a valid surface-expressed receptor ('transcriptionally non-productive'). This raises the possibility that AS might be able to function to suppress the expression of valid TCR rearrangements, perhaps to prevent the expression of autoreactive receptors.

These findings could also impact on the design of repertoire sequencing experiments. All protocols of which I am aware (including the one I developed and employed in this thesis) involve targeted sequencing around the recombination junction to extract clonotypic information. Many even use V-gene specific primers to amplify just this region. While it is not important for this work, it is feasible other groups may be trying to extract specific rearrangements to use functionally downstream; researchers should remain aware of the fact that just because the cDNA describing the CDR3 appears productive, alternative splicing events upstream could be preventing that rearrangement from being expressed as working protein. Sequencing technologies permitting longer reads than are currently available would be required to ensure transcripts are functional based on their entire lengths. Similarly, construction of TCR sequences for gene-therapy purposes might benefit from modulation of cryptic splice sites to prevent unwanted alternative splicing of transgenes.

### 3.4 Chapter discussion

High-throughput sequencing of TCR repertoires effectively consists of the application of novel disruptive technologies to a relatively new field. It is clearly a very powerful tool for research, yet as a field we are still in the early stages of exploring what we might accomplish using these methods. In this chapter, I have displayed an overview of some of the kinds of experiments and analyses one can employ in repertoire sequencing efforts, indicating the range of research questions that we might begin to address. Working from small samples of peripheral blood from healthy individuals, I have shown it is possible to extract a great deal of information about various aspects of TCR biology. At the most simple level, I have described the basic clonotypic and germline gene distributions in my data. These findings recapitulate those of the earliest TCR repertoire sequencing efforts, particularly with respect to the striking nature of the relative abundances of the lowest and highest frequency sequences and the differential usage of different V and J genes (75, 76). This provides us an impression of the 'shape' of the repertoire, which allows us the chance to use this as a reference point when looking for perturbations to the repertoire, such as in disease. An important related aspect of this appreciation of repertoire configuration is how to score and interpret diversity.

I have also looked in my data for the presence of shared or public sequences, as well as those belonging to the so-called 'invariant' TCR T-cell subpopulations. Even without focusing on antigen-specific cell populations I can readily show the presence of a large number of sequences that are shared between multiple donors. Some of these sequences are even found in all donors: while these mostly consist of sequences known to be found in MAIT cells, at least one other sequence was found to be equally public, with several others not far behind. As more donors and cell compartments are sequenced, I anticipate that we will find a far larger number of shared sequences, blurring the lines between what we currently consider to be public and private, a prediction supported by the observation of yet more low-frequency but highly shared sequences even when only priming off a small number of V genes. I also expect that in line with current findings (138), cells matching the phenotypic profiles of the non-pMHC restricted TCR populations but with increasing deviation from the exact canonical sequences will be found, which might eventually necessitate the need to reassess usage of the word 'invariant'. I also demonstrated how by sorting cells into subpopulations prior to sequencing one can not only gain information about those sorted samples,

but apply sequence-matching analyses to infer properties about unsorted populations. Finally I have shown how we might use repertoire sequencing data to learn about aspects of TCR biology unrelated to the clonal composition of the repertoire in revealing the presence of multiple novel splice events that occur within the TCR V gene sequences, which might represent an under-appreciated level of TCR regulation.

Several themes that emerge from this chapter are worthy of particular note. The first is how these experiments speak to the richness of the data produced by repertoire sequencing. However – due to the age of the technique – there is still little consensus about what information is most important and how best to present it, which presents a challenge for both analysis and interpretation. Indeed, as is the case for all HTS, with decreasing costs and greater access to sequencers we could be heading towards a scenario in which data analysis instead of generation becomes the limiting step (207).

Another important aspect of this data is the frequency with which we find uncommon phenomena (such as alternative-splicing or use of unexpected V-genes) that might not be detected or be more readily overlooked from studies using lower-throughput or -resolution repertoire methods. It is probable that as the field investigates increasing numbers of large, multi-dimensional datasets more thoroughly we will discover increasing numbers of such novel events or contradictions to existing rules, and we should be prepared to obtain and assimilate such information. For example, the IMGT databases contain a great deal of well-annotated and easily accessible information regarding variable lymphocyte receptors from a number of species, and their classification and numbering systems allow sensible interpretation of data between studies (19, 101, 102). This has made it in invaluable resource for repertoire studies, demonstrable by the fact that almost every high-throughput repertoire paper either cites the resource or makes use of its nomenclature. However, as we have seen in this chapter, it certainly contains incorrect classification of some genes, therefore – at least in some small way – experiments which are based exclusively upon their annotations are flawed from the outset. One could imagine a scenario in which looking for a particular specific TCR clone could be important, such as tracking minimum residual disease in a leukaemia patient: if this clone happens to be using a V gene that was not included in the multiplex primer panel or rearrangement analysis software then it unlikely to feature or be detected in the data. This chapter serves as a reminder that we must remain open-minded in our approach, and that even the most well-curated databases are still subject to errors. I imagine that IMGT must soon adopt a system whereby the findings of such repertoire sequencing experiments can be incorporated. Another area where this might be useful

is the identification of new alleles of different germline genes, which has already been reported in one RepSeq analysis (183).

There are a number of caveats and limitations both to repertoire sequencing generally and the analyses presented here specifically. One of the major issues is that of sampling: in any one experiment we can only observe a fraction of total cells taken typically from one tissue, usually blood<sup>10</sup>. These tissues are themselves subject to sampling processes as cells enter and leave, divide and die. This will affect the nature of the questions that can be answered using repertoire sequencing. One is only able to reliably study cells which appear frequently appear above the rate of detection of the experiment: one would not be able to study say GEM cells at the depth of sampling used in this study without great cost. Given the conservation between samples, it is however probable that the general TCR repertoire features and population structures obtained from a small but random sample are indeed likely to be representative of the whole tissue. In a similar vein, Eugster et al were able to detect antigen-specific T-cell responses from sequential small blood samples, however the exact clones used were often different between samples (208). It must also be stressed that in this thesis I was only making use of a fraction of RNA derived from just two to three millilitres of blood, yet was still able to generate this wealth of data: if greater depth is required one could easily take more blood, process more RNA or sequence more of the library.

Sampling leads on to another important consideration for repertoire sequencing, that of diversity. Diversity is a key parameter of the T-cell response – arguably even the 'purpose' of TCRs – as it corresponds on some level to the number and nature of antigens that can be responded to (209). However we cannot compare sample diversity in its purest sense (i.e. number of different sequences, or richness) due to the sampling effect. Instead, in this thesis I have adopted two diversity indices to measure different aspects of TCR repertoire diversity, which can then be compared between samples without knowing the true size of the repertoire. Being able to measure repertoire diversity in healthy individuals provides a baseline against which to compare repertoires from patients who we might expect to display altered T-cell populations.

The final major consideration is that of expression. With the exception of Harlan Robins' group, and anyone who makes use of data generated either in collaboration with them or through the company he helped found, the majority of groups currently using deep-sequencing to study TCR repertoires use RNA as their starting material. This has a number of technical benefits, such as the downregulation of non-productive sequences

<sup>&</sup>lt;sup>10</sup>This problem affects mouse studies to a lesser extent, as more T-cells can be sampled, i.e. a greater proportion of a given mouse's total T-cells can be sampled by splenectomy than can be obtained from a human by phlebotomy.

and the ability to perform unbiased pan-TCR amplification (discussed in detail in Section 5.1.2 in Chapter 5), yet also brings some caveats, the foremost being that mRNA expression levels are mutable. TCR expression has a further complication, as not only can mRNA levels change through modulation of transcription or turnover as with any transcript, but such processes could theoretically differentially affect TCRs as each V gene brings its own promoter and regulatory regions to a rearrangement (210, 211). Therefore we must not confuse sequence frequencies with cell frequencies. However, the two are related, if not linearly. While not extensively studied, TCR expression has been shown in several *in vitro* model systems to increase upon T-cell stimulation prior to cell division (212, 213, 214, 215), even as TCR molecules are internalised from the cell surface (216, 217). This implies that there are two forces driving the frequency of TCR mRNA molecules in a blood sample (and thus the probability of their detection by sequencing): the number of cells of that clonotype, and their activation status, which could be advantageous in certain experiments. Using gDNA should allow description of relative abundance of cells, assuming an unbiased amplification protocol. However, if one requires absolute quantitation of cells from repertoire sequencing data, it would require using a barcoded amplification approach analogous to the one described in this thesis, but using gDNA as a substrate, a strategy that to my knowledge has not yet been employed.

In spite of these limitations, in this chapter I have demonstrated how one can apply TCR repertoire sequencing to even small samples of peripheral blood and extract a great deal of diverse information relating to T-cell biology.

## Perturbations to the T-cell receptor repertoire during HIV infection

### 4.1 Introduction

#### 4.1.1 HIV and AIDS

June 1981 saw the beginnings of the Western outbreak of acquired immunodeficiency syndrome (AIDS) in the United States of America, which made it to the United Kingdom by the end of that year (218, 219). The syndrome consists of increased likelihood of pathology from opportunistic infections, often from typically non-pathogenic microbes, as well as certain uncommon malignancies and autoimmune conditions, in patients with no history of immunodeficiency (220). Pneumonia from *Pneumocystis jiroveci* (then called *Pneumocystis carinii*) and Kaposi's sarcoma (later found to be caused by the eponymous herpesvirus (221, 222)) were particularly common forms of AIDS associated disease (223, 224). Moreover the syndrome was initially enriched among certain demographics, particularly men that have sex with other men, intravenous drug users, haemophiliacs and Haitians (220). In the following years, several groups – notably those headed by Luc Montagnier and Robert Gallo – isolated cytopathic T-cell tropic retroviruses from patients with AIDS (225, 226, 227). These were later discovered to all be the same virus and the causative agent of AIDS, which was eventually renamed human immunodeficiency virus (HIV) (228, 229, 230).

HIV has been the subject of a great deal of research, thus much is now known about it and its life cycle. It belongs to the lentivirus family of retroviruses (231). Primary infection can often be asymptomatic or produce mononucleosis symptoms, and initial peak viraemia will typically be contained as the cellular and humoral responses to the virus develops (232, 233). Patients usually then enter a phase of seeming good health with minimal viraemia, which can last from months to years, although the virus remains highly active in lymphoid tissues (234, 235). As the disease progresses, patients suffer from a gradual loss of CD4<sup>+</sup> T-cells, which prompted research that identified CD4 itself as HIV's primary receptor for cell entry (236, 237). This CD4 cell death was previously thought to be the result of apoptosis of uninfected bystander T-cells, likely as a result of activation-induced cell death or by action of some viral component (238). However recent evidence suggests that the bulk of CD4 cell loss in lymphoid tissues may be due to unsuccessfully infected cells undergoing pyroptosis in response to aborted reverse transcription products, creating an inflammatory focus that in turn could recruit more T-cells to the site of infection (239, 240). This loss of CD4 cells and the chronic systemic inflammation seen during HIV are thus considered the driving forces of the immunodeficiency that causes AIDS, making CD4 count a useful clinical marker of disease progression (241).

HIV also perturbs the CD8<sup>+</sup> compartment, both through driving antigen-specific clonal expansion as well as the general bystander activation and exhaustion observed during infection (242, 243). CD8<sup>+</sup> cytotoxic T-cells (CTL) are important mediators of anti-HIV immunity. Strong CD8 responses correlate with control of initial viraemia, while subsequent loss of or escape from viral-specific CTLs associates with increasing viral titres and clinical progression (244, 245, 246, 247). Qualitative differences in the CD8 T-cells of individuals who achieve good virological control correlate with a delay in progression to AIDS (248, 249, 250). Furthermore certain HLA class haplotypes associate with either greater or lesser ability to control disease progression (251).

3'-azido-3'-deoxythymidine (AZT), a thymidine analogue that inhibits HIV reverse transcriptase (RT) by chain termination, was shown to possess antiviral activity both in vitro and in vivo, providing the first antiretroviral therapy (ART) for HIV (252, 253). Inspection of viral sequences from treated patients with decreasing sensitivity to treatment however revealed that the RT gene could acquire resistance mutations (254). The development of additional drugs, and particularly new classes of drugs that target different stages of the HIV life cycle, allowed the development of combination ART (cART, also called highly active antiretroviral therapy, HAART) drug regimens (255). HAART reduces viral loads, improves CD4 cell levels and reduces the risk of opportunistic infections, while reducing the risk of resistance as the virus would need to mutate multiple residues simultaneously (256). Accordingly cART has become the standard of care for patients with HIV (257).

Current therapeutic options are however unable to completely remove virus from an individual as HIV integrates its genome into that of the host, allowing it to persist in the face of treatment, necessitating constant medicating (233). Moreover patients on HAART still remain at greater risk of certain non-AIDS related diseases, which are typically those often associated with aging, displaying increased rates of cardiovascular, kidney, liver and bone disease, cancer and various neurological syndromes (reviewed in (241)). Despite increasing life expectancy of  $HIV^+$  individuals substantially, these treated patients still have lower life expectancies than  $HIV^-$  people (258). This suggests that HIV is able to cause some physiological damage (either before or during therapy) that treatment is unable to fully rectify, or that the drugs themselves have off-target effects. At the end of 2013 there were 35 million people estimated to be infected with HIV globally, of whom approximately 12.9 million were receiving ART (259), therefore the burden both of the disease and any therapy-associated pathology remain important subjects of research.

#### 4.1.2 HIV TCR repertoire analysis

There have been a number of studies using various techniques investigating TCRs in HIV-infected individuals. Flow cytometry, DNA hybridisation and quantitative PCR have shown decreased expression of certain V genes, with certain families being depleted to variable extents in different individuals (260, 261, 262, 263). Furthermore spectratyping analysis revealed a skewing of CDR3 length distributions in certain V families, indicative of clonal expansions (264, 265, 266). A decrease in TCR diversity associated with HIV infection has also been measured using a number of experimental approaches (267, 268, 269), although such observations have typically been limited to either antigen- or T-cell subset-specific populations. The overall impact on the clonal structure of the T-cell population of the virus depleting CD4<sup>+</sup> T-cells while simultaneously being targeted by (and eventually escaping from) CD8 cells remains incompletely understood.

As discussed in Chapter 3, the introduction of high-throughput DNA sequencing (HTS) allows analysis of the TCR repertoire at much greater depth than was previously possible. The wide ranging T-cell effects and large global disease burden make HIV-infection an eminently suitable testing ground for such analysis. Therefore in this chapter I describe the sequencing of TCR repertoires from a cohort of ART-naïve HIV patients. RNA was extracted from the peripheral blood of 16 adult HIV-positive patients due to start ART at two time-points, both immediately prior to ('v1') and three months after commencing therapy ('v2'). Patients were excluded if they were currently showing symptoms for AIDS related illness, or if they had been recently vaccinated. Each RNA sample was processed using the error-correcting barcoded-cDNA protocol that was used in Chapter 3 (detailed later in Section 5.2.6). The patient data was compared to the healthy repertoires of the ten healthy donors explored in Chapter 3, four of whom were also sampled twice after a three month interval.

## 4.2 HIV infection associates with a dramatic loss of TCR diversity

Given the variety of reports of reduced diversity in the CDR3 repertoires of HIV patients I used the indexes employed in Chapter 3 to assess the diversity in the pre-treatment patient samples. Figure 4.1A and B display a significant difference between the Gini index and Shannon entropy of healthy and HIV-positive repertoires respectively. The reduced diversity reflected by the difference in these indexes was manifest in both  $\alpha$  and  $\beta$  chains, although the complexity of  $\alpha$  chain repertoires was less than those of  $\beta$  (Figure 4.1B), consistent with the lack of a diversity region and consequently less junctional diversity (78). The high inequality and low complexity of repertoires from HIV patients could be considered equivalent to a relatively oligorlonal population in comparison to healthy individuals. In order to assess whether this was the case I measured the average proportional frequency of the 50 largest CDR3s in each donor: in patients these common sequences accounted for approximately ten times the proportion of their respective donors than in healthy donors (Figure 4.1C). I performed a number of linear regressions to see whether any clinical scores correlated with the repertoire diversity metrics. Surprisingly, the most significant correlation was between the CD8 cell count and the Gini index (Figure 4.1D), suggesting that particularly unevenly distributed repertoires might be produced in part by large clonal expansions in the CD8 pool. Despite the low CD4 T-cell counts in these patients, CD4 values did not correlate with Gini index (see Figure B.1A in Appendix B). Multiple regressions of both indexes against various clinical scores (performed by my colleague Katharine Best) confirmed that CD8 cell count and Gini index were the two parameters that correlated best (Figure B.1B).

These findings seemed to run in opposition to the simplest potential explanation, that the decrease of diversity was solely attributable to the loss of CD4 T-cells as a result of viral infection. I wished to further explore the hypothesis of CD8 expansions being a pressure driving down diversity, however we did not have access to fresh or sorted blood samples from these patients. In order to overcome this limitation, an alternative assay was devised that requires neither access to patient samples nor sorting of HIV-positive cells. TCR sequences were randomly sampled from the CD4<sup>+</sup> and CD8<sup>+</sup> repertoires sorted from healthy donors (discussed earlier in Section 3.2.1) in silico, and combined in different amounts to simulate mixed repertoires from donors with different CD4:CD8 ratios. The diversity indexes of these simulated repertoires are shown in Figure 4.2. As the proportion of CD8 cells increases, so does the repertoire



Figure 4.1: TCR-repertoires of HIV-infected patients are typified by lower clonal diversity and higher inequality than healthy donors. A: Evenness of healthy and HIV-infected whole, unsampled CDR3 repertoires as measured by Gini index. HIV<sup>+</sup> Gini indexes were significantly greater than healthy donors' (p = 7.937e-15, the lower bound for this test in R, the statistical software used). B: Information content (as a diversity index) measured by Shannon entropy of CDR3 repertoires. HIV<sup>+</sup> Shannon entropies were significantly lower than healthy donors' (p = 7.937e-15). Between both healthy donors and HIV patients, alpha repertoires have significantly lower Shannon entropies (p = 2.582e-05). C: Mean proportional frequencies of the 50 most common CDR3s for healthy (blue) and HIV-infected (red) donors: HIV<sup>+</sup> mean frequencies are significantly higher than healthy donors (p = 7.937e-15). P-values in A, B and C are all Wilcoxon rank sum tests. D: Linear regression of CD8 T-cell count during the v1 bleed versus Gini index ( $R^2 = 0.51$ , p = 4.30e-6).



Figure 4.2: Simulating TCR repertoires with different CD4:CD8 ratios. CD4<sup>+</sup> and CD8<sup>+</sup> T-cells were sorted from PBMC extracted from two healthy donors (HV01 and HVD4) and sequenced. Fixed numbers of TCR classifiers were randomly sampled from each population and mixed in different ratios to simulate populations with different ratios, and then Gini index (A) and Shannon entropy (B) were calculated. Horizontal lines indicate the mean index values of the healthy (blue) and HIV-infected (red) repertoires in our dataset  $\pm$  standard deviation. The mean ratio observed in our patient clinical data (0.31:1) is denoted by  $\bar{x}$  (vertical red line,  $\pm$  standard deviation)

inequality (Figure 4.2A), while the complexity falls (Figure 4.2B). The diversity indices of the simulated repertoires intersect the mean values observed in the sequenced data of the healthy repertoires at ratios of 2.5:1 to 2:1, within typical expected ranges for a healthy population (270). However, when we reconstitute simulated repertoires with a ratio of 0.31 – the average CD4:CD8 ratio seen in the HIV patient's clinical data – we see that the reconstituted repertoires are still yet more diverse, showing both lower Gini indexes and higher Shannon entropies. These findings demonstrate that the low diversity observed in the HIV<sup>+</sup> samples does not appear to solely be due to a decreasing CD4:CD8 ratio; instead, CD8 expansions as well as CD4 cell loss must be driving the inequality and low complexity in HIV<sup>+</sup> patients.

## 4.3 TCR repertoire population structures are perturbed in HIV patients

In spite of the stochastic nature of V(D)J recombination, there are a number of repertoire features that are reported shared between individuals, potentially reflecting constraints of the recombination machinery (77, 179, 271). These properties are characteristic of, and potentially contributory towards, what we consider to be 'healthy' TCR repertoires. Therefore I wished to establish whether these features were altered during HIV infection.

One such property is the sharing of TCR clonotypes across multiple individuals (discussed above in Section 3.1.4), possibly as a result of convergent or biased recombination processes (130, 131, 181, 184, 271). Given the dramatic loss of diversity observed in HIV patients, I hypothesised that we should expect to see less sharing. I calculated the publicness of those CDR3s that were only found either in healthy or HIV-infected patients; those that were limited to HIV patients were far less public, occurring less often and in fewer donors for both  $\alpha$  (Figure 4.3A) and  $\beta$  chains (Figure B.2A). Calculation of the Jaccard index – a normalised measure of sharing – between pairs of different donors revealed that any two HIV patients share fewer CDR3s than do any two uninfected controls (Figure 4.3B). I also observed that the largest CDR3s in the healthy donors are also encoded by a greater number of nucleotide rearrangements than those in HIV-infected donors (Figures 4.3C and B.2B). This loss in translational convergence occurs even though the most common CDR3s in HIV patients account for a far greater proportion of their respective repertoires, reflecting the extreme expansions of individual rearrangements.



Figure 4.3: HIV<sup>+</sup> TCR repertories are less shared, public, and translationally convergent than healthy controls, and are highly divergent in their V and J gene use. A:  $\alpha$  chain CDR3s are less public in HIV-infected than healthy repertoires. The number of CDR3s shared by different numbers of individuals is shown, considering only those CDR3s that were found exclusively in either healthy (blue) or HIV<sup>+</sup> (red) donor repertoires. This analysis was applied to whole samples (solid lines) and a random subsample of 5000 CDR3s (dotted lines). To account for the greater number of patients than controls, ten HIV<sup>+</sup> samples were randomly selected and publicness measured: this was repeated 100 times, shaded areas represent standard deviation. B: Inter-donor Jaccard index measuring sharing between pairwise combinations of healthy control (HC) or HIV-infected CDR3 repertoires, for 5000 randomly chosen CDR3s. Horizontal lines indicate medians. Inter-donor Jaccard scores were significantly higher in healthy donors than  $HIV^+$  patients for both chains (p = 2.2e-16 and 1.147e-13 respectively, unpaired Wilcoxon rank sum test). C:  $\alpha$  CDR3s display a lower mean translational convergence in HIV<sup>+</sup> donors than healthy, i.e. were encoded by fewer nucleotide sequences, determined by counting the number of unique Decombinator identifiers that produce the same CDR3. Blue and red lines indicate the mean translational convergence for each group (healthy/HIV respectively) for each CDR3 rank; green and purple lines show a sliding window averaging across 100 ranks. D: Barplot showing the difference in TRAV usage, by Jensen-Shannon Divergence (JSD). Donor-specific proportional gene usage matrices, before each pair of healthy or HIV<sup>+</sup> donors was compared. Only interquartile ranges and medians are shown.

I also assessed the differences between the V and J gene expression profiles of the two groups, by comparing frequency distributions of pairs of donors using the Jensen-Shannon measure of divergence (JSD). Values for  $\alpha$  V genes are shown in Figure 4.3C, while the remaining genes' divergence values are in Appendix B, Figure B.3. Healthy donors' repertoires diverge from one another by a relatively small, consistent amount, while any two HIV<sup>+</sup> patients differ by greater, more variable amounts. Comparing healthy donor to HIV-patient distributions reveals an intermediate level of divergence (data not shown for clarity). These data suggest a scenario in which there exists a healthy core of gene usage, where any two individuals can differ from one another by a small amount. HIV infection then drives an individual's usage profiles away from this healthy core, but it does so in a different direction in different people, producing idiosyncratic profiles that differ somewhat from typical values, but greater from other patients. I also note that there is no difference between infected and uninfected donor profiles for TRBJ (Figure B.3C), perhaps due to the smaller number of genes or the confounding effect of the D region. Examination of the raw gene usage profiles themselves (Figure B.4) reveals that most genes on average are used an equivalent amount, just with greater variance among patients; a small number of genes in each locus, however, show a consistent difference in frequency between healthy controls and HIV-infected individuals (e.g. TRAJ33 and TRBV6-1, Figure B.4B and C).

# 4.4 Effect of antiretroviral therapy on TCR repertoires of HIV<sup>+</sup> individuals

By comparing the treatment-naïve samples with those taken after an average of 14 weeks of therapy I was able to assess the impact of short-term ART upon the TCR repertoires of HIV patients. The four healthy donors who were also sampled after a comparable time period act as controls of typical repertoire variation over time and between samples.

By inspecting the clinical parameters we can infer that the ART had the expected effects: viral loads drop in all patients, typically beneath the limits of detection, while absolute CD4 cell levels and CD4:CD8 ratios increase significantly (Figure B.5). The only major lymphocytic parameter to not significantly change over the course of the treatment is the CD8 T-cell count (Figure B.5B), which could indicate that the CD8 expansions hypothesised above are persistent in the face of treatment. This analysis also revealed an interesting caveat of our patient data: the length of time elapsed before



Figure 4.4: ART partially reduces repertoire inequality, but general diversity and idiosyncratic gene usage do not recover. A: Intra-donor Jaccard indices, describing overlap between v1 and v2 bleeds for both healthy control (HC) and HIV $^+$  donors for 5000 CDR3s subsampled from whole repertoires. Horizontal lines indicate medians. Healthy controls have significantly lower  $\alpha$  and  $\beta$  chain intra-donor sharing by Jaccard (p = 0.02912) and 0.01367 respectively, one-sided unpaired Wilcoxon signed rank test). B: Log2 fold change in median TRAV Jensen-Shannon Divergence (JSD) from v1 to v2 (red). Blue bars indicate the log2 change in median JSD that would be required for that patient to reach the mean value of the median JSDs for the ten healthy controls, i.e. return them to the average level of divergence healthy controls display in Figure 4.3D. C: Gini indexes of whole CDR3 repertoires of healthy donors, and both v1 and v2 bleeds of  $HIV^+$  patient repertoires. Horizontal lines indicate medians. Pre-ART bleeds have significantly higher Gini indexes than post-ART (p = 0.0001673, paired one-sided Wilcoxon test). D: Shannon entropies of whole CDR3 repertoires of healthy donors, and both v1 and v2 bleeds of HIV<sup>+</sup> patients. Horizontal lines indicate medians. There is no significant difference between pre- and post-ART Shannon entropies (p = 0.1609, paired one-sided Wilcoxon test). Fold changes are shown in Figure B.8.

each patient began therapy correlated with their CD4:CD8 cell ratio at time of the pretreatment bleed (Figure B.6). This might be explained by differences in viral control influencing the likelihood of patients seeking rapid treatment.

In contrast to the inter-donor Jaccard indexes shown in Figure 4.3B, calculation of intra-donor CDR3 repertoires revealed that the overlap between two samples from the same donor is greater for the HIV patients than the healthy donors (Figure 4.4A), i.e. more sequenced are retained within an HIV-patient over the same time period. Given the various levels of subsampling in such sequencing efforts – including cells sampled during phlebotomy, mRNAs processed during amplification and amplicons recorded in the sequencer – it is only ever possible to obtain data for a fraction of the entire TCR repertoire. That these HIV patients have far fewer unique CDR3s at both time points but still retain more sequences over therapy suggests that clonal expansions are persistent, even in the face of therapy. However there are two patients that have exceedingly low Jaccard indexes in both chains, indicating that they have undergone dramatic repertoire remodelling events. Post-treatment gene usage divergence values were calculated as before: not only did divergence not return to healthy ranges, in the majority of patients it actually increased (Figures 4.4B and B.7). The majority of patient repertoires (11 out of 16) did observe a decrease in Gini index on ART, indicating a partial recovery in repertoire equality (Figures 4.4C and B.8A). However, the information content as measured by Shannon entropy did not significantly change over the course of treatment (Figures 4.4D and B.8B). Thus three months of ART is sufficient to reduce viral load to undetectable levels and begin to replenish CD4 cell numbers, but does not restore the properties of the TCR repertoire to the ranges observed in healthy donors.

### 4.5 Tracking sequence-specific CDR3s on ART

It is not yet possible to infer the quaternary structure of TCRs from primary sequence, much less their binding specificities. Moreover most high-throughput experiments, including this one, do not yet routinely obtain paired  $\alpha$  and  $\beta$  chain sequence and MHC haplotype information. However there are a number of reports that CDR3s from T-cells reactive against the same antigens often share sequence identity or features, and some TCRs can recognise the same peptide in different HLA backgrounds (272, 273, 274, 275). There are also reports from several groups of public TCR sequences reactive against HIV antigens (276, 277, 278), sometimes even finding the same CDR3 in combination with different V and J genes (249). This lead me to question whether our data contains any CDR3 sequences of known specificity, as a proxy for such specificities



Figure 4.5: Dynamics of viral-associated and invariant MAIT cell CDR3s. A: Proportional frequencies of the published, antigen specific CDR3s that were also found in the whole CDR3 repertoires pre- and post-treatment. CDR3s were only included if they appeared twice or more in both of a patient's bleeds. Horizontal lines indicate medians. HIV-related CDR3s significantly decreased after three month of therapy, while CMV CDR3s increased (p = 0.01799 and 0.04919 respectively, paired Wilcoxon signed rank test). HIVassociated CDR3s were only found in six of the 16  $HIV^+$  patients, with the majority (7/13) all found in one donor (P017). The two HIV-associated CDR3s found in healthy volunteers were from the same donor (HV04). CMV-associated CDR3s were found across nine HIV<sup>+</sup> donors (with four, the most, being contributed by P027), while the single healthy response was found in HV03. B: Proportional frequencies of  $\alpha$  chain sequences that match published MAIT cell VJ-CDR3 sequences, for healthy control (left) and HIV donors (right). Horizontal lines indicate medians. Healthy donors' median MAIT CDR3 frequencies are significantly higher than those of HIV patients at both time points (p = 0.006366 and 0.03558, unpaired one-sided Wilcoxon signed rank test). There is no significant change in frequency for either healthy or HIV<sup>+</sup> results from bleed v1 to v2 (p = 0.2929 and 0.1403 respectively, paired one-sided Wilcoxon signed rank test).

potentially existing in the patients. I therefore harvested the amino acid sequences of published CDR3s known to be specific to HIV antigens from the literature (referenced in Appendix Table B.1). These were typically obtained through isolating HIV-specific pMHC-tetramer positive T-cells from infected patients. A comparable set of sequences were also collected for CMV- and EBV-specific CDR3s was also collected, as these are other well-studied, widespread chronic viruses known to drive large oligoclonal T-cell expansions (81, 279, 280).

The relative change and raw frequency values for those published viral-associated CDR3s that were found in our data (present at least twice in both of an individual's bleeds) are shown in Figures 4.5A and B.9. Thirteen HIV-associated CDR3s were present in the HIV patients' sequence data, coming from six of the 16 individuals sampled. These sequences included three  $\alpha$  chain CDR3s, one of which was present in two donors while the third differed by only a single amino acid. Seventeen CMVassociated CDR3s were detected from nine patients, while six EBV-associated sequences were found across five patients. Interestingly, the frequency of HIV-associated CDR3s decreased in frequency over the course of therapy, while the majority of CMV- and EBV-associated sequences grew more frequent. These viral-associated sequences were almost completely absent from the healthy controls, with only one or two low-frequency responses identified per virus (although this will partially reflect the greater number of HIV patients sequenced). If we assume that these viral-associated sequences may indeed represent T-cells playing a role in specific antiviral responses, this data would suggest that during ART anti-HIV clones are able to contract as viral load falls, which would leave 'space' in the repertoire into which clones of other specificities could expand. The nature of this 'space' would likely be decreased competition for some shared or limiting resource. The extent of the herpesvirus response is particularly noteworthy. The rate of return for this analysis was much greater than for HIV – despite finding more sequences, I only searched for 265 published CMV CDR3s compared with 871 specific for HIV – and the resultant CDR3s account for greater proportions of their repertoires (Figure B.9A). This is consistent with observations that CMV- and EBV-specific TCRs often belong to very large and highly shared clones (81, 275).

I also looked for the presence of the invariant  $\alpha$  chain TCRs of the non-pMHC restricted T-cell subsets discussed in Section 3.1.5. MAIT cell VJ-CDR3 sequences had been highly prevalent in the healthy controls (see Figure 3.9), and are still often detectable in the HIV-patient repertoires (Figure 4.5B). However there was a significant reduction in the proportional frequency of MAIT cell TCR sequences detected in untreated HIV patients, that did not significantly change after three months of treatment. This depletion of MAIT cell sequences is so profound that we can see it at the gene level: the V and J genes for the canonical MAIT alpha chain (TRAV1-2 and TRAJ33) as well as the V genes known to contribute to their beta chains (TRBV6-1, TRBV6-4 and TRBV20-1 (281)) were among the minority of genes that were used significantly less in the HIV patient data (Figure B.4A-C). Similarly, despite there being no published iNKT VJ-CDR3 sequences observed in this analysis, the genes encoding their invariant TCRs (TRAV10 and TRAJ18 (282)) were also less common in the repertoires from HIV-infected samples. NKT cells are known to be reduced in HIV patients (283, 284, 285); it is possible that a far wider range of iNKT CDR3s exists than is currently known and some are being depleted in these patients, which would explain a decrease in the associated V genes. No GEM sequences were found using these critera, consistent with their low detection even in healthy individuals.

#### 4.6 Discussion

Through application of the unbiased, error-correcting TCR amplification and sequencing protocol developed in the course of my PhD, I have explored the effect of HIV infection and subsequent antiretroviral treatment upon TCR repertoires. HIV patients' repertoires differ from those of healthy individuals in every regard examined. They are typified by marked clonal inequality, wherein a relatively small number of highly expanded, private rearrangements occupy a large proportion of the repertoire. Furthermore this inequality correlated with the patients' CD8<sup>+</sup> cell counts. However neither Gini index values nor CD8 counts correlated with CD4 counts, suggesting that the two compartments are being altered by different processes. Cell counts of both populations failed to correlate with the length of time individuals remained infected but untreated, which varied widely in this cohort. These correlations, combined with the fact that in silico simulation of repertoires of decreasing CD4:CD8 ratios failed to reproduce the low inequality observed in the patients, speak to a model in which CD8<sup>+</sup> clonal expansions help drive the loss of repertoire evenness, independent of CD4 cell death. These two components of T-cell perturbation drive reductions in repertoire diversity, translational convergence and inter-donor TCR sharing, endowing patients with highly idiosyncratic gene usage profiles that lie far outside the ranges observed in healthy uninfected individuals.

A current weakness of TCR sequencing approaches is the inability to assign specificity, as predicting protein structures and behaviours of large multi-protein complexes remains extremely challenging in general, and particularly difficult for TCR-pMHC interactions (reviewed in (286)). Even were there mechanisms to infer TCR antigens

from sequence, this current study would not be geared towards this end as we obtained neither paired  $\alpha\beta$  chain information nor patient HLA haplotypes. I nevertheless identified a small number of CDR3s in the HIV patients' repertoires that have previously been reported as belonging to viral-specific T-cell clones. The frequency of these CDR3s undergo changes that we might expect if they were reactive against their associated viruses: HIV-associated CDR3s got rarer while CMV- and EBV-associated CDR3s became more prevalent. This matches a variety of reports that found a decrease in HIV-reactive clone frequency during ART while CMV-reactive clones increased (268, 287, 288), which is in turn reinforced by the frequent loss of CMV viraemia during treatment for HIV (289, 290). These findings lend weight to the possibility that the viral-associated CDR3s in this data might actually be derived from TCRs which share those specificities, and the changes observed in Figure 4.5A represent the contraction of HIV-reactive clones as ART depletes the source of antigen, allowing re-emergence of protective responses and pathogen control. If so, it is possible these shared viralassociated CDR3s could represent members of a subset of receptors in which antigenic specificity is concentrated in one of the two chains (291, 292).

Questions as to the nature of the expanded sequences remain. These could represent  $CD8^+$  clones that are yet to contract in response to diminished antigen levels, or still be responding to residual infection. These need not consist exclusively or even primarily of HIV-reactive clones. For example, of the viral-associated CDR3s I found, those associated with CMV were found at much higher frequencies than those associated with HIV. Indeed, Betts *et al* reported that only an average 6.31% of total CD8 cells from treatment-naïve HIV-patients were specific for HIV antigens (293). The rest of the expanded clones may be driven to proliferate in an antigen-independent manner, or by exposure to other non-HIV pathogens as a result of increasing immunosuppression (294, 295). For instance it has been shown that HIV induces gastrointestinal damage that leads to increased translation of gut microbial antigens to the circulation (296, 297), which stands to act as a substantial source of persistent antigens for T-cells to react against. Another potential explanation for such prevalent, persistent clones is the more recent discovery that the integration of HIV at different genomic sites associates with clonal expansion and retention (298), even when the integration consists of defective provirus (299), presumably as a result of interacting with the expression of neighbouring host genes. However due to the lack of correlation between CD4 count and either diversity index it seems unlikely that this phenomena causes widespread effects in this data.

Not only does this data describe a profound perturbation to the normal TCR repertoire in the context of chronic HIV infection, these abnormal population structures persist despite otherwise effective ART. Three months of ART is often associated with a reversal of AIDS-associated pathology, clinical improvement and decreased risk of morbidity, even though CD4<sup>+</sup> T-cell counts may remain well below normal levels (300, 301). Nevertheless, it is apparent that the recovery of the CD8<sup>+</sup> T-cell repertoire remains incomplete despite effective treatment. Large CD8<sup>+</sup> clonal expansions are often found to persist after years of therapy (reviewed in (302)), and significant 'holes' in the repertoire may remain. Insufficient epitope coverage as a result of profound CD8 clonal inequality may also explain why HIV patients with low CD4:CD8 ratios are at a greater risk of developing non-HIV related morbidity and mortality (270, 303), even when viral replication is controlled.

In this data I have described that the invariant alpha chain TCRs of MAIT cells – which can comprise in excess of 10% of the circulating CD8 T-cell pool under normal circumstances (304) – are profoundly depleted in this HIV<sup>+</sup> cohort, and remain so over the course of treatment. This is in agreement with similar findings from recent efforts that described the same phenotype using multiparameter flow cytometry (143, 304, 305). It is unclear whether the under-representation of these cells in the blood occurs as a result of increased cell death or trafficking to another tissue, although they do not appear to be preferentially targeted by the virus (143), consistent with their primarily  $CD8^+$  or double negative profile (306). However, given their contribution to bacterial ligand recognition at mucosal sites, their absence in even treated  $HIV^+$  patients may represent gaps in the immune protection from certain microbes. Invariant TCR chain NKT cells have also been reported to be selectively depleted early during HIV infection (284, 285, 307), which may similarly lead to gaps in viral protection, although these sequences remain below the thresholds of consistent detection in this study. The loss of these non-pMHC restricted T-cell subsets are thus prime examples of potential holes in the repertoires of HIV patients.

Further longitudinal samples from the patients in this and other cohorts at later time-points after starting therapy will be instructive in assessing the extent and repertoire correlates of long-term immunological recovery. Sequencing RNA from a larger blood sample would also increase the range of detection in these assays. Sorting of patient blood into different populations – particularly  $CD4^+$  and  $CD8^+$  – before sequencing would allow much greater resolution in the descriptive and predictive power of the experiment, although handling and cell sorting of blood from HIV patients presents its own technical and safety challenges. In this vein, an ideal experiment would likely
involve high-throughput single-cell whole transcriptome sequencing, from which one could extract both paired TCR information and expression of T-cell subset markers. There is obviously more to T-cell immunity than just which clonotypes are present at what frequencies. In the case of HIV, particular clonotypes may be present but exhausted as a result of chronic stimulation (243), and remain unable to offer further protection. Thus transcriptomic profiling of cells with known TCRs would allow qualitative descriptions from which we might estimate phenotype.

In conclusion, my data contribute to the field by describing in detail how HIV infection profoundly alters the TCR repertoire. Partial reversal of inequality may be rapid and lead to improved immune function, coinciding with the first more rapid phase of T-cell recovery (308). Long-term damage to the repertoire may be much more difficult to repair, particularly in adults where naive peripheral T-cell pools are maintained by homeostatic cell division as opposed to thymic production of new cells, which may contribute to long-term morbidity (309, 310). TCR sequencing and similar approaches offer a chance to gain new insight into such processes, which in larger studies may provide useful biomarkers for patient stratification.

# **Protocol development**

# 5.1 Introduction

High-throughput DNA sequencing (HTS) of variable antigen receptor repertoires (or 'Rep-Seq' (74)) represents a tremendous opportunity to dissect immunological phenomena with the utmost breadth and accuracy. However, while the transition to such platforms offers great throughput and resolution, it also introduces a number of technical hurdles which must be overcome in order to produce valid, good quality data. This chapter deals with the nature of these hurdles, and how I have addressed them during my PhD. Herein I detail the amplification procedures and sequence-based metrics that were used to judge efficacy and appropriateness of different strategies. The biological interpretation of the data output from these protocols was covered above in Chapters 3 and 4.

#### 5.1.1 DNA sequencing technologies

In order to fully appreciate the development of protocols for sequencing TCR repertoires, we must first understand how sequencing technologies work, and how highthroughput sequencers are different to earlier generation machines.

#### 5.1.1.1 First-generation DNA sequencing

Despite Watson and Crick solving the three-dimensional structure of DNA in 1953  $(311)^1$ , the ability to 'read' or sequence DNA did not follow for some time. Initial efforts focused on sequencing the genomes of single-stranded RNA bacteriophages and bacterial ribosomal RNA subunits – which were the most readily available pure populations of single RNA species at the time – through what we would now consider to be extremely arduous techniques, involving a great deal of radioactive labelling and two-dimensional fractionation (313, 314, 315). It was by using such techniques that Walter Fiers' laboratory was able to produce the first complete sequence of a gene encoding a protein, that of the coat protein of bacteriophage MS2 (316), followed four years

<sup>&</sup>lt;sup>1</sup>Notably working from crystallographic data produced by Rosalind Franklin and Maurice Wilkins (312).

later by the complete genome (317). Around this time various researchers<sup>2</sup> began to develop methods for sequencing DNA, again making use of radiolabelled nucleotides. Original observations that one could make use of DNA polymerase to incorporate nucleotides at the ends of bacteriophage genomes that possessed 5' overhangs (318) lead researchers to create their own partially double-stranded DNA molecules through the addition of synthetic oligonucleotides, enabling sequencing by incubating with mixtures of variably-labelled nucleotides (319, 320, 321, 322). A notable alternative technique involved using a number of different chemical techniques to cleave DNA preferentially at different bases, allowing inference of the original sequence by comparison of the resultant molecule lengths (323). However the major breakthrough that forever altered the progress of DNA sequencing technology came in 1977, with the development of Sanger's 'chain-termination' or dideoxy technique, in the same year that his group sequenced the first DNA genome, that of bacteriophage  $\phi X174$  (or 'PhiX') (324, 325). The chaintermination technique makes use of chemical analogues of the deoxyribonucleotides (dNTPs) that are the monomers of DNA strands; dideoxynucleotides (ddNTPs) lack the 3' hydroxyl group that is required for extension of DNA chains (as it forms the bond with the 5' phosphate of the next dNTP) (326). Radiolabelled ddNTPs mixed into a DNA extension reaction at a fraction of the concentration of standard dNTPs results in DNA strands of each possible length being produced, as the dideoxy nucleotides get randomly incorporated as the strand extends, halting further progression. By performing four parallel reactions containing individual ddNTP bases and running the results on four lanes of a polyacrylamide gel, one is able to use autoradiography to infer what the nucleotide sequence in the original template was, as there will a radioactive band in the corresponding lane at that position of the gel. While working on the same principle as other techniques (that of producing all possible incremental length sequences and labelling the ultimate nucleotide), the accuracy, robustness and ease of use led to the dideoxy chain-termination method – or simply, Sanger sequencing – to become the most common technology used to sequence DNA for years to come.

A number of improvements were made to Sanger sequencing in the following years, which primarily involved the replacement of phospho- or tritrium-radiolabelling with fluorometric based detection (allowing the reaction to occur in one vessel instead of four) and improved detection through capillary based electrophoresis, both of which contributing to the development of increasingly automated DNA sequencing machines (327, 328, 329, 330, 331, 332, 333). These sequential alterations produced the first

<sup>&</sup>lt;sup>2</sup>Principally Ray Wu, Walter Gilbert and Fred Sanger, the latter of whom had been very instrumental in the earlier development of RNA sequencing techniques.

crop of commercial DNA sequencing machines (334) which were used to sequence the genomes of increasingly complex species. These 'first-generation' DNA sequencing machines produce reads slightly less than one kilobase (kb) in length: in order to analyse longer fragments researchers made use of techniques such as 'shotgun sequencing' where overlapping DNA fragments were cloned and sequenced separately, and then stitched together in silico (335, 336). The development of techniques such as polymerase chain reaction (PCR) (61, 62) and recombinant DNA technologies (337, 338) further aided the genomics revolution by providing means of generating the high concentrations of pure DNA species required for sequencing. These advances also aided sequencing by less direct routes. For instance, the Klenow fragment DNA polymerase<sup>3</sup> had originally been used for sequencing due to its ability to incorporate ddNTPs efficiently; however more sequenced genomes and tools for genetic manipulation provided the resources to find polymerases that were better able to cope with the additional chemical moeities of the increasingly modified dNTPs used for sequencing (340). Eventually, newer dideoxy sequencers – such as the ABI PRISM range developed from Leroy Hood's research, produced by Applied Biosystems (341), which allowed simultaneous sequencing of hundreds of samples (342) – came to be used in the Human Genome Project, helping to produce the first draft of that mammoth undertaking years ahead of schedule (343, 344).

#### 5.1.1.2 Second-generation DNA sequencing

Concurrent to the development of large-scale dideoxy sequencing efforts, another technique appeared that set the stage for the first wave in the next generation of DNA sequencers. This method markedly differed from existing methods in that it did not infer nucleotide identity through using radio- or fluorescently-labelled dNTPs or oligonucleotides before visualising with electrophoresis. Instead researchers utilised a recently discovered luminescent method for measuring pyrophosphate synthesis<sup>4</sup> to infer sequence by measuring dNTP incorporation when each nucleotide is washed through the system in turn, over the template DNA which was affixed to a solid phase (346). Note that despite the differences, both Sanger's dideoxy and this pyrosequencing method are 'sequence-by-synthesis' (SBS) techniques, as they both require the direct action of DNA polymerase to produce the observable output (in contrast to say the Maxam-Gilbert technique, which uses chemical degradation to produce different sized radi-

 $<sup>^{3}</sup>$ A fragment of the *E. coli* DNA polymerase that lacks 5' to 3' exonuclease activity, produced through protease digestion of the native enzyme (339).

<sup>&</sup>lt;sup>4</sup>This consisted of a two-enzyme process in which ATP sulphurylase is used to convert pyrophosphate into ATP, which is then used as the substrate for luciferase, thus producing light proportional to the amount of pyrophosphate (345).

olabelled fragments). This pyrosequencing technique, pioneered by Pål Nyrén and colleagues, possessed a number of features that were considered beneficial: it could be performed using natural nucleotides (instead of the heavily-modified dNTPs used in the chain-termination protocols), and observed in real time (instead of requiring lengthy electrophoreses) (347, 348, 349). Later improvements included attaching the DNA to paramagnetic beads, and enzymatically degrading unincorporated dNTPs to remove the need for lengthy washing steps. The major difficulty posed by this technique is finding out how many of the same nucleotide there are in a row at a given position: the intensity of light released corresponds to the length of the homopolymer, but noise produced a non-linear readout above four or five identical nucleotides (349). Pyrose-quencing was later licensed to 454 Life Sciences, a biotechnology company founded by Jonathan Rothburg, where it evolved into the first major successful 'next-generation sequencing' (NGS) technology.

The sequencing machines produced by 454 (which was later purchased by Roche) were a paradigm shift in that they allowed the mass parallelisation of sequencing reactions, greatly increasing the amount of DNA that can be sequenced in any one run (350). Libraries of DNA molecules are first attached to beads via adapter sequences, which then undergo a water-in-oil emulsion PCR (emPCR) (351) to coat each bead in a clonal DNA population (where ideally on average one DNA molecule ends up on one bead which amplifies in its own droplet in the emulsion). These DNA-coated beads are then washed over a plate with picoliter reaction wells that only fit one bead each; pyrosequencing then occurs as smaller bead-linked enzymes and dNTPs are washed over the plate, and pyrophosphate release is measured using a charged couple device (CCD) sensor beneath the wells. This set up was capable of producing reads around 400-500 base pairs (bp) long, for up to the million or so wells that contained suitably clonally-coated beads (350). This parallelisation increased the yield of sequencing efforts by orders of magnitudes, for instance allowing researchers to completely sequence a single human's genome – that belonging to DNA structure pioneer, James Watson - far quicker and cheaper than a similar effort by DNA-sequencing entrepreneur Craig Venter's team using Sanger sequencing the preceding year (352, 353). The original 454 machine (and first HTS option widely available to consumers), the GS 20, was later superceded by the 454 GS FLX, which offered a greater number of reads (by having more wells in the 'picotiter' plate) as well as better quality data (354). This principle of performing huge numbers of parallel sequencing reactions on a micrometer scale is what came to define the second-generation of DNA sequencing, often made possible as a result of improvements in microfabrication and high-resolution imaging (355).

A number of technologies sprung up following the success of 454, however the most important among them is arguably the Solexa method of sequencing, later acquired by Illumina (354). Instead of parallelising by performing bead-based emPCR, adapterbracketed DNA molecules are passed over a lawn of complementary oligonucleotides bound to a flowcell; a following solid phase PCR produces neighbouring clusters of clonal populations from each of the individual original flow-cell binding DNA strands (356, 357). This process has been dubbed 'bridge amplification', due to replicating DNA strands having to arch over to prime the subsequent round of polymerisation off neighbouring surface-bound oligonucleotides (354). Sequencing itself is achieved in a SBS manner using fluorescent 'reversible-terminator' dNTPs, which cannot immediately bind further nucleotides as the fluorophore occupies the 3' hydroxyl position; this must be cleaved away before polymerisation can continue (358). These modified dNTPs and DNA polymerase are washed over the primed, single-stranded flow-cell bound clusters in cycles; each cluster incorporates the reverse complement of the base at the position of the cycle number – which can be monitored with a CCD by exciting the fluorophores with appropriate lasers – before enzymatic removal of the blocking fluorescent moieties and continuation to the next cycle and hence next position in the sequencing template. While the first Genome Analyzer machines were initially only capable of producing very short reads (up to 35 bp long) they had an advantage in that they could produce paired-end (PE) data, in which the orientation of the molecules could be reversed, allowing observation of the sequence at both ends of each DNA cluster. The standard Genome Analyzer version (GAIIx) was later followed by the HiSeq, a machine capable of even greater read length and depth, and then the MiSeq, which was a lower-throughput (but lower cost) machine with faster turnaround (359, 360).

A number of other sequencing companies, each hosting their own novel methodologies, have also appeared and had variable impacts upon both what experiments are feasible and the market at large. In the early years of second-generation sequencing perhaps the third major option (alongside 454 and Solexa/Illumina sequencing) (361) was the sequencing by oligonucleotide ligation and detection (SOLiD) system from Applied Biosystems (which became Life Technologies following a merger with Invitrogen) (362). As its name suggests, SOLiD sequenced not by synthesis (i.e. catalysed with a polymerase), but by ligation, using a DNA ligase, building on principles established previously with the open-source 'polony' sequencing developed in George Church's group (363). In brief, following a bead emPCR, clonal DNA populations are interrogated using cycles of hybridising and ligating short degenerate, labelled oligonucleotides: each fluorescent colour corresponds to a particular dinucleotide at a given position in the

template, which can be observed (giving '2 base encoding') before cleaving the fluorescent moeity off and continuing the ligation cycle. Capable of producing paired-end reads - highly accurately, due to measuring pairs of nucleotides rather than single monomers, meaning an error would have to occur on both observations – the SOLiD platform is not able to produce the read length and depth of Illumina machines (364), although it remains competitive on a cost per base basis (365). Another notable technology based on sequence-by-ligation was Complete Genomic's 'DNA nanoballs' technique, where sequences are obtained similarly from probe-ligation but the clonal DNA population generation is novel: instead of a bead or bridge amplification, rolling circle amplification is used to generate long DNA chains consisting of repeating units of the template sequence bordered by adapters, which then self assemble into nanoballs, which are affixed to a slide to be sequenced (366). The last remarkable second-generation sequencing platform is that developed by Jonathan Rothburg after leaving 454. Ion Torrent (another Life Technologies product) is the first 'post-light sequencing' technology, as it uses neither fluorescence nor luminescence (367). In a manner analogous to 454 sequencing, beads bearing clonal populations of DNA fragments (produced via an emPCR) are washed over a picowell plate, followed by each nucleotide in turn; however nucleotide incorporation is measured not by pyrophosphate release, but the difference in pH caused by the release of protons (H<sup>+</sup> ions) during polymerisation, made possible using the complementary metal-oxide-semiconductor (CMOS) technology used in the manufacture of microprocessor chips (367). This technology allows for very rapid sequencing during the actual detection phase (365), although as with 454 (and all other pyrosequencing technologies) it is less able to readily interpret homopolymer sequences due to the loss of signal as multiple matching dNTPs incorporate (368).

This oft-described 'genomics revolution' has drastically altered the cost and ease associated with DNA sequencing. This growth occurs in excess of that seen in the computing revolution described by Moore's law – that the complexity of microchips (measured by number of transitors per unit cost) doubles approximately every two years – as sequencing capabilities between 2004 and 2010 doubled every five months (369). The various offshoot technologies are diverse in their chemistries, capabilities and specifications, providing researchers with a diverse toolbox from which to design experiments. However in recent years the Illumina sequencing platform has been the most successful, to the point of near monopoly  $(370)^5$  and thus can probably considered to have made the greatest contribution to the second-generation of DNA sequencers.

 $<sup>{}^{5}</sup>$ In 2012 the board of Illumina claimed that approximately 90% of the world's total sequencing output is produced using their machines (371).

#### 5.1.1.3 Third-generation DNA sequencing

There is some discussion about what defines the generations of DNA sequencing technology, particularly regarding the division from second to third (372, 373, 374, 375). Arguments are made that single molecule sequencing (SMS), real-time sequencing, and simple divergence (from what are currently accepted as second-generation technologies) should be the defining characteristics of third-generation; it is also feasible that a particular technology might straddle the boundary. For the purposes of this discussion, I am taking the most consistently agreed upon characteristic, in which third-generation technologies are those capable of sequencing single molecules, negating the need for the clonal amplification techniques that typify the previous generation. The first SMS technology was developed in the lab of Stephen Quake (376, 377), later commercialised by Helicos BioSciences, and worked broadly in the same manner that Illumina does, but without any bridge amplification; DNA templates become attached to a planar surface, and then propriety fluorescent reversible terminator dNTPs (so-called 'virtual terminators' (378)) are washed over one base a time and imaged, before cleavage and cycling the next base over. While relatively slow and expensive (and producing relatively short reads), this was the first technology to allow sequencing of non-amplified DNA, thus avoiding all associated biases and errors (372, 374). As Helicos filed for bankruptcy early in 2012 (379) other companies took up the third-generation baton.

At the time of writing, the most widely used third-generation technology is the single molecule real time platform (SMRT) from Pacific Biosciences (380), available on the PacBio range of machines. During SMRT runs DNA polymerisation occurs in arrays of microfabricated nanostructures called zero-mode waveguides (ZMWs) – essentially tiny holes in a metallic film over a chip – which exploit the decay properties of light passing through very small apertures to allow visualisation of single fluorophores (due to the zone of laser excitation being so small), even over the background of neighbouring molecules in solution (381). By depositing single DNA polymerase molecules inside the ZMWs and washing over the DNA library of interest along with fluorescent dNTPs, the extension of DNA chains by single nucleotides can be monitored in real time; after incorporation the fluorescent dye is cleaved away, ending the signal for that position (382). In addition to the benefits of sequencing single molecules in a very short amount of time (additionally producing kinetic data, allowing for detection of modified bases (383)), PacBio machines are capable of producing incredibly long reads, up to and exceeding 10 kb in length, which are ideal for de novo genome assemblies (372, 380).

Perhaps the most anticipated arena for third-generation DNA sequencing development is the promise of nanopore sequencing, an offshoot of the larger field of using

nanopores for the detection and quantification of all manner of chemicals and molecules, beyond the scope of just biology (384). The potential for nanopore sequencing was first established even before second-generation sequencing had emerged, when researchers demonstrated that single-stranded RNA or DNA could be driven across a lipid bilayer through  $\alpha$ -hemolysin molecules<sup>6</sup> by electrophoresis; moreover, passage through the channel blocks ion flow, decreasing the current for a length of time proportional to the length of the nucleic acid (385). There is also the potential to use non-biological, solid-state technology to generate suitable nanopores, which might also provide the ability to sequence double stranded DNA molecules (386, 387). Oxford Nanopore Technologies, the first company offering nanopore sequencers, has generated a great deal of excitement over their nanopore platforms GridION and MinION (388, 389), the latter of which is a small (mobile phone sized) USB device which was released to end users in an early access trial in 2014 (390). Despite the admittedly poor quality profiles currently observed, it is hoped that such sequencers represent a genuinely disruptive technology in the DNA sequencing field, producing incredibly long read (non-amplified) sequence data far cheaper and faster than we ever could before (384, 389, 391).

#### 5.1.2 Considerations for repertoire sequencing

Sequencing variable immune receptor repertoires presents different technological and scientific challenges to more typical sequencing efforts, such as genome sequencing or RNA-seq, which must be addressed.

The region of interest and target of the sequencing will most often be that encoding the complementarity determining region 3 (CDR3), due to its hypervariability and its role in antigen determination. By targeting the CDR3 and its flanking sequences one is able to obtain information about both the non-templated sequence and germline genes used in a particular rearrangement. While there may indeed be biological information located outside of this region – such as polymorphisms further towards the gene termini – this is usually less sought after, and would require a different sequencing approach.

Having selected the target, there are two further considerations: choice of nucleic acid substrate, and method of amplification. The latter consideration is required as the current generation of commercial DNA sequencing platforms (discussed in Section 5.1.1) require higher concentrations of nucleic acid than is easily obtained from appropriate biological samples.

<sup>&</sup>lt;sup>6</sup>Large ion channels from *Staphylococcus aureus*, that can remain open for extended periods.

TCR sequences differ markedly between its genomic DNA (gDNA) and messenger RNA (mRNA) forms (Figure 5.1A). Primarily, gDNA contains introns, the most significant of which resides between the splice donor site at the end of the J gene and the splice donor at the start of the constant region: depending on the TCR chain and genes involved, this sequence spans anything from 2.5 to 70 kb (inferred from IMGT LocusView (104)). After removal of introns by splicing during transcript maturation, the constant region forms a contiguous sequence with the end of the rearranged J gene. There will also be differences in copy number; the gDNA version will be present once per cell, while the RNA could be present at greater amounts (or even be absent). Indeed a number of papers have reported both temporary up- and down-regulation of TCR mRNA in T-cell cultures following treatment with different strength activation signals<sup>7</sup> (212, 213, 392, 393). As V(D)J recombination is predicted to only produce in-frame sequences one third of the time (due to the random addition and removal of nucleotides), sequencing gDNA would be expected to sequence more non-productive rearrangements than sequencing RNA, since mechanisms exist to prevent or downregulate expression of such failed rearrangements (discussed in Section 3.1.1). Therefore we would expect strategies that target mRNA as opposed to gDNA to be relatively enriched in productive TCR rearrangements. There are additional practical considerations to the choice of nucleic acid, such as the relative increased stability of DNA compared to RNA, or the availability of banked biological and clinical samples.

The challenge with amplifying TCRs lies in the fact that the different V genes can differ substantially in their sequence, thus providing no shared region from which to prime. There are two typical solutions. The first is to generate panels of different primers targeting each of the V genes in one direction, and either all of the J genes (Figure 5.1B, arrows) or the constant region (Figure 5.1B and C) in the other direction. These primers would then be pooled to run a multiplex PCR, in which multiple oligonucleotide pairs prime multiple targets in one reaction. The alternative option is to add a known adapter sequence to the 5' end of all cDNA molecules (with respect to the TCR coding sequence). This allows amplification of all rearrangements with a single primer pair complementary to both the adapter and the constant region (Figure 5.1C). The addition of a known adapter sequence can be achieved using a number of molecular biochemical techniques; such reactions are generally termed rapid amplification of cDNA ends, or RACE (394). Both techniques have merits and disadvantages. By targeting the J genes instead of the constant region in a multiplex PCR one is able

<sup>&</sup>lt;sup>7</sup>Although it is difficult to know what impact changes in cell size and number have had on these data.



Figure 5.1: How the choice of nucleic acid impacts on choice of TCR amplification protocol. A: Schematic of a rearranged TCR genetic locus, indicating the panels of V and J genes, the site of interest (CDR3) and introns (dotted lines indicate splicing events), including the large J-C intron, which can reach up to 68 kb (the distance from the furthest functional alpha chain J gene (TRAJ57) splice donor site to the splice acceptor site of the constant region (TRAC). B: Schematic of a multiplex PCR reaction to amplify rearranged TCR chains from gDNA, using a panel of V gene and J gene specific primers (green and yellow arrows respectively). C: Schematic of amplifying TCRs from cDNA (reverse transcribed either from constant region specific or poly-T primers), either in a multiplex PCR from V primers to J or constant region and a RACE adapter added to the 5' end of all rearrangements (blue and purple arrows).

to amplify directly from gDNA (avoiding the large J-C intron). A multiplex PCR also allows more control over how long PCR products (or 'amplicons') will be, which is important as different sequencing technologies have different sequence length requirements and capabilities. However when mixtures of primers are used there is potential for 'primer bias', wherein different primer pairs can amplify with different efficiencies, which would lead to bias in the output data (395, 396). There can be no primer bias in the a RACE approach, as each polymerisation event uses the same primers. Multiplex PCR approaches could additionally fail to capture the full spectrum of rearrangements of a given chain depending on whether or not all V genes have been included in the primer panel.

#### 5.1.3 Published TCR methodologies

In the few years in which adaptive immune repertoires have been explored using HTS both approaches have been applied to TCRs. Interestingly, despite the capability of multiplex PCRs to start with gDNA templates, a large number of groups with published protocols use RNA (83, 126, 130, 198, 271, 397). There is one notable exception to this, the protocol developed in the laboratory of Dr Harlan Robins, whose TCR repertoire sequencing papers using gDNA templates make up a substantial proportion of the literature in the field to date (398). They have additionally performed a number of experiments in a model TCR $\gamma$  multiplex system using synthetic DNA libraries covering all possible V-J combinations, allowing measurement and minimisation of the effect of primer bias by altering primer concentrations (399). As for 5' RACE techniques, the vast majority of published techniques make use of the 'template switching' effect (75, 77, 88, 97, 99, 127, 400). A template switch reaction exploits the terminal deoxynucleotidyl transferase activity of certain reverse transciptase (RT) enzymes, which under certain conditions will add short non-templated homopolymers to the end of the nascent cDNA chain (401, 402). Addition of oligonucleotides that contains a complementary 3' homopolymer sequence can then bind act as a primer, allowing the RT to swap strands and produce a DNA strand in the same orientation as the original  $mRNA^{8}$  (404).

When I began developing the TCR sequencing protocols used in this thesis there were very few published TCR repertoire papers, with even fewer publicly accessible datasets. The few publications that provided both protocol and data have been extremely useful, both as illustrations of how different groups have addressed the chal-

 $<sup>^{8}</sup>$ This is analogous to how the retrovirus from which the RT is derived replicates its own RNA genome (403).

lenges of repertoire sequencing, and as a source of data to use in benchmarking my own protocols. Accordingly I have described the process by which three such datasets were produced in Section C.1 of Appendix C.

#### 5.1.3.1 Our protocol; initial decisions

Our group elected to work towards developing a RACE protocol, i.e. extracting RNA from some source of T-cells, adding a known sequence to the V-proximal end of TCR cDNAs and then amplifying between that adapter and the constant region. The main justification for this was a desire to minimise all potential sources of technical bias. Additionally, as we are interested in the effects of infection upon the repertoire – rather than say studying the process of V(D)J recombination itself – we do not expect to need the non-productively rearranged sequences that will not be contributing to binding.

There were a number of other criteria for our eventual protocol to fulfil. It should be sensitive, in that it permits the detection of low-frequency clonotypes. It should be robust, and able to avoid or overcome error, which will always be produced at some level in a high-throughput system; an early paper that sequenced transgenic mice bearing single TCRs detected numerous sequences produced from through the accumulation of errors (405). Furthermore, if such a technology is ever to be of clinical use, it should also be as rapid and cost-effective as possible.

#### 5.1.4 A note on FASTQ quality scores

The analyses presented in this chapter deal with investigating and comparing different sets of high-throughput DNA sequencing data. A necessary component of such comparisons involves considering sequence quality scores. These are values that give an estimate as to the veracity of the individual DNA base calls. Although different sequencing platforms calculate their quality scores differently, the values produced are theoretically variants on the same principle established by the program Phred<sup>9</sup>. In this scheme, the quality (Q) of a given base call is a value logarithmically related to the probability (P) of that base call being correct, so that  $Q = -10 \times log_{10}(P)$  (407). In this framework, a quality score of 10 (Q10) means the probability of that base call being incorrect is 1 in 10, while Q20 corresponds to 1 in 100 and so on, allowing predicted accuracies to be calculated from quality scores. While different platforms still

<sup>&</sup>lt;sup>9</sup>Phred was developed alongside automated base-calling during the Human Genome Project, before even the invention of second-generation sequencing platforms (406, 407).

use this basic premise, due to the differences in error profiles calculation by the respective base-calling softwares, quality scores are not always directly comparable between technologies.

# 5.2 Sequencing development

In the course of my PhD I performed a number of iterations of sequencing protocol development, which informed and culminated in the protocol I used to produce the data described in Chapters 3 and 4.



# 5.2.1 Comparison results from published datasets

Figure 5.2: Example V-J usage plots for the previously published datasets Produced by PlotDCR (a modified version of the plotting modules from Decombinator v1.4). A and B: Alpha and beta distributions from the Wang *et al* paper (SRR030416). C and D: Beta VJ distributions from the Warren *et al* (SRR060699) and Bolotin *et al* (SRR494403) papers respectively.

I analysed three existing, published TCR-HTS datasets (described in Section C.1) using the bioinformatic pipeline I applied to the data I generated, in order to provide

controls. Table C.1 in Section C.2 of Appendix C shows the raw read and analysis numbers for three FASTQ files downloaded from each dataset, and example V-J usage information is shown in Figure 5.2.

## 5.2.2 Simple V amplification



#### 5.2.2.1 Protocol

Figure 5.3: Schematic detailing the amplification and library preparation steps involved in our initial simple V amplification experiment. A: TCR mRNA molecule schematic, with different regions approximately to scale. Rearrangements are reversetranscribed using oligonucleotides directed against the J-region proximal (5') end of the constant region (blue arrow). B: Rearrangements that make us of particular V regions are specifically targeted with specific V gene primers (green arrow) and relatively nested constant region primers (blue arrow). Partially double-stranded 'Y-adapters' for adapter-ligation library preparation also shown (right): each consists of two oligonucleotides which are only complementary at one end (red), which makes up the ligation interface. C to G: Standard adapter-ligation library preparation. Y-adapters are ligated to both ends of the now doublestranded amplicons (C and D), which results in different adapter sequences being present at either end of the resultant molecules (E), permitting paired-end sequencing from either end. Indexing and flow-cell binding adapter sequences can then be added by PCR (F), before paired-end sequencing of final libraries (arrows, G).

My initial HTS experiment was kept relatively simple. One TCR $\alpha$  and one TCR $\beta$ V gene – TRAV8-4 and TRBV12-3 – were selected, and amplified with specific V-region primers. These genes were chosen as they are found in the TCR chains of the Jurkat cell line<sup>10</sup>. Jurkat is a widely used, well-studied and easily grown leukaemic cell-line (410, 411), providing a ready supply of clonal TCRs with which I could validate the developing protocol. I therefore designed a multiplex PCR that would amplify all TCR rearrangements that use the Jurkat V regions (Figure 5.3A and B). After amplification, purified TCR amplicons underwent standard adapter-ligation library preparation (Figure 5.3B to G).

I also trialled using magnetic assisted cell sorting (MACS) to isolate different T-cell subsets. This technique relies on attaching paramagnetic beads to specific cell types via antibodies specific for particular cell surface markers, allowing retention of known cell types in a magnetic field (412, 413). I performed negative selection MACS on peripheral blood mononuclear cells (PBMC), whereby all cell types aside from those of interest are bound by antibodies and retained in the magnetic field, providing the sorted cell population in the eluate. Two donors' PBMC were sorted into CD4<sup>+</sup>CD45RA<sup>+</sup> and CD4<sup>+</sup>CDR45R0<sup>+</sup> populations, broadly representative of naïve and memory phenotypes (see Section 3.2.2.1). Two rounds of magnetic sorting were required to achieve the enrichment shown in Figure 5.4A, wherein  $CD45RA^+$  cells were sorted to 87.5% purity. and  $CD45RO^+$  cells to 98.2%. In addition to the two sorted donors, an additional donor's unsorted PBMC underwent the multiplex PCR. Figure 5.4B shows the products of the specific-V RT-PCR by agarose gel electrophoresis. Amplified bands were excised, purified, and sent to Professor Paul Kellam's group at the Wellcome Trust Sanger Institute, where the adapter-ligation was performed. Fully processed libraries were then sequenced on the Illumina HiSeq platform, producing two 100bp paired end (PE) reads.

#### 5.2.2.2 Sequencing results

Sequences produced were tested for the presence of recombined TCRs (explained later in Section 5.3). As Table C.2 shows, this protocol produced a wealth of data, averaging almost 13 million reads per sample, or approximately 1.4 gigabases (gB) of data per sample. I was however only able to detect rearranged TCRs in a very small proportion of those reads: on average only 0.32% of TCR $\alpha$  and 0.77% of TCR $\beta$  input reads seemed to contain TCRs, far below the observed assignment rate for the existing published data sets (Table C.1). Closer examination of the reads revealed that the read length of the data (100bp) is simply too short to detect the majority of rearrangements. In order to try and salvage as many reads as possible, I ran the data through FLASH, a program that merges overlapping reads in paired-end FASTQ data (117). This should allow

 $<sup>^{10}</sup>$ GenBank accession numbers X02592 (408) and K02885 (409) respectively.



TCR assignation for those clones whose V region could not be identified in 100bp, but whose total amplicon length is short enough to overlap enough to merge, giving enough sequence to make the assignation. While the rates of reads that merged was not large (approximately 15% on average, see Table C.2), they were greater than expected. These libraries were produced from gel-purified amplicons whose migration through agarose suggested sizes of approximately 300bp (Figure 5.4B). Given that the sequencing produced PE reads of 100bp each (and factoring in the 10bp minimum overlap that FLASH requires by default) the upper bound read length we should expect to see should be 190bp, significantly shorter than the apparent size of our bands and outside the area of the gel-purification. Plotting the distribution of the lengths in the overlapped reads reveals that 190bp is indeed the maximum length observed (Figure 5.4C). Due to the nature of this distribution, I would conjecture that either some nucleolytic process is occurring in between gel purification and library preparation, truncating amplicons at some low rate, or a fraction of shorter amplicons are migrating at faster than expected rates so as to be included in the gel purification, allowing them to be included in this analysis.

Whatever the cause of the short sequences, the rate of TCR assignation in the merged reads is greater than those reads that did not overlap, by at least two orders

Figure 5.4 (preceding page): Results of simple V amplification and HiSeq sequencing experiment. A: Example purities of 'naïve' and 'memory' MACS-sorted T-cell populations, by flow cytometry. Cell populations were first gated on CD3 and CD4. B: Agarose gels showing the PCR products that were excised and purified for sequencing. D =donor; N = naïve, M = memory. Left-hand lane in each gel shows Hyperladder I (Bioline): blue and green triangles indicate 600bp and 400bp bands respectively. Red arrows indicate the  $\alpha$  and  $\beta$  amplicons that were gel excised for purification (based on Jurkat sequences, approximate expected band sizes were 360bp and 300bp for  $\alpha$  and  $\beta$  respectively). C: Length distributions of extended reads merged by FLASH. D: Representative quality score barplot for the D2 raw FASTQ file, as produced by FastQC. E: Representative quality score barplot for the D2  $\alpha$  inter-tag FASTQ file, i.e. the FASTQ data for the reads that contribute to the Decombinator-assigned rearrangement, delimited by the outermost bases of the V and J tag sequences. FastQC plot features are: central red line = median, blue line = mean, upper and lower whiskers = 10% and 90% points, yellow box = 25-75% inter-quartile range. Green, amber and red backgrounds represent typical values deemed as 'good', 'passable' or 'poor' quality. F and G: Frequency plots produced by PlotDCR (a modified version of the plotting modules of Decombinator) showing the V-J pairing frequency for  $\alpha$  and  $\beta$  chain rearrangements respectively for a representative donor (D2). H: Example read indicating mis-priming (off-target amplification as a result of the bF3 primer binding a non-targeted site), resulting in presence of additional V genes in the HiSeq data. Green indicates V region identity, grey indicates primer sequence identity and asterisks indicate exact matches.

of magnitude (Table C.3). This corroborates the hypothesis that poor assignation in this HiSeq data is a result of the short read length, as longer reads contained more detectable rearrangements. I also observed that the overlapped reads displayed a lower average predicted frequency of errors, when considering the quality of the basecalls used to identify the hypervariable region (p-value = 0.0012, one-sided Wilcoxon rank sum test)<sup>11</sup>. This will partially be due to the nature of the FLASH merging algorithm, as lower quality scores can be discarded in favour of higher ones at positions covered by both reads (117). Overall read statistics for combined samples are shown in Table C.4. In order to assess the distribution of potential errors across the reads I visualised the quality scores using FastQC (108), a widely used FASTQ data checking tool. Both raw sequences and 'inter-tag' region FASTQ files – those that contain the sequence used by Decombinator to assign a TCR – were processed (Figure 5.4D and E). The inter-tag sequences show much higher and more evenly distributed quality scores. Illumina read quality is known to fall towards the 3' of the read (414), from a combination of loss of phasing of molecules within the cluster, loss of fluorescent signal intensity over time and changes in the cross-talk observed between fluorophores at later cycles (415, 416). The increasing quality at the start of the read is due to the Illumina base-calling software using the information from the first few cycles to calibrate the quality-determination for later cycles. Thus part of the higher and more even quality seen in the inter-tag is due to selecting from a higher quality internal subsequence, as well as the mitigation of quality loss from merging reads. Additionally, low quality 3' tails would be expected to contain an increased proportion of errors; due to the read length this is the area where the V tag for potential TCR arrangements would fall; we are thus selecting for higher quality reads as those with errors are less likely to be detected by Decombinator.

Figure 5.4F and G show representation frequency distribution heatmaps of  $\alpha$  and  $\beta$  V-J gene usage to provide an overview of the spread of TCR clonotypes. As expected we see a polyclonal response within the V genes targeted by the primers. Low frequency clones that do not use the targeted V genes are presumed to largely represent instances of mispriming. An example of such mispriming is shownin Figure 5.4H, which was identified as a rearrangement using TRAV9-1 amplified from the primer directed against TRBV12-3 (bF3). Mis-assignment by Decombinator could also produce such reads.

<sup>&</sup>lt;sup>11</sup>Note that there is no significant difference between the mean quality scores of these different groups, which is likely due to the fact that the Q score value is a linear representation of logged data.

#### 5.2.2.3 Verdict

While the Illumina HiSeq was able to produce in excess of ten million reads for each of our samples, the read length prevents it from being useful for our purposes. However, the quality of the data and the rate of assignation in merged (and therefore longer) reads suggests that the Illumina technology shows promise as a suitable platform could we but attain longer reads.

# 5.2.3 Poly-T RACE

#### 5.2.3.1 Protocol



Figure 5.5: Poly-T RACE strategy schematic. A: TCR cDNA, produced by reverse transcription as displayed in Figure 5.3A. B: Tdt supplied with dATP is used to polyadenylate the 3' end of the cDNA, to which a primer bearing a 3' poly-T sequence can anneal. Amplification occurs between the poly-T RACE primer and a relatively nested constant region primer. A 'VN-hook' is used to ensure that the poly-T primer anneals at the V-region end of the poly-A tract. C: Amplicons produced using poly-T RACE.

Having established various protocol parameters with the simple V amplification technique, I next moved towards a protocol that would allow amplication of all possible TCR rearrangements in a non-biased manner. This involved switching to a rapid amplification of cDNA ends (RACE) protocol, which allows amplification of a cDNA molecule in which only a single priming site (e.g. a TCR constant region) needs to be known through addition of another priming site to the appropriate terminus of the cDNA (394). In the first RACE trial I performed this adapter was added through treatment with Tdt<sup>12</sup>, supplied with a single deoxyribose nucleoside triphosphate (dNTP), which results in the homopolymer 'tailing' of the 3' end of the cDNA (and thus the 5' end of the original transcript) (25, 417). All such labelled transcripts can then be

<sup>&</sup>lt;sup>12</sup>This enzyme is coincidentally the same one required for non-templated addition of nucleotides during the TCR recombination process.

amplified using one primer against the original known sequence (the constant region) and another that contains in its 3' a homopolymer of the complement nucleotide to that which was used in the tailing reaction, which in my case was dATP (Figure 5.5).

Initial attempts to translate the reaction conditions of the single V PCR to the poly-T RACE PCR were unsuccessful, so I made a number of alterations. These included purifying DNA products in between various stages of the reaction, and addition of a 'VN' dinucleotide 'hook' to the 3 end of the poly-T primer – where V is any nucleotide apart from T, and N is any nucleotide – encouraging the annealing of the poly-T primer at the 3' end of the poly-T tail. The most significant improvement was seen in the switch to and subsequent optimisation of a reaction using Phusion DNA polymerase (New England Biolabs), replacing the Taq polymerase variant used previously. Phusion is a 'second generation' recombinant polymerase, engineered to increase processivity and speed through addition of a DNA binding protein domain (418). In addition to this increased processivity (i.e. how many nucleotides a given polymerase molecule incorporates before dissociation from the DNA) Phusion reportedly has higher fidelity, making fewer mistakes (419), which should reduce sequence errors. Phusion has even been reported to produce libraries which retain higher post-homopolymer quality sequences (by Sanger sequencing) (420), making it a credible suitable polymerase.

#### 5.2.3.1.1 Poly-T RACE cloning



Figure 5.6: Agarose gel scans showing amplicons produced using the preliminary poly-T RACE protocol. Original genetic substrate taken from either lysed Jurkat cells (A) or two healthy donors' PBMC (B). Left-hand lane in each gel shows 100bp Hyperladder IV (Bioline); blue and green triangles indicate 600bp and 400bp bands respectively. Red arrows in B indicate those bands that were excised and re-amplified before cloning into pUC19 for Sanger sequencing (see Appendix Figure C.1). Control lanes were: -RT (minus reverse transcriptase); -Tdt (minus homopolymer-tailing); W (PCR water control).

In order to test the efficacy of the optimised Phusion protocol I first amplified the TCR from RNA extracted from Jurkat cells. As Figure 5.6A shows, I obtained a single band for each chain, representing the alpha and beta rearrangements: Sanger sequencing of these PCR products corresponded to the published sequences for these chains. Application of this protocol to PBMC samples from two healthy donors revealed a ladder of bands, representing clonotypes of differing mRNA length and frequency (Figure 5.6B). In order to test whether this protocol was reliably amplifying genuine rearrangements, I excised bands from the gel shown in Figure 5.6B (red arrows), purified the DNA, re-amplified and cloned into plasmids to allow Sanger sequencing.

All normally sized TCR $\beta$  chains cloned from donor 7 (D7) were legitimate, productive TCR sequences (Appendix Figure C.1A and B). However, the D7 and D8 alpha chains (which had been chosen due to their aberrant band sizes) proved less typical.  $D7\alpha$  (Figure C.1C) produced the longest band from the poly-T RACE PCR, which turned out to be due to the V region being used by this rearrangement possessing a unexpectedly long 5 untranslated region  $(UTR)^{13}$ . This sequence does not appear to represent gDNA contamination, as the leader intron immediately upstream of the V region has been correctly spliced. Conversely, the D8 $\alpha$  amplicon (Figure C.1D) was the smallest distinct band, which yielded another peculiarly primed but correctly rearranged receptor. In this instance a molecule of the TRAC primer aRC1 seems to have lost the five 3'-most nucleotides (likely through the strong 3' to 5' exonucleolytic activity of Phusion), allowing amplification between two aRC1 moleculs (Figure C.1E). These results indicate that this protocol is capable of selectively amplifying TCR sequences. However they also indicate both the presence of unexpectedly long mRNA species and erroneous priming events. I also observed that two of the cloned sequences – Figure C.1B and another, non-productive rearranged clone (not shown) – contain more thymidine residues in the RACE homopolymer than were present in the primer. This is likely caused by polymerase slippage, a form of errors in which repetitive sequences cause the enzyme to dissociate, allowing the newly produced strand to shift in place before re-association, forming insertions and deletions (421).

#### 5.2.3.2 Sequencing results

#### 5.2.3.2.1 454

<sup>&</sup>lt;sup>13</sup>Note that this sequence belongs to a canonical MAIT cell sequence, underlining how frequent these cells are, as discussed in Section 3.1.5. Furthermore, this cDNA could potentially have been even longer; it contains a poly-A tract 668bp upstream of the start of its V region to which the our poly-T primer could bind, relatively nested to a potential more 5' Tdt-added adenine tract.



Figure 5.7: Summary data for the poly-T RACE amplification products as sequenced on the 454 GS FLX. A: Length distributions of the 454 data, for the  $\alpha$  and  $\beta$ read files pooled from all donors. The upper limit read length of 1600bp is the number of different 'flows' performed by the machine (where each nucleotide is washed over the wells of the plate in turn plate) (422). It is unlikely these rare sequences represent genuine TCR transcripts. B: Quality score barplot for the raw FASTQ file of a representative donor (D10,  $\alpha$ file), produced by FastQC (see Figure 5.4 legend). C: Quality score for the reads (sorted from all files) that contain the RACE homopolymer at the start of the read, showing the impact of the homopolymer on the quality of the read. Red line indicates approximate position of the homopolymer in the read (which is variable, due to the indels the homopolymer causes). D: Quality score for the inter-tag FASTQ sequence for D10  $\alpha$ . E and F: Frequency plots produced by PlotDCR showing the V-J pairing frequency for  $\alpha$  and  $\beta$  chain rearrangements respectively for D10.

After the low rate of TCR assignment observed using the Illumina HiSeq, I next opted to try Roche's 454 sequencing platform, which has been used in a number of early TCR repertoire sequencing strategies (83, 97, 129, 198). This technology is capable of much longer read lengths (350). Purified amplicons were produced from four healthy donors and again sent to collaborators at the Wellcome Trust Sanger Institute for sequencing. Amplicons pools were randomly fragmentated by sonication and subjected to standard library preparation (analogous to that illustrated in Figure 5.3B-G).

Read statistics from this run are displayed in Table C.5. The samples contained an average 19,000 reads each, below the 25-30,000 expected given the capacity of the plate<sup>14</sup> and how many samples we multiplexed. Of these reads, an average 41% were assigned TCRs by Decombinator. The 454 produces reads of differing lengths; the length distributions are displayed in Figure 5.7A. The quality score distribution in 454 reads (Figure 5.7B) mirrors that of the HiSeq data (Figure 5.4D). 454 qualities are also known to decrease towards the 3' of the read, and for similar reasons (accumulation of noise and loss of signal) (416, 423).

Due to the mechanism of the 454 base-calling software, its output is particularly prone to insertion and deletion errors ('indels') following homopolymer tracts (361, 368), therefore I assessed the impact of these sequences in this data. I pooled the reads from all donors and divided them into either orientation, i.e. those that contain the RACE homopolymer either in their 5' or 3' end. Despite theoretically having been produced in equal proportions during the PCR, reads that contained a 5' RACE homopolymer only accounted for 19% of the total. These reads showed a distinctly poorer quality profile than whole reads (Figure 5.7C), with quality scores declining precipitously after the approximate location of the poly-T. As in the HiSeq data, the quality distribution for the inter-tag FASTQ sequence is similarly improved relative to the whole read scores (Figure 5.7D), again suggesting that Decombinator does indeed select out higher quality reads. Figure 5.7E and F show the  $\alpha$  and  $\beta$  VJ gene usages for a representative donor; despite a spread of clonalities these distributions were different between donors, and did not resemble those observed in published datasets (Figure 5.2).

#### 5.2.3.2.2 MiSeq – adapter-ligation

In parallel to the 454, I also explored the then-newly available Illumina MiSeq platform. This makes use of the same Illumina reversible-terminator chemistry as the HiSeq, but with longer reads and a faster turnaround (424). TCRs were amplified with

<sup>&</sup>lt;sup>14</sup>Each run can produce up to  $1 \times 10^6$  reads per run, of which we were allocated a quarter.

the poly-T RACE amplification (Figure 5.5) followed by NEBNext adapter-ligation (Figure 5.3B-G) and run on the MiSeq.

Table C.6 shows the summary data for these original MiSeq test samples. Initial trials spiking small amounts of TCR library into runs again produced both fewer reads than expected and a lower rate of Decombinator assignation than in the 454 data (31% versus 41%). The read quality was also lower than the reported capabilities of the machine, lower even than unrelated samples sequenced in the same run. This appears to be due to the RACE homopolymer having a markedly deleterious effect on the quality of a given read; post-RACE-homopolymer sequences rapidly fall to the lower limit (Figure  $(5.8A)^{15}$ . The quality of reads that originate from the constant region of the amplicon show greater quality scores (Figure 5.8B), but still below the reported accuracy of the machine. The distribution of homopolymer tract length (Figure 5.8D) further indicates polymerase slippage. V-J pairing frequency (Figure 5.8E and F) again revealed a preference for the  $J\beta 1$  cluster, in contrast to previous published datasets (Figure 5.2B-D). This lead to an alteration of the beta constant region primers: instead of the one original primer ( $\beta$ RC1), which used two degenerate bases to cover the sequences of both of the TRBC genes<sup>16</sup>, two primers exactly specific for each TRBC were designed and mixed in equal concentrations.

#### 5.2.3.2.3 MiSeq – PCR library preparation

Neither 454 nor Illumina platforms appear able to efficiently sequence libraries containing the homopolymer introduced during the poly-T RACE protocol. However I observed that reads starting from the constant region end of the amplicons – those that contain the V(D)J information – tended to not suffer from this poor quality. Therefore we theorised that by sequencing the libraries unidirectionally from the constant region one could obtain all the clonal information without wasting reads by sequencing through the RACE homopolymer. I modified the library preparation protocol to add the Illumina sequencing and flowcell-binding elements specifically to both ends of TCR cDNA via a stepwise PCR protocol (as opposed to the random addition of sequences through ligation, as depicted in Figure 5.3B-G). This also offered the opportunity to use the constant region primer itself as the sequencing primer, permitting sequencing

 $<sup>^{15}</sup>$ The lower limit of Illumina quality scores is Q2, although this does not refer to a specific measured error rate. Instead it is used if successive base calls towards the end of a read are consistently low quality (e.g. Q15 or lower) to indicate that such sequences are probably not suitable for downstream analysis (425).

<sup>&</sup>lt;sup>16</sup>Accession numbers M12887 and M12888.



Figure 5.8: Results for the first poly-T RACE amplicons being sequenced on the MiSeq, having undergone NEBNext adapter-ligation library preparation. A: Quality barplots of sorted reads that were found to contain the poly-T RACE homopolymer, produced by FastQC (see Figure 5.4 legend). Red line indicates approximate location of homopolymer. B: Quality distribution of the sorted reads that did not contain the homopolymer. C: Quality barplots for the inter-tag FASTQ sequence, i.e. the FASTQ data that contributed to Decombinator-assigned TCRs. The quality of the inter-tag data increases because the Decombinator-delimited TCR FASTQ data has been converted to the correct orientation for transcription (i.e.  $V \rightarrow (D) \rightarrow J \rightarrow C$ ). D: Frequency of different length homopolymers of the sorted reads that made up the data summarised in A. The first 70 nucleotides of each read were searched for a 'seed' homopolymer consisting of eight matching nucleotides (setting the lower bound), and then any additional identical bases were counted. 90113 out of the total 292096 reads (31%) contained homopolymers of equal length to that of the primer. E and F: Frequency plots produced by PlotDCR showing the V-J pairing frequency for alpha and beta chain rearrangements.



Figure 5.9: Clustering on the MiSeq flowcells. A to C: Thumbnail images of DNA clusters produced by bridge amplification during different MiSeq runs. Images are taken from the adenosine channel for the first cycle of each respective run, but are representative of all channels. Images are produced by the MiSeq software, and sample different sections of the tile for an overview of cluster formation. A: An example of a successful run performed using the same versions kits on the same MiSeq machine, with a mean raw cluster density of 964k/mm<sup>2</sup>. B: The clustering of the first MiSeq run in which stepwise PCR was used to add the required sequencing elements  $(158 \text{k/mm}^2)$ . These clusters are diffuse and low-intensity, which likely contributed to low levels of cluster detection. C: The second run had PhiX genome spiked in to provide additional diversity produced greater density (391k/mm<sup>2</sup>). Given the poor rate of TCR assignment many of these clusters likely correspond to PhiX reads. D: Average raw cluster densities (black) and clusters that pass Illumina filters (e.g. chastity and phasing filters, red) for each iteration of initial MiSeq runs. 'AL' = adapter-ligated, i.e. the runs presented in Table C.6. TCR libraries constituted a minor proportion of these AL runs, so the majority of the clusters should be formed from the unrelated libraries (which were randomly sheared viral genomes). 'PCR' runs had Illumina adapter sequences introduced by stepwise PCR. 'PCRv1' was the initial attempt (sequencing off the TCR constant regions with custom primers), 'PCRv2' introduced PhiX spike-ins (and reverted to standard SP1 oligos) and 'PCRv3' introduced random nucleotides between SP1 and TCR constant regions, to augment initial nucleotide diversity for every cluster. The image shown in A was kindly provided by Anthony Brooks at UCL Genomics.

further into the V gene. Initial attempts suffered from low cluster density, and only a small proportion of those clusters passed all requisite filters. However visual inspection of images taken by the sequencer did not appear to show loss of clusters. Instead there was a relative increase in low-intensity, more diffuse clusters (Figure 5.9A and B), which might be less readily accepted by the detection software. Illumina basecalling software calibrates cluster detection based on the results of the first few cycles of imaging, assuming that there should be approximately equivalent proportions of each nucleotide at each step. This has lead to reports of problems when sequencing so-called 'low-diversity' libraries (426). In order to overcome such low nucleotide diversity, six random nucleotides were added between the constant region and sequencing primers, and highly diverse 'PhiX' libraries<sup>17</sup> were spiked in. Cluster densities improved following these alterations (Figure 5.9C and D), as did quality scores and rates of TCR assignation, although all values remained far below the reported capabilities of the machine. Binding sites for the primary Illumina sequencing primer (SP1) were also reintroduced, so as to be able to sequence under the most controlled and tested conditions. However even as number of raw reads per run increased, the proportion of reads that Decombined remained low; large amounts of the data was made up of reads that mapped to the PhiX genome, or were seemingly noise.

One hypothesis that could explain the lack of cluster generation is that the sequences required for bridge PCR and Illumina sequencing were not being correctly incorporated during library preparation. Individual amplicons were cloned from final sequenced libraries, revealing that all requisite sequences were present in the appropriate locations and orientations (Appendix Figure C.2). Therefore this method of library preparation does not appear to be responsible for the low cluster density.

Consultation with Illumina sequencing experts lead us to the conclusion that the most probable cause of the consistently low cluster density numbers (and hence poor read depths) was the RACE homopolymer. One possible explanation posited is that the RACE homopolymer might interact with the poly-T oligonucleotide sequences that are used to attach the P5 and P7 binding elements to the flowcell. This issue is seemingly separate from the poor quality in sequences downstream of homopolymers, and is independent of library diversity, as the problem was maintained even after adding leading random nucleotides and spiking in PhiX DNA. Thus the inclusion of such long homopolymers in the libraries likely precludes one from achieving optimal output.

 $<sup>^{17}</sup>$  Genomic libraries of the bacteriophage  $\phi X$  174 – a common control library in the sequencing industry – produced through random shearing and adapter ligation.

#### 5.2.3.2.4 Effect of native TCR homopolymers

Even without considering the RACE poly-T tract, we would expect there to be a certain number of homopolymer sequences occurring within TCR transcripts. Therefore I also assessed the extent of the these 'native' homopolymers and whether they affect the quality of our reads, for both the 454 and MiSeq data. I counted the number of homopolymers of four bases or longer in the prototypical alleles of the IMGT-described functional TCR genes, revealing that homopolymers occur in the majority of germline genes: over 350 were found spread across 132 of the 155 germline genes (Figure 5.10A). A similar search applied to a published, rearranged  $\alpha\beta$  TCR data set that did not use homopolymer RACE – which would contain any additional homopolymers produced during recombination – revealed that 88% of reads contained homopolymers of four bases or longer, and that those reads typically contained at least two such sequences (Figure 5.10B). This indicates that even without considering the effect of the RACE homopolymer the majority of reads in a TCR library can be expected to contain homopolymer sequences. Any TCR sequencing approach must therefore be able to cope with them. In order to assess whether these native homopolymers impacted upon quality, I first combined all of the reads for all samples of a given run. For the 454 data, reads containing the RACE homopolymer in their 5' were filtered out. A 200bp window downstream of the sequencing primer – that should contain any V(D)J information without reaching the poorest quality at the 3' of reads – was then extracted and searched for homopolymers of five nucleotides or longer. The last five nucleotides of any such homopolymer, and the next 15 bases, were then extracted. Random sequences of comparable length were also extracted from the same window of the reads for comparison. The median quality of 454 reads during a homopolymer are much lower compared to random sequences, requiring almost ten subsequent base calls to return to baseline (Figure 5.10C). In contrast, the median quality of the MiSeq data within and following homopolymers is unperturbed relative to the random sequence control.

#### 5.2.3.3 Verdict

The convenience of homopolymer-tailing 5' RACE in studying TCRs was realised shortly after its development (427), and was still being used for lower-throughput analyses at the time of our adopting it as a possible protocol for high-throughput TCR repertoire sequencing (428). Therefore in my PhD I trialled sequencing TCR libraries amplified using this approach on both Roche's 454 GS FLX platform, and Illumina's MiSeq platform.



Figure 5.10: The extent and effect of 'native' homopolymers (HP) in TCR repertoire sequence data. A: The number of homopolymers of four nucleotides or longer in the germ-line V and J genes of the human  $\alpha$  and  $\beta$  chain TCR loci. Only the prototypical alleles for functional genes have been included. 356 total homopolymers were detected, across 132 out of 155 germ-line genes. B: The number of homopolymers of four nucleotides or longer in the pooled data from Wang et al (83) (see Section C.1), which was not produced using a homopolymer approach, and sequenced using 454 technology. 1.634,849 homopolymers were detected, across 747,829 of the total 847,610 reads. C: The effect of native TCR homopolymers on the quality of the data. Homopolymers of five nucleotides or longer were detected, and the last five nucleotides plus the next 15 bases were harvested into separate FASTQ files, for both the pooled 454 data and that of the third, most productive MiSeq run (PCRv3). Due to having more reads (and homopolymers), the post-homopolymer subsequences from the MiSeq data were subsampled to match the number of detected 454 subsequences (31285 reads). An equal number of reads were then randomly subsampled from the initial pooled FASTQ files, and then random 20-mers were subsampled from these (from between positions 40 and 240, thus only sampling the higher-quality section of the read that should contain any V(D)J information while skipping the initial primer regions). Solid black lines represent median quality scores for each position, grey shaded areas represent median absolute diversity, dotted black lines indicate the end of the homopolymer sequences.

454 seems to be an inappropriate technology for the sequencing of TCR libraries, primarily due to a propensity for quality scores to decline following homopolymer sequences – indicative of a lower probability of base calls being correct. Given the regularity with which homopolymers occur natively within the TCR transcriptome, this represents a substantial source of potential error, theoretically producing a less accurate assessment of a sample repertoire. While producing reads that are long enough for our purposes, there are a number of other technical and practical reasons that make 454 sequencing a less viable option. 454 sequencing requires more DNA per library preparation than Illumina methods (several  $\mu$ gs as opposed to less than one  $\mu$ g), necessitating either more PCR cycles (increasing risk of error) or more substrate (which might be limiting in clinical samples). It is also consistently more expensive than Illumina sequencing (355) and has a slower turnaround. In controlled comparisons, Illumina based technologies typically produce fewer errors than 454, partially as a function of the greater depth afforded by Illumina<sup>18</sup> (360, 361, 368). The 454 propensity towards indels makes it additionally less suitable for repertoire sequencing. Such errors are less readily detected by Decombinator (which has mechanisms for correcting substitution errors located in V or J tag sites) and are more likely to turn a functional TCR sequence into a non-functional one (appearing as frameshift mutations), as opposed to just a different TCR sequence (which might then still translate to the same CDR3 sequence) (361, 368). This relative abundance of indel errors in raw TCR sequence data from the 454 platform (as well as the Ion Torrent platform, which operates in a similar pyrosequencing manner) has been noted previously, further suggesting Illumina as a more appropriate technology (126). Finally, even had the 454 quality profile been better, the choice to not pursue this technology in our sequencing program proved serendipitous, as in 2013 Roche revealed that it planned to stop selling and ultimately discontinue the technology, instead focusing on its later acquisition PacBio (429).

Sequence data produced using the MiSeq lacked the post-homopolymer decline in quality observed in the 454 results. It also possessed a number of features that are highly suited to our needs: it produces reads that are long enough for us to capture both V and J tags (permitting Decombinator assignation), doing so on runs that are both very quick to run and relatively cost-effective. However reads originating from the end of the amplicons that contain the RACE homopolymer – as in the 454 data – were of very poor quality, and therefore of little practical use. Efforts to rectify this

<sup>&</sup>lt;sup>18</sup>It is worth noting that the quality scores of the two technologies are not directly comparable due to differences in how the respective softwares calculate error: each is designed to detect their major source of errors, which is indels for 454 and substitutions for Illumina (354, 355, 416).

problem through unidirectional sequencing from the constant region failed to produce the volume of data that the machine should be capable of. In the case of the MiSeq, this problem was not due to low diversity libraries, as it persisted even with the addition of PhiX spike-ins and random nucleotides downstream of the sequencing primer.

Despite its ability to amplify a polyclonal population of sequences, TCR amplification through homopolymer-RACE consistently produced libraries refractory to both 454 and Illumina sequencing, and thus is not a suitable amplification process for high throughput repertoire analysis. This finding has implications for those groups who may have employed such strategies in the past for lower throughput cDNA analyses and are considering the transition to NGS.

#### 5.2.4 Ligation RACE

#### 5.2.4.1 Protocol



Figure 5.11: Ligation RACE strategies. Starting with TCR mRNA (A), we explored a number of ligation RACE strategies, including ligation of adapters to mRNA 5' caps directly (Ai and B), or reverse-transcription followed by adapter-ligation to cDNA 3's (Aii and C). Both produce equivalent adapter-labelled cDNA molecules (D).

In order to retain the ability of the poly-T PCR to amplify a wide range of Vbearing TCR rearrangements whilst improving on the efficiency of clustering in the bridge amplification, alternative 5' RACE strategies were tested. A number of 5' RACE strategies employ some form of oligonucleotide ligation to add a known sequence, offering an alternative to homopolymer tailing (430). The first such strategy we tried used RNA-ligation, wherein RNA oligonucleotides bearing the PCR priming site are ligated specifically onto fully processed mRNAs using an RNA ligase (431). Treatment of whole RNA from a sample of lysed cells with phosphatase removes the 5' phosphate group from any RNA molecules that lack a 7-methylguanosine (m7G) cap. Tobacco acid pyrophosphatase (TAP) is then used to 'decap' the fully-processed mRNAs, removing the m7G while leaving a 5' phosphate group, a requirement for ligation. Then addition of an RNA oligonucleotide adapter that contains a 3' hydroxyl group (but itself lacking a 5' phosphate) in the presence of an RNA ligase results in the attachment of the adapter to the 5' end of previously-capped mRNAs. Concurrent with the testing of the RNA-ligation RACE, various members of the laboratory – Prof. Chain, Dr Oakes and myself – also trialled several DNA ligation techniques, the most successful of which involved using an RNA ligase to attach a DNA adapter to the 5' UTR of our cDNA molecules (432, 433). Both of the two major tested ligation-RACE strategies are illustrated in Figure 5.11. Amplicons then underwent a comparable PCR-based library preparation technique to the initial MiSeq attempts.

#### 5.2.4.2 Sequencing results

Table C.7 shows the read statistics of the two major ligation protocols tested. Despite very high rates of TCR assignation using the RNA ligase protocol, quality remained relatively low. Moreover, the TCRs assigned displayed skewed V and J gene usage and pairing, particularly in the RL1 sample (Figure 5.12A and B). In contrast, the reads from the DNA ligation protocol are of better quality, with a median that remains above Q30 well past the 150 or so nucleotides typically required for Decombinator assignation (Figure 5.12C and D), with a V-J usage pattern far more in keeping both with our own and others' previous findings (c.f. Figures 5.8 and 5.2). We also note that the percentage of reads filtered out from the Decombined subset (i.e. those that contained ambiguous base-calls in their inter-tag region, or whose tags were too far apart to encode a genuine receptor) has dropped relative to that seen in earlier runs, again confirming a decline in the occurrence of errors.

## 5.2.4.3 Verdict

Initial trials with RNA and DNA ligation RACE approaches seemed to remedy the poor clustering and low quality reads that characterised the poly-T libraries. Despite giving slightly lower rates of TCR assignation by Decombinator than RNA ligation, DNA ligation appears to offer better quality data, with a gene usage profile that better fits those of previous methods and other laboratories' published data<sup>19</sup>. Therefore the DNA ligation RACE approach appeared to be the best candidate for further development.

<sup>&</sup>lt;sup>19</sup>It should also be noted that concurrent with testing these ligation protocols, Illumina modified both the sequencing kits and base-calling software to the effect that the technology copes better with low-diversity libraries (434).



Figure 5.12: Performance of the basic RNA and DNA ligation RACE protocols. A and B: Proportional V-J gene usage in the TRA and TRB data for sample RL1, prepared using the RNA ligation strategy. C and D: The whole read and inter-tag quality scores for the  $\alpha$  FASTQ files of sample DL2, prepared with the DNA ligation protocol. E and F: Proportional V-J gene usage in the  $\alpha$  and  $\beta$  data for DL2, produced by PlotDCR. The RNA ligation strategy was performed by Prof. Chain, while the DNA ligation sequencing was performed by Prof. Chain and Dr. Theres Oakes.

# 5.2.5 Randomly-barcoded V amplification (BV)

#### 5.2.5.1 Protocol

I next addressed the twin issues of sequence error and PCR duplication. No matter how effective an amplification procedure, all raw sequencing data will retain some errors (substitutions, insertions or deletions) that differ from the sequence of the original input molecules. While the nature of Decombinator analysis partially mitigates some errors (discussed later in Section 5.3), errors could still remain within the inter-tag region. PCR duplicates are an even harder problem to address: the heavy sampling intrinsic to deep-sequencing libraries followed by logarithmic amplification makes it very hard to know whether and how the frequency of reads belonging to a given TCR measured after sequencing reflect the true frequency of that TCR in the sample. Some groups have addressed this issue by using algorithms which cluster TCR sequences simply based on their similarity (76, 126). While this will reduce the number of artifactual TCRs, it could also combine genuine TCRs that just happen to differ by a small amount, thus removing actual diversity.

Beyond merely increasing diversity at the start of our reads, the introduction of random nucleotides to our libraries afforded an opportunity to identify and remove errors that accumulate during PCR and sequencing. The addition of random sequences prior to amplification can effectively label each initial input molecule, provided there are enough random sequences. After sequencing, these random sequences can be used as molecular barcodes: differences between TCRs sharing a barcode represent accumulated errors, while the number of different barcodes associated with a given TCR reflects the actual number of different TCR cDNA molecules that had daughter amplicons detected in the sequencing run. Such approaches have begun to be explored across fields that utilise HTS technology (435, 436, 437). The first paper to use the current generation of deep-sequencing platforms to sequence variable antigen receptor repertoires actually made use of such a technique (438), but the practice has not been widely adopted in subsequent protocols.

In order to test the feasibility of including such a technique in our protocols I again reverted to a simpler amplification, selecting a restricted number of V genes from which to prime, so that we could assess the results of 'barcoding' our libraries in a restricted dataset (see Figure 5.13). I selected the two V regions targeted in our initial simple V amplification protocol (TRAV8-4 and TRBV12-3), as well as TRAV21 and TRBV20-1, the genes which appear to often be used most frequently in the repertoires yielded from earlier experiments. Not only would this simplify the amplification procedure,



Figure 5.13: Schematic of the barcoded V (BV) amplification protocol. A: As in previous protocols, TCR RNA is reverse transcribed using oligonucleotides directed against the 5' of the  $\alpha$  and  $\beta$  constant regions (blue arrow). Red circle represents mRNA 5' cap. **B**: Second strand synthesis is achieved using oligonucleotides complementary to one of four specific V genes. Sequences 5' to the V-primer sequence add a random hexamer (6N) and an Illumina sequencing primer (SP2). C: A 'third strand' single round extension reaction incorporates another random hexamer at the other end of the amplicons (as well as SP1, the other sequencing primer, and an index for demultiplexing), completing the 12-mer barcoding of TCR cDNA. D: A four-cycle PCR is used to add: another index (that which is sequenced in the dedicated Illumina indexing read) and P7 at one end (i) and P5 at the other (ii). P5 and P7 are the elements required for cluster generation, as they bind to the oligonucleotides that coat the flow-cell, permitting bridge amplification. E: A final PCR directed against the P5 and P7 elements (for 22 cycles) amplifies full-length amplicons to sufficient concentrations for sequencing. Amplicons are finally purified, quantified, sized and normalised before sequencing on the MiSeq. Three sequencing reads are obtained per library: the first (R1) from SP1, the second (I1) from the reverse complement of SP2 (producing the standard Illumina index read) and finally from SP2 itself (R3). R1 contains the V(D)J information, and therefore is Decombined. R2 contains information pertaining to the other end of the amplicons, which includes 5' UTR (untranslated region) information. The concatenated first six bases of R1 and R2 form the random dodecamer barcode which is used to error-correct the sequence. R1 and I1 both contain index sequences, which allow multiplexing of multiple samples per run.
but would allow us deeper insights into the clonal hierarchy within a small number of V families.

Briefly, following reverse transcription a second strand extension reaction was performed, which selectively adds Illumina sequencing primer 2 (SP2) followed by a random hexamer to the targeted V regions. After removing unbound primers, a novel 'third strand' reaction is performed, wherein a single cycle extension is performed using an oligonucleotide bearing Illumina sequencing primer 1 (SP1) followed by a random hexamer and the J-region proximal constant region sequence, followed by removal of unbound primers. In this manner each molecule of cDNA (bearing the correct V regions) is labelled with a dodecamer of random nucleotides. There are approximately 16.7 million different combinations of random nucleotides in a 12-mer ( $4^6 \times 4^6$ ). Therefore when analysing RNA from small populations of T-cells<sup>20</sup> the probability of any two TCRs being labelled with the same barcode should be very small. Following 'barcoding' a short PCR introduces the final required sequencing elements and amplifies.

### 5.2.5.2 Sequencing results

This protocol produced a large number of reads, with good quality scores and high rates of Decombinator assignation (Table C.8); an average 61% of reads from each FASTQ file contained a rearranged TCR. The quality remains high well into the read (Figure 5.14A) and throughout the inter-tag region (Figure 5.14B). As with the original simple V amplification experiment that we ran on the HiSeq (Figure 5.4), I specifically observed the targeted V regions in the output data, albeit with trace amounts of other, non-primed Vs (Figure 5.14C and D).

### 5.2.5.3 Verdict

Only a small number of V genes were amplified and analysed in this manner so as to simplify the protocol, both to allow testing of the barcoding and gain a deeper insight into clonal structure within specific V genes. By efficiency of Decombinator assignation alone, barcoded single V amplification – combined with PCR library preparation and MiSeq sequencing – could be regarded as the most efficient protocol discussed yet, producing a large number of good quality, identifiably TCR reads. The major improvement however appears to be the error- and frequency-correction capabilities provided by the unique incorporation of random barcodes, which is discussed below in Section 5.3.

<sup>&</sup>lt;sup>20</sup>The few ml of blood from which RNA is extracted in these experiments likely contains fewer than five million T-cells, and only a fraction of the total RNA is used for sequencing.



Figure 5.14: Overview of barcoded simple V amplication protocol results. A and B: Representative FastQC quality score barplots for whole and inter-tag sequence FASTQs for a representative sample (HV4), sequenced on the Illumina MiSeq. C and D: V-J proportional usage heatmaps for sample HV4. V genes targeted in the PCR were TRAV8-4, TRAV21, TRBV12-3 and TRBV20-1. The effect of adding and 'collapsing' our sequences based on these molecular barcodes is discussed later in Section 5.3.

### 5.2.6 Barcoded Ligation RACE (BLR)

### 5.2.6.1 Protocol

The final step in producing a TCR sequencing pipeline that fulfilled the initial goals was to add the error-correcting functionality of barcoding to the pan-TCR amplifying DNA-ligation RACE protocol, combining the best aspects of the more successful of the previous protocols tested.

This new barcoded-ligation RACE (BLR) strategy consists of reverse transcribing mRNA, ligating on a barcoded-SP2 oligonucleotide, performing second and third strand syntheses (adding SP1 and a second random hexamer) and then proceeding with the second and final PCR steps as in the randomly-barcoded specific-V protocol. This protocol is shown in detail in Figure 5.15.

### 5.2.6.2 Sequencing results

The BLR amplification procedure proved highly effective, producing a large number of reads per sample, a large percentage of which contained TCRs (Table C.9). Furthermore, these reads are of very good quality, with a sequencing error predicated to occur on average only once per 3000 base calls, within the inter-tag region.

The quality profiles for representative files again show high quality data for the entirety of the portion of the reads that contain the V(D)J sequence (Figure 5.16A and B). Plotting the VJ gene usage reveals polyclonal repertoire usage (Figure 5.16C and D), comparable to that of samples produced by other laboratories (Figure 5.2). Moreover, these libraries are uniquely barcoded, permitting further error-correction and frequency estimation by 'collapsing' (see Section 5.3), which will further improve upon the quality and accuracy of the data.



Figure 5.15: Schematic of the barcoded ligation RACE (BLR) protocol. A: As in previous protocols, TCR RNA is reverse transcribed using oligonucleotides directed against the 5' of the  $\alpha$  and  $\beta$  constant regions (blue arrow). **B**: RACE is achieved through ligation of DNA adapter to the 3' end of the cDNA; adapter consists of Illumina sequencing primer SP2 and a hexamer of random nucleotides (6N). C: Separate single round second (i) and third strand (ii) reactions allow incorporation of another random hexamer at the other end of the amplicons (as well as SP1, the other sequencing primer, and an index for demultiplexing), completing the random 12-mer barcoding of TCR cDNA. D: A four-cycle PCR is used to add: another index (that which is sequenced in the dedicated Illumina indexing read) and P7 at one end (i) and P5 at the other (ii). E: A final PCR directed against the P5 and P7 elements (for 23 cycles) amplifies full-length amplicons to sufficient concentrations for sequencing. Amplicons are finally purified, quantified, sized and normalised before sequencing on the MiSeq. Three sequencing reads are obtained per library: the first (R1) from SP1, the second (I1) from the reverse complement of SP2 (producing the standard Illumina index read) and finally from SP2 itself (R3). R1 contains the V(D)J information, and therefore is Decombined. R2 contains information pertaining to the other end of the amplicons, which includes 5' UTR (untranslated region) information. The concatenated first six bases of R1 and R2 form the random dodecamer barcode which is used to error-correct the sequence. R1 and I1 both contain index sequences, which allow multiplexing of multiple samples per run. This is the protocol used to amplify and sequence the majority of the samples processed in this thesis.



Figure 5.16: Quality score boxplots and VJ gene usage heatmaps for a representative whole blood healthy donor sample (HV07), produced using the BLR strategy. A: Quality score distribution for the raw, complete FASTQ file. B: Quality score distribution for the inter-tag region (i.e. from the start of the V tag to the end of the J tag). C and D: VJ proportional frequencies for  $\alpha$  and  $\beta$  chain rearrangements.

### 5.2.6.2.1 Assess bias

While this protocol should be exempt from primer bias – as all rearrangements are amplified using the same primers - it is possible that bias gets introduced during the ligation step, perhaps dependent on terminal nucleotide or V gene identity. In order to address whether this was the case I first compared V gene usage profiles between three healthy donor repertoires produced using the BLR strategy and those of previous publications (summarised earlier in Table C.1 and Figure 5.2). The proportional usage of each TRBV gene was calculated for each repertoire sample, and these distributions were compared against all other samples both within and between different protocols, using the Jensen-Shannon measure of divergence (JSD, Figure 5.17A). Intra-protocol comparisons (i.e. comparing the results of different samples produced with the same method) produce low and consistent levels of divergence. Of the intra-protocol comparisons, the Warren and Bolotin samples are the most internally conserved (yellow and black), the Wang samples are the most divergent (red), while the BLR samples are intermediate (cyan). We would expect the BLR samples to be more divergent as each sample came from a different donor, while all of the data from the previously published papers represent repeats from the same bleed of one donor; the Wang data therefore appears to be the most inconsistent with regards to V gene usage. These findings are reflected in the inter-protocol comparisons: the Warren and Bolotin samples are the least divergent from one another (gold), either of these compared against the BLR data are slightly more divergent (green and navy), and then any comparison against the Wang data is yet still more divergent (purple, orange and brown). Therefore we can deduce that in terms of usage profiles of the different TRBV genes, my data appears to be within the divergence range of existing published data sets.

In order to look specifically at whether the DNA ligation process engineered a specific bias, I made use of the data produced from the second of the paired-end Illumina reads<sup>21</sup> (see Figure 5.15E). As the ligation oligonucleotide consists of the binding site for the R2 sequencing primer (SP2) followed by six random nucleotides, the first six bases of the actual reads will correspond to the ligation-introduced barcode, while the seventh base onwards will represent the 5' end of that TCR transcript. I measured the proportions with which each base is used at each position across the start of R2 (Figure 5.17B). I observed a noted bias in the prevalence of different bases in the random hexamer, particularly in the first and last positions (-6 and -1). There seems to be very little usage of guanosine, which corresponds to cytosine being used infrequently

 $<sup>^{21}\</sup>mathrm{The}$  'R2' (read 2) data.



in the adapter molecules at these positions. Similarly there exist large preferences in the bases at the start of the actual transcript read, particularly in the first position – which contains a guanosine in over half of the reads – which normalises around position seven or eight. By measuring the proportion with which different dinucleotide pairs occurred across the ligation junction, I could assess which bases were actually being ligated together. Comparing these pairings to dinucleotides at other sites either at the start of the random hexamer or within the early transcript read revealed a markedly different pattern of dinucleotide usage across the different positions, with a particular preference at the junction site for dinucleotides ending with a G residue (Figure 5.17C). This was likely explained by the exceeding abundance of Gs at the +1 position observed in Figure 5.17B. To control for this, the frequency of dinucleotide pairs was normalised by dividing by the frequencies of the individual bases at each position, giving a truer representation of how likely a base is to pair with another. Figure 5.17D reveals that at the start of the random barcode (-6 to -5), there is practically no preference in base pairing, and thus the prominent inequality of bases at the -6 position observed in Figure 5.17B is likely some artefact of the oligonucleotide generation. At the ligation junction (-1 to +1) there is a slight loss of uniformity in the base pairing observed for dinucleotides that begin with a guanosine, although as Gs make up less than 10% of bases at the last position of the adapter this finding could reflect a sampling artefact. The two positions from later in the read show a similar pattern of dinucleotide usage

Figure 5.17 (preceding page): Assessment of bias in the BLR protocol. A: Inference of potential bias by comparing differences in V gene usage distributions, as measured by Jensen-Shannon Divergence (JSD). Samples used were three healthy volunteer (HV) samples produced using the BLR protocol, as well as three samples each from three previously published data sets (see Section C.1). All possible pairwise sample combinations were compared and JSD calculated (with a greater JSD meaning a greater difference in V gene usage). Horizontal lines indicate group means. B: Proportional usage of each nucleotide position in the read 2 (R2) data (see Figure 5.15E), for those paired-reads that successfully assigned a TCR in their R1 reads (for the same HV samples used in A). Note that the first six base-calls of every read correspond to the random hexamer barcode sequence incorporated at that end of the molecule (-6 to -1), which would then be followed by the 5' UTR sequence of the TCR (1+). C: Raw proportions of all the different possible dinucleotides, at the four sites underlined with black bars in  $\mathbf{B}$ , corresponding to the start of the random hexamer (-6 to -5), the interface between the ligation oligo and the cDNA molecule (-1 to +1) and two sites within the 5' of the transcript (4 to 5 and 10 to 11). D: Normalised proportions of the dinucleotide proportions displayed in **C**, where the value of each dinucleotide is divided by the proportion of each of the individual nucleotides at their respective positions (and divided again by 16, the number of dinucleotide combinations, in order to provide an equivalent scale to  $\mathbf{C}$ ). All error bars indicate standard deviation.

to each other that is different from the other two positions, likely reflecting the innate biases and motifs of the human transcriptome. For example we observe a very low frequency of CG dinucleotides, which is a well-documented feature of animal genomes due to the propensity of methylated thymines to deaminate to thymidine residues (439). Therefore at the site of ligation we do not appear to detect a particular bias that should preclude or influence our protocol from detecting any particular TCR transcript based on its 5' primary sequence.

### 5.2.6.3 Verdict

The barcoded ligation RACE strategy fulfils the required criteria: it produces high quality data (in the expected ranges of the machine), from which one can obtain a large number of polyclonal TCRs. Furthermore, as each cDNA molecule gets uniquely labelled with random barcodes one is able to pass output Decombinator data through an error-correction process, further improving the qualitative and quantitative aspects of the repertoire data.

I have demonstrated that the ligation occurs without an obvious bias. The observed imbalance in the non-templated nucleotides in the ligation adapter does not necessarily reflect on the likelihood of bias in our system, as the production of 'random' sequences in synthetic oligonucleotides is not truly random (440), a feature that is often observed in transcriptome sequencing efforts that use random primers for reverse-transcription (441). Similarly the differential base preferences we observe here at the start of TCR transcripts, particularly in the first position – where there is an extreme preference for guanosine – are not themselves indicative of bias. Despite a great deal of heterogeneity, there are a number of biased motifs that appear in or around transcriptional start sites (TSSs), which often result in an increase production of G residues in the first bases transcribed (442, 443, 444). There is also an additional methodological explanation for the pronounced G bias in the first base of our transcripts, as a result of using a Moloney Murine Leukemia Virus (MMLV) derived reverse transcriptase (RT) enzyme. MMLV derived RTs possess terminal deoxynucleotidyl transferase activity, which results in the addition of several non-templated bases after reverse transcribing an mRNA; however this activity has a strong bias for the inclusion of cytosine residues<sup>22</sup> (401, 402). Even in reaction conditions such as these which aren't optimised for such extension there is often a single or double C residue added (401). As the sequences measured in Figure

<sup>&</sup>lt;sup>22</sup>Incidentally this enzyme activity provides the basis for template-switching amplification protocols used in other TCR repertoire RACE studies, as a poly-G containing oligonucleotide can anneal to this tail and prime the second strand reaction.

5.17 are in the complementary strand with respect to the cDNA, any C residues that the RT incorporates to the cDNA would contribute to the G bias we see in the +1 and +2 positions. When I normalise for the frequency of each nucleotide either at the end of the ligation adapter or the start of the transcript we observe that there is very little bias in nucleotide pairing (and no pairing that does not occur at all). This is supported by a report which detailed that T4 RNA ligase does not show preferential bias when considering the primary structure of cDNA adapters 3'-ligated to RNA molecules (a process which was significantly improved through the addition of random nucleotides at the end of the adapter) (445). Therefore all and any TCR cDNA species could theoretically be labelled without preference using this protocol.

Due to the success of this protocol, I applied it to the major sample sets of this thesis, those investigating the global TCR repertoires of healthy individuals and HIV patients, as described in Chapters 3 and 4. However, the protocol has subsequently been further refined within the group. Improvements include combining certain steps to reduce the number of bead purifications that are required, which should reduce the number of lowfrequency TCR cDNAs lost throughout the protocol. We are additionally exploring the use of quantitative, real-time PCRs in the amplification protocol, to allow us to halt the reaction after an optimal number of samples. In this way we can reduce the number of cycles to the absolute minimum required to produce sufficient concentrations for sequencing: the fewer cycles of PCR, the less chance there is for any errors to occur.

## 5.3 Bioinformatic development

"Screening peripheral blood T cells for the expression of up to 20 different  $V\beta$  families ... will provide an extensive characterisation of TCR  $V\beta$  expression. However, this analysis would also require large numbers of cells ... and generate massive amounts of data, which could quickly make the analysis impossible to perform."

Jeff Faint et al

From 'Quantitative flow cytometry for the analysis of T cell receptor  $V\beta$  chain expression', Journal of Immunological Methods, 1999 (56).

### 5.3.1 TCR repertoire analysis with Decombinator

At the outset of this project, there existed very few published descriptions of TCR analyses, and even fewer publicly available programs. Where described, most methodologies primarily made use of pairwise alignment algorithms to detect the presence of particular V, J (and sometime D) regions, such as: DNAPlot (as used in IMGT/HIGHV-QUEST) (446, 447), BLAT (198, 448), BLAST, (97, 449) and SWA (83, 271, 450). Whilst effective, these methodologies suffer from a relative lack of speed when compared to certain other techniques which might be applied, limiting the rate of analysis of large data sets. To overcome this hurdle, then-PhD student Dr. Niclas Thomas in our group developed a novel TCR analysis suite named Decombinator (111), which has been used for all of the TCR rearrangement analyses in this thesis. This makes use of an extremely efficient string matching algorithm described by Aho and Corasick (451), which is widely used across both bioinformatics and numerous other computing-related fields (452).



Figure 5.18: Schematic detailing an Aho-Corasick string matching automaton. The trie is generated prior to searching, based on the four keyword patterns (or tags) to search for. The trie is then applied to the search string (or read). The traversal of the trie is illustrated in red. Starting at the leftmost node (the root), the branch to navigate to is decided based on matching the next character in the string. If the criteria to continue along the branch is not met, the automaton instead navigates along the failure functions (leftwards arrows), using the longest suffix so far (e.g. the GC on the keyword 3) to find the longest prefix match (the C at the start of pattern 2). In this example, one pass of the search string will yield matches for patterns 2 and 1, in that order.

In our implementation, Decombinator uses the Aho-Corasick (AC) finite state automaton to rapidly search a given sequencing read for the presence of one of a list of 'tag' sequences, the presence of which should uniquely identify a particular V or J gene. Finding both a V and J tag allows full description of a TCR rearrangment using a five-part classifier consisting of: which V and J genes were used; how many nucleotides were exonucleolytically removed from each of these genes, and what the intervening 'insert' DNA sequence is (from the end of the V to the beginning of the J). From these five characteristics the entire DNA sequence of the rearranged chain can be recapitulated. Note that we do not explicitly assign a D region (for  $\beta$  or  $\delta$  chains), as these are short sequences that are often typically too deleted to unambiguously assign (75); any remaining D region sequences will be contained within the insert sequence. Compared to raw sequencing data, the classifiers produced by Decombinator are easily stored, transferred and manipulated in downstream processes.

The speed of AC string matching lies in how it finds sequences: before traversing a string, it uses the panel of keyword patterns (i.e. the V or J tags) to generate a trie (Figure 5.18). When passing through the string to be searched (the sequencing read) the finite state automaton navigates the tree node by node, depending on which nucleotide is present at the next position. The AC algorithm simultaneously looks for each subsequence in one pass of the read, and in the event of a mismatch uses the longest suffix matched at any one time to skip to the longest possible prefix match in another branch of the trie. These features combine to produce the incredible rate of V/J gene assignment we see with Decombinator, orders of magnitude above that seen using pairwise alignment (111). As a direct pattern-matching algorithm, an error within the tag sequence itself would ordinarily preclude assignation. To overcome this limitation, if complete V or J tags are not found, Decombinator splits each tag into two half-tag sequences and searches for these. Successfully finding a half-tag is followed by checking the neighbouring sequence where the rest of the full tag would ordinarily lie against the relevant partner half-tag: if the comparison of these two sequences produces a Hamming distance of one (i.e. a one nucleotide mismatch) the V or J can be assigned.

#### 5.3.1.1 Generation of new Decombinator tag sequences

In the course of this project I made several contributions to Decombinator, including the generation of human V and J  $\alpha$  chain tag sequences.  $\alpha$  tags were designed in a similar method to that used by Dr. Thomas for generation of the  $\beta$  tags (and later  $\gamma/\delta$  tags): complete sequences for the prototypic allele of all functional TRAV and TRAJ genes were downloaded from IMGT/GENE-DB (19), and then potential 20mer windows located 20 nucleotides away from the recombining edge (to allow room for deletion) were assessed for uniqueness. If potential tags for multiple genes were identical, the tag window was shifted and uniqueness re-assessed. Developing tags for the  $\alpha$  chain is simpler than for  $\beta$  in some respects, as the V genes share less identity, demonstrable as fewer TRAV genes fall into subgroups<sup>23</sup>: there are only seven subgroups containing 20 members in the TRA locus compared to 16 subgroups totalling 60 members in TRB (19, 453). Therefore in the tag set originally published with Decombinator, only two TRAV genes – TRAV8-2 and TRAV8-4 – are included under one tag. I also produced tags for the three  $\delta$  chain V genes that reportedly do not recombine with  $\alpha$  J genes, for reasons discussed in Section 3.1.2.

Tags were originally designed to cover only the prototypical alleles (i.e. the first one discovered) for all TCRs with 'functional' gene status in IMGT, thereby ignoring those that were not thought to contribute. However, when investigating sequences that failed to be assigned by Decombinator, I observed that a proportion contained rearranged TCRs that make use of these other genes (as described above in Section 3.1.2). I therefore designed a new set of tags that covered the prototypical alleles of every  $\alpha$  and  $\beta$  V or J gene for which IMGT has an entry, regardless of functionality. As shown earlier in Figure 3.2, a number of these are indeed used in multiple donors, and are thus worthy of inclusion for recombination level analyses at least. In the design of these extra tags, I also strove to maintain at least a minimum distance of 20 nucleotides between the start of the tag and the end of the gene (where possible) to minimise the possibility of tags being altered during recombination. By being able to move further into the V region (thanks to longer read lengths from the MiSeq) I was also able to design tags with increased resolution between related genes, lowering the chance that errors could result in misassignment. In order to check whether this new extended tag set was still able to recognise the same original clones, I ran Decombinator on sample healthy individuals and compared the frequency at which I found each gene. As Figures 5.19A and B and C.4 show, the extended tags are typically able to find either equivalent or greater numbers of sequences. There are some exceptions, such as TRAV1-1, the loss of which was found to be due to intra-V gene splicing to a cryptic acceptor site (see Section 3.3). Comparison of the frequency of individual clones in each tag set is remarkably conserved (Figure 5.19C and D). The new tag set therefore largely recapitulates the findings of the original tags, save with increased resolution in

 $<sup>^{23}\</sup>mathrm{Subgroups}$  are classified as families of germ-line genes which posses greater than 75% nucleotide identity.



a small number of genes while permitting detection of genes which may be involved in recombinations, regardless of their official IMGT status or expression at the cell surface.

#### 5.3.1.2 Limitations of standard Decombinator analysis

As with any analysis tool, Decombinator suffers from a number of small limitations which must be taken into consideration when reviewing data produced using it. The foremost is one of accuracy. Whilst Dr. Thomas demonstrated a 98.8% accuracy rate when using in silico sequences allowed a 0.1% error rate (111) – a value on the order of that which we might expect in an Illumina run (361) – errors are not equally distributed among reads (405). Therefore it is probable that certain reads accrue enough errors in the tag regions so as to be assigned incorrect V or J genes. Alternatively erroneous production of a tag sequence at an inappropriate location in the read might result in a sequence that contains a genuine, error-free recombination elsewhere in the read being overlooked. Moreover, an error in the insert sequence cannot be corrected by any comparison to germline, as this region contains non-germline-encoded information. A 2013 publication by Bolotin et al showed that in their hands Decombinator produced a number of erroneous additional CDR3 sequences upon translation, despite scoring very accurately on V and J assignation, implying that errors within the insert sequence are artificially inflating diversity with false sequences (454), although this analysis made use of an early version that lacks more recently added error-filters.

Tag-based gene assignation results in specific limitations based on the design of the tags. With the inclusion of the extended tag set described in this thesis, Decombinator is now able to detect rearrangements using a larger panel of genes, regardless of functionality. However many TCR genes have multiple known alleles. If single nucleotide polymorphisms (SNPs) occur within tag regions certain alleles may be less likely to be detected. Conversely, alleles might differ in their ability to recognise antigen (recorded for at least one TRAV and one TRBV gene (455, 456)) but remain conserved in their tag region, which might affect correct data interpretation. Tag placement presents another problem: despite being typically at least 20 nucleotides away from the recombining

Figure 5.19 (preceding page): Comparison of original and extended **Decombinator tags.** Plots of a representative healthy donor (HV09) showing the comparative performance of Decombinator in combination with the original and extended tag sets. A and B: Frequency of Decombinator rearrangements using particular TRAV and TRBV genes, when using either the original or new extended tag sets. C and D: Correlation of recombination frequency for  $\alpha$  and  $\beta$  chain rearrangements.

edge of the germline gene, it is possible that a rearrangement could still have a greater number of deletions and remain productive, yet be undetectable to Decombinator currently. A related issue is that some V genes are simply too homologous in their 3' sequence, and thus one is unable to assign tag regions that are able to distinguish them.

These problems are largely soluble through the production of different panels of tags. For instance, if one wished to distinguish between different allelic forms of V and J genes, tags could be generated for each allele. Similarly, recognition of hyper-deleted genes could be achieved through selection of tags further away from the recombining edges. Both of these approaches would however require longer reads of higher quality than are routinely achievable. Other problems remain, such as how to deal with potential sequence errors remaining in the insert sequences. I have addressed this in this work through the unique labelling of transcript molecules, as described in Section 5.3.2.

# 5.3.2 'Collapsing' TCRs: error-correction and improved frequency estimation



**Figure 5.20:** Schematic detailing the need for error-correction. Following processing, a labelled transcript pool (left) accrues PCR and sequencing errors (right), and differential amplification or sampling may result in changes to the frequency hierarchy.

Both PCR and deep-sequencing (which invariably requires additional PCR steps) produce sequence errors that may artificially inflate the diversity and disrupt the heirar-

chy of a repertoire sample (Figure 5.20). As discussed in Sections 5.2.5 and 5.2.6 above, the addition of random molecular barcodes to cDNA molecules prior to PCR amplification provides us the ability to mitigate for PCR and sequencing errors and duplications. In order to take advantage of this feature, I developed an alternative TCR analysis pipeline. The first step involves concatenating the two random hexamer sequences from the start of both sequencing reads (see Figure 5.15E) to form the 12-mer barcode. Next, a modified version of Decombinator analyses each read, outputting the sequence and the quality information of this barcode as well as the inter-tag region, in addition to the standard five fields. The product of this 'verbose' Decombinator (vDCR) can then be fed into the final script in the pipeline that performs the errorcorrection and frequency-estimation. 'Collapsing' in this context refers to the reduction both in frequency and number of clones, by both mitigating PCR duplication and removing clones produced by PCR or sequencing errors. This is achieved in two steps. First, all assigned TCRs that share the same barcode are examined in turn; the most frequent nucleotide sequence is presumed to be that which most likely represents the genuine transcript. Any reads that share the same barcode sequence yet do not match this prototypical sequence are thus likely to be errors and can be ignored. The second step is to count the number of different barcodes that associate with a given rearrangement – after performing some clustering to account for errors in the barcodes themselves - thus providing a far more accurate estimation of the true frequency of that transcript than the read count. A flowchart detailing these steps is shown in Appendix C, in Figure C.3.

With the addition of the error-correction software – named CollapseTCR – the entire bioinformatic pipeline as used in this thesis is summarised in Figure 5.21A. Collapsing repertoire data leads to approximately twofold and fourfold reductions in the the number of unique TCR sequences for the BLR and BV protocols respectively due to removal of erroneous TCRs, and a 40-fold and 170-fold decrease in total number of rearrangements, due to mitigated PCR duplication (Figure 5.21B). An example of the effect this might have for a given rearrangement is illustrated in Figure 5.21C, where I have shown all of the unique Decombinator assignations that were associated with a particular random barcode sequence before collapsing. Due to the the large number of possible barcodes (which is far in excess of the number of TCR cDNA molecules we would expect in samples of this size) the expectation would be that all of these classifiers therefore represent the PCR progeny of one original founder molecule. In the pattern typically displayed in all samples, there is one rearrangement that is far more frequent than all others, with a number of lower frequency but highly similar

A i	<pre>@HWI-M01520:79:00000000-a7tuF:1:1101:15622:1335 CCCTCCTACCTCCTTGTAGGAACTGGTGGGAACACGTTTTTCAGGTCCTCTGTGACCGT +</pre>	Amplified libraries sequenced on Illumina
	1>>AA13>>>AFAFFFFFCCCBBFGGGGGGGGGGGGGHFGHHGGHHEFHHHHHHHGFGHCG	MiSeq, FASTQ output
ii	6, 12, 1, 5, GTGGGGGGGGCCG, [ID], [TCR-seq], [TCR-qual], [barcode-seq], [barcode-qual]	TCRs assigned using verbose DCR, which additionally ouputs TCR and barcode (6N+6N) sequence information
iii	6, 12, 1, 5, GTGGGGGGGCCG, 560	Assigned TCRs error- corrected/frequency- estimated based on their barcodes, using CollapseTCR
iv	CASSLGGGADEQYFGPG	Error-corrected TCRs can further be translated and CDR3 regions extracted



# Unique rearrangements

С	69x:	З,	33,	8,	7,	CCAGGCGGGGGGGG
	2x:	З,	33,	8,	9,	CCAGGCGGGGGGGGGCA
	2x:	З,	33,	8,	7,	CCGGGCGGGGGGGG
	2x:	З,	33,	8,	7,	CCAGGGGGGGGGGGG
	1x:	4,	33,	9,	7,	TCCAGGGGGGGGGGGG
	1x:	З,	33,	9,	7,	GCCAGGGGGGGGGGG
	1x:	З,	33,	9,	7,	CCCAGGCGGGGGGGGG
	1x:	З,	33,	8,	7,	CCAGGGGGGGGGGG
	1x:	З,	33,	8,	7,	CCAGGGGGGGGGG
	1x:	З,	33,	8,	7,	CCAGGCGGGGGGGG
	1x:	З,	33,	8,	7,	CAAGGCGGGGGGGGG
	1x:	З,	33,	11,	7,	GGTCCAGGCGGGGGGGGG

rearrangements. In this particular example, there were 83 reads all associated with the same barcode, which between them comprised 20 different inter-tag DNA sequences producing 12 unique Decombinator classifiers. Therefore instead of the 12 different clonotypes we would naïvely interpret as having a cumulative frequency of 83, collapsing allows us to disregard the sequences that have emerged through PCR amplification and error, revealing the single, accurate TCR representation.

Plotting the relationship between raw and collapsed frequency – at least for those genuine sequences that survive collapsing – reveals that the overall correlation is strong (Figure 5.22). However, variance is much greater for lower frequency clones, so that one is less able to reliably infer the collapsed frequency of a TCR if its raw frequency is below a certain threshold. Depending on the sample, this threshold seems to lie somewhere between 100 and 1000 copies, which likely reflects the various sampling processes involved. The frequency collapsing data can also be expressed as a duplication rate, calculated by dividing the raw observed frequency of a TCR chain by its eventual error-corrected, collapsed frequency, telling us to what extent that sequence was multiplied. I observed a similar pattern of duplication rates across the different samples; a representative example is shown in Figure 5.23A. Low frequency recombinations exhibit a wide variance in duplication rates, as might be expected given the large spread of low frequency clones illustrated in Figure 5.22. Moreover these low frequency data fall into one of a number of gradients, which is determined by their collapsed frequency (the denominator of the duplication rate). However, similar to previous findings, above a certain size (an approximate raw frequency between 500 and 1000, or a collapsed

Figure 5.21 (preceding page): The global effect of collapsing TCR repertoires. A: Schematic of the various steps of the full bioinformatic pipeline used in this thesis. i: FASTQ data as produced by sequencing our amplified libraries on the Illumina MiSeq, which contains both nucleotide sequence and base-call quality information. ii: TCRs are assigned by vDCR, containing the typical Decombinator five-part classifier in addition to the sequence and quality information for both the inter-tag region as well as the barcode region (a concatenation of the two random hexamer sequences that were uniquely incorporated prior to amplification). iii: CollapseTCR uses barcode information to remove erroneous TCR classifiers and produce a better estimate of true cDNA frequency in sample. iv: Collapsed repertoires can then be translated and CDR3 regions extracted for further analysis, using the **CDR3ulator** script. **B**: Overview effect of collapsing by showing the decrease in frequency of unique and total TCRs detected from the raw, unprocessed samples of the healthy peripheral blood controls from both the barcoded V (BV) and barcoded ligation race (BLR) protocols. C: All of the Decombinator classifiers that were associated with a particular unique molecular barcode (ACCGCCGGGGAC) in the  $\alpha$  TCR vDCR output file for healthy peripheral blood sample HV01v1.



Figure 5.22: Correlation between raw and collapsed TCR frequencies. A and B:  $\alpha$  and  $\beta$  TCR samples produced using the BV protocol (sample HV5). C and D:  $\alpha$  and  $\beta$  samples produced using the BLR protocol (sample HV01v1). R<sup>2</sup> values for all samples in order: 0.96, 0.96, 0.92, 0.92. P-values were all lower than 2.2e-16, the smallest value calculable using the R statistical analysis software linear regression module.



frequency above 15) duplication rates began to normalise. Therefore clones above a certain size are amplified at a rate proportional to their prevalence (explaining the good correlation of high frequency clones in Figure 5.22). I also observed that the different barcoded protocols produce different duplication rates, with BV amplification showing a fourfold median increase relative to BLR (Figure 5.23B). Close examination revealed that the different V genes targeted in the BV protocol do not always display the same average duplication rates, while the BLR protocol, which amplifies all rearrangements using the same primers, does (Figure 5.23C and D). This difference is particularly obvious when comparing the difference in TRAV and TRBV gene duplication rates in the BV protocol (where  $\alpha$  genes are duplicated approximately 17% more), but holds true when comparing the two different specific  $\beta$  genes targeted (TRBV12-3 and TRBV20-1, Figure 5.23D and Table C.10). Some of the increased variance in the BLR duplication rates may be due to these data coming a greater number of batches than the BV data: BLR results are the product of four preparation batches sequenced across five different sequencing runs, while all BV data is the product of two experiments.

Figure 5.23 (preceding page): Use of unique molecular identifiers allows measurement of duplication rates of different TCR rearrangements. Duplication rate = raw rearrangement frequency / collapsed frequency. A: Representative plot illustrating the typical distribution of duplication rates within a given TCR sample (sample HV04b, BV protocol). Clonotypes are coloured by their collapsed frequency (red dotted line hightlights broken x axis at 1000, which switches to a logged axis). Lower frequency clones display highly variant duplication rates along paths determined by their actual collapsed frequency (as this value is the denominator in the calculation of the rate), while larger clones (those with collapsed frequencies greater than 15, or raw values greater than 1000, approximately) tend to show approximately equivalent duplication rates. Grey horizontal bar indicates the mean frequency of the rearrangements with collapsed frequencies greater than 15. B: Duplication rates for all the healthy control barcoded sample rearrangements with collapsed frequencies greater than 15 (apart from BV HV09, as this had a very small sample with a very 'flat' repertoire, with very little duplication observed). Median group values are shown as horizontal lines (104.15 for BV, 24.0 for BLR). C: Example plot section showing the difference in duplication rate between the two different V regions primed for in the BV  $\beta$  amplification protocol. Analysis limited to those rearrangements that have a collapsed frequency greater than 15 (shown with larger and darker plot points); TRBV12-3/4 duplication rate is significantly higher than that of TRBV20-1 (p-value = 0.002, unpaired one-sided Welch's t-test). **D**: Duplication rates of all rearrangements with collapsed frequencies greater than 15 for each of the different V genes targeted in the BV protocol. Horizontal lines indicate group means.

#### 5.3.2.1 vDCR/Collapsing verdict

The pipeline described in this thesis employs a modified version of Decombinator to detect and classify TCR rearrangements, followed by error- and frequency-correction made possible by the introduction of random molecular barcodes to transcript sequences prior to amplification. The modifications made to Decombinator itself largely fall into two categories: those that increase the breadth and specificity of TCRs which it can detect, through the design of new sets of V and J tag sequences, and those that remove specific kinds of errors, through the addition of simple filters. The other major change made was to make Decombinator output some additional fields, containing sequence and quality information for the TCR and barcode sequences, which allows us to collapse the data and obtain a more accurate representation of what the actual repertoire in our sample was.

Collapsing is probably the single most important contribution to the robustness of the protocol. In developing this methodology I have demonstrated that using raw high-throughput TCR sequence data is flawed both qualitatively and quantitatively. Over half of the unique TCRs measured are actually the product of sequencing or PCR errors. Moreover the raw frequency of these TCRs only correlates well with the the collapsed frequency above a given frequency: this stochasticity of low-frequency clonotypes (which make up the majority each sample) would correspond to an inability to correctly assign the clonal hierarchy or perform any number of quantitative assessments accurately. These lower frequency clones display a highly variable rate of duplication throughout our pipeline, which is perhaps not well explained by the standard paradigm of PCR amplification. It is perhaps possible that the random sampling that occurs throughout the various purifications and dilutions throughout the protocol play a role, and that larger clones simply rise above the noise.

I also observed that average duplication rates varied between the two protocols that employ cDNA-barcoding. The BV protocol rearrangements unexpectedly displayed larger average duplication rates than the BLR protocol, which seems counter-intuitive given that the BLR protocol samples underwent one more PCR cycle in total than the BV samples (27 rather than 26). A possible explanation is that the final duplication rate is related to the amount of input nucleic acid: equivalent amounts were used in both protocols, however BV specifically only targets a fraction of the overall repertoire, and thus will produce a smaller amount of cDNA from a given blood RNA sample. It is feasible that this lesser amount of input material has therefore more 'room' during the PCR, and can attain higher duplication rates than protocols in which there might be more competition in the reaction. Other members of the group have performed quantitative PCR of subsequent versions of the BLR protocol and demonstrated that these reactions are highly efficient (in terms of the fold increase in number of molecules per cycle, giving values close to the maximum of two, data not shown), which makes it difficult to attribute the difference in duplication rates to differential amplification. In addition to these differences between the protocols, I also observed a marked difference in duplication rates between rearrangements making use of the different V genes targeted in the BV compared to the BLR protocol. This is not unexpected: primer bias – where amplicons using different primers are amplified to different extents – has been reported to occur as a result of differences in both DNA binding energies (395, 396) and certain DNA motifs being preferentially favoured for priming by different DNA polymerases (457). This is indeed the reason why we strove to develop an unbiased RACE based protocol. It is possible that with the addition of barcoding a large portion of primer bias might be mitigated.

There are two recognised limitations to the collapsing methodology I have employed. The first is that there will be a subset of erroneous reads remaining: those reads which contained incorrect base-calls in both their (inter-tag) TCR and barcode sequences will not be collapsed together, as currently it relies on a match of either one in order to recognise that clonotypes are likely derived from the same original molecule. This could be addressed in later versions of the software, however given the throughput we typically work with in our samples the number of these sequences are probably very low. The second limitation is that the clustering of barcode sequences (so as to estimate the actual frequency of the rearrangement) allows for a relatively large number of edits when considering the length of the string (all barcodes that are up to three bases different from one another are considered to represent one sequence that has accrued errors). It is for this reason that we only claim to be estimating, not exactly measuring the true frequency of that cDNA. This could be addressed by including a greater number of random bases in the barcode sequences and placing these barcodes in sections of the read from which one obtains higher quality sequence data, which would allow us to be more confident that we are both observing the barcode accurately and not artificially shrinking a rearrangement's frequency by clustering.

I have demonstrated that the current pipeline is able to accurately detect and describe polyclonal TCR repertoires in a high-throughput manner, producing robust quantitative and qualitative data, a feature that has largely been enabled through the addition of random barcodes thus allowing us to collapse our data. Furthermore this analysis has highlighted some of the flaws and biases that will presumably be present in similar repertoire sequencing efforts that have not been comparably processed; therefore these findings should be of great interest to any group that seeks to accurately deepsequence receptors of the adaptive immune system.

## 5.4 Protocol development discussion

In the course of this thesis I have developed and refined a protocol that permits unbiased amplification, effective high-throughput DNA sequencing and robust bioinformatic analysis of TCR repertoires. In the course of developing this protocol I have gleaned various findings that would aid others undergoing a similar process. In my data an unbiased RACE approach, sequenced on the Illumina platform (and avoiding the addition of homopolymers wherever possible) produced the most reliable, unbiased repertoires with the highest sequencing quality. Furthermore I have shown that a large proportion of errors unavoidably produced during the process of amplification and sequencing can be mitigated through incorporation of random molecular barcodes, which allows one to correct errors after sequencing. This pipeline therefore meets the major criteria that an effective repertoire sequencing effort should.

One might argue that the principle of barcoding could be combined with a multiplex repertoire PCR (such as I have shown using the BV protocol) and achieve comparable results to an unbiased RACE, cancelling out the effect of primer bias. I contend that this is not the case. Even if one uses barcodes to negate the effect of differential amplification rates in a multiplex PCR (meaning that the data you end up with is more accurate in terms of what TCRs are there in what frequencies), the very fact that there has been an imbalance could mean that any disfavoured target molecules are less likely to survive the various sampling events (e.g. carrying over from one PCR to the next, or sampling to the correct concentration for the sequencing machine). Such reads would therefore not be represented in the final data; collapsing can obviously only work on TCRs for which sequences are observed. Multiplex PCRs for TCR repertoires also require generation of panels of primers targeting different V (and potentially J) genes; typically this involves selecting the functional gene sections from a repository such as IMGT and selecting primer sites by some criteria. This could be expected to fail to amplify some TCR rearrangements in two ways: allelic differences at primer binding sites could result in failure to prime, which would cause an apparent reduction or loss in gene use, and inappropriately curated or inaccurate references could result in certain genes being omitted when they do in fact make up part of the repertoire (as I have demonstrated is the case with various supposedly non-functional or not- $\alpha$ -rearranging TRDV genes). This more philosophical failing of multiplex strategies can be extended further: in priming for each V gene (typically as close to the V(D)J

border as will still allow V gene identification) one loses a great deal of biological information which might be useful in elucidating further TCR biology (such as the investigation of alternative splicing I undertook in Section 3.3). It is for this reason that I would especially recommend a RACE based approach when sequencing either immunoglobulin genes (which due to somatic hypermutation might lose priming sites) and variable antigen receptor repertoires from other species (whose germline loci are likely to be less well annotated than say human or mouse).

We are not the first group to have made use of such a cDNA labelling approach. The first group to deep-sequence adaptive immune repertoires – measuring the antibody repertoires of zebrafish – used such an approach (438), and it has been explored for use in general RNA sequencing (435, 458). However for reasons unknown to this researcher, the field at large has not adopted the practice, with one exception: the Russian laboratory that produced the Bolotin *et al* data used for comparison throughout this chapter has since developed and published new protocols that make use of a comparable barcoded RACE technique (and similarly observe the associated improvements following normalisation) (454, 459, 460). An interesting corollary of the impact of collapsing TCR repertoire data is that existing repertoire data that does not make use of such an approach could well be intrinsically flawed with respect to its error profile and frequency distributions. Some groups reduce sequencing error with 'nearest neighbour' or similar clustering algorithms (76, 126, 183). However this potentially could mistakenly conflate genuine rearrangements that happen to be highly similar.

The barcoded ligation RACE strategy is still being further honed and developed in the group, focusing particularly on how to drive down the PCR duplication rate. Even though we can still account for said duplication by collapsing on barcodes, a lower rate would effectively free up space on the sequencing flow-cell, allowing us to theoretically sequence a greater proportion of low-frequency clones. It would also be highly beneficial to reduce the number of PCR cycles each sample receives to the bare minimum, not only to reduce duplication but the range of accumulative errors than can occur (461, 462, 463). Reducing the cycle number could be achieved by performing the final amplification in a real-time quantitative PCR in comparison to samples of known concentration; this should allow users to halt the amplification reaction at the exact cycle at which there is enough DNA to sequence. Other protocol alterations currently being tested involve incorporation of longer random stretches, solely in the ligation adapter, reducing the number of purifications required and increasing the resolution between similar barcodes.

In this study I have primarily made use of the Illumina MiSeq platform for sequencing our repertoires, as it produced a sufficient number of reads of lengths more than suitable for our purposes, with quality score distributions that are consistently found to best rival technologies. Moreover it does so with less cost than many alternatives and in a time frame that is conducive to research, and potentially clinical applications as well. These features make it the most suitable technology for most repertoire sequencing applications, which is reflected in the fact that the major commercial entities offering repertoire sequencing make use of it (464, 465). However, like all second-generation DNA sequencing technologies (which are defined by amplifying clonal populations of DNA fragments in a hugely parallel manner, followed by independent sequencing), DNA amplification is required at one if not more stages (as most machines require relatively high input DNA concentrations). This requirement can theoretically be dispensed with when working with third-generation sequencing platforms as these technologies are defined by being able to sequence single DNA molecules (372, 373, 375). Adoption of such technology (such as the SMRT platform from Pacific Biosciences (382) or the MinIon from Oxford Nanopore Technologies (388)) for sequencing un-amplified immune repertoire libraries could provide the first truly unbiased repertoire data. Furthermore these technologies are able to sequence both very long molecules (which would allow sequencing of unrearranged and genomic TCR loci in one pass) and modified nucleic acids (which might elucidate further sites of regulation of TCR expression).

In conclusion, the research that I have undertaken suggests that in order to obtain the highest quality sequence data that contains the greatest amount of biological information one should bear in mind three main points: an unbiased amplification procedure (e.g. 5' RACE) should be employed, to ensure coverage of all genuine rearrangements in the sample; an appropriate sequencing technology must be used (which in the current market will most likely involve an offering from Illumina), and final output data can be significantly improved through the inclusion of random barcode sequences prior to amplification. High-throughput DNA sequencing of variable antigen receptors represents a powerful tool in being able to dissect and understand how vertebrate immune systems operate; it is imperative that our protocols for doing so are fit enough for the task.

# Conclusions

The T-cell receptor repertoire is an incredibly complex and dynamic biological system, and the ability to accurately measure and model it could greatly aid the understanding of a diverse range of immune processes. However this very complexity makes repertoires intrinsically difficult to measure. Technologies born out of the genomics revolution are enabling greater interrogation of the clonal make up of TCR repertoires then ever before.

In this thesis I demonstrate the development of a protocol that uses unbiased amplification, random molecular barcoding and error-correcting analysis software to sequence the TCR repertoires of RNA samples. I have used this protocol both to characterise elements of healthy donor repertoires and measure how some of these features are perturbed during HIV infection.

A number of themes emerge from the interpretation of this data and related published findings. Perhaps the most significant is the degree of sharing of TCR sequences and profiles between unrelated individuals, which might seem counter-intuitive for a system that is generally regarded to exist as a means for producing receptor diversity. Data such as these begin to redefine our conception of variable antigen receptor generation, in which the stochasticity produced by V(D)J recombination does indeed produce a tremendously diverse repertoire, but within pre-determined bounds. This seems especially true of so-called 'invariant' TCR chains, which appear in multiple individuals in phenotypically defined cell subsets. These bounds likely reflect a combination of the constraints of the recombination machinery and selection processes (both thymic and peripheral), although the contribution that each of these makes is still yet to be dissected.

The other predominant theme of this work is that of unexpected complexity. As our ability to measure TCR rearrangements increases, the limit of detection for rare and aberrant events recedes. Furthermore using an unbiased amplification reaction removes the confirmation bias associated with targeted multiplex primer approaches. These effects have manifested several times in this thesis, such as in the detection of high-frequency rearrangements using purportedly non-functional gene segments and low-frequency alternative splicing events. While the biological significance of these phenomena remain to be established, these findings demonstrate the diverse potential applications of the technology and inform the design of future experiments.

Application of TCR RepSeq to HIV patient blood samples demonstrates the utility of the technology in understanding the pathogenesis of clinically relevant infections. The data presented complements and elaborates on the established hypotheses, producing a model in which the drastic loss of TCR diversity – as a result of both CD4<sup>+</sup> T-cell depletion and considerable expansion of a subset of CD8<sup>+</sup> T-cells – perturbs all features of the repertoire. In this data I observed significant alterations in the frequencies of CDR3s associated with antiviral responses identified in previous studies. These observations were made despite this study only sequencing a small sample of blood at relatively low depth; were we to sequence more blood or increase our read depth (and harvest or otherwise obtain more pathogen-specific TCR sequences) I predict that such analyses would prove even more fruitful.

I contend that I have demonstrated the power of deep-sequencing as a tool to investigate the repertoire, being able to both answer a variety of biological questions and generate hypotheses for future studies. I anticipate that repertoire sequencing will become an invaluable tool in a wide variety of immune studies, that will continue to grow both more commonplace and more powerful as sequencing and related technologies progress. Used in conjunction with other technologies (such as cell sorting or transcriptomics) would permit gathering of linked rearrangement and phenotypic information, increasing the dimensionality and potential information content of output data.

### 6.1 Future work

There remain a great number of experiments and improvements that could be made both in this work and the field of TCR receptor repertoires more generally.

With respect to the data presented here, the majority of future experiments would involve collection of both more sequences and more samples. Greater numbers of sequences per sample – by taking more blood, amplifying from more RNA or sequencing more of our libraries – would allow greater detection of rare and potentially shared sequences. Taking samples from more healthy donors and HIV patients would allow us to greater characterise the differences in TCR features both within and between groups, while taking more longitudinal samples from the same donors would provide data on the dynamics and diversity of the peripheral pool. From a clinical perspective, with enough patients and longer follow up this could be extended to measure whether and which TCR properties might show prognostic value or aid in the stratification of patients for treatment regimes. Cell sorting, particularly of CD4<sup>+</sup>/CD8<sup>+</sup> populations in the context of HIV patients, would be expected to permit greater understanding of the dynamics of the repertoire in these subsets. The major factor limiting these works is simply one of cost; given sufficient funds all of these experiments are possible today.

On a practical level, there are a number of improvements which could be made to increase the yield of information from the data produced by our (and similar) deepsequencing protocols. Primarily, barcoding of some sort should become far more prevalent than it already is, as it is the only solution that I am aware of that counters all aspects of PCR- and sequencing-introduced errors. Modulation of primer concentrations in multiplex PCRs may well be able to effectively counteract primer bias at the gene level (399), but it is still unable to normalise read frequencies back to nucleotide frequency, and nor can it correct nucleotide sequences save by clustering, which may conflate genuinely different rearrangements that happen to have similar sequences. The barcoding protocol detailed in this thesis could itself be improved through the addition of longer barcode sequences, in order to provide better resolution between barcode sequences during the estimation of true rearrangement frequency. Improvement of sequencing quality of both TCR and barcode sequences would greatly minimise the chance of developing or retaining erroneous sequences. A related constraint is that of read length. An ideal situation might involve reading the entire variable domain of all TCR mRNA molecules in a sample (or even the whole message). This would remove all requirements for sequence inference, allowing unequivocal interpretation as to the nature and productivity of a given mRNA. This might produce enough sequence to allow development of Decombinator tags that would be able to detect and distinguish all TCR V gene families and alleles, and would allow far greater investigation of TCR splicing. These practical issues are largely limited by the nature of the sequencing technology available to us: improvements in read length and quality would help address these concerns.

# Postscript

In 1989 Charles Janeway wrote the seminal paper 'Approaching the Asymptote? Evolution and Revolution in Immunology' (466). In addition to expounding a variety of predictions and frameworks that bore fruit across the range of immunological fields, he made the following statement:

"... I believe it is safe to state that our understanding of immunological recognition is approaching some sort of asymptote, where future experiments are obvious, technically difficult to perform, and aim to achieve ever higher degrees of precision rather than revolutionary changes in our understanding."

This could arguably be applied to the practice of investigating immune repertoires, where the majority of improvements in the last two decades largely correspond to technical advances permitting analyses with greater throughput and resolution. However, I contest the notion that increasing precision and making changes in our understanding cannot go hand in hand.

Immunology is in some respects still a comparatively young field, consisting of the study of one of the most complex biological systems known (second perhaps only to that of the nervous system). Tremendous advances have been made identifying the physiological, cellular and molecular components of the immune system – only a tiny portion of which have even been touched upon in this thesis – yet large gaps in our understanding remain. Findings can be difficult to translate between models or from basic to clinical research. and the effects of interventions remain hard to predict both quantitatively and qualitatively.

However, higher-dimensional, higher-throughput technologies such as those described in this thesis can help us move towards system level understandings. By gaining greater ability to measure the constituents of the immune system, at all scales, we increase our ability to model emergent properties. A Grand Unified Theory of Immunology is still a long way off, but we can still move towards it, even if we are only increasing precision. As Jacques Miller concluded in a review article recounting his incredible contributions to the field (38):

"There is still a great future for immunology!"

# A

# Appendix A - exploring the TCR repertoire of healthy individuals

### A >TRDV1-TRAJ44-TRAC

 CCATTTCAGCCTTACAGCTAGAAGATTCAGCAAAGTACTTTTGTGCT

 CTTGGGGAACT
 GGCAGGCACTGCCAGTAAACTCACCTTTGGGACTGGAAC

 AAGACTTCAGGTCACGCTCG
 ATATCCAGAACCCTGACCCTGCCGTGT...

B >TRDV2-TRAJ44-TRAC

...GAGAGAGATGAAGGGTCTTACTACTGTGCCTGTGACACCCGACCTAAAATACCGGCACTGCCAGTAAACTCACCTTTGGGACTGGAACAAGACTTCAGGTCACGCTCGATATCCAGAACCCTGACCCTGCCGTGT...

c >TRDV3-TRAJ43-TRAC

CCCAGAAAGCCTTTCACTTGGTGATCTCTCCAGTAAGGACTGAAGAC
 AGTGCCACTTACTGCTGCCCT
 CCAATGACATGCGCTTTGGAGCAGGGAC
 CAGACTGACAGTAAAACCAAATATCCAGAACCCTGACCCTGCCGTGT...
 TRDV gene
 TRAJ gene
 TRAJ gene
 TRAC gene

Figure A.1: Detection of TRDV usage in  $\alpha$  chain TCR data. Identification of  $\alpha$  chain rearrangements (i.e. containing TRAJ and TRAC) using the IMGT-defined  $\delta$  chain specific V genes TRDV1 (A), TRDV2 (B) and TRDV3 (B), in publicly available TCR sequence data (from Wang *et al* (83), accession number SRA010149 ).

Gene * Allele	Sequence
TRBD1*01	GGGACAGGGGGC
TRBD2*01	GGGACTAGCGGGGGGG
TRBD2*02	GGGACTAGCGGGAGGG

Table A.1: Germline sequences of the human TRBD genes. Obtained from IMGT GENE-DB (19).



Figure A.2: Purity of CD4 and CD8 sorted T-cells. A: Purity of CD4<sup>+</sup> sorted cells before lysis, having sorted first for lymphocytes based on FSC/SCC (not shown), then CD3<sup>+</sup> (left) and CD4<sup>+</sup>CD8<sup>-</sup>(right). B: Purity of CD8<sup>+</sup> sorted cells before lysis, having sorted first for lymphocytes based on FSC/SCC (not shown), then CD3<sup>+</sup> (left) and CD8<sup>+</sup>CD4<sup>-</sup> (right).



Figure A.3: Alpha chain VJ pairing frequencies for HV01. VJ pairing frequencies for HV01 VJ pairing for CD4 (A) and CD8 (B) sorted repertoires. 10,000 rearrangements were sampled from whole files.



Figure A.4: Alpha chain VJ pairing frequencies for HVD4. VJ pairing frequencies for HVD4 VJ pairing for CD4 (A) and CD8 (B) sorted repertoires. 10,000 rearrangements were sampled from whole files.



Figure A.5: Beta chain VJ pairing frequencies for HV01. VJ pairing frequencies for HV01 VJ pairing for CD4 (A) and CD8 (B) sorted repertoires. 20,000 rearrangements were sampled from whole files.


Figure A.6: Beta chain VJ pairing frequencies for HVD4. VJ pairing for CD4 (A) and CD8 (B) sorted repertoires. 20,000 rearrangements were sampled from whole files.



Figure A.7: Adult and cord blood  $\beta$  chain TCR length distributions. A: Box plot of  $\beta$  chain rearrangement insert lengths. B: Box plot of  $\beta$  chain CDR3 lengths. Data produced using the BV protocol, and therefore only targets rearrangements using TRAV8-2/TRAV8-4, TRAV21, TRBV12-3 and TRBV20-1.

Invariant Population	Reference	$\alpha$	$\beta$
GEM	Van Rhijn <i>et al</i> , 2013 (140)	$\checkmark$	$\checkmark$
MAIT	Gold <i>et al</i> , 2010 (467)	$\checkmark$	
MAIT	Greenaway $et al$ , 2013 (282)	$\checkmark$	
MAIT	Reantragoon $et al$ , 2013 (281)	$\checkmark$	
MAIT	Van Rhijn <i>et al</i> , 2013 (140)	$\checkmark$	$\checkmark$
MAIT	Lepore <i>et al</i> , 2014 (138)	$\checkmark$	$\checkmark$
MAIT	Van Schaik <i>et al</i> , 2014 (141)	$\checkmark$	
NKT	Exley et al, 1997 (468)		$\checkmark$
NKT	Han <i>et al</i> , 1999 (469)	$\checkmark$	
NKT	Demoulins $et al$ , 2003 (153)	$\checkmark$	
NKT	Sanderson $et al$ , 2012 (470)		$\checkmark$
NKT	Greenaway <i>et al</i> , 2013 (282)	$\checkmark$	
NKT	Van Rhijn <i>et al</i> , 2013 (140)	$\checkmark$	$\checkmark$
Putative	Van Schaik $et al$ , 2014 (141)	$\checkmark$	

Table A.2: Sources of the TCRs of T-cell subpopulations bearing invariant  $\alpha$  chain and semi-invariant  $\beta$  chain rearrangements.

TRAV gene	TRAJ gene	CDR3	Number donors	Found in CD4	Found in CD8
TRAV13-1	TRAJ6	CAASKGGSYIPTF	10	$\checkmark$	
TRAV12-2	TRAJ15	CAVNNQAGTALIF	9	$\checkmark$	$\checkmark$
TRAV13-1	TRAJ53	CAASSGGSNYKLTF	9	$\checkmark$	$\checkmark$
TRAV14/DV4	TRAJ3	CAMREGGYSSASKIIF	9	$\checkmark$	$\checkmark$
TRAV8-3	TRAJ6	CAVGAASGGSYIPTF	9	$\checkmark$	$\checkmark$
TRAV12-1	TRAJ15	CVVNNQAGTALIF	8	$\checkmark$	$\checkmark$
TRAV12-2	TRAJ5	CAVNNTGRRALTF	8	$\checkmark$	$\checkmark$
TRAV12-3	TRAJ8	CAMMNTGFQKLVF	8	$\checkmark$	$\checkmark$
TRAV13-2	TRAJ52	CAENNAGGTSYGKLTF	8	$\checkmark$	$\checkmark$
TRAV16	TRAJ37	CALSGGSGNTGKLIF	8	$\checkmark$	$\checkmark$
TRAV2	TRAJ5	CAVEEDTGRRALTF	8	$\checkmark$	
TRAV21	TRAJ11	CAVMNSGYSTLTF	8	$\checkmark$	$\checkmark$
TRAV21	TRAJ15	CAVPNQAGTALIF	8	$\checkmark$	$\checkmark$
TRAV29/DV5	TRAJ52	CAASGAGGTSYGKLTF	8	$\checkmark$	$\checkmark$
TRAV9-2	TRAJ13	CALMNSGGYQKVTF	8	$\checkmark$	
TRAV12-1	TRAJ23	CVVNNQGGKLIF	7	$\checkmark$	$\checkmark$
TRAV12-1	TRAJ20	CVVNNDYKLSF	7		$\checkmark$
TRAV12-1	TRAJ30	CVVNMNRDDKIIF	7	$\checkmark$	$\checkmark$
TRAV12-2	TRAJ17	CAVMIKAAGNKLTF	7	$\checkmark$	$\checkmark$
TRAV12-2	TRAJ8	CAVNMNTGFQKLVF	7	$\checkmark$	$\checkmark$
TRAV12-2	TRAJ48	CAVNNFGNEKLTF	7	$\checkmark$	$\checkmark$
TRAV12-3	TRAJ17	CAMMIKAAGNKLTF	7	$\checkmark$	$\checkmark$
TRAV12-3	TRAJ23	CAMMIYNQGGKLIF	7	$\checkmark$	$\checkmark$
TRAV13-2	TRAJ53	CAENKNSGGSNYKLTF	7	$\checkmark$	$\checkmark$
TRAV17	TRAJ9	CATDGTGGFKTIF	7	$\checkmark$	$\checkmark$
TRAV2	TRAJ26	CAVEEDNYGQNFVF	7	$\checkmark$	$\checkmark$
TRAV21	TRAJ36	CAVKTGANNLFF	7	$\checkmark$	$\checkmark$
TRAV21	TRAJ17	CAVMIKAAGNKLTF	7	$\checkmark$	$\checkmark$
TRAV21	TRAJ8	CAVMNTGFQKLVF	7	$\checkmark$	$\checkmark$
TRAV21	TRAJ58	CAVKETSGSRLTF	7	$\checkmark$	
TRAV21	TRAJ42	CAVMNYGGSQGNLIF	7	$\checkmark$	$\checkmark$
TRAV21	TRAJ50	CAVMKTSYDKVIF	7	$\checkmark$	$\checkmark$
TRAV23/DV6	TRAJ12	CAASRMDSSYKLIF	7	$\checkmark$	
TRAV29/DV5	TRAJ45	CAASEGGGADGLTF	7	$\checkmark$	$\checkmark$
TRAV29/DV5	TRAJ53	CAASGSGGSNYKLTF	7	$\checkmark$	
TRAV29/DV5	TRAJ43	CAASANNDMRF	7	$\checkmark$	
TRAV3	TRAJ7	CAVRDDYGNNRLAF	7	$\checkmark$	
TRAV3	TRAJ8	CAVRDNTGFQKLVF	7	$\checkmark$	$\checkmark$
TRAV8-3	TRAJ52	CAVGAAGGTSYGKLTF	7	$\checkmark$	
TRAV8-3	TRAJ10	CAVGFTGGGNKLTF	7	$\checkmark$	

Table A.3: Table of putative invariant  $\alpha$  chain TCRs from the BLR data. To qualify, sequences had to be present in seven out of ten of our healthy donor samples and not belong to any other pre-existing known or putative invariant TCR group. Presence of these sequences in sorted CD4 and CD8 repertoires (see Section 3.2.1) is indicated.

V	J	CDR3	CB	Adult	BV	BLR	Sum	CD4	CD8
TRAV8-2	TRAJ42	CVVMNYGGSQGNLIF	5	4	9	6	15	$\checkmark$	
TRAV8-2	TRAJ13	CVVMNSGGYQKVTF	5	4	9	6	15	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ36	CVVSDQTGANNLFF	6	3	9	2	11	$\checkmark$	$\checkmark$
TRAV21	TRAJ13	CAVMNSGGYQKVTF	5	3	8	5	13	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ23	CVVMIYNQGGKLIF	6	2	8	5	13	$\checkmark$	
TRAV21	TRAJ23	CAVMIYNQGGKLIF	4	4	8	4	12	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ32	CVVMNYGGATNKLIF	4	4	8	3	11		
TRAV21	TRAJ20	CAVNDYKLSF	5	3	8	2	10		$\checkmark$
TRAV21	TRAJ11	CAVMNSGYSTLTF	5	2	7	8	15	$\checkmark$	$\checkmark$
TRAV21	TRAJ42	CAVMNYGGSQGNLIF	6	1	7	7	14	$\checkmark$	$\checkmark$
TRAV21	TRAJ36	CAVKTGANNLFF	4	3	7	7	14	$\checkmark$	$\checkmark$
TRAV21	TRAJ39	CAVMNNNAGNMLTF	5	2	7	6	13	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ4	CVVSLSGGYNKLIF	4	3	7	6	13	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ7	CVVSDDYGNNRLAF	4	3	7	5	12	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ4	CVVSRSGGYNKLIF	4	3	7	5	12	$\checkmark$	
TRAV8-2	TRAJ3	CVVSEGYSSASKIIF	4	3	7	4	11	$\checkmark$	
TRAV8-2	TRAJ10	CVVSFTGGGNKLTF	4	3	7	3	10	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ13	CVVTLSGGYQKVTF	5	2	7	3	10	$\checkmark$	
TRAV8-2	TRAJ5	CVVSGDTGRRALTF	2	5	7	3	10	$\checkmark$	
TRAV8-2	TRAJ12	CVVSRMDSSYKLIF	3	4	7	3	10	$\checkmark$	
TRAV8-2	TRAJ6	CVVSGSGGSYIPTF	3	4	7	3	10	$\checkmark$	
TRAV8-2	TRAJ17	CVVMIKAAGNKLTF	6	1	7	3	10	$\checkmark$	
TRAV21	TRAJ32	CAVMNYGGATNKLIF	4	3	7	2	9	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ3	CVVTGYSSASKIIF	4	3	7	2	9		
TRAV8-2	TRAJ4	CVVSFSGGYNKLIF	4	3	7	2	9	$\checkmark$	
TRAV8-2	TRAJ4	CVVTFSGGYNKLIF	5	2	7	2	9	$\checkmark$	$\checkmark$
TRAV8-2	TRAJ11	CVVSVNSGYSTLTF	4	3	7	1	8	$\checkmark$	
TRAV8-2	TRAJ15	CVVSEGQAGTALIF	4	3	7	1	8	$\checkmark$	
TRAV8-2	TRAJ13	CVVSYSGGYQKVTF	5	2	7	0	7		
TRAV21	TRAJ15	CAVPNQAGTALIF	3	3	6	8	14	$\checkmark$	$\checkmark$
TRAV21	TRAJ17	CAVMIKAAGNKLTF	3	3	6	7	13	$\checkmark$	$\checkmark$
TRAV21	TRAJ8	CAVMNTGFQKLVF	5	1	6	7	13	$\checkmark$	$\checkmark$
TRAV21	TRAJ58	CAVKETSGSRLTF	2	2	4	7	11	$\checkmark$	
TRAV21	TRAJ50	CAVMKTSYDKVIF	3	1	4	7	11	$\checkmark$	$\checkmark$

Table A.4: Table of putative invariant  $\alpha$  chain TCRs from the BV data. Upper section: VJ-CDR3 sequences that were present in the results of seven or more donors' data produced using the barcoded and targeted V (BV) protocol (see Sections 3.2.2.2 and 5.2.5), with the number of donors given. Lower section: Presence of sequences identified in Table A.3, using the barcoded ligation RACE (BLR) strategy (see Sections 3.1.5 and 5.2.6) in the BV data. Column headers: CB = number of cord blood samples containing the sequence; Adult = peripheral blood healthy volunteer samples; BV = total number of samples amplified using the barcoded V protocol with the sequence; BLR = total number of barcoded ligation RACE samples with the sequence (in bold if seven or greater); Sum = total number of donors across both protocols, and CD4 and CD8 ticks designate whether that sequence was detected in the two sorted healthy donor repertoires (see Section 3.2.1).

Gene	Splice	Sequence	Number donors
TRAV1-1	D1A2	TGTTTCCATGAAGATGGGAG   GAGCTCCAGATGAAAGACTC	10
TRAV1-1	D2A2	AGCCATTGTCCAGATAAACT   GAGCTCCAGATGAAAGACTC	10
TRAV1-1	D3A2	GGGTTTTATGGGCTGTCCTG   GAGCTCCAGATGAAAGACTC	10
TRAV1-1	D4A2	CACATTTCTTTCTTACAATG   GAGCTCCAGATGAAAGACTC	8
TRAV8-1	D2A2	GGAACTGTTAATCTCTTCTG   GAAACCCTCTGTGCAGTGGA	1
TRAV8-3	D2A2	GCAACACCTTATCTCTTCTG   GAAACCCTCTGTGCATTGGA	1
TRAV13-2	D2A2	CAAGCAAGAATCTGGAAAAG CTACTCAACCTGGAGACTCA	1
TRAV14/DV4	D2A2	CAAAGTTATGGTCTATTCTG   AAGGCAAGAAAATCCGCCAA	1
		CCAAGTTATGGTCTATTCTG   AAGGCAAGAAAATCCGCCAA	
TRAV22	D2A2	GTGAGTTGGGGCTGTCCAAG   GGACAAAACAGAATGGAAGA	1
TRAV41	D2A2	CAATCAACTGCAGTTACTCG   GAAAAGCACAGCTCCCTGCA	9
TRDV1	D2A2	GCAGTCACCCTGAACTGCCT   GGTTCTGATGAACAGAATGC	10
TRDV1	D2A3	GCAGTCACCCTGAACTGCCT   AATGCAAAAAGTGGTCGCTA	3
TRBV2	D2A2	TAATCACTTATACTTCTATT   GTTGAAAGGCCTGATGGATCA	10
TRBV7-2	D2A2	GGGAAAGGATGTAGAGCTCA   GGCAACAGTGCACCAGACAAA	9
TRBV7-2	D2A3	GGGAAAGGATGTAGAGCTCA   GTGCACCAGACAAATCAGGGC	9
TRBV7-2	D2A4	GGGAAAGGATGTAGAGCTCA   GAGAGGACTGGGGGATCCGTC	3
TRBV7-3	D2A2	GGGAAAATATGTAGAGCTCA   GGCACGGGTGCGGCAGATGAC	7
TRBV7-3	D2A3	GGGAAAATATGTAGAGCTCA   GTCAGGCCTGAGGGATCCGTC	9
TRBV7-6	D-1A2	GCTCACAGTGACACTGATCT   GAGAGGCCTGAGGGATCCATC	3
TRBV7-6	D1A2	TCCTGGGTTTCCTAGGGACA   GAGAGGCCTGAGGGATCCATC	4
TRBV7-6	D2A2	TGGAGTCTCCCAGTCTCCCA   GAGAGGCCTGAGGGATCCATC	10
TRBV7-6	D3A2	GGGACAGGATGTAGCTCTCA   GAGAGGCCTGAGGGATCCATC	10
TRBV7-8	D1A3	TCCTGGGTTTCCTAGGGACA   GAAAGGCCTGAGGGATCCGTC	8
TRBV7-8	D2A2	AGGACAGGATGTAGCTCTCA   GAATGAAGCTCAACTAGACAA	10
TRBV7-8	D2A3	AGGACAGGATGTAGCTCTCA   GAAAGGCCTGAGGGATCCGTC	9
TRBV7-9	D2A2	GGGACAGAATGTAACTTTCA   GAATGAAGCTCAACTAGAAAA	10
TRBV7-9	D2A3	GGGACAGAATGTAACTTTCA   GAGAGGCCTAAGGGATCTTTC	2
TRBV10-2	D2A2	CAAGATCACAGAGACAGGAA   GATAAAGGAGAAGTCCCCGAT	7
TRBV10-2	D3A3	GAGCCACAGCTATATGTTCT   GATCCAAGACAGAGAATTTCC	3
TRBV11-2	D1A2	CCCTCTGTCTCCTGGGAGCA   GAGAGGCTCAAAGGAGTAGAC	10
TRBV11-2	D2A3	TGGCCATGCTACCCTTTACT   GGCTCAAAGGAGTAGACTCCA	1
TRBV18	D2A2	ATCGGCAGCTCCCAGAGGAA   GAGGGCCCCAGCATCCTGAGG	10

Table A.5: List of intra-V gene splicing events. Paired-end reads for one donor were overlapped and sequences that successfully merged – thus containing whole transcripts – and contained valid rearrangements as determined by Decombinator were mapped against the germline loci. Alternative splicing was determined by the presence of sequences mapping to non-canonical TCR exon boundaries, i.e. anything that was not joining L-PART1 to L-PART2 to complete the leader sequence. Splice terminology: the splice donor and acceptor sites of the leader exon were termed D1 and A1, then all novel sites were numbered relatively based on their position. '|' characters indicate splice junctions. Note that there are two TRAV14/DV4 - D2A2 entries, as this region happened to fall on a single nucleotide polymorphism.

## Appendix B - the effect of HIV infection upon the TCR repertoire

 $\mathbf{B}$ 



Figure B.1: Clinical score-diversity correlations. A: The CD4 cell count does not correlate with the Gini index of the untreated v1 bleed in HIV patients ( $R^2 = -0.03317$ , p = 0.9458). B: P-values of the various possible explanatory v1 variables included in a multiple regression, showing that CD8+ lymphocyte count explains the low diversity (by either Gini index or Shannon entropy) better than other variables. Multiple regression analysis performed by group colleague Katharine Best. See Figure 4.1.

187

Reference	Specificity
Pantaleo et al. 1994 (471)	HIV
Wilson et al. 1998 (242)	HIV
Kolowos et al. 1999 $(472)$	HIV
Kou et al. 2000 (473)	HIV
Bourcier et al. 2001 (156)	HIV
Lee et al. 2002 (474)	HIV
Dong et al. 2004 (248)	HIV
Gillespie et al. $2006$ (274)	HIV
Lichterfeld et al. $2006$ (475)	HIV
Meyer-Olson et al. $2006$ (476)	HIV
Almeida et al. 2007 (249)	HIV
Lichterfeld et al. 2007 (477)	HIV
Yu et al. 2007 (478)	HIV
Simons et al. 2008 (479)	HIV
Berger et al. 2011 (276)	HIV
Iglesias et al. $2011$ (480)	HIV
Chen et al. 2012 (481)	HIV
Conrad et al. 2012b (482)	HIV
Mendoza et al. 2012 $(277)$	HIV
Stewart-Jones et al. 2012 (483)	HIV
Motozono et al. $2014$ (278)	HIV
Sun et al. 2014 (166)	HIV
List collated by Zvyagin et al. 2014 (484)	EBV
List collated by Zvyagin et al. 2014 (484)	$\mathrm{CMV}$

Table B.1: Sources of published viral- or subset-specific CDR3s. Collated CDR3 amino acid sequences are available for download (see Section 2.5) (485)



Figure B.2: HIV infection is associated with a loss of publicness and translational convergence in  $\beta$  chain CDR3s. A:  $\beta$  CDR3s are less public in HIV-infected than healthy comparator donors, measured by counting in how many donors each group specific (i.e. found in only HIV-infected or healthy donors) CDR3 appears. This applies for both whole samples (solid lines) and a subsample of 5000 CDR3s (dotted lines). To account for the greater number of HIV<sup>+</sup> patients than healthy controls, for any given analysis ten HIV samples were randomly selected from the total pool, and publicness measured: this was repeated 100 times, shaded areas represent standard deviation. B:  $\beta$  CDR3s display a lower mean translational convergence in HIV<sup>+</sup> donors than healthy, i.e. were encoded by fewer nucleotide sequences (and thus recombination events), as determined by counting unique Decombinator identifiers that produce the same CDR3. Blue and red lines indicate the mean translational convergence for each group (healthy/HIV respectively) for each CDR3 rank; green and purple lines show a sliding window averaging across 100 ranks. See Figure 4.3A and C.



Figure B.3: Healthy donor and HIV-patient gene usage divergence. Barplots showing the difference in TRAJ (A), TRBV (B) and TRBJ (C) usage, by Jensen-Shannon Divergence (JSD). Donor-specific proportional gene usage matrices, before each pair of healthy or  $HIV^+$  donors were compared. Only interquartile ranges and medians are shown. See Figure 4.3D.



Figure B.4: Raw TCR proportional gene usage in healthy and HIV-infected samples. Raw mean (solid lines) and standard deviations (shaded areas) for proportional gene usage of TRAV (A), TRAJ (B), TRBV (C) and TRBJ (D) TCR genes. Asterisks indicate significance comparing all healthy donor values to HIV-patient values for that gene in an unpaired Wilcoxon rank sum test.



Figure B.5: Clinical features of the HIV patient samples. A: Change in absolute count of  $CD3^+CD4^+$  lymphocytes, which significantly increased during this course of ART (p = 0.0039). B: Change in absolute count of  $CD3^+CD8^+$  lymphocytes; no significant change observed (p = 0.17). C: Change in CD4:CD8 cell ratio; a significant increase was obvserved (p = 0.0046). D: Significant decrease in viral load on therapy, as measured by qPCR (p = 1.526e-05). ND = not detectable, which corresponds to 50 copies or lower. All p-values in this figure measured by paired, one-sided Wilcoxon signed rank tests.



Figure B.6: Relationship between years before treatment and CD4:CD8 ratio. An unexpected feature of our patient data is that there is a correlation between the amount of time passed before starting antiretroviral treatment and the CD4:CD8 ratio at time of start therapy ( $R^2 = 0.1967$ , p = 0.04844).



Figure B.7: Change in HIV patients' gene usage divergence in the first three months of ART. Log2 fold change in median TRAJ (A), TRBV (B) and TRBJ (C) Jensen-Shannon Divergence (JSD) from v1 to v2 (red). Blue bars indicate the log2 change in median JSD that would be required for that patient to reach the mean value of the median JSDs for the ten healthy controls, i.e. return them to the average level of divergence healthy controls display in Figure B.3. See Figure 4.4B.



Figure B.8: HIV patients' change in diversity indexes on ART. A and B: Log2 fold-change in whole CDR3 repertoire Gini indexes and Shannon entropies from v1 to v2. Horizontal lines indicate group (healthy/HIV) medians. HIV<sup>+</sup> patients show a significantly lower median fold change in Gini index compared to healthy controls (p = 0.005427, unpaired one-sided Wilcoxon test). See Figure 4.4C and D.



Figure B.9: HIV-associated CDR3s decrease in frequency on ART, while CMVand EBV-associated CDR3s increase. A: Log2 fold-change in frequency of HIV- and CMV-associated CDR3s detected in all of the HIV-infected donors. Most HIV-related CDR3s decreased proportionally (9/13, left), while the majority of CMV-related CDR3s increased (13/17, right). CDR3s were only included in this analysis if they occurred at least twice in both v1 and v2 bleeds of a donor. B: Left: proportional frequencies of the published, EBVspecific CDR3s that were also found in our whole CDR3 repertoires pre- and post-treatment. Right: Log2 fold-change in frequency of EBV-implicated CDR3s. EBV-associated responses were found in five HIV<sup>+</sup> and two uninfected donors. See Figure 4.5.

# Appendix C - developing a TCR sequencing pipeline

#### C.1 Previously published datasets

The earliest of the three publicly available data sets used in this thesis was produced by Wang et al (83). This paper sequenced the repertoires of various T-cell subsets, but I have here only analysed their 'pan T-cell' group in which CD3<sup>+</sup> peripheral blood mononuclear cells (PBMCs) were obtained using magnetic assorted cell sorting (MACS). As with all of the papers with available datasets, RNA was chosen as the input nucleic acid, and so cells were lysed, RNA extracted and mRNA reverse transcribed into cDNA. TCR rearrangements were then amplified with a so-called amplicon rescued multiplex (ARM)-PCR, which is a modified version of the Templex PCR (TEM-PCR) system developed by the same group (486). In these protocols, there is first a multiplex PCR using primers directed against the constant region (of the alpha and beta constant regions in this case) and against the different V genes. In this first PCR, there are actually two nested sets of primers for each target (V and C regions), with the inner set at a higher concentration: the lower-concentration out primers purportedly enrich the target molecules for the inner primers in the early cycles of the reaction (487). The inner primers contain one of a pair of 'universal' priming sites 5' of the V or C specific sections. Amplicons are then 'rescued' (an ambiguous term which presumably reflects a sampling, purification or dilution event) and amplified with primers directed against the universal sites, allowing amplification of all TCR rearrangements (that have been labelled with these universal sites in the previous PCR) using the same primers. The authors claim that this produces semi-quantitative results, as all rearrangements have been amplified using the same primers for the exponential phase of the PCR, however as the patent describing their technique reports that 10 to 15 cycles are used in the first PCR this statement is open to debate (488). Libraries produced by ARM-PCR were then sequenced on the Roche 454 platform.

The second dataset came from Warren et al (77). These authors reverse transcribe lysed PBMC RNA using TRBC specific primers, and perform a template switch reaction. Next they amplified from their TRBC primer to their TS primer in a first round of PCR, and purified rearranged beta chain transcripts by purifying excised bands from agarose gel electrophoresis. A second round of PCR was performed using nested TS and TRBC primers, the latter of which contained 5' biotin moieties. The products of this PCR were then purified and randomly sheared by sonication, before using streptavidin beads to purify sequences that contain the biotinylated TRBC sequence; after another gel excision to size-select the authors have thus obtained shorter fragments (125-175bp) that contain just the TRBC proximal sequence, which contains the V(D)J information. These amplicons then underwent standard adapter-ligation library preparation (illustrated in Figure 5.3), and were sequenced on the Illumina GAIIx platform. Note that the sonication step was required to bring the amplicon lengths within the parameter range that could be sequenced on this platform at this time.

The final paper that has uploaded appropriate was from Bolotin *et al* (126). This paper actually tried several different combinations of protocols, however I have only used the data from one of these in my comparisons. In this technique PBMC RNA was reverse-transcribed and amplified using a template-switch approach as was performed by Warren *et al.* A nested PCR was then performed on the products of this PCR, in which 5' overhanging sequences added adapter regions for 454 sequencing.

There were two additional data sets that were available during the time frame in which we were developing and testing our own protocols, however these were not suitable for comparison with my data. Freeman *et al* (75) produced very short read data, which would need to be assembled into longer contigs in order for to Decombinator to process it, while Mamedov *et al* (97) sequenced blood from a patient with ankylosing spondylitis before and after stem cell therapy, and thus would likely not produce repertoires consistent with those of typical, healthy individuals.

#### C.2 Protocol development supplementary figures

#### **A** >D7β1

#### **B** >D7β2

**C** >D7α

#### **D** >D8α

	📃 L region	V region	D region	J region	C region	Poly-T primer
Е	aRC1		ACGGCAGGGT	CAGGGTTCTGGA	TAT	
	D8α		ACGGCAGGGT	CAGGGTTCTGCT	GAGGTGCAACT.	ATTCAT
	GermlineTRAV8-	-4 .	CTGAAGGAGC	CTGGTTCTGCT	GAGGTGCAACT	ACTCAT
	GermlineTRAV8-	-4 .	CTGAAGGAGC	CCTGGTTCTGCT	GAGGTGCAACT.	ACTCAT

Figure C.1 (preceding page): Example amplicons cloned from the DNA amplified from two healthy donors' PBMC using our poly-T RACE PCR. Bands both expected and unexpected sizes were selected. TCR genes used in these rearrangements are A (D7 $\beta$ 1): TRBV12-3 – TRBD2 – TRBJ1-5; B (D7 $\beta$ 2): TRBV12-3 – TRBD1 – TRBJ1-4; C (D7 $\alpha$ ): TRAV1-2 – TRAJ33; D (D8 $\alpha$ ): TRAV8-4 – TRAJ33. Bold residues indicate mismatches relative to expected sequences. A and C contain the expected number of thymidines in their RACE homopolymer (23). B contains 24, while a clone relating to a non-productive rearrangement (not shown) was found to contain 25. E shows the overlap between the 5' end of the D8 $\alpha$  clone,  $\alpha$ RC1 primer and germline TRAV8-4 sequences, revealing the presumed exonucleation of the primer and subsequent mis-priming. Note that C is a MAIT-cell TCR sequence (see Section 3.1.5).

Sample	Reads	%	%	Mean	Mean	Predicted
		Decombined	Filtered	Quality	P Error	Accuracy
Wang $1\alpha$	37,097	63.63%	0.26%	36.2	$1.77\times10^{-4}$	99.98%
Wang1 $\beta$	$37,\!097$	30.92%	1.28%	36.0	$2.63  imes 10^{-4}$	99.97%
Wang $2\alpha$	15,614	62.86%	0.32%	36.5	$1.52\times10^{-4}$	99.98%
$Wang2\beta$	$15,\!614$	32.22%	0.75%	36.3	$2.42\times 10^{-4}$	99.98%
Wang $3\alpha$	143,944	55.59%	0.88%	36.7	$1.40\times10^{-4}$	99.99%
Wang $3\beta$	$143,\!944$	37.11%	4.06%	36.6	$1.97 \times 10^{-4}$	99.98%
Warren1 $\beta$	1,893,628	80.15%	1.61%	36.8	$3.92\times 10^{-4}$	99.96%
Warren $2\beta$	$1,\!625,\!571$	79.71%	0.11%	36.7	$3.36\times 10^{-4}$	99.97%
Warren $3\beta$	1,794,595	80.33%	0.02%	36.8	$3.04\times10^{-4}$	99.97%
Bolotin1 $\beta$	12,300	59.46%	1.10%	26.1	$2.44\times10^{-3}$	99.76%
Bolotin $2\beta$	10,990	56.93%	0.78%	25.7	$2.55\times 10^{-3}$	99.74%
Bolotin $3\beta$	8,347	44.79%	1.71%	24.1	$3.22\times10^{-3}$	99.68%

Table C.1: Overview read statistics for three sets of publicly available healthy human datasets that were available from papers published prior to completion of our own protocols. FASTQ files were downloaded from the Sequence Read Archive (SRA), with accession numbers in the displayed order: SRR030416, SRR030417 and SRR030709 (83); SRR060699, SRR060700 and SRR060701 (77); SRR494403, SRR494404 and SRR494405 (126). For each paper, the three files represent separate amplifications from different aliquots of either blood or RNA, from the same bleed of one donor. Both Wang *et al* and Bolotin *et al* FASTQ files were produced using 454 technology, while the Warren *et al* paper used the Illumina Genome Analyser (GA) IIx, the precursor machine to the HiSeq. Note that the Bolotin *et al* paper also published sequences from comparable libraries sequenced on both the GAIIx and Ion Torrent platforms, but were not included in this analysis; the reads from their GAIIx run were too short (72bp) for reliable TCR assignation by Decombinator, and the V gene distributions of their Ion Torrent data were inconsistent with the other platforms, suggesting a possible bias.

А	>Pa1
	CAAGCAGAAGACGGCATACGAGAT <mark>GGCTAC</mark> GTGACTGGAGTTCAGACGTGTGCTCTTCCAATCTTTTTTTTTTT
	GCAGCTGCAATGGGTGAGCAGCAAACAGGAGGTGACGCAGATTCCTGCAGCTCTGAGTGTCCCAGAAGGAGAAAACTTGGTTCTCAACTG
	CAGTTTCACTGATAGCGCTATTTACAACCTCCAGTGGTTTAGGCAGGACCCTGGGAAAGGTCTCACATCTCTGTTGCTTATTCAGTCAAG
	TGAUTCAGCCACCTACCTCTGTGCTGTAGACGACTACAAGCTCAGCTTTGGAGCCGGAACCACAGTAACTGTAAGAGCAA <u>ATATCCAGAA</u>
	<u>CCCTGACCCTGCCGTAGTGTAGAT</u> CTCGGTGGTCGCCGTATCATT
в	>Pa2
_	
	ACACAGCGAAAGAGCCTAGAACCTGGGTCCTAGTTTGCACCTAGAATATGAG <mark>GTACTTTGAGG</mark> CTTGCTAAGACCACCCAGCCCATCTCC
	ATGGACTCATATGAAGGACAAGAAGTGAACATAACCTGTAGCCACAACAACATTGCTACAAATGATTATATCACGTGGTACCAACAGTTT
	${\tt cccagccaaggaccacgatttattattcaaggatacaagacaaagttacaagacgagatggcctccctgtttatccctgccgacagaagttacaagattacaagacaagattacaagacgagatggcctccctgtttatccctgccgacagaagttacaagacgagatggcctccctgtttattcctgccgacagaagttacaagacgagatggcctccctgtttattcctgccgacagaagttacaagacgagatggcctccctgtttattcctgccgacagaagttacaagacgacgagatggcctccctgtttattcctgccgacagaagttacaagacgacgagatggcctccctgtttattcctgccgacagaagttacaagacgacagaagttacaagacgacgagatggcctccctgtttattcctgccgacagaagttacaagacgacgagatggcctccctgtttattcctgccgacagaagttacaagacgacaggacgagatggcctccctgtttattcctgccgacagaagttacaagacgacgacgacgacgacgacgacgacgacgacgac$
	TCCAGCACTCTGAGCCTGCCCCGGGTTTCCCTGAGCGACACTGCTGTGTACTACTGCCTCGTGGGTGACTGGGATAACTATGGTCAGAAT
	TTTGTCTTTGGTCCCGGAACCAGATTGTCCCGTGCCCCT <mark>ATATCCAGACCCTGACCCGT</mark> AGTGTAGATCTCG <b>A</b> TGGTCGCCGTA
	ICAT
С	>Pa3
	CAAGCAGAAGACGGCATACGAGAT <mark>GGCTAC</mark> GTGACTGGAGTTCAGACGTGTGCTCTTCCGATCTTTTTTTTTTT
	ATGCCTTTCCAGAGCACTCTGGGGGCGCCGTTCTCGCTAGGCCTTCTCCTCCAACTCTCTCGGAAGGTTCCCAGAAAGC <mark>AAGGACCAACTCTT</mark>
	CTTCACTTCCCGGGATGTGCACCAAGACTCCTTGTTAAAGGCTCAAAGCCTTCTCAGCAGGGACGAT <mark>ACAACATGACCTATGAACGGTTC</mark>
	TCTTCATCGCTGCTCATCCTCCAGGTGCGGGAGGCAGATGCTGCTGTTTACTACTGTGCTGGTAGGAAGCTACATACCTACATTT
	GGAAGAGGAACCAGCCTTATTGTTCATCCGT <mark>ATATCCAGAACCCTGACCCTGCCGT</mark> AGTGTAGATCTCGGTGGTCGCCGTATCATT
Б	Npc7
U	
	CAAGCAGAAGACGGCATACGAGAT <mark>GGCTAC</mark> 5TGACTGGAGTTCAGACGTGTGCTCTTCCGATC7TTTTTTTTTT
	TTTTGGCQ <mark>ATGGCTTTGCAGAGCACTCTGGGGGGGGGGTGTGGCTAGGGCTTCTCCTCAACTCTCTGGAAGGTTGCAGAAAGQAAGGACC</mark>
	AAGTGTTTCAGCCTTCCACAGTGGCATCTTCAGAGGGAGCTGTGGTGGAAATCTTCTGTAATCACTCTGTGTGCCAATGCTTACAACTTCT
	TCTGGTACCTTCACTTCCCGGGATGTGCACCAAGACTCCTTGTTAAAGGCTCAAAGCCTTCTCAGCAGGGACGATACAACATGACCTATG
	<u>▶ ₦ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ ₽ </u>
	CAAGGCAACTGACCTTTGGGTCTGGGACACAATTGACTGTTTTACCTG <u>ATATCCAGAACCCCTGACCCTGCCGT</u> AGTGTAGAT
	TCGCCGTATCATT
Е	>Pb1
	CARCARA CARCARA TACTARA CARA TACTARA CARCARA CARCARA CARCARA CARCARA CARCARA CARA CARA CARA CARA CARA CARA CARA
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTTCTGCTGCTCTGGGGCCCAGGCTTCGGGGCTT</mark> GGTGCTGTCGTCTCTCAACATCCGAGCTGG
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCGGGCCAGGCTCGGGCTT</mark> GGTGCTGTCGTCTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTTCCCCG
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCGGGCCAGGCTCGGGCTT</mark> GGTGCTGTCGTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTCCCG AAACAGAGTCTCATGCTGATGGCAACTTCCCAATGAGGGCTCCAAGGCCACATACGAGGCGTCGAGAAGGACAAGTTTCTCATCAAC
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGGCCAGGGCTCGGGGCTT</mark> GGTGCTGTCGTCTCTCAACATCCGAGGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTCCCG AAACAGAGTCTCATGCTGATGGCAACTTCCCAATGAGGGCTCCAAGGCCACAATACGAGCGAG
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCCGGGGCTT</mark> GGTGCTGTCGTCTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTCCCG AAACAGAGTCTCATGCTGATGGCAACTTCCCAATGAGGGCTCCAAGGCCACAATACGAGCAAGGCGCTCGAGAAGGACAAGTTTCTCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAGCAGCTTCTACATCTGCAGTGCCATTCCGGGACTA GCGGCAAGGGCCGGCAGTACTTCGGGCCGGGC
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCCGGGGCTTGGTGGTGGTGCTGTCGTCACAACATCCGAGGTGG</mark> GTTATTGTAAGASTGGAACCTCTGTGAAGATCGASTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTGGTATCGTCAGTCCCG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCCCCAACATACGAGCAGGCGTCGAGAAGGACAAGTTTCTCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAAGCATCTGCAGTGCCATTCCGGGACTA GCGGGAAGGGG <mark>CAGCAGTACTTCGGGCCGGGCACCAGGCTCACGGTCACAGAGACAAGGTGTTCCCACCAGTGACATCCGGGCCACCAGGTCACAGGACCTGAACAAGGTGTTCCCACCAGTGACACCAGGTCACCGGTCACAGAGACCAGACCAGGTGTCCCCCCGGGCACCAGGTCACGGTCACAGGACCTGAACAAGGTGTTCCCCCCCC</mark>
	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCCGGGCTT</mark> GGTGCTGTCGTCTCTCAACATCCGAGGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTCCCG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCTCCAAGGCCACATACGAGGCGTCGAGAAGGACAAGTTTCTCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGGCCCCATCCTGAAGACAGGCGTCGAGAAGGACAAGTTTCCGGGACTA CCGGGAAGGGCCGGACCAGTACTTCGGGCCGGGC
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTCGGGCTTGGTGCTGTCGTCTCTCAACATCCGAGGCTGGGTTACCTGTAGGGGAACCTCTGTGAAGAACCGAGTGCCGTTCCCTGGACTTCCAGGCCACAACTATGTTTTGGTATCGTCAGTCCAGGTCCCGGAAACAGGGCTCCAAGGCCACAACTATGTTTGGTATCGTCAGTTCCCACGAGAAGGGCTCCAAGGCCACATACGAGGCCACAAGGCCTGAGAAGGACAAGGTTTCCCACGGACAAGGCCCGGGCCACACGAGGCCTGAAGAGGGCCGCGGGCACATCCGGGGCACCAGGCCCACGGCCACACGAGGCCTGAACAAGGTGTCCCACCAGGCCACGACCAGGCCCGAGAAGGGCGCGCGC</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTTGGTGCTGTCGTCGTCTCAACATCCGAGGCTGGGTTATCGTAAGAGTGGAACCTCGTGGAGAGGACCGGGCCGTGGGGCCACAACTATGTTTTGGTATCGTCAGTCCAGTTCCGGGGCCGAGAGGGCCACAACTATGTTTGGTATCGTCAGTCCAGTTCCAACACAGGGCTCCAAGGCCACAACTATGTTTGGTATCGTCAGTTCCAACACACAGGGCTCCAAGGCCACACGAGGCCACAACTATGTTCTCCAACGAGTGACAAGGCCTGAAGAGGCCTGAGAAGGACAAGGTCTCCACGGGACTA CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAGCAGCTTCTACATCTGCAGTGCCATTCCGGGACTA GCGGGAAGGGGCGGAGCAGTACTTCGGGCCGGGC</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCCGGGCTTGGTGGTGCTGTCGTCGTCTCAACATCCGAGCTGG</mark> GTTATTGTAAGASTGGAACCTCTGTGAAGATCGAGAGTGCCGTTCCCGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTTCCGG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCCCACATCCGGACCACAGCAGGCGTCGAGAAGGACAAGGTTTCCTCATCAACA CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAGCAGCGCTCTACATCTGCAGTGCCATTCCGGGACTA GCGGGAAGGGGCGCGAGCAGTACTTCGGGCCGCCCACGCCCACGCTCACGGCACCAGGCGCTCACGAGCGTGCTCCACCCAGTGTAGGGCGCCGGCGACCAGGCGCGCGC
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTCGGGCGTGGGGGGGG</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCCGGGCTT</mark> GTGCTGTCGTCTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGGGCGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTGGTATCGTCAGTCCGG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCTCCAAGGCCACATACGAGCGAG
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCCGGGCTT</mark> GTGCTGTCGTCTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTCCGG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGGCTCCAAGGCCACATACGAGCCAGGAGGAAGGA
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCGGGCTTGGTGCTGTCGTCGTCTCTCAACATCCGAGCTGG</mark> GTTATTGTAAGASTGGAACCTCTGTGAAGATCGAGAGTGCCGTTCCCTGGACTTTCAGGCCACAACTATGTTTTGGTATCGTCAGTTCCGGACTGCCAACAAGGCCCGACAACTATGTTTGGTATCGTCAGTTCCCAGGCCCACACAGGCCCCACACTAGGCCGCACAAGGCCTCAACAAGGTCTCCACAGTTCCCAGGCCACACAGGCCCCACACTGGAGCAGGCCTCACAGTGCCACTCCGGGACTA GCGGGAAGGGCCGGACGTACCTCGGGCCGGCCACCAGGCCCACGGCCACACAGGCGCCCGACCAGGCGCTCTACATCTGGAGCGCCCACCAGGGCCCGGGCCACCAGGCCCGGGCCACGGCGCCGACCAGGCGCCCAGGCCCCAGGCCCCAGGCCCGGCCCAGGCCCCAGGCCCGGCCGACCAGGGCGCCCCAGGCCCCAGGCCCCGGCCCCGGCCCGGCCCCGGCCCCGGCCCGGCCCGGCCCGGCCCGGCCCGGCCCGGCCCGGCCGGCGC
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTTGGGGGGGG</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTTGGTGCGTGTCGTCTCTCAACATCCGAGGCTGGGTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGGTGCCGTTCCCAGGCCACATCCGGGCCACAACTATGTTTGGTATCGTCAGTCCGGGACTGCAGGTCCCAGGCCCCGGGCACAGGCTCCAAGGCCACACTATGTTTGGTATCGTCATCGTCATCCAGTGCCACTCCAGGGCCCCAGGCCCCAGGCCACACGAGGCCTCGAGAAGGACAAGGTTTCCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAAGCAGCTTCTACATCTGCAGTGCCATTCCGGGGCACCAGGCCCAGGCCACGGCCACGGCCACGGCCAGGCCTGAACAAGGTGTTCCCACCCA</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCCGGGCTT</mark> GTGGTGTCGTCTCTCAACATCCGAGCTGG GTTATCTGTAAGAGTGGAACCTCTGTGAAGATCGAGGGCCCCTTCCAGGCCACATACGAGCCACAACTATGTTTGGTATCGTCAGTTCCAGTCCAG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCCTCCAAGGCCACATACGAGCAGCATCGAGAAGGACAAGTTTCTCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAGGCGTTCTACATCTGCAGTGCCATTCCGGGACTA CCGGGAAGGGCCGGACCAGTACTTCGGGCCGGGC
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCCGGGGCTTGGTGGTGCTGTCGTCGTCTCAACATCCGAGCTGGGTTATTGGTAAGAAGCACGAGGCTGGAGGGGGGGG</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGCTTGGTGCTGTCGTCGTCTCAACATCCGAGGCTGGGCTTCTGAGAGAGCGTGGAGGACCTGTGTGGTGTCGTCTCAACATCCGAGGCGGGGCGCGGGGGGGG</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCCAGGGTGCGTGTCGTCGTCGTCTCAACATCCGAGGCTGGGGCCAGGGGGGGTGTCGTGTGTGT</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCTCGGGGCTGGGGGGCGGGGCCTGCACACACA</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCTGCGGCCAGGGCTCGGGGCTTGGTGGTGTCGTCGTCAACAACACACCGGGGCTGGAGGGCCGGGGCCCGGAGGAGGACCTCGAGGTCGCGGGCCGGGGCCCGGGCCAACTACGGCCGGACCAACTATGTTTGGTATCGTCAGGTCCGGGCCGGACAACTACGGCCGGACCAACTATGGTCGGGCCGCGGCCAACTCCGGGGCCACACTCCGGGCCCACACTACGGCCGGC</mark>
F	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCGGGGCTTGGTGGTGTCGTCTCTCAACATCCGAGGCTGGGTTATTGGTAAGAGTGGAACCTCTGGAGGATCGAGGTGCCGTGCCCGGGCACAACTATGTTTGGTATGGTCAGTTCCGAGTGCCG AAACAGAGTCTCATGCTGATGGCAACTTCCAATGAGGGCTCCAAGGCCACATACGAGCAAGGCTCGAGAAGGACAAGTTTCTCATCAAC CATGCAAGCCTGACCTTGTCCACTCTGACAGTGACCAGTGCCCATCCTGAAGACAAGGCTTCTACATCTGCAGTGCCATTCCGGGGCT GCGGGAAGGGCGCAGCAGTACTTCGGCCGGGCACCAGGCTCACGGTCACAGAGGACCTGAACAAGGTGTTCCCCACCAGTGTAGATCTCG GTGGTCGCCGTATCATT &gt;pb3 CAAGCAGAAGACGGCATACGAGATACTTGACTGACTGACGAGGTCGGGCGCGGTGTGGGGGGGCGCGGGTGGCCCGGCCCTTTTTT</mark>
F	GTGAGGCCATCACGGAAGATGCTGCTGCTGCTGCTGCTGCGGCGCAGGCCCGGGCTTGGGGGCTGGGGGCTGAGAAGACACACGGCGGGGCGAGAAGGCCGGAGAAGGCCGGAGAAGGCCGAGAAGGCCGAGAAGGCCGAGAAGGCCGAGAAGGCCGAGAAGGCCGAGGCGGGGCAAGGGCCGGGGGCAGAGGGCCGAGGGGGG
F	GTGAGGCCATCACGGAACATCTCGCTGCTTCTGCGGCTTCTGGGGCCAGGCTCCGGGCTTTGGTGCTGTGTGTG
F	GTGAGGCCATCACGGAAGATGCTGCTGCTTCTGCTGCTTCTGGGGCCAGGCTCCGGGCTTGGTGCTGTGTGTG
F G	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTGCTGCTGCTGCGGGCCAGGCTCCGGGGCTGGGGTGTGTGT</mark>
F G	GTGAGGCCATCACGGAAG <mark>ATGCTGCTGCTGCTTCTGCGCGCTGGGGCCAGGGCTTGGTGCTGTCGTCGTCTTCAACATCCGAGGTGGGGTTGCCTGAAGAGGGCCTGAACAACCTGTGCAAGTCGCGGGGCGCGGTTCCTGGAGGAGGCACCAAGTCTGCTGGAGGAGGACGAAGGACGTCTGAAGAGTCTCGAAGTCGCGGGGCGCGGCGAGGGCGCGGGCACCAGGGCCCGAGGCCCAGGGCCACAGGGGCCGGGCACCAGGGCCCGGGCACCAGGGCCCGGGCACCAGGGCCCGGGCACCAGGCCCGGGCACCAGGCCGGGCACCAGGCCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCACCAGGCCGGGCGCGGCG</mark>
F G	GTGAGGCCATCACGGAAG <mark>ATGGTGCTGCTGCTGTGGGGCGCTTGGGGCCAGGCTTGGTGCTGTGGTGTGTGT</mark>
F G	GTGAGGCCATCACGGAGAGATGCTGTGTGTGTGCTGCTGCTGCTGCTGCTGCGGGCTCGGGGCTTGGGGGCAACAAGTGTGTGT
F G	GTGAGGCCATCACGGAGAGATGCTGTGTGTGTGTGTGTGT
F G	GTGAGGCCATCACGGAAGATGCTGGCGCGCGTCTCTGCTGCTTCTGGGGCCAGGCCAGGCCTTGGGGGCCTATGTTTGGTAAGAGGCGCGAACACGCAGTTCTGGAGGCGCGAGCAGGCCGGAGCAGCTCTGGAGGGCCGAGCGCGAGCAGTTCCGGGGGCCACGGCCCAGGCCCCAGGCCCCGGCCCCTTTGTCGCCAGGCCCGGCCCCGGCCCCTTGTGGCCCGGCCCGGCCCCTTGTGGCCCGGCCCGGCCCCTTGTGGGCCCCTTGTGGGGGG
F G	GTGAGGCCATCACGGAAGATGCTGCTGCTGCTGCTGCTGCTGCTGCTGGGGCAAGGCCACAAGGCGCTGCGAGAAGGACAAGGTGCAAGCAGGGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCAGAAGGCGCGAGAAGGCGCAGAGGCGGGGCGGGGCGGGGCGGGGCGGGGCGGGGCGGGGCGGGG

Figure C.2 (preceding page): Sanger sequenced clones produced using the stepwise PCR protocol. Presence and orientation of sequences indicates that the stepwise PCR protocol is adding the required elements in the correct positions, and thus theoretically libraries should be able to be sequenced. TCR genes used in each rearrangement are A: TRAV21 – TRAJ20; B: TRAV4 – TRAJ26; C: TRAV2 – TRAJ6; D: TRAV2 – TRAJ22; E: TRBV20-1 - TRBJ2-7; F: TRBV5-1 - TRBJ1-2; G: TRBV7-9 - TRBJ2-7. H shows an example sequence cloned from a non-TCR library (genomic varicella-zoster virus) that sequenced with expected run output, to demonstrate matching adapter sequences. Note that SP1 sequences are absent in A-G as these clones were produced with the initial protocol, in which custom sequencing oligonucleotides were targeted against the end of the constant regions (underlined). Sequencing of clones from later libraries showed that the SP1 element was similarly correctly incorporated (analysis performed by other laboratory members, data not shown). Poly-T homopolymer lengths for clones A to H (not counting the thymidine that makes up the ultimate base of the SP2 sequence) are: 21; 17; 22; 22; 22; 22 and 24 (compared to the 23 T residues found in the primer). Base-calls that differ to reference sequences are in bold. VZV library kindly donated by Dr. Daniel Depledge.

Sample	Reads	Extended	%
D2	$10,\!875,\!543$	$1,\!943,\!685$	17.87%
D3 naïve	$17,\!493,\!702$	$2,\!207,\!696$	12.62%
D3 memory	16,410,241	1,786,562	10.89%
D4 naïve	$13,\!068,\!337$	$1,\!988,\!492$	15.22%
D5 memory	10,553,778	1,702,220	16.13%

**Table C.2: Merging HiSeq reads.** Number and percentages of HiSeq FASTQ reads that were successfully merged using FLASH, which uses overlapping portions of paired-end data to recapitulate entire target amplicons contained within the sequencing primers.



Figure C.3 (preceding page): Flowchart overview of how the error- and frequency-correcting code of CollapseTCR works. A: Files decombined with vDCR are read into memory; all rearrangements whose barcode is of sufficiently high FASTQ quality is stored in the Clone dictionary. **B**: The script loops through all the entries in the Clone dictionary, which looks within each barcode sequence to establish what the 'prototypical' rearrangement is, i.e. the most frequent, which is assumed to represent the true receptor chain. Also performs several other checks and assesses whether there are any instances of 'barcode clash', i.e. TCR chains that are not related detected in conjunction with the same barcode. This section is responsible for collapsing the actual Decombinator identifiers down, i.e. reducing the number of unique TCRs by merit of erroneous sequences being removed. C: Having removed erroneous sequences, the script then loops through all of the barcodes that were associated with a given rearrangement. In order to account for errors in the barcode itself, barcodes that are highly similar to one another (in this instance having an edit distance of three or less) are clustered together; the number of unique clusters are then taken as the truer estimated frequency of that TCR. Finally results are output as collapsed Decombinator files. Flowchart produced using yEd (yWorks).



Figure C.4: TRAJ and TRBJ usage difference between original and extended **Decombinator tag sets.** Plots of a representative healthy donor (HV09) showing the comparative performance of Decombinator in combination with the original and extended tag sets. A and B: Frequency of Decombinator rearrangements using particular TRAJ and TRBJ genes, when using either the original or new extended tag sets.

		04	07	7.0	2.0	D 11 / 1
		%	%	Mean	Mean	Predicted
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy
$D2 ext \alpha$	$1,\!943,\!685$	9.93%	0.16%	33.5	$2.01 \times 10^{-3}$	99.80%
$\mathrm{D2}\;\mathrm{ext}eta$	$1,\!943,\!685$	21.91%	0.22%	35.3	$1.41 \times 10^{-3}$	99.86%
D2 R1 $\alpha$	$8,\!931,\!858$	0.05%	0.23%	34.6	$2.25  imes 10^{-3}$	99.78%
$D2 R1\beta$	$8,\!931,\!858$	0.14%	0.29%	34.7	$1.87 \times 10^{-3}$	99.81%
$D2 R2\alpha$	$8,\!931,\!858$	0.04%	0.89%	34.9	$2.68\times10^{-3}$	99.73%
$\mathrm{D2}~\mathrm{R}2eta$	$8,\!931,\!858$	0.07%	0.24%	34.8	$2.55  imes 10^{-3}$	99.75%
D3 naïve $\mathrm{ext}\alpha$	2,207,696	7.70%	0.16%	33.5	$2.06\times 10^{-3}$	99.79%
D3 naïve $\mathrm{ext}\beta$	2,207,696	26.98%	0.21%	35.3	$1.42 \times 10^{-3}$	99.86%
D3 naïve R1 $\alpha$	15,286,006	0.15%	0.24%	35.0	$1.84 \times 10^{-3}$	99.82%
D3 naïve R1 $\beta$	15,286,006	0.29%	0.26%	34.8	$1.79 \times 10^{-3}$	99.82%
D3 naïve R2 $\alpha$	15,286,006	0.10%	0.97%	35.2	$2.35\times10^{-3}$	99.77%
D3 naïve R2 $\beta$	15,286,006	0.17%	0.17%	34.8	$2.56\times10^{-3}$	99.74%
D3 memory $ext\alpha$	1,786,562	7.73%	0.25%	33.2	$2.29\times 10^{-3}$	99.77%
D3 memory $ext\beta$	1,786,562	24.83%	0.33%	35.1	$1.53 \times 10^{-3}$	99.85%
D3 memory $R1\alpha$	14,623,679	0.04%	0.42%	34.9	$1.96 \times 10^{-3}$	99.80%
D3 memory $R1\beta$	$14,\!623,\!679$	0.06%	0.45%	34.7	$1.88 \times 10^{-3}$	99.81%
D3 memory $R2\alpha$	14,623,679	0.03%	1.11%	35.0	$2.49\times10^{-3}$	99.75%
D3 memory $R2\beta$	14,623,679	0.04%	0.35%	34.8	$2.55\times10^{-3}$	99.74%
D4 naïve $ext\alpha$	1,988,492	14.89%	0.20%	33.6	$1.96  imes 10^{-3}$	99.80%
D4 naïve $\mathrm{ext}\beta$	1,988,492	20.93%	0.26%	35.2	$1.45 \times 10^{-3}$	99.85%
D4 naïve R1 $\alpha$	11,079,845	0.07%	0.33%	34.4	$2.33  imes 10^{-3}$	99.77%
D4 naïve R1 $\beta$	11,079,845	0.10%	0.24%	34.3	$2.19 \times 10^{-3}$	99.78%
D4 naïve R2 $\alpha$	11,079,845	0.06%	1.03%	34.7	$2.87 \times 10^{-3}$	99.71%
D4 naïve R2 $\beta$	11,079,845	0.08%	0.25%	34.5	$2.70  imes 10^{-3}$	99.73%
D4 memory $ext\alpha$	1,702,220	17.44%	0.15%	33.7	$1.92 \times 10^{-3}$	99.81%
D4 memory $ext\beta$	1,702,220	20.47%	0.19%	35.2	$1.42 \times 10^{-3}$	99.86%
D4 memory $R1\alpha$	8,851,558	0.11%	0.10%	34.7	$2.02 \times 10^{-3}$	99.80%
D4 memory $R1\beta$	8,851,558	0.09%	0.14%	34.8	$1.86 \times 10^{-3}$	99.81%
D4 memory $R2\alpha$	8,851,558	0.09%	0.95%	35.1	$2.43  imes 10^{-3}$	99.76%
D4 memory $R2\beta$	8,851,558	0.07%	0.20%	34.9	$2.50 \times 10^{-3}$	99.75%

Table C.3: Read and Decombinator output data for the Simple V amplification protocol, as sequenced on the Illumina HiSeq. Reads that were extended (overlapped) by FLASH are highlighted in grey. Records: number of FASTQ reads; number of those reads successfully assigned by Decombinator; how many assignations survived the N/length filter; percentage of total reads that were assigned TCRs; percentage of assigned TCRs removed by N/length filter; average quality score of inter-tag region (i.e. fastq record from the start of Decombinator-assigned V tag to the end of the J tag); average per base probability of an error having occurred in that inter-tag FASTQ file, and the related predicated accuracy of that sequence data.

		%	%	Mean	Mean	Predicted
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy
$D2\alpha$	10,875,543	1.85%	0.18%	33.6	$2.03\times10^{-3}$	99.80%
$\mathrm{D}2eta$	10,875,543	4.09%	0.23%	35.3	$1.44 \times 10^{-3}$	99.86%
D3 naïve $\alpha$	17,493,702	1.19%	0.23%	33.8	$2.06\times10^{-3}$	99.79%
D3 naïve $\beta$	17,493,702	3.81%	0.21%	35.2	$1.49 \times 10^{-3}$	99.85%
D3 memory $\alpha$	16,410,241	0.90%	0.28%	33.3	$2.28\times10^{-3}$	99.77%
D3 memory $\beta$	16,410,241	2.80%	0.33%	35.0	$1.55  imes 10^{-3}$	99.84%
D4 naïve $\alpha$	13,068,337	2.37%	0.22%	33.6	$1.99\times10^{-3}$	99.80%
D4 naïve $\beta$	13,068,337	3.33%	0.25%	35.1	$1.49 \times 10^{-3}$	99.85%
D4 memory $\alpha$	$10,\!553,\!778$	2.98%	0.17%	33.7	$1.94\times10^{-3}$	99.81%
D4 memory $\beta$	10,553,778	3.44%	0.19%	35.2	$1.45 \times 10^{-3}$	99.86%

Table C.4: Overall read statistics for the Simple V amplification protocol Sequenced on the Illumina HiSeq. Fields are as in Table C.3.

		%	%	Mean	Mean	Predicted	Mean
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy	Length
D2 $\alpha$	13,330	42.52%	0.12%	37.7	$4.05\times10^{-4}$	99.96%	421.9
D2 $\beta$	9,714	48.45%	0.53%	36.4	$6.64 \times 10^{-4}$	99.93%	480.4
D3 $\alpha$	23,456	61.22%	0.33%	38.0	$3.60\times10^{-4}$	99.96%	413.3
D3 $\beta$	$25,\!686$	28.83%	0.16%	37.2	$5.88 \times 10^{-4}$	99.94%	282.5
D9 $\alpha$	12,587	33.32%	0.00%	37.4	$4.73\times10^{-4}$	99.95%	375.7
D9 $\beta$	12,432	46.45%	0.36%	35.4	$8.63  imes 10^{-4}$	99.91%	507.3
D10 α	12,587	33.32%	0.00%	37.4	$4.73\times10^{-4}$	99.95%	375.7
D10 β	38,044	33.89%	0.45%	37.0	$6.19 \times 10^{-4}$	99.94%	331.6

Table C.5: Overall read statistics for the poly-T RACE protocol, sequenced on the Roche 454 GS FLX. Fields are as in Table C.3, with an additional column reporting the average read length of the original FASTQ files. Note that D9 and D10 report identical results for alpha chain files; on closer inspection, the FASTQ files were identical. One file must have been accidentally overwritten somewhere between sequencing and final processing.

		%	%	Mean	Mean	Predicted
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy
$D3\alpha$	113,002	35.27%	0.00%	24.9	$2.04\times10^{-2}$	97.96%
$D3\beta$	452,284	25.27%	0.02%	31.4	$6.38 \times 10^{-3}$	99.36%
D10 $\alpha$	16,014	36.51%	0.03%	25.8	$1.75\times10^{-2}$	98.25%
D10 $\beta$	28,326	26.74%	0.03%	32.6	$4.72\times10^{-3}$	99.53%

Table C.6: Overall read statistics for the poly-T RACE protocol, with NEBNext adapter-ligation library preparation. Sequenced on the Illumina MiSeq (2 x 150bp paired end (PE), version 1 chemistry kits). Fields are as in Table C.3.

		%	%	Mean	Mean	Predicted
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy
RL1 $\alpha$	611,664	76.88%	0.00%	37.4	$4.47\times10^{-4}$	99.96%
$\mathrm{RL1}eta$	491,388	61.11%	0.01%	37.4	$3.90 \times 10^{-4}$	99.96%
$RL2\alpha$	827,843	62.13%	0.01%	38.1	$2.47\times 10^{-4}$	99.98%
RL3 $\beta$	1,079,562	68.47%	0.00%	37.5	$3.84 \times 10^{-4}$	99.96%
$DL1\alpha$	3,437,933	51.84%	0.02%	37.8	$3.48\times10^{-4}$	99.97%
$DL1\beta$	3,237,576	54.13%	0.01%	37.6	$3.67  imes 10^{-4}$	99.96%
$DL2\alpha$	2,477,170	31.14%	0.00%	37.9	$3.15\times10^{-4}$	99.97%
$\mathrm{DL}2\beta$	2,116,423	64.44%	0.00%	37.6	$3.48  imes 10^{-4}$	99.97%

Table C.7: Summary statistics for the ligation RACE protocols, sequenced on the Illumina MiSeq. RL = RNA ligation, DL = DNA ligation, numbers indicate test run number per protocol. RNA and DNA ligation protocols were performed by Prof. Chain and Dr Oakes, and are included in this thesis to explain the downstream protocol choices.

		%	Total	%	Mean	Mean	Predicted
Sample	Reads	Decombined	%	Filtered	Quality	P Error	Accuracy
HV4 $\alpha$	2,844,497	42.56%	66.78%	0.00%	37.6	$3.80  imes 10^{-4}$	99.96%
HV4 $\beta$	$2,\!844,\!497$	24.22%		0.00%	37.8	$3.29\times 10^{-4}$	99.97%
HV5 $\alpha$	2,465,640	39.65%	65.50%	0.01%	37.6	$3.98  imes 10^{-4}$	99.96%
$HV5\beta$	$2,\!465,\!640$	25.85%		0.00%	37.7	$3.37  imes 10^{-4}$	99.97%
HV7 $\alpha$	1,610,847	27.26%	44.92%	0.00%	33.5	$1.78  imes 10^{-3}$	99.82%
HV7 $\beta$	1,610,847	17.66%		0.00%	34.6	$1.36 \times 10^{-3}$	99.86%
HV9 $\alpha$	1,533,948	22.58%	35.79%	0.00%	33.5	$1.78 \times 10^{-3}$	99.82%
$HV9\beta$	$1,\!533,\!948$	13.21%		0.00%	34.6	$1.33  imes 10^{-3}$	99.87%
HV10 $\alpha$	$1,\!199,\!385$	28.18%	47.59%	0.00%	33.4	$1.80 \times 10^{-3}$	99.82%
HV10 $\beta$	$1,\!199,\!385$	19.42%		0.00%	34.6	$1.37\times 10^{-3}$	99.86%

Table C.8: Summary statistics for the randomly barcoded simple V amplification protocol, sequenced on the Illumina MiSeq. Fields are as in Table C.3, with an additional total percentage Decombined column, as both alpha and beta reads for the same donors were sequenced under the same index, and thus were demultiplexed together into one output FASTQ file. Files were sequenced in two batches along with other samples, with ten samples multiplexed per run. Thanks go to James Opzoomer, an intern in the laboratory who performed the initial processing on the blood samples once the protocol had been sufficiently refined.

		%	%	Mean	Mean	Predicted
Sample	Reads	Decombined	Filtered	Quality	P Error	Accuracy
HV07 $\alpha$	1,530,742	77.23%	0.01%	37.5	$4.18\times10^{-4}$	99.96%
HV07 $\beta$	1,692,780	68.36%	0.00%	37.7	$3.24 \times 10^{-4}$	99.97%
HV08 $\alpha$	1,674,080	85.50%	0.01%	37.5	$4.20\times10^{-4}$	99.96%
$HV08\beta$	1,431,872	73.98%	0.00%	37.8	$3.18 \times 10^{-4}$	99.97%
HV09 $\alpha$	1,549,771	82.02%	0.01%	37.7	$3.68\times10^{-4}$	99.96%
$\mathrm{HV09}\beta$	1,296,475	76.72%	0.00%	37.7	$3.27  imes 10^{-4}$	99.97%
HV10 $\alpha$	1,482,059	82.15%	0.00%	37.8	$3.54\times10^{-4}$	99.96%
$\mathrm{HV10}\beta$	1,006,028	79.66%	0.00%	37.8	$3.03  imes 10^{-4}$	99.97%
HV11 $\alpha$	1,377,301	85.72%	0.01%	38.0	$2.90\times10^{-4}$	99.97%
HV11 $\beta$	1,283,645	82.40%	0.00%	37.7	$3.29 \times 10^{-4}$	99.97%
HV13 $\alpha$	1,136,590	85.15%	0.00%	38.0	$2.84\times10^{-4}$	99.97%
HV13 $\beta$	1,521,705	75.66%	0.00%	37.7	$3.23 \times 10^{-4}$	99.97%

Table C.9: Summary statistics for the randomly barcoded ligation RACE (BLR) amplification protocol, sequenced on the Illumina MiSeq. Fields are as in Table C.3.

		TRAV8-24		TRAV21		TRBV12-34		TRBV20-1	
		BV	BLR	BV	BLR	BV	BLR	BV	BLR
TRAV8-24	BV								
	BLR	***							
TRAV21	BV	ns	*						
	BLR	***	ns	*					
TRBV12-34	BV	***	***	ns	***				
	BLR	***	ns	*	ns	***			
TRBV20-1	BV	***	***	*	***	**	***		
	BLR	***	ns	*	ns	***	$\mathbf{ns}$	***	

Table C.10: Differential amplification observed between V genes during multiplex PCR. Levels of significance observed when comparing all of the different groups shown in Figure 5.23 with one another. All comparisons were made using Welch's t-test (unpaired Student's t-test without assumption of equal variance). P-values ; 0.001 = \*\*\*, ; 0.01 = \*\*, ; 0.05 = \*, ;= 0.05 = not significant (ns). BV = barcoded V protocol, BLR = barcoded ligation RACE protocol. Purple cells indicate BV-BV comparisons, blue are BLR-BLR, while yellow indicates inter-protocol comparisons. Grey cells represent redundant combinations.

### References

- HYUNG WOOK PARK. Germs, Hosts, and the Origin of Frank Macfarlane Burnets Concept of Self and Tolerance, 1936 1949. Journal of the History of Medicine, 61(4):492-534, 2006.
- [2] ZEEV PANCER AND MAX D COOPER. The evolution of adaptive immunity. Annual review of immunology, 24:497-518, January 2006.
- [3] ANNE GALY, MARILYN TRAVIS, DAZHI CEN, AND BENJAMIN CHEN. Human T, B, Natural Killer, and Dendritic Cells Arise from a Common Bone Marrow Progenitor Cell Subset. Immunity, 3:459–473, 1995.
- [4] MOTONARI KONDO, IRVING L WEISSMAN, AND KOICHI AKASHI. Identification of Clonogenic Common Lymphoid Progenitors in Mouse Bone Marrow. Cell, 91:661– 672, 1997.
- [5] B Y VERONIKA GROH, STEVE PORCELLI, I MARINA FABBI, LEWIS L LANIER, LOUIS J PICKER, TERRI ANDERSON, ROGER A WARNKE, ATUL K BHAN, JACK L STROMINGER, AND MICHAEL B BRENNER. Human lymphocytes bearing T cell receptor gamma/delta are phenotypically diverse and evenly distributed throughout the lymphoid system. The Journal of experimental medicine, 169(April):1277-1294, 1989.
- [6] NOBUMICHI HOZUMI AND SUSUMU TONEGAWA. Evidence for somatic rearrangement of immunoglobulin genes coding for variable and constant regions. Proceedings of the National Academy of Sciences of the United States of America, 73(10):3628-3632, 1976.
- [7] MAUREEN GILMORE-HEBERT, KATHLEEN HERCULEST, MICHAEL KOMAROMY, AND RANDOLPH WALL. Variable and constant regions are separated in the 10-kbase transcription unit coding for immunoglobulin K light chains. Proceedings of the National Academy of Sciences of the United States of America, 75(12):6044-6048, 1978.
- [8] J G SEIDMAN, A Y A LEDER, MARSHALL H EDGELL, FRED POL-SKY, SHIRLEY M TILGHMAN, DAVID C TIEMEIER, AND PHILIP LEDER. Multiple related immunoglobulin variableregion genes identified by cloning and sequence analysis. Proceedings of the National Academy of Sciences of the United States of America, 75(8):3881-3885, 1978.
- [9] H SAKANO, K HUPPI, G HEINRICH, AND S TONEGAWA. Sequences at the somatic recombination sites of immunoglobulin light-chain genes. Nature, 280(July):288-294, 1979.
- [10] EDWARD E MAX, J G SEIDMAN, AND PHILIP LEDER. Sequences of five potential recombination sites encoded close to an immunoglobulin kappa constant region gene. Proceedings of the National Academy of Sciences of the United States of America, 76(7):3450-3454, 1979.
- [11] F ALT, E OLTZ, F YOUNG, J GORMAN, G TACCIOLI, AND J CHEN. VDJ recombination. Immunology Today, 13(8):306– 314, 1992.
- [12] D G SCHATZ, M A OETTINGER, AND D BALTIMORE. The V(D)J recombination activating gene, RAG-1. Cell, 59(6):1035-48, December 1989.

- [13] MARJORIE A OETTINGER, DAVID G SCHATZ, CAROLYN GORKA, AND DAVID BALTIMORE. RAG-1 and RAG-2, Adjacent Genes That Synergistically Activate V(D)J Recombination. Science, 248(4962):1517–1523, 1990.
- [14] SUSUMU TONEGAWA. Somatic generation of antibody diversity. Nature, 302(April):575-581, 1983.
- [15] Y. CHIEN, M. IWASHIMA, K.B. KAPLAN, J.F. ELLIOTT, AND M.M. DAVIS. A new T-cell receptor gene located within the alpha locus and expressed early in Tcell differentiation. *Nature*, **327**, 1987.
- [16] R BAER, T BOEHM, H YSSEL, H SPITS, AND T H RAB-BITTS. Complex rearrangements within the human Jdelta-Cdelta / Jalpha-Calpha locus and aberrant recombination between Jalpha segments. The EMBO journal, 7(6):1661–1668, 1988.
- [17] SHINGO HATA, MARTHA CLABBY, PETER DEVLIN, HERGEN SPITS, JAN E DE VRIES, AND MICHAEL S. KRANCEL. Diversity and organization of human T cell receptor delta variable gene segments. The Journal of experimental medicine, 169 (January):41-57, 1989.
- [18] YOSHIHIRO TAKIHARA, ERIC CHAMPAGNE, ERMANNO CICCONE, LORENZO MORETTA, MARK MINDEN, AND TAK W MAK. Organization and orientation of a human T cell receptor delta chain V gene segment that suggests an inversion mechanism is utilized in its rearrangement. European journal of immunology, 19:571-574, 1989.
- [19] VÉRONIQUE GIUDICELLI, DENYS CHAUME, AND MARIE-PAULE LEFRANC. IMGT/GENE-DB: a comprehensive database for human and mouse immunoglobulin and T cell receptor genes. Nucleic acids research, 33(Database issue):D256-61, January 2005.
- [20] IAKES EZKURDIA, DAVID JUAN, JOSE MANUEL RODRIGUEZ, ADAM FRANKISH, MARK DIEKHANS, JENNIFER HARROW, JESUS VAZQUEZ, ALFONSO VALENCIA, AND MICHAEL L TRESS. Multiple evidence strands suggest that there may be as few as 19 000 human protein-coding genes. Human molecular genetics, 23(22):5866-5878, 2014.
- [21] DONNA KAROLCHIK, GALT P. BARBER, JONATHAN CASPER, HI-RAM CLAWSON, MELISSA S. CLINE, MARK DIEKHANS, TIMO-THY R. DRESZER, PAULINE A. FUJITA, LUVINA GURUVADOO, MAXIMILIAN HAEUSSLER, RACHEL A. HARTE, STEVE HEITNER, ANGIE S. HINRICHS, KATRINA LEARNED, BRIAN T. LEE, CHIN H. LI, BRIAN J. RANEY, BROOKE RHEAD, KATE R. ROSENBLOOM, CRICKET A. SLOAN, MATTHEW L. SPEIR, ANN S. ZWEIG, DAVID HAUSSLER, ROBERT M. KUHN, AND W. JAMES KENT. The UCSC Genome Browser database: 2014 update. Nucleic Acids Research, 42(November 2013):764-770, 2014.
- [22] MARTIN GELLERT. V(D)J recombination: RAG proteins, repair factors, and regulation. Annual review of biochemistry, 71(D):101-32, January 2002.
- [23] YUNMEI MA, ULRICH PANNICKE, KLAUS SCHWARZ, AND MICHAEL R LIEBER. Hairpin opening and overhang processing by an Artemis/DNA-dependent protein kinase complex in nonhomologous end joining and V(D)J recombination. Cell, 108(D):781– 794, 2002.
- [24] FREDERICK W ALT AND DAVID BALTIMORE. Joining of immunoglobulin heavy chain gene segments : Implications from a chromosome with evidence of three D-JH fusions. Proceedings of the National Academy of Sciences of the United States of America, 79(July):4118– 4122, 1982.

- [25] STEPHEN V. DESIDERIO, GEORGE D. YANCOPOULOS, MICHAEL PASKIND, ELISE THOMAS, MICHAEL A. BOSS, NATHANIEL LAN-DAU, FREDERICK W. ALT, AND DAVID BALITIMORE. Insertion of N regions into heavy-chain genes is correlated with expression of terminal deoxytransferase in B cells. Nature, 311:752-755, 1984.
- [26] TOSHIHISA KOMORI, AMI OKADA, VALERIE STEWART, AND FRED-ERICK W ALT. Lack of N regions in antigen receptor variable region genes of TdT-deficient lymphocytes. Science, 261(5125):1171-1175, 1993.
- [27] SUSAN GILFILLAN, ANDRÉE DIERICH, MARIANNE LEMEUR, CHRISTOPHE BENOIST, ANDREE DIERICH, AND DIANE MATHIS. Mice lacking TdT: mature animals with an immature lymphocyte repertoire. Science, 261(5125):1175-1178, 1993.
- [28] OLEG Y BORBULEVYCH, KURT H PIEPENBRINK, AND BRIAN M BAKER. Conformational melding permits a conserved binding geometry in TCR recognition of foreign and self molecular mimics. Journal of immunology (Baltimore, Md. : 1950), 186(5):2950-8, March 2011.
- [29] R M ZINKERNAGEL AND P C DOHERTY. Restriction of in vitro T cell-mediated cytoxicity in lymphocytic chroiomeningitis within a syngeneic or semiallogeneic system. Nature, 248(April):701-702, 1974.
- [30] EMIL R UNANUE. Antigen-presenting function of the macrophage. Annual review of immunology, 2:395–428, 1984.
- [31] PJ BJORKMAN, MA SAPER, B SAMRAOUI, WS BENNETT, JL STROMINGER, AND DC WILEY. Structure of the human class I histocompatibility antigen, HLA-A 2. Nature, 329(October):506-512, 1987.
- [32] PAUL M ALLEN, GARY R MATSUEDA, EDGAR HABER, AND EMIL R UNANUE. Specificity of the T cell receptor: two different determinants are generated by the same peptide and the I-Ak molecule. The Journal of Immunology, 135(1):368-373, 1985.
- [33] A R M TOWNSEND, J ROTHBARD, F M GOTCH, G BAHADUR, D WRAITH, AND A J MCMICHAEL. The Epitopes of Influenza Nucleoprotein Recognized by Cytotoxic T Lymphocytes Can Be Defined with Short Synthetic Peptides. Cell, 44:959–968, 1986.
- [34] PJ BJORKMAN, MA SAPER, B SAMRAOUI, WS BENNETT, JL STROMINGER, AND DC WILEY. The foreign antigen binding site and T cell recognition regions of class I histocompatibility antigens. Nature, 329(October):512-518, 1987.
- [35] ALAIN TOWNSEND AND HELEN BODMER. Antigen recognition by class I-restricted T lymphocytes. Annual review of immunology, 7:601-624, 1989.
- [36] ROSE ZAMOYSKA. CD4 and CD8: modulators of T-cell receptor recognition of antigen and of immune responses? Current opinion in immunology, 10:82–87, 1998.
- [37] MARKUS G RUDOLPH, ROBYN L STANFIELD, AND IAN A WILSON. How TCRs bind MHCs, peptides, and coreceptors. Annual review of immunology, 24:419–66, January 2006.
- [38] JACQUES FAP MILLER. Events that led to the discovery of T-cell development and function a personal recollection. *Tissue antigens*, 1959(March):509-517, 2004.

- [39] MARIA CIOFANI AND JUAN CARLOS ZÚÑIGA PELÜCKER. Determining γδ versus αβT cell development. Nature reviews. Immunology, 10(9):657–63, September 2010.
- [40] RONALD N GERMAIN. T-cell development and the CD4-CD8 lineage decision. Nature reviews. Immunology, 2(5):309-22, May 2002.
- [41] SF SCHLOSSMAN. Antigen recognition: the specificity of T cells involved in the cellular immune response. Immunological Reviews, 10:97–111, 1972.
- [42] MONNA CRONE, CLAUS KOCH, AND MORTEN SIMONSEN. The elusive T cell receptor. Immunological Reviews, 10:36-57, 1972.
- [43] JAMES P ALLISON, BRADLEY W MCINTYRE, AND DAVID BLOCH. Tumor-specific Antigen Of Murine T-lymphoma Defined Antibody With Monoclonal Antibody. Journal of Immunology, 129(5):2293-2300, 1982.
- [44] B Y STEFAN C MEUER, KATHLEEN A FITZGERALD, REBECCA E HUSSEY, JAMES C HODGDON, STUART F SCHLOSSMAN, AND EL-LIS L REINHERZ. Clonotypic Structures Involved In Antigen-Specific Human T Cell Function. Journal of experimental medicine, 157 (February):705-719, 1983.
- [45] K HASKINS, JOHN KAPPLER, AND P MARRACK. The major histocompatibility complex-restricted antigen receptor on T cells. Journal of Experimental Medicine, 157(4):1149–1169, 1983.
- [46] B W MCINTYRE AND J P ALLISON. The mouse T cell receptor: structural heterogeneity of molecules of normal T cells defined by xenoantiserum. Cell, 34(3):739-46, October 1983.
- [47] J KAPPLER, R KUBO, K HASKINS, J WHITE, AND P MARRACK. The mouse T cell receptor: comparison of MHCrestricted receptors on two T cell hybridomas. *Cell*, 34(3):727–37, October 1983.
- [48] STEPHEN M. HEDRICK, DAVID I. COHEN, ELLEN A. NIELSEN, AND MARK M. DAVIS. Isolation of cDNA clones encoding T cell-specific membrane-associated proteins. Nature, 308:149–153, 1984.
- [49] YUSUKE YANAGI, YASUNOBE YOSHIKAI, KATHLEEN LEGGETT, STEPHEN P CLARK, INGRID ALERSANDER, AND TAK W MAK. A human T cell-specific cDNA clone encodes a protein having extensive homology to immunoglobulin chains. Nature, 308:145–149, 1984.
- [50] STEPHEN M HEDRICK, ELLEN A NIELSEN, JOSHUA KAVALER, DAVID I COHEN, AND MARK M DAVIS. Sequence relationships between putative T-cell receptor polypeptides and immunoglobulins. Nature, 308:153-158, 1984.
- [51] G SIU, S P CLARK, Y YOSHIKAI, M MALISSEN, Y YANAGI, E STRAUSS, T W MAK, AND L HOOD. The human T cell antigen receptor is encoded by variable, diversity, and joining gene segments that rearrange to generate a complete V gene. *Cell*, 37(2):393–401, June 1984.
- [52] H SAITO, DM KRANZ, Y TAKAGAKI, AND AC HAYDAY. Complete primary structure of a heterodimeric T-cell receptor deduced from cDNA sequences. *Nature*, 309:757, 1984.
- [53] Y. CHIEN, D.M. BECKER, T. LINDSTEN, M. OKAMURA, D.I. CO-HEN, AND M.M. DAVIS. A third type of murine T-cell receptor gene. Nature, 312:31–35, 1984.

- [54] NANCY JONES, JEFFREY LEIDEN, DENO DIALYNAS, JOHN FRASER, MARTHA CLABBY, JACK L STROMINGER, DAVID AN-DREWS, WILLIAM LANE, AND JAMES WOODY. Partial Primary Structure of the Alpha and Beta Chains of Human Tumor T-Cell Receptors. Science, 227(4684):311-314, 1985.
- [55] STEFAN C MEUER, ORESTE ACUTO, REBECCA E HUSSEY, JAMES C HODGDON, KATHLEEN A FITZGERALD, STUART F SCHLOSSMAN, AND ELLIS L REINHERZ. Evidence for the T3-associated 90K heterodimer as the T-cell antigen receptor. Nature, 303:808-810, 1983.
- [56] J M FAINT, D PILLING, A N AKBAR, G D KITAS, P A BACON, AND M SALMON. Quantitative flow cytometry for the analysis of T cell receptor Vbeta chain expression. Journal of immunological methods, 225:53-60, 1999.
- [57] STANCA M CIUPE, BLYTHE H DEVLIN, MARY LOUISE MARKERT, AND THOMAS B KEPLER. Quantification of total T-cell receptor diversity by flow cytometry and spectratyping. BMC immunology, 14(1):35, August 2013.
- [58] SEAN C BENDALL, GARRY P NOLAN, MARIO ROEDERER, AND PRATIP K CHATTOPADHYAY. A deep profiler's guide to cytometry. Trends in immunology, 33(7):323-332, 2012.
- [59] ISABELL BRETSCHNEIDER, MICHAEL J CLEMENTE, CHRISTIAN MEISEL, MANUEL GUERREIRO, MATHIAS STREITZ, WERNER HOPFENMÜLLER, JAROSLAV P MACIEJEWSKI, MARCIN W WLO-DARSKI, AND HANS-DIETER VOLK. Discrimination of Tcell subsets and T-cell receptor repertoire distribution. Immunologic research, 58(1):20-7, January 2014.
- [60] ANITA DIU, FRANÇOIS ROMAGNÉ, CATHERINE GENEVÉE, CORINNE ROCHER, JEAN-MICHEL BRUNEAU, ALAIN DAVID, FRANÇOISE PRAZ, AND THIERRY HERCEND. Fine specificity of monoclonal antibodies directed at human T cell receptor variable regions : comparison with oligonucleotide-driven amplification for evaluation of Vbeta expression. European journal of immunology, 23:1422-1429, 1993.
- [61] RANDALL K SAIKI, STEPHEN SCHARF, FRED FALOONA, KARY B MULLIS, AND GLENN T HORN. Enzymatic Amplification of β-Globin Genomic Sequences and Restriction Site Analysis for Diagnosis of Sickle Cell Anemia. Science, 230(4732):1350-1354, 1985.
- [62] RANDALL K SAIKI, DAVID H GELFAND, SUSANNE STOFFEL, STEPHEN J SCHARF, GLENN T HORN, KARY B MULLIS, AND HENRY A ERLICH. Primer-Directed Enzymatic Amplification of DNA with a Thermostable DNA Polymerase. Science, 239(4839):487-491, 1988.
- [63] Y W CHOI, B KOTZIN, L HERRON, J CALLAHAN, P MARRACK, AND J KAPPLER. Interaction of Staphylococcus aureus toxin "superantigens" with human T cells. European journal of immunology, 86(November):8941– 8945, 1989.
- [64] WILLIAM M C ROSENBERG, PAUL A H MOSS, AND JOHN I BELL. Variation in human T cell receptor Vβ and Jβ repertoire: analysis using anchor polymerase chain reaction. European journal of immunology, 22(2):541-549, 1992.
- [65] AI-HONG LI, ERIK FORESTIER, RICHARD ROSENQUIST, AND GÖRAN ROOS. Minimal residual disease quantification in childhood acute lymphoblastic leukemia by real-time polymerase chain reaction using the SYBR green dye. Experimental hematology, 30:1170– 1177, 2002.

- [66] MADELEINE COCHET, CHRISTOPHE PANNETIER, ARMELLE REG-NAULT, SYLVIE DARCHE, CLAUDE LECLERC, AND PHILLIPPE KOURLEKY. Molecular detection and in vivo analysis of the specific T cell response to a protein antigen. European journal of immunology, 22:2639-2647, 1992.
- [67] J. GORSKI, MARYAM YASSAI, XIAOLEI ZHU, B. KISSELA, B. KISSELLA, CAROLYN KEEVER, AND N. FLOMENBERG. Circulating T cell repertoire complexity in normal individuals and bone marrow recipients analyzed by CDR3 size spectratyping. Journal of Immunology, 152(10):5109, 1994.
- [68] JESSICA GORSKI, TERESA PIATEK, MARYAM YASSAI, AND KRYSTYNA MASLANKA. Improvements in repertoire analysis by CDR3 size spectratyping. Annals of the New York Academy of Sciences, 756:99–102, 1995.
- [69] MALA K MAINI, GIULIA CASORATI, PAOLO DELLABONA, AN-DREAS WACK, AND PETER C L BEVERLEY. T-cell clonality in immune responses. *Immunology Today*, 5699(6):262–266, 1999.
- [70] KAZUHIKO YAMAMOTO, HIROKO SAKODA, TOSHIHIRO NAKAJIMA, TOMOHIRO KATO, MITSUO OKUBO, MAKOTO DOHI, YUTAKA MIZUSHIMA, KOJI ITO, AND KUSUKI NISHIOKA. Accumulation of multiple T cell clonotypes in the synovial lesions of patients with rheumatoid arthritis revealed by a novel clonality analysis. International immunology, 4(11):1219–1223, 1992.
- [71] MÁIRE F QUIGLEY, JORGE R ALMEIDA, DAVID A PRICE, AND DANIEL C DOUEK. Unbiased molecular analysis of T cell receptor expression using template-switch anchored rt-PCR. Current protocols in immunology, Chapter 10:10.33, August 2011.
- [72] PRADYOT DASH, JENNIFER L MCCLAREN, THOMAS H OGUIN III, WILLIAM ROTHWELL, BRANDON TODD, MELISSA Y MOR-RIS, JARED BECKSFORT, CORY REYNOLDS, SCOTT A BROWN, PE-TER C DOHERTY, AND PAUL G THOMAS. Paired analysis of TCRα and TCRβ chains at the single-cell level in mice. The Journal of Clinical Investigation, 121(1):288– 295, 2011.
- [73] SONG-MIN KIM, LATIKA BHONSLE, PETRA BESGEN, JENS NICKEL, ANNA BACKES, KATHRIN HELD, SIGRID VOLLMER, KLAUS DORNMAIR, AND JOERG C PRINZ. Analysis of the paired TCR α- and β-chains of single human T cells. PloS one, 7(5):e37338, January 2012.
- [74] JENNIFER BENICHOU, ROTEM BEN-HAMO, YORAM LOUZOUN, AND SOL EFRONI. Rep-Seq: uncovering the immunological repertoire through next-generation sequencing. *Immunology*, 135(3):183–91, March 2012.
- [75] J DOUGLAS FREEMAN, RENÉ L WARREN, JOHN R WEBB, BRAD H NELSON, AND ROBERT A HOLT. Profiling the T-cell receptor beta-chain repertoire by massively parallel sequencing. Genome research, 19(10):1817-24, October 2009.
- [76] HARLAN S ROBINS, PAULO V CAMPREGHER, SANTOSH K SRIVAS-TAVA, ABIGAIL WACHER, CAMERON J TURTLE, ORSALEM KAHSAI, STANLEY R RIDDELL, EDUS H WARREN, AND CHRISTOPHER S CARLSON. Comprehensive assessment of T-cell receptor beta-chain diversity in alphabeta T cells. Blood, 114(19):4099-107, November 2009.
- [77] RENÉ L WARREN, J DOUGLAS FREEMAN, THOMAS ZENG, GINA CHOE, SARAH MUNRO, RICHARD MOORE, JOHN R WEBB, AND ROBERT A HOLT. Exhaustive T-cell repertoire sequencing of human peripheral blood samples reveals signatures of antigen selection and a directly measured repertoire size of at least 1 million clonotypes. Genome research, 21(5):790-7, May 2011.

- [78] T. PETTERI ARSTILA, ARMANDA CASROUGE, VÉRONIQUE BARON, JOS EVEN, JEAN KANELLOPOULOS, AND PHILIPPE KOURILSKY. A Direct Estimate of the Human T Cell Receptor Diversity. Science, 286(5441):958–961, October 1999.
- [79] QIAN QI, YI LIU, YONG CHENG, JACOB GLANVILLE, DAVID ZHANG, JI-YEUN LEE, RICHARD A OLSHEN, CORNELIA M WEYAND, SCOTT D BOYD, AND JÖRG J GORONZY. Diversity and clonal selection in the human T-cell repertoire. Proceedings of the National Academy of Sciences of the United States of America, August 2014.
- [80] OLGA V BRITANOVA, EKATERINA V PUTINTSEVA, MIKHAIL SHUGAY, EKATERINA M MERZLYAK, MARIA A TURCHANI-NOVA, DMITRIY B STAROVEROV, DMITRIY A BOLOTIN, SERGEY LUKYANOV, EKATERINA A BOGDANOVA, ILGAR Z MAMEDOV, YURIY B LEBEDEV, AND DMITRIY M CHUDAKOV. Agerelated decrease in TCR repertoire diversity measured with deep and normalized sequence profiling. Journal of immunology (Baltimore, Md. : 1950), 192(6):2689–98, March 2014.
- [81] P L KLARENBEEK, E B M REMMERSWAAL, I J M TEN BERGE, M E DOORENSPLEET, B D C VAN SCHAIK, R E E ESVELDT, S D KOCH, A TEN BRINKE, A H C VAN KAMPEN, F J BEMEL-MAN, P P TAK, F BAAS, N DE VRIES, AND R A W VAN LIER. Deep Sequencing of Antiviral T-Cell Responses to HCMV and EBV in Humans Reveals a Stable Repertoire That Is Maintained for Many Years. PLoS pathogens, 8(9):e1002889, September 2012.
- [82] ANNA M SHERWOOD, CINDY DESMARAIS, ROBERT J LIVINGSTON, JESSICA ANDRIESEN, MAXIMILIAN HAUSSLER, CHRISTOPHER S CARLSON, AND HARLAN ROBINS. Deep sequencing of the human TCR $\gamma$  and TCR $\beta$  repertoires suggests that TCR $\beta$  rearranges after  $\alpha\beta$  and  $\gamma\delta$  T cell commitment. Science translational medicine, **3**(90):90ra61, July 2011.
- [83] CHUNLIN WANG, CATHERINE M SANDERS, QUNYING YANG, HARRY W SCHROEDER, ELIJAH WANG, FARBOD BABRZADEH, BABACK GHARIZADEH, RICHARD M MYERS, JAMES R HUDSON, RONALD W DAVIS, AND JIAN HAN. High throughput sequencing reveals a complex pattern of dynamic interrelationships among human T cell subsets. Proceedings of the National Academy of Sciences of the United States of America, 107(4):1518-23, January 2010.
- [84] JOSEPHJ.C. THOME, NAOMI YUDANIN, YOSHIAKI OHMURA, MASARU KUBOTA, BORIS GRINSHPUN, TAHERI SATHALIYAWALA, TOMOAKI KATO, HARVEY LERNER, YUFENG SHEN, AND DONNAL. FARBER. Spatial Map of Human T Cell Compartmentalization and Maintenance over Decades of Life. Cell, 159(4):814–828, November 2014.
- [85] PINA F I KRELL, SUSANNE REUTHER, UTE FISCHER, THOMAS KELLER, STEPHAN WEBER, MICHAEL GOMBERT, FRIEDHELM R SCHUSTER, CORINNA ASANG, POLINA STE-PENSKY, BRIGITTE STRAHM, ROLAND MEISEL, AND JENS STOVE. Next-generation-sequencing-spectratyping reveals public T-cell receptor repertoires in pediatric very severe aplastic anemia and identifies a b chain CDR3 sequence associated with hepatitisinduced pathogenesis. Haematologica, 98(9), 2013.
- [86] JUNFENG WU, DAWEI LIU, WENWEI TU, WENXIA SONG, AND XIAODONG ZHAO. T-cell receptor diversity is selectively skewed in T-cell populations of patients with Wiskott-Aldrich syndrome. The Journal of allergy and clinical immunology, August 2014.
- [87] AMY E O'CONNELL, STEFANO VOLPI, KERRY DOBES, CLAU-DIA FIORINI, ERDYNI TSITSIKOV, HELEN DE BOER, ISIL B BARLAN, JENNY M DESPOTOVIC, FRANCISCO J ESPINOSA-ROSALES, I CELINE HANSON, MARIA G KANARIOU, ROX-ANA MARTÍNEZ-BECKERAT, ALVARO MAYORGA-SIRERA, CARMEN

MEJIA-CARVAJAL, NESRINE RADWAN, AARON R WEISS, SUNG-YUN PAI, YU NEE LEE, AND LUIGI D NOTARANGELO. Next generation sequencing reveals skewing of the T and B cell receptor repertoires in patients with wiskott-Aldrich syndrome. Frontiers in immunology, 5(July):340, January 2014.

- [88] XIAOMIN YU, JORGE R ALMEIDA, SAM DARKO, MIRJAM VAN DER BURG, SUK SEE DERAVIN, HARRY MALECH, ANDREW GENNERY, IVAN CHINN, MARY LOUISE MARKERT, DANIEL C DOUEK, AND JOSHUA D MILNER. Human syndromes of immunodeficiency and dysregulation are characterized by distinct defects in T-cell receptor repertoire development. The Journal of allergy and clinical immunology, 133(4):1109-15, April 2014.
- [89] D. WU, A. SHERWOOD, J. R. FROMM, S. S. WINTER, K. P. DUNSMORE, M. L. LOH, H. A. GREISMAN, D. E. SABATH, B. L. WOOD, AND H. ROBINS. High-Throughput Sequencing Detects Minimal Residual Disease in Acute T Lymphoblastic Leukemia. Science Translational Medicine, 4(134):134ra63-134ra63, May 2012.
- [90] MALEK FAHAM, JIANBIAO ZHENG, MARTIN MOORHEAD, VIC-TORIA E. H. CARLTON, PATRICIA STOW, ELAINE COUSTAN-SMITH, CHING-HON PUI, AND DARIO CAMPANA. Deepsequencing approach for minimal residual disease detection in acute lymphoblastic leukemia. Blood, 120(26):5173-5181, 2012.
- [91] STEPHAN A GRUPP, MICHAEL KALOS, DAVID BARRETT, RICHARD APLENC, DAVID L PORTER, SUSAN R RHEINGOLD, DAVID T TEACHEY, ANNE CHEW, BERND HAUCK, J FRASER WRIGHT, MICHAEL C MILONE, BRUCE L LEVINE, AND CARL H JUNE. Chimeric antigen receptor-modified T cells for acute lymphoid leukemia. The New England journal of medicine, 368(16):1509–18, April 2013.
- [92] W.-K. WENG, R. ARMSTRONG, S. ARAI, C. DESMARAIS, R. HOPPE, AND Y. H. KIM. Minimal Residual Disease Monitoring with High-Throughput Sequencing of T Cell Receptors in Cutaneous T Cell Lymphoma. *Science Translational Medicine*, 5(214):214ra171, December 2013.
- [93] SCOTT D BOYD, ELEANOR L MARSHALL, JASON D MERKER, JAY M MANIAR, LYNDON N ZHANG, BITA SAHAF, CAROL D JONES, BIRGITTE B SIMEN, BOZENA HANCZARUK, KHOA D NGUYEN, KARI C NADEAU, MICHAEL EGHOLM, DAVID B MIKLOS, JAMES L ZEHNDER, AND ANDREW Z FIRE. Measurement and Clinical Monitoring of Human Lymphocyte Clonality by Massively Parallel V-D-J Pyrosequencing. Science Translational Medicine, 1(12):1-8, 2009.
- [94] AARON C LOGAN, HONG GAO, CHUNLIN WANG, BITA SAHAF, CAROL D JONES, ELEANOR L MARSHALL, ISMAEL BUÑO, RAN-DALL ARMSTRONG, ANDREW Z FIRE, KENNETH I WEINBERG, MICHAEL MINDRINOS, JAMES L ZEHNDER, SCOTT D BOYD, WEN-ZHONG XIAO, RONALD W DAVIS, AND DAVID B MIKLOS. Highthroughput VDJ sequencing for quantification of minimal residual disease in chronic lymphocytic leukemia and immune reconstitution assessment. Proceedings of the National Academy of Sciences of the United States of America, 108(52):21194-9, December 2011.
- [95] AARON C LOGAN, NIKITA VASHI, MALEK FAHAM, VICTORIA CARLTON, KATHERINE KONG, ISMAEL BUÑO, JIANBIAO ZHENG, MARTIN MOORHEAD, MARK KLINGER, BING ZHANG, AMNA WAQAR, JAMES L ZEHNDER, AND DAVID B MIKLOS. Immunoglobulin and T cell receptor gene highthroughput sequencing quantifies minimal residual disease in acute lymphoblastic leukemia and predicts post-transplantation relapse and survival. Biology of blood and marrow transplantation : journal of the American Society for Blood and Marrow Transplantation, 20(9):1307-13, September 2014.

- [96] JONATHAN A SCHUMACHER, ERIC J DUNCAVAGE, TIMOTHY L MOSBRUGER, PHILIPPE M SZANKASI, AND TODD W KELLEY. A comparison of deep sequencing of TCRG rearrangements vs traditional capillary electrophoresis for assessment of clonality in T-Cell lymphoproliferative disorders. American journal of clinical pathology, 141(3):348-59, March 2014.
- [97] ILGAR Z MAMEDOV, OLGA V BRITANOVA, DMITRIY A BOLOTIN, ANNA V CHKALINA, DMITRIY B STAROVEROV, IVAN V ZVYA-GIN, ALEXEY A KOTLOBAY, MARIA A TURCHANINOVA, DE-NIS A FEDORENKO, ANDREW A NOVIK, GEORGE V SHARONOV, SERGEY LUKYANOV, DMITRIY M CHUDAKOV, AND YURI B LEBE-DEV. Quantitative tracking of T cell clones after haematopoietic stem cell transplantation. EMBO molecular medicine, 3(4):201-7, April 2011.
- [98] O V BRITANOVA, A G BOCHKOVA, D B STAROVEROV, D A FEDORENKO, D A BOLOTIN, I Z MAMEDOV, M A TURCHANI-NOVA, E V PUTINTSEVA, A A KOTLOBAY, S LUKYANOV, A A NOVIK, Y B LEBEDEV, AND D M CHUDAKOV. First autologous hematopoietic SCT for ankylosing spondylitis: a case report and clues to understanding the therapy. Bone marrow transplantation, 47(11):1479–81, November 2012.
- [99] JEROEN W J VAN HEIJST, IZASKUN CEBERIO, LAUREN B LIPUMA, DANE W SAMILO, GLORIA D WASILEWSKI, ANNE MARIE R GONZALES, JIMMY L NIEVES, MARCEL R M VAN DEN BRINK, MIGUEL A PERALES, AND ERIC G PAMER. Quantitative assessment of T cell repertoire recovery after hematopoietic stem cell transplantation. Nature medicine, 19(3):372–373, February 2013.
- [100] EVERETT H MEYER, ANDRO R HSU, JOANNA LILIENTAL, L AN-DREA, MAREIKE FLOREK, JAMES L ZEHNDER, SAM STROBER, PHILIP LAVORI, DAVID B MIKLOS, DAVID S JOHNSON, AND ROBERT S NEGRIN. A distinct evolution of the Tcell repertoire categorizes treatment refractory gastrointestinal acute graft-versus-host disease. Blood, 121(24):4955-4963, 2013.
- [101] MARIE-PAULE LEFRANC, VÉRONIQUE GIUDICELLI, CHANTAL GINESTOUX, JOUMANA JABADO-MICHALOUD, GÉRALDINE FOLCH, FATENA BELLARCERE, YAN WU, ELODIE GEMROT, XAVIER BRO-CHET, JÉRÔME LANE, LAETITIA REGNIER, FRANÇOIS EHREN-MANN, GÉRARD LEFRANC, AND PATRICE DUROUX. IMGT, the international ImMunoGeneTics information system. Nucleic acids research, 37(Database issue):D1006-12, January 2009.
- [102] MARIE-PAULE LEFRANC, CHRISTELLE POMMIÉ, MANUEL RUIZ, VÉRONIQUE GIUDICELLI, ELODIE FOULQUIER, LISA TRUONG, VALÉRIE THOUVENIN-CONTET, AND GÉRARD LEFRANC. IMGT unique numbering for immunoglobulin and T cell receptor variable domains and Ig superfamily Vlike domains. Developmental and comparative immunology, 27 (1):55-77, January 2003.
- [103] MARIE-PAULE LEFRANC. From IMGT-ONTOLOGY CLASSIFICATION Axiom to IMGT standardized gene and allele nomenclature: for immunoglobulins (IG) and T cell receptors (TR). Cold Spring Harbor protocols, 2011(6):627-32, June 2011.
- [104] MARIE-PAULE LEFRANC, VÉRONIQUE GIUDICELLI, QUENTIN KAAS, ELODIE DUPRAT, JOUMANA JABADO-MICHALOUD, DO-MINIQUE SCAVINER, CHANTAL GINESTOUX, OLIVER CLÉMENT, DENYS CHAUME, AND GÉRARD LEFRANC. IMGT, the international ImMunoGeneTics information system. Nucleic acids research, 33(Database issue):D593-7, January 2005.
- [105] ELODIE FOULQUIER, CHANTAL GINESTOUX, AND MARIE-PAULE LEFRANC. IMGT Scientific chart: color menu. http://www.imgt.org/IMGTScientificChart/

RepresentationRules/colormenu.php. Accessed: 2015-01-11.

- [106] JOHN T. LIS. Fractionation of DNA Fragments by Polyethylene Glycol Induced Precipitation. Methods in Enzymology, 65(1974):347-353, 1980.
- [107] ILLUMINA. Preparing DNA Libraries for Sequencing on the MiSeq, 2013.
- [108] SIMON ANDREWS. FastQC: A quality control tool for high throughput sequence data. http://www. bioinformatics.babraham.ac.uk/projects/fastqc/. Accessed: 2014-06-16.
- [109] PETER J A COCK, TIAGO ANTAO, JEFFREY T CHANG, BRAD A CHAPMAN, CYMON J COX, ANDREW DALKE, IDDO FRIEDBERG, THOMAS HAMELRYCK, FRANK KAUFF, BARTEK WILCZYNSKI, AND MICHIEL J L DE HOON. Biopython: freely available Python tools for computational molecular biology and bioinformatics. Bioinformatics, 25(11):1422-3, June 2009.
- [110] GREGORY J HANNON. FASTX-Toolkit: FASTQ/A short-reads pre-processing tools. http: //hannonlab.cshl.edu/fastx\_toolkit/index.html. Accessed: 2015-01-10.
- [111] NICLAS THOMAS, JAMES HEATHER, WILFRED NDIFON, JOHN SHAWE-TAYLOR, AND BENJAMIN CHAIN. Decombinator: a tool for fast, efficient gene assignment in T-cell receptor sequences using a finite state machine. *Bioinformatics*, 29(5):542-50, March 2013.
- [112] Decombinator source code. https://github.com/ uclinfectionimmunity/Decombinator/, 2015. Accessed: 2015-01-09.
- [113] MARIE-PAULE LEFRANC. Immunoglobulin and T Cell Receptor Genes: IMGT and the Birth and Rise of Immunoinformatics. Frontiers in Immunology, 5(February):1-22, 2014.
- [114] ALEXANDER DOBIN, CARRIE A DAVIS, FELIX SCHLESINGER, JORG DRENKOW, CHRIS ZALESKI, SONALI JHA, PHILIPPE BATUT, MARK CHAISSON, AND THOMAS R GINGERAS. STAR: ultrafast universal RNA-seq aligner. Bioinformatics, 29(1):15-21, January 2013.
- [115] HENG LI, BOB HANDSAKER, ALEC WYSOKER, TIM FENNELL, JUE RUAN, NILS HOMER, GABOR MARTH, GONCALO ABECASIS, AND RICHARD DURBIN. The Sequence Alignment/Map format and SAMtools. Bioinformatics, 25(16):2078–9, August 2009.
- [116] JAMES T ROBINSON, HELGA THORVALDSDOTTIR, WENDY WINCK-LER, MITCHELL GUTTMAN, ERIC S LANDER, GAD GETZ, AND JILL P MESIROV. Integrative genomics viewer. Nature biotechnology, 29(1):24-26, 2011.
- [117] TANJA MAGOČ AND STEVEN L SALZBERG. FLASH: fast length adjustment of short reads to improve genome assemblies. *Bioinformatics (Oxford, England)*, 27(21):2957–63, November 2011.
- [118] JONATHAN FRIEDMAN. Jensen-Shannon Divergence function. http://stats.stackexchange.com/posts/ 94618/revisions. Accessed: 2015-01-11.
- [119] JAMES M. HEATHER, KATHARINE BEST, NICLAS THOMAS, AND BENJAMIN CHAIN. TCR analysis scripts. http://dx. doi.org/10.6084/m9.figshare.1190861, 2014.

- [120] ADRIEN SIX, MARIA ENCARNITA MARIOTTI-FERRANDIZ, WAHIBA CHAARA, SUSANA MAGADAN, HANG-PHUONG PHAM, MARIE-PAULE LEFRANC, THIERRY MORA, VÉRONIQUE THOMAS-VASLIN, ALEKSANDRA M. WALCZAK, AND PIERRE BOUDINOT. The Past, Present, and Future of Immune Repertoire Biology The Rise of Next-Generation Repertoire Analysis. Frontiers in Immunology, 4(November):1-16, 2013.
- [121] LIDIA CERIANI AND PAOLO VERME. The origins of the Gini index: extracts from Variabilità e Mutabilità (1912) by Corrado Gini. The Journal of Economic Inequality, 10(3):421-443, June 2011.
- [122] C. E. SHANNON. A Mathematical Theory of Communication. Bell System Technical Journal, XXVII(3):379, 1948.
- [123] MARIE-PAULE LEFRANC. Nomenclature of the Human T Cell Receptor Genes. Current protocols in immunology, S40:1-23, 2000.
- [124] MARIE-PAULE LEFRANC. IMGT Locus on focus. Experimental and clinical immunogenetics, 5535(15):1–7, 1998.
- [125] SANTOSH K SRIVASTAVA AND HARLAN S ROBINS. Palindromic nucleotide analysis in human T cell receptor rearrangements. PloS one, 7(12):e52250, January 2012.
- [126] DMITRY A BOLOTIN, ILGAR Z MAMEDOV, OLGA V BRITANOVA, IVAN V ZVYAGIN, DMITRIY SHAGIN, SVETLANA V USTYUGOVA, MARIA A TURCHANINOVA, SERGEY LUKVANOV, YURY B LEBE-DEV, AND DMITRIY M CHUDAKOV. Next generation sequencing for TCR repertoire profiling: platformspecific features and correction algorithms. European journal of immunology, 42(11):3073-83, November 2012.
- [127] PEIPEI LIU, DI LIU, XI YANG, JING GAO, YAN CHEN, XUE XIAO, FEI LIU, JING ZOU, JUN WU, JUNCAI MA, FANGQING ZHAO, XUYU ZHOU, GEORGE F GAO, AND BAOLI ZHU. Characterization of human αβTCR repertoire and discovery of D-D fusion in TCRβ chains. Protein & cell, May 2014.
- [128] JANET M MURRAY, TERRI MESSIER, JAMI RIVERS, J PATRICK O'NEILL, VERNON E WALKER, PAMELA M VACEK, AND BARRY A FINETTE. VDJ recombinase-mediated TCR β locus gene usage and coding joint processing in peripheral T cells during perinatal and pediatric development. Journal of immunology (Baltimore, Md. : 1950), 189(5):2356-64, September 2012.
- [129] VANESSA VENTURI, MÁIRE F QUIGLEY, HUI YEE GREENAWAY, PAULINE C NG, ZACHARY S ENDE, TINA MCINTOSH, TEDI E ASHER, JORGE R ALMEIDA, SAMUEL LEVY, DAVID A PRICE, MILES P DAVENPORT, AND DANIEL C DOUEK. A Mechanism for TCR Sharing between T Cell Subsets and Individuals Revealed by Pyrosequencing. Journal of immunology, March 2011.
- [130] HANJIE LI, CONGTING YE, GUOLI JI, XIAOHUI WU, ZHE XI-ANG, YUANYUE LI, YONGHAO CAO, XIAOLONG LIU, DANIEL C DOUEK, DAVID A PRICE, AND JIAHUAI HAN. Recombinatorial biases and convergent recombination determine interindividual TCR $\beta$  sharing in murine thymocytes. Journal of immunology, 189(5):2404–13, September 2012.
- [131] ASAF MADI, ERIC SHIFRUT, SHLOMIT REICH-ZELIGER, HILAH GAL, KATHARINE BEST, WILFRED NDIFON, BENJAMIN CHAIN, IRUN R COHEN, AND NIR FRIEDMAN. T-cell receptor repertoires share a restricted set of public and abundant CDR3 sequences that are associated with self-related immunity. Genome research, 24:1603–1612, 2014.

- [132] S PORCELLI, C E YOCKEY, M B BRENNER, AND S P BALK. Analysis of T cell antigen receptor (TCR) expression by human peripheral blood CD4-8- alpha/beta T cells demonstrates preferential use of several V beta genes and an invariant TCR alpha chain. The Journal of experimental medicine, 178(1):1-16, July 1993.
- [133] O LANTZ AND A BENDELAC. An invariant T cell receptor alpha chain is used by a unique subset of major histocompatibility complex class I-specific CD4+ and CD4-8- T cells in mice and humans. The Journal of experimental medicine, 180(3):1097-106, September 1994.
- [134] ALBERT BENDELAC, OLIVIER LANTZ, MARY E QUIMBY, JONATHAN W YEWDELL, JACK R. BENNINK, RANDY R BRUTKIEWICZ, AND JACK R BENNINK. CD1 Recognition by Mouse NK1 + T Lymphocytes. Science, 268(5212):863-865, 1995.
- [135] FLORANCE TILLOY, EMMANEUL TREINER, SE-HO PARK, CORINNE GARCIA, FRANCOIS LEMONNIER, HENRI DE LA SALLE, ALBERT BENDELAC, MARC BONNEVILLE, AND OLIVIER LANTZ. An Invariant T Cell Receptor Alpha Chain Defines a Novel TAP-independent Major Histocompatility Complex Class Ib-restricted alpha/beta T Cell Subpopulation in Mammals. Journal of Experimental Medicine, 189(12):1907–1921, 1999.
- [136] EMMANUEL TREINER, LIVINE DUBAN, SEIAMAK BAHRAM, MIR-JANA RADOSAVLJEVIC, VALERIE WANNER, FLORENCE TILLOY, PIERRE AFFATICATI, SUSAN GILFILLAN, AND OLIVIER LANTZ. Selection of evolutionarily conserved mucosalassociated invariant T cells by MR1. Nature, 422(March):1-7, 2003.
- [137] LARS KJER-NIELSEN, ONISHA PATEL, ALEXANDRA J. CORBETT, JEROME LE NOURS, BRONWYN MEEHAN, LIGONG LIU, MUGDHA BHATI, ZHENJUN CHEN, LYUDMILA KOSTENKO, RANGSIMA RE-ANTRAGOON, NICHOLAS A. WILLIAMSON, ANTHONY W. PUR-CELL, NADINE L. DUDEK, MALCOLM J. MCCONVILLE, RICHARD A. J. OHAIR, GEORGE N. KHAIRALLAH, DALE I. GODFREY, DAVID P. FAIRLIE, JAMIE ROSSJOHN, AND JAMES MCCLUSKEY. MR1 presents microbial vitamin B metabolites to MAIT cells. Nature, 491:717, 2012.
- [138] MARCO LEPORE, ARTEM KALINICHENKO, ALESSIA COLONE, BHAIRAV PALEJA, AMIT SINGHAL, ANDREAS TSCHUMI, BERNETT LEE, MICHAEL POIDINGER, FRANCESCA ZOLEZZI, LUCA QUAGLI-ATA, PETER SANDER, EVAN NEWELL, ANTONIO BERTOLETTI, LUIGI TERRACCIANO, GENNARO DE LIBERO, AND LUCIA MORI. **Parallel T-cell cloning and deep sequencing of human MAIT cells reveal stable oligocional TCR\beta** repertoire. *Nature communications*, 5(May):3866, January 2014.
- [139] ANNE G KASMAR, ILDIKO VAN RHIJN, TAN-YUN CHENG, MARIE TURNER, CHETAN SESHADRI, ANDRE SCHIEFNER, RAVI C KALATHUR, JOHN W ANNAND, ANNEMIEKE DE JONG, JOHN SHIRES, LUIS LEON, MICHAEL BRENNER, IAN A WILSON, JOHN D ALTMAN, AND D BRANCH MOODY. **CD1b tetramers bind**  $\alpha\beta$  **T cell receptors to identify a mycobacterial** glycolipid-reactive **T cell repertoire in humans**. The Journal of experimental medicine, **208**(9):1741-7, August 2011.
- [140] ILDIKO VAN RHIJN, ANNE KASMAR, ANNEMIEKE DE JONG, STEPHANIE GRAS, MUGDHA BHATI, MARIEKE E DOORENSPLEET, NIEK DE VRIES, DALE I GODFREY, JOHN D ALTMAN, WILCO DE JAGER, JAMIE ROSSJOHN, AND D BRANCH MOODY. A conserved human T cell population targets mycobacterial antigens presented by CD1b. Nature Immunology, (June):1-10, June 2013.
- [141] B. VAN SCHAIK, P. KLARENBEEK, M. DOORENSPLEET, A. VAN KAMPEN, D. B. MOODY, N. DE VRIES, AND I. VAN RHIJN. Discovery of Invariant T Cells by Next-Generation Sequencing of the Human TCR -Chain Repertoire. The Journal of Immunology, October 2014.
- [142] CALMAN PRUSSIN AND BARBARA FOSTER. TCR V alpha 24 and V beta 11 coexpression defines a human NK1 T cell analog containing a unique Th0 subpopulation. The Journal of Immunology, 159:5862–5870, 1997.
- [143] CORMAC COSGROVE, JAMES E. USSHER, ANDRI RAUCH, KATH-LEEN GARTNER, AYAKO KURIOKA, MICHAEL H. HUHN, KRISTA ADELMANN, YU-HOI KANG, JOANNAH R. FERGUSSON, PETER SIMMONDS, PHILIP GOULDER, TED H. HANSEN, JULIE FOX, HULDRYCH F. GUNTHARD, NINA KHANNA, FIONA POWRIE, ALAN STEEL, BRIAN GAZZARD, RODNEY E. PHILLIPS, JOHN FRATER, HOLM UHLIG, AND PAUL KLENERMAN. Early and nonreversible decrease of CD161++/MAIT cells in HIV infection. Blood, 121(6):951-961, 2013.
- [144] MARIE MALISSEN, JEANNINE TRUCY, EVELYNE JOUVIN-MARCHE, PIERRE-ANDRE CAZENAVE, ROLAND SCOLLAY, AND BERNARD MALISSEN. Regulation of TCR alpha and beta gene allelic exclusion during T-cell development. Immunology Today, 13(8):315–322, 1992.
- [145] M. S. CARTER, J. DOSKOW, P. MORRIS, S. LI, R. P. NHIM, S. SANDSTEDT, AND M. F. WILKINSON. A Regulatory Mechanism That Detects Premature Nonsense Codons in T-cell Receptor Transcripts in Vivo Is Reversed by Protein Synthesis Inhibitors in Vitro. Journal of Biological Chemistry, 270(48):2895– 29003, December 1995.
- [146] BRENNA L BRADY, NATALIE C STEINEL, AND CRAIG H BASSING. Antigen receptor allelic exclusion: an update and reappraisal. Journal of immunology (Baltimore, Md. : 1950), 185(7):3801-8, October 2010.
- [147] SALVATORE SPICUGLIA, ALEKSANDRA PEKOWSKA, JOAQUIN ZACARIAS-CABEZA, AND PIERRE FERRIER. Epigenetic control of Tcrb gene rearrangement. Seminars in immunology, 22(6):330-6, December 2010.
- [148] SHULIN LI AND MILES F WILKINSON. Nonsense surveillance in lymphocytes? Immunity, 8:135-141, 1998.
- [149] JOACHIM WEISCHENFELDT, INGE DAMGAARD, DAVID BRYDER, KIM THEILGAARD-MÖNCH, LINA A THOREN, FINN CILIUS NIELSEN, STEN EIRIK W JACOBSEN, CLAUS NERLOV, AND BO TORBEN PORSE. NMD is essential for hematopoietic stem and progenitor cells and for eliminating by-products of programmed DNA rearrangements. Genes & development, 22:1381–1396, 2008.
- [150] NICHOLAS R J GASCOIGNE AND S MUNIR ALAM. Allelic exclusion of the T cell receptor alpha-chain : developmental regulation of a post-translational event. Seminars in immunology, 11:337-347, 1999.
- [151] JAYANTHI P GUDIKOTE AND MILES F WILKINSON. Tcell receptor sequences that elicit strong downregulation of premature termination codonbearing transcripts. The EMBO journal, 21:125–134, 2002.
- [152] BEN F KOOP, LEE ROWEN, KAI WANG, CHIA LAM KUO, DONALD SETO, JOHANNES A LENSTRA, STEPHEN HOWARD, WEI SHAN, PURNIMA DESHPANDE, AND LEROY HOOD. The Human T-Cell Receptor TCRAC/TCRDC (Calpha/Cdelta) Region: Organization, Sequence, and Evolution of 97.6 kb of DNA. Genomics, 19:478-493, 1994.
- [153] T DEMOULINS. A biased Valpha24+ T-cell repertoire leads to circulating NKT-cell defects in a multiple sclerosis patient at the onset of his disease. *Immunology Letters*, 90(2-3):223-228, December 2003.

- [154] J. J. MILES, N. A. BORG, R. M. BRENNAN, F. E. TYNAN, L. KJER-NIELSEN, S. L. SILINS, M. J. BELL, J. M. BURROWS, J. MCCLUSKEY, J. ROSSIOHN, AND S. R. BURROWS. TCR Genes Direct MHC Restriction in the Potent Human T Cell Response to a Class I-Bound Viral Epitope. The Journal of Immunology, 177(10):6804– 6814, November 2006.
- [155] MONICA BOITA, GIUSEPPE GUIDA, PAOLA CIRCOSTA, ANGELA RITA, STEFANIA STELLA, ENRICO HEFFLER, IULIANA BADIU, DAVIDE MARTORANA, SARA MARIANI, GIOVANNI ROLLA, AND ALESSANDRO CIGNETTI. The molecular and functional characterization of clonally expanded CD8 + TCR BV T cells in eosinophilic granulomatosis with polyangiitis (EGPA). Clinical Immunology, 152(1-2):152-163, 2014.
- [156] KD BOURCIER, DG LIM, AND YH DING. Conserved CDR3 Regions in T-Cell Receptor (TCR) CD8+ T Cells That Recognize the Tax11-19/HLA-A\*0201 Complex in a Subject Infected with Human T-Cell Leukemia Virus Type 1: Relationship of T-Cell Fine Specificity and Major Histocompatibility Complex/Peptide. Journal of virology, 75(20):9836– 9843, 2001.
- [157] JOHAN VERHAGEN, RAPHAËL GENOLET, GRAHAM J BRITTON, BRIAN J STEVENSON, CATHERINE A SABATOS-PEYTON, JU-LIAN DYSON, IMMANUEL F LUESCHER, AND DAVID C WRAITH. CTLA-4 controls the thymic development of both conventional and regulatory T cells through modulation of the TCR repertoire. Proceedings of the National Academy of Sciences of the United States of America, December 2012.
- [158] SOBHAN ROY, DALAM LY, NAN-SHENG LI, JOHN D ALTMAN, JOSEPH A PICCIRILI, D BRANCH MOODY, AND ERIN J ADAMS. Molecular basis of mycobacterial lipid antigen presentation by CD1c and its recognition by αβ T cells. Proceedings of the National Academy of Sciences of the United States of America, October 2014.
- [159] C. MIOSSEC, F.FAURE, L. FERRADINI, S. ROMAN-ROMAN, S. JITSUKAWA, S. FERRINI, A. MORETTA, F. TRIEBEL, AND T. HERCEND. Further analysis of the T cell receptor gamma/delta+ peripheral lymphocyte subset: The Vdelta1 Gene Segment is Expressed with either Calpha or Cdelta. Journal of Experimental Medicine, 171(April), 1990.
- [160] C MIOSSEC, A CAIGNARD, L FERRADINI, S ROMAN-ROMAN, F FAURE, H MICHALAKI, F TRIEBEL, AND T HERCEND. Molecular characterization of human T cell receptor alpha chains including a V delta 1-encoded variable segment. European journal of immunology, 21(4):1061– 4, April 1991.
- [161] I BANK, M BOOK, L COHEN, A KNELLER, E ROSENTAL, M PRAS, I B BASSAT, AND A BEN-NUN. Expansion of a unique subpopulation of cytotoxic T cells that express a C alpha V delta 1 T-cell receptor gene in a patient with severe persistent neutropenia. Blood, 80(12):3157-63, December 1992.
- [162] C CASTELLI, A MAZZOCCHI, S SALVI, A ANICHINI, AND M SENSI. Use of the V delta 1 variable region in the functional T-cell receptor alpha chain of a WT31+ cytotoxic T lymphocyte clone which specifically recognizes HLA-A2 molecule. Scandinavian journal of immunology, 35(4):487-94, April 1992.
- [163] TAKAMASA UENO, HIROKO TOMIYAMA, MAMORU FUJIWARA, SHINICHI OKA, AND MASAFUMI TAKIGUCHI. HLA class Irestricted recognition of an HIV-derived epitope peptide by a human T cell receptor alpha chain having a Vdeltal variable segment. European journal of immunology, 33(10):2910-6, October 2003.

- [164] I BANK, L COHEN, A KNELLER, N K DE ROSBO, M BOOK, AND A BEN-NUN. Aberrant T-cell receptor signalling of interferon-gamma- and tumour necrosis factor-alpha-producing cytotoxic CD8+ Vdelta1/Vbeta16 T cells in a patient with chronic neutropenia. Scandinavian journal of immunology, 58(1):89-98, July 2003.
- [165] XIAOMING SUN, MASUMICHI SAITO, YOSHINORI SATO, TAKAYUKI CHIKATA, TAKUYA NARUTO, TATSUHIKO OZAWA, ELII KOBAYASHI, HIROYUKI KISHI, ATSUSHI MURAGUCHI, AND MASAFUMI TAKIGUCHI. Unbiased Analysis of TCRα/β Chains at the Single-Cell Level in Human CD8(+) T-Cell Subsets. PloS one, 7(7):e40386, January 2012.
- [166] XIAOMING SUN, MAMORU FUJIWARA, YI SHI, NOZOMI KUSE, HI-ROYUKI GATANAGA, VICTOR APPAY, GEORGE F GAO, SHINICHI OKA, AND MASAFUMI TAKIGUCHI. Superimposed epitopes restricted by the same HLA molecule drive distinct HIV-specific CD8+ T cell repertoires. Journal of immunology (Baltimore, Md. : 1950), 193(1):77– 84, July 2014.
- [167] GÉRALDINE FOLCH AND MARIE-PAULE LEFRANC. Assignment of rearranged cDNAs and gDNAs to germline genes: Human TRDV, 2001. Accessed: 2014-07-05.
- [168] GÉRALDINE FOLCH, CHANTAL GINESTOUX, AND MARIE-PAULE LEFRANC. Gene table: human (Homo sapiens) TRDV. http://www.imgt.org/IMGTepertoire/index. php?section=LocusGenes&repertoire=genetable& species=human&group=TRDV, 2011. Accessed: 2014-07-05.
- [169] S YOKOTA, T E HANSEN-HAGGE, AND C R BARTRAM. Tcell receptor delta gene recombination in common acute lymphoblastic leukemia: preferential usage of V delta 2 and frequent involvement of the J alpha cluster. Blood, 77(1):141-8, January 1991.
- [170] T E HANSEN-HAGGE, C MAHOTKA, R E PANZER-GRÜMAYER, H J REUTER, K SCHWARZ, J J VAN DONGEN, AND C R BAR-TRAM. Alternative splicing restricts translation of rearrangements at the T-cell receptor delta/alpha locus. Blood, 85(7):1888-96, April 1995.
- [171] ANN M CARROLL, JILL K SLACK, AND XIAOCHUN MU. V(D)J Recombination Generates a High Frequency of Nonstandrd TCR Ddelta-Associated Rearrangements in Thymocytes. The Journal of Immunology, 150(6):2222-2230, 1993.
- [172] FERENC LIVAK. In vitro and in vivo studies on the generation of the primary T-cell receptor repertoire. Immunological reviews, 200:23-35, 2004.
- [173] ROBERT E TILLMAN, ANDREA L WOOLEY, MAUREEN M HUGHES, AND BARRY P SLECKMAN. Regulation of T-cell receptor beta-chain gene assembly by recombination signals : the beyond 12 / 23 restriction. Immunological reviews, 200:36-43, 2004.
- [174] P A H Moss, R J MOOTS, W M C ROSENBERG, S J ROWLAND-JONES, H C BODMER, A J MCMICHAEL, AND J I BELL. Extensive conservation of alpha and beta chains of the human T-cell antigen receptor recognizing HLA-A2 and influenza A matrix peptide. Proceedings of the National Academy of Sciences USA, 88(October):8987-8990, 1991.
- [175] RICARDO CIBOTTI, JEAN-PIERRE CABANIOLS, CHRISTOPHE PAN-NETIER, CHRISTIANE DELARBRE, ISABELLE VERGNON, JEAN M KANELLOPOULOS, AND PHILIPPE KOURLISKY. Public and private V beta T cell receptor repertoires against hen egg white lysozyme (HEL) in nontransgenic versus HEL transgenic mice. The Journal of experimental medicine, 180 (September):862-872, 1994.

- [176] VICTOR P ARGAET, CHRISTOPHER W SCHMIDT, SCOTT R BURROWS, SHARON L SILINS, MIKE G KURILLA, DENISE L DOOLAN, ANDREAS SUHRBIER, DENIS J MOSS, ELLIOT KIEFF, TOM B SCULLEY, AND IHOR S MISKO. Dominant selection of an invariant T cell antigen receptor in response to persistent infection by Epstein-Barr virus. The Journal of experimental medicine, 180(December):2335-2340, 1994.
- [177] STEPHEN J TURNER, PETER C DOHERTY, JAMES MCCLUSKEY, AND JAMIE ROSSJOHN. Structural determinants of Tcell receptor bias in immunity. Nature Reviews Immunology, 6(12):883-94, December 2006.
- [178] VANESSA VENTURI, DAVID A PRICE, DANIEL C DOUEK, AND MILES P DAVENPORT. The molecular basis for public T-cell responses? Nature Reviews Immunology, 8:231– 238, 2008.
- [179] JOHN J MILES, DANIEL C DOUEK, AND DAVID A PRICE. Bias in the αβ T-cell repertoire: implications for disease pathogenesis and vaccination. Immunology and cell biology, 89(3):375-87, March 2011.
- [180] HANJIE LI, CONGTING YE, GUOLI JI, AND JIAHUAI HAN. Determinants of public T cell responses. Cell research, 22(1):33-42, January 2012.
- [181] VANESSA VENTURI, KATHERINE KEDZIERSKA, DAVID A PRICE, PETER C DOHERTY, DANIEL C DOUEK, STEPHEN J TURNER, AND MILES P DAVENPORT. Sharing of T cell receptors in antigen-specific responses is driven by convergent recombination. Proceedings of the National Academy of Sciences of the United States of America, 103(49):18691-6, December 2006.
- [182] MÁIRE F QUIGLEY, HUI YEE, VANESSA VENTURI, ROSS LINDSAY, KYLIE M QUINN, AND ROBERT A SEDER. Convergent recombination shapes the clonotypic landscape of the naive T-cell repertoire. Proceedings of the National Academy of Sciences USA, 107(45):19414-19419, 2010.
- [183] A. MURUGAN, T. MORA, A. M. WALCZAK, AND C. G. CALLAN. Statistical inference of the generation probability of T-cell receptors from sequence repertoires. Proceedings of the National Academy of Sciences, September 2012.
- [184] YUVAL ELHANATI, ANAND MURUGAN, CURTIS G CALLAN, THIERRY MORA, AND ALEKSANDRA M WALCZAK. Quantifying selection in immune receptor repertoires. Proceedings of the National Academy of Sciences of the United States of America, 111(27):9875–9880, June 2014.
- [185] DALE I GODFREY AND MITCHELL KRONENBERG. Going both ways : immune regulation via CD1d-dependent NKT cells. The Journal of Clinical Investigation, 114(10):1379-1388, 2004.
- [186] THE MHC SEQUENCING CONSORTIUM, B. AGUADO, S. BAHRAM, S. BECK, R.D.CAMPBELL, S. A. FORBES, D. GERAGHTY, T. GUILLAUDEUX, L. HOOD, R. HORTON, H. INNON, M. JANER, C. JASONI, A. MADAN, S. MILNE, M. NEVILLE, A. OKA, S. QIN, G. RIBAS-DESPUIG, J. ROGERS, L. ROWEN, T. SHIINA, T. SPIES, G. TAMIYA, H. TASHIRO, J. TROWSDALE, Q. VU, L. WILLIAMS, AND M. YAMAZAKI. Complete sequence and gene map of a human major histocompatibility complex. The MHC sequencing consortium. Nature, 401(6756):921-3, October 1999.
- [187] ROGER HORTON, LAURENS WILMING, VIKKI RAND, RUTH C LOVERING, ELSPETH A BRUFORD, VARSHA K KHODIYAR, MICHAEL J LUSH, SUE POVEY, C CONOVER TALBOT, MATHEW W WRIGHT, HESTER M WAIN, JOHN TROWSDALE, ANDREAS ZIEGLER, AND STEPHAN BECK. Gene map of the extended human MHC. Nature reviews. Genetics, 5(12):889–99, December 2004.

- [188] MIRJANA RADOSAVLJEVIC AND SEIAMAK BAHRAM. In vivo immunogenetics: from MIC to RAET1 loci. Immunogenetics, 55(1):1–9, April 2003.
- [189] DONNA G ALBERTSON, RITA FISHPOOLL, PAUL SHERINGTON, ELISABETH NACHEVA, AND CESAR MILSTEIN. Sensitive and high resolution in situ hybridization to human chromosomes using biotin labelled probes: assignment of the human thymocyte CD1 antigen genes to. The EMBO journal, 7(9):2801-2805, 1988.
- [190] KEIICHIRO HASHIMOTO, MOMOKI HIRAI, AND YOSHIKAZU KURO-SAWA. A gene outside the human MHC related to classical HLA class I genes. Science, 269(5224):693-695, 1995.
- [191] A N AKBAR, L TERRY, A TIMMS, P C BEVERLEY, AND G JANOSY. Loss of CD45R and gain of UCHL1 reactivity is a feature of primed T cells. Journal of immunology (Baltimore, Md. : 1950), 140(7):2171-8, April 1988.
- [192] M MERKENSCHLAGER AND P C BEVERLEY. Evidence for differential expression of CD45 isoforms by precursors for memory-dependent and independent cytotoxic responses: human CD8 memory CTLp selectively express CD45RO (UCHL1). International immunology, 1(4):450-9, January 1989.
- [193] FEDERICA SALLUSTO, JENS GEGINAT, AND ANTONIO LANZAVEC-CHIA. Central memory and effector memory T cell subsets: function, generation, and maintenance. Annual review of immunology, 22:745-63, January 2004.
- [194] M K MAINI, L R WEDDERBURN, F C HALL, A WACK, G CASO-RATI, AND P C BEVERLEY. A comparison of two techniques for the molecular tracking of specific Tcell responses; CD4+ human T-cell clones persist in a stable hierarchy but at a lower frequency than clones in the CD8+ population. Immunology, 94(4):529–35, August 1998.
- [195] K. E. FOULDS, L. A. ZENEWICZ, D. J. SHEDLOCK, J. JIANG, A. E. TROY, AND H. SHEN. Cutting Edge: CD4 and CD8 T Cells Are Intrinsically Different in Their Proliferative Responses. The Journal of Immunology, 168(4):1528-1532, February 2002.
- [196] HUNG SIA TEH, PAWEL KISIELOW, BERNADETTE SCOTT, HI-ROYUKI KISHI, YASUSHI UEMATSU, HORST BLUTHMANN, AND HARALD VON BOEHMER. Thymic major histocompatibility complex antigens and the  $\alpha\beta$  T-cell receptor determine the CD4/CD8 phenotype of T cells. Nature, 335(September):229–233, 1988.
- [197] SUSAN CHAN, MARGARIDA CORRELA-NEVES, CHRISTOPHE BENOIST, AND DIANE MATHIS. CD4/CD8 lineage commitment: matching fate with competence. Immunological reviews, 165:195-207, 1998.
- [198] PAUL L KLARENBEEK, PAUL P TAK, BARBERA D C VAN SCHAIK, AEILKO H ZWINDERMAN, MARJA E JAKOBS, ZHUOLI ZHANG, AN-TOINE H C VAN KAMPEN, RENÉ A W VAN LIER, FRANK BAAS, AND NIEK DE VRIES. Human T-cell memory consists mainly of unexpanded clones. Immunology letters, 133(1):42-8, September 2010.
- [199] MITCHELL S CAIRO AND JOHN E WAGNER. Placental and/or umbilical cord blood: an alternative source of hematopoietic stem cells for transplantation. Blood, 90(12):4665-4678, 1997.
- [200] L GARDERET, N DULPHY, C DOUAY, N CHALUMEAU, V SCHA-EFFER, M T ZILBER, A LIM, J EVEN, N MOONEY, C GELIN, E GLUCKMAN, D CHARRON, AND A TOUBERT. The umbilical cord blood alphabeta T-cell repertoire: characteristics of a polyclonal and naive but completely formed repertoire. Blood, 91(1):340-6, January 1998.

- [201] EREZ RECHAVI, ATAR LEV, YU NEE LEE, AMOS J SIMON, YOAV YINON, SCHLOMO LIPITZ, NINETTE AMARIGLIO, BOAZ WEISZ, LUIGI D NOTARANGELO, AND RAZ SOMECH. Timely and spatially regulated maturation of B and T cell repertoire during human fetal development. Science translational medicine, 7(276):1-12, 2015.
- [202] M. A. HALL, J L REID, AND J S LANCHBURY. The distribution of human TCR junctional region lengths shifts with age in both CD4 and CD8 T cells. International immunology, 10(10):1407–19, October 1998.
- [203] MOLLY BOGUE, SUSAN GILFILLAN, CHRISTOPHE BENOIST, AND DIANE MATHIS. Regulation of N-region diversity in antigen receptors through thymocyte differentiation and thymus ontogeny. Proceedings of the National Academy of Sciences USA, 89(November):11011– 11015, 1992.
- [204] A BONATI, P ZANELLI, S FERRARI, A PLEBANI, B STARCICH, M SAVI, AND T M NERI. T-cell receptor beta-chain gene rearrangement and expression during human thymic ontogenesis. Blood, 79(6):1472-83, March 1992.
- [205] BRENDAN MARSHALL, RUTH SCHULZ, MIN ZHOU, AND ANDREW MELLOR. Alternative Splicing and Hypermutation of a Nonproductively Rearranged TCR α -Chain in a T Cell Hybridoma. The Journal of Immunology, 162:871-877, 1999.
- [206] NICHOLAS J MCGLINCY AND CHRISTOPHER W J SMITH. Alternative splicing resulting in nonsense-mediated mRNA decay: what is the meaning of nonsense? *Trends in biochemical sciences*, 33(8):385–93, August 2008.
- [207] ANDREA SBONER, XINMENG JASMINE MU, DOV GREENBAUM, RAYMOND K AUERBACH, AND MARK B GERSTEIN. The real cost of sequencing: higher than you think! Genome biology, 12(8):125, January 2011.
- [208] ANNE EUGSTER, ANNETT LINDNER, ANNE-KRISTIN HENINGER, CARMEN WILHELM, SEVINA DIETZ, MARA CATANI, ANETTE-G ZIEGLER, AND EZIO BONIFACIO. Measuring T cell receptor and T cell gene expression diversity in antigen-responsive human CD4(+) T cells. Journal of immunological methods, 400-401:13-22, November 2013.
- [209] JANKO NIKOLICH-ZUGICH, MARK K SLIFKA, AND ILHEM MES-SAOUDI. The many important facets of T-cell repertoire diversity. Nature reviews. Immunology, 4(2):123– 32, February 2004.
- [210] JEFFREY M. LEIDEN. Transcriptional regulation of T cell receptor genes. Annual review of immunology, 11(1):539-570, 1993.
- [211] LEE ROWEN, BEN F KOOP, AND LEROY HOOD. The complete 685-kilobase DNA sequence of the human  $\beta$  T cell receptor locus. *Science*, 272(5269):1755–1762, 1996.
- [212] T LINDSTEN, C H JUNE, AND C B THOMPSON. Transcription of T cell antigen receptor genes is induced by protein kinase C activation. Journal of Immunology, 141(5):1769-74, September 1988.
- [213] FLORENCE PAILLARD, GHISLAINE STERKERS, GEORGES BIS-MUTH, ELISABETH GOMARD, AND CATHERNE VAQUERO. Lymphokine mRNA and T cell multireceptor mRNA of the Ig super gene family are reciprocally modulated during human T cell activation. European journal of immunology, 18:1643-1646, 1988.

- [214] G. P. LENNON, J. E. SILLIBOURNE, E. FURRIE, M. J. DOHERTY, AND R. A. KAY. Antigen Triggering Selectively Increases TCRBV Gene Transcription. The Journal of Immunology, 165(4):2020-2027, August 2000.
- [215] ALEXANDRA GALLARD, GILLES FOUCRAS, CHRISTIANE COUREAU, AND JEAN-CHARLES GUÉRY. Tracking T cell clonotypes in complex T lymphocyte populations by real-time quantitative PCR using fluorogenic complementarity-determining region-3specific probes. Journal of immunological methods, 270(2):269-80, December 2002.
- [216] YASUHIRO MINAMI, LAWRENCE E SAMELSON, AND RICHARD D KLAUSNER. Internalization and Cycling of the T Cell Antigen Receptor. The Journal of biological chemistry, 262(27):13342-13347, 1987.
- [217] VICTORIA L CROTZER, ALLAN S MABARDY, ARTHUR WEISS, AND FRANCES M BRODSKY. T cell receptor engagement leads to phosphorylation of clathrin heavy chain during receptor internalization. The Journal of experimental medicine, 199(7):981–91, April 2004.
- [218] MS GOTTLIEB, HM SCHANKER, PT FAN, A SAXON, AND JD WEISMAN. Pneumocystis pneumoniaLos Angeles. Morbidity and Mortality Weekly Report, 30(21):250-252, 1981.
- [219] R M DU BOIS AND M A BRANTHWAITE. Primary Pneumocystis carinii and cytomegalovirus infections. The Lancet, 12:1339, 1981.
- [220] VINCENT QUAGLIARELLO. The Acquired Immunodeficiency Syndrome: Current Status. The Yale Journal of Biology and Medicine, 55:443-452, 1982.
- [221] YUAN CHANG, ETHEL CESARMAN, MELISSA S PESSIN, FRANK LEE, JANICE CULPEPPER, M KNOWLES, PATRICK S MOORE, AND DANIEL M KNOWLES. Identification of herpesvirus-like DNA sequences in AIDS-associated Kaposi's sarcoma. Science, 266(5192):1865-1869, 1994.
- [222] PS MOORE AND Y CHANG. Detection of herpesvirus-like DNA sequences in Kaposi's sarcoma in patients with and those without HIV infection. New England Journal of Medicine, 332(18):1181-1185, 1995.
- [223] CDC. Epidemiologic Notes and Reports Update on Kaposi's Sarcoma and Opportunistic Infections in Previously Healthy Persons. Morbidity and Mortality Weekly Report, 31(19):1-2, 1982.
- [224] CDC. Current Trends Update on Acquired Immune Deficiency Syndrome (AIDS). Morbidity and Mortality Weekly Report, 31(37), 1982.
- [225] F. BARRÉ-SINOUSSI, J. C. CHERMANN, F. REY, M. T. NUGEYRE, S. CHAMARET, J. GRUEST, C. DAUGUET, C. AXLER-BLIN, F. VÉZINET-BRUN, C. ROUZIOUX, W. ROZENBAUM, AND L. MON-TAGNIER. Isolation of a T-Lymphotropic Retrovirus from a Patient at Risk for Acquired Immune Deficiency Syndrome (AIDS). Science, 220(4599):868– 871, 1983.
- [226] MIKULAS POPOVIC, M. G. SARNGADHARAN, ELIZABETH READ, AND ROBERT C. GALLO. Detection, Isolation, and Continuous Production of Cytopathic Retroviruses (HTLV-III) from Patients with AIDS and Pre-AIDS. Science, 224(4648):497-500, 1984.
- [227] JAY A. LEVY, ANTHONY D. HOFFMAN, SUSAN M. KRAMER, JILL A. LANDIS, JONI M. SHIMABUKURO, AND LYNDON S. OS-HIRO. Isolation of Lymphocytopathic Retroviruses from San Francisco Patients with AIDS. Science, 225(4664):840-842, 1984.

- [228] L RATNER, R.C. GALLO, AND F. WONG-STAAL. HTLV-III, LAV, ARV are variants of same AIDS virus. Nature, 313:636-637, 1985.
- [229] JEAN L. MARX. A Virus by Any Other Name. Science, 227(4693):1449–1451, 1985.
- [230] JOHN COFFIN, ASHLEY HAASE, JAY A LEVY, LUC MONTAGNIER, STEVEN OROSZLAN, NATALIE TEICH, HOWARD TEMIN, KUMAO TOYOSHIMA, HAROLD VARMUS, PETER VOGT, AND ROBIN WEISS. What to call the AIDS virus? Nature, 321:10, 1986.
- [231] PIERRE SONIGO, MARC ALIZON, KATHERINE STASKUS, DAVID KLATZMANN, STEWART COLE, OLIVIER DANOS, ERNEST RETZEL, PIERRE TIOLLAIS, ASHLEY HAASE, AND SIMON WAIN-HOBS. Nucleotide sequence of the visna lentivirus: relationship to the AIDS virus. Cell, 42(August):369–382, 1985.
- [232] ANDREW R MOSS AND PETER BACCHETTI. Natural history of HIV infection. AIDS, 3:55–61, 1989.
- [233] SALETA SIERRA, BERND KUPFER, AND ROLF KAISER. Basics of the virology of HIV-1 and its replication. Journal of clinical virology, 34(4):233-44, December 2005.
- [234] JANET EMBRETSON, MARY ZUPANCIC, JORGE L RIBAS, ALLEN BURKE, PAUL RACZ, KLARA TENNER-RACZ, AND ASHLEY T HAASE. Massive covert infection of helper T lymphocytes and macrophages by HIV during the incubation period of AIDS. Nature, 362(359-362), 1993.
- [235] GIUSEPPE PANTALEO, CECILIA GRAZIOSI, JAMES F DEMAREST, LUCA BUTINI, MARIA MONTRONI, CECIL H FOX, JAN M OREN-STEIN, DONALD P KOTLER, AND ANTHONY S FAUCI. HIV infection is active and progressive in lymphoid tissue during the clinically latent stage of disease. Nature, 362(March):355–358, 1993.
- [236] ANGUS G DALGLEISH, PETER CL BEVERLEY, PAUL R CLAPHAM, DOROTHY H CRAWFORD, MELVYN F GREAVES, AND ROBIN A WEISS. The CD4 (T4) antigen is an essential component of the receptor for the AIDS retrovirus. Nature, 312(December):763-767, 1984.
- [237] PAUL JAY MADDON, ANGUS G DALGLEISH, J STEVEN MCDOU-GAL, PAUL R CLAPHAM, ROBIN A WEISS, AND RICHARD AXEL. The T4 Gene Encodes the AIDS Virus Receptor and Is Expressed in the Immune System and the Brain. Cell, 47:333-348, 1986.
- [238] J. B. ALIMONTI. Mechanisms of CD4+ T lymphocyte cell death in human immunodeficiency virus infection and AIDS. Journal of General Virology, 84(7):1649–1661, July 2003.
- [239] GILAD DOITSH, MARIELLE CAVROIS, KARA G LASSEN, ORLANDO ZEPEDA, ZHIYUAN YANG, MARIO L SANTIAGO, ANDREW M HEBBELER, AND WARNER C GREENE. Abortive HIV infection mediates CD4 T cell depletion and inflammation in human lymphoid tissue. Cell, 143(5):789– 801, November 2010.
- [240] GILAD DOITSH, NICOLE L K GALLOWAY, XIN GENG, ZHIYUAN YANG, KATHRYN M MONROE, ORLANDO ZEPEDA, PETER W HUNT, HIROYU HATANO, STEFANIE SOWINSKI, ISA MUÑOZ ARIAS, AND WARNER C GREENE. Cell death by pyroptosis drives CD4 T-cell depletion in HIV-1 infection. Nature, 505:509-514, December 2013.
- [241] STEVEN G DEEKS. HIV infection, inflammation, immunosenescence, and aging. Annual review of medicine, 62:141-55, January 2011.

- [242] JAMIE D K WILSON, GRAHAM S OGG, RACHEL L ALLEN, PHILIP J R GOULDER, ANTHONY KELLEHER, ANDY K SEWELL, CHRISTO-PHER A O O'CALLAGHAN, SARAH L ROWLAND-JONES, MAR-GARET F C CALLAN, AND ANDREW J MCMICHAEL. Oligoclonal Expansions of CD8+ T Cells in Chronic HIV Infection Are Antigen Specific. Journal of experimental medicine, 188(4):785-790, July 1998.
- [243] VIJAY K KUCHROO, ANA C ANDERSON, AND CONSTANTINOS PETROVAS. Coinhibitory receptors and CD8 T cell exhaustion in chronic infections. Current opinion in HIV and AIDS, 9(5):439-45, September 2014.
- [244] PERSEPHONE BORROW, HANNA LEWICKI, BEATRICE H HAHN, GEORGE M SHAW, AND MICHAEL B A OLDSTONE. Virusspecific CD8+ cytotoxic T-lymphocyte activity associated with control of viremia in primary human immunodeficiency virus type 1 infection. Journal of virology, 68(9):6103-6110, 1994.
- [245] MICHEL R. KLEIN, CARD A. VAN BAALEN, AGNES M. HOLWERDA, SUSANA R. KERKHOF GARDE, RICHARD J. BENDE, IRENEUS P. M. KEET, JAN-KAREL M. EEFTINCK-SCHATTENKERK, ALBERT D. M. E. OSTERHAUS, HANNEKE SCHUTEMAKER, AND FRANK MIEDEMA. Gag-specific cytotoxic T lymphocyte responses during the clinical course of HIV-1 infection: a longitudinal analysis of rapid progressors and long-term asymptomatics. The Journal of experimental medicine, 181(April):1365-1372, 1995.
- [246] J. CAO, J. MCNEVIN, U. MALHOTRA, AND M. J. MCELRATH. Evolution of CD8+ T Cell Immunity and Viral Escape Following Acute HIV-1 Infection. The Journal of Immunology, 171(7):3837-3846, September 2003.
- [247] HENDRIK STREECK AND DOUGLAS F NIXON. T cell immunity in acute HIV-1 infection. The Journal of infectious diseases, 202 Suppl:S302-8, October 2010.
- [248] TAO DONG, GUILLAUME STEWART-JONES, NAN CHEN, PHILIPPA EASTERBROOK, XIAONING XU, LAURA PAPAGNO, VICTOR APPAY, MICHAEL WEEKES, CHRIS CONLON, CELSA SPINA, SUSAN LIT-TLE, GAVIN SCREATON, ANTON VAN DER MERWE, DOUGLAS D RICHMAN, ANDREW J MCMICHAEL, E YVONNE JONES, AND SARAH L ROWLAND-JONES. HIV-specific cytotoxic T cells from long-term survivors select a unique T cell receptor. The Journal of experimental medicine, 200(12):1547-57, December 2004.
- [249] JORGE R ALMEIDA, DAVID A. PRICE, LAURA PAPAGNO, ZAÏNA AÏT ARKOUB, DELPHINE SAUCE, ETHAN BORNSTEIN, TEDI E ASHER, ASSIA SAMRI, AURÉLIE SCHNURIGER, IOANNIS THEODOROU, DOMINIQUE COSTAGLIOLA, CHRISTINE ROUZIOUX, HENRI AGUT, ANNE-GENEVIÈVE MARCELIN, DANIEL DOUEK, BRIGITTE AUTRAN, AND VICTOR APPAY. Superior control of HIV-1 replication by CD8+ T cells is reflected by their avidity, polyfunctionality, and clonal turnover. The Journal of experimental medicine, 204(10):2473-2485, October 2007.
- [250] ASIER SÁEZ-CIRIÓN, CHRISTINE LACABARATZ, OLIVIER LAM-BOTTE, PIERRE VERSMISSE, ALEJANDRA URRUTIA, FAROUDY BOUFASSA, FRANÇOISE BARRÉ-SINOUSSI, JEAN-FRANÇOIS DEL-FRAISSY, MARTINE SINET, GIANFRANCO PANCINO, AND ALAIN VENET. HIV controllers exhibit potent CD8 T cell capacity to suppress HIV infection ex vivo and peculiar cytotoxic T lymphocyte activation phenotype. Proceedings of the National Academy of Sciences of the United States of America, 104(16):6776-81, April 2007.
- [251] MARY CARRINGTON AND STEPHEN J O'BRIEN. The influence of HLA genotype on AIDS. Annual review of medicine, 54:535-51, January 2003.

- [252] HIROAKI MITSUYA, KENT J WEINHOLD, PHILLIP A FURMAN, MARTY H ST. CLAIR, SANDRA NUSINOFF LERRMAN, ROBERT C GALLO, DANI BOLOGNESI, DAVID W BARRY, AND SAMUEL BRODER. 3'-Azido-3'-deoxythymidine (BW A509U): An antiviral agent that inhibits the infectivity and cytopathic effect of human Tlymphotropic virus type III / lymphadenopathyassociated virus in vitro. Proceedings of the National Academy of Sciences USA, 82(October):7096-7100, 1985.
- [253] ROBERT YARCHOAN, KENT J WEINHOLD, H K I M LYERLY, EDWARD GELMANN, ROBERT M BLUM, GENE M SHEARER, HI-ROAKI MITSUYA, JERRY M COLLINS, CHARLES E MYERS, RAY-MOND W KLECKER, PHILLIP D MARKHAM, DAVID T DURACK, S NUSINOFF LEHRMAN, DAVID W BARRY, MARGARET A FISCHL, ROBERT C GALLO, DANI P BOLOGNESI, AND SAMUEL BRODER. Administration of 3'azido-3'-deoxythymidine, an inhibitor of HTLV-III/LAV replication, to patients with AIDS or AIDS-related complex. The Lancet, 327(8481):575-580, 1986.
- [254] BRENDEN A LARDER AND SHARON D KEMP. Multiple Mutations in HIV-1 Reverse Transcriptase Confer High-Level Resistance to Zidovudine (AZT). Science, 246(4934):1155-1158, 1989.
- [255] ROBERT YARCHOAN, HIROAKI MITSUYA, AND SAMUEL BRODER. Strategies for the Combination Therapy of HIV. Journal of acquired immune deficiency syndromes, 3(Suppl 2):S99–S103, 1990.
- [256] R D MOORE AND R E CHAISSON. Natural history of HIV infection in the era of combination antiretroviral therapy. AIDS, 13(14):1933-42, October 1999.
- [257] IAN WILLIAMS, DUNCAN CHURCHILL, JANE ANDERSON, MARTA BOFFITO, MARK BOWER, GUS CAIRNS, KATE CWYNARKSI, SIMON EDWARDS, SARAH FIDLER, MARTIN FISHER, ANDREW FREEDMNA, ANNA MARIA GERETTI, YVONNE GILLEECE, ROB HORNE, MAR-GARET JOHNSON, SAYE KHOO, CLIFFORD LEEN, NEAL MAR-SHALL, MARK NELSON, CHLOE ORKIN, NICHOLAS PATON, FRANK POST, ANTON POZNIAK, CAROLINE SABIN, ROY TREVELION, ANDREW USTIANOWSKI, JOHN WALSH, LAURA WATERS, ED-MUND WILKINS, ALAN WINSTON, AND MIKE YOULE. British HIV Association guidelines for the treatment of HIV-1-positive adults with antiretroviral therapy 2012. HIV Medicine, 15(Suppl 1):1-85, 2014.
- [258] ANTIRETROVIRAL THERAPY COHORT COLLABORATION. Life expectancy of individuals on combination antiretroviral therapy in high-income countries: a collaborative analysis of 14 cohort studies. Lancet, 372(9635):293-299, 2008.
- [259] WORLD HEALTH ORGANIZATION. Global Update on the health sector response to HIV, 2014. Technical report, 2014.
- [260] L IMBERTI, A SOTTINI, A BETTINARDI, M PUOTI, AND D PRIMI. Selective depletion in HIV infection of T cells that bear specific T cell receptor V beta sequences. *Science (New York, N.Y.)*, 254(5033):860-2, November 1991.
- [261] JEFFREY LAURENCE, ANDREW S. HODSTEV, AND DAVID N. POS-NETT. Superantigen implicated in dependence of HIV-1 replication in T cells on TCR V beta expression. Nature, 358:255-259, 1992.
- [262] P BAHADORAN, F RIEUX-LAUCAT, F LE DEIST, S BLANCHE, A FISCHER, AND J P DE VILLARTAY. Lack of selective V beta deletion in peripheral CD4+ T cells of human immunodeficiency virus-infected infants. European journal of immunology, 23(8):2041-4, August 1993.

- [263] NAGET REBAI, GIUSEPPE PANTALEO, JAMES F DEMAREST, CRISTINA CIURLI, HUGO SOUDEYNS, JOSEPH W ADELSBERGER, MAURO VACCAREZZA, ROBERT E WALKER, RAFICK P SEKALY, AND ANTHONY S FAUCI. Analysis of the T-cell receptor beta-chain variable-region repertoire in monozygotic twins discordant for human immunodeficieny virus: Evidence for perturbations of specific Vbeta segments in CD4+ T cells of the viruspositive twins. Proceedings of the National Academy of Sciences, 91(February):1529–1533, 1994.
- [264] M. CONNORS, J.A. KOVACS, S. KREVAT, J.C. GEA-BANACLOCHE, M.C. SNELLER, M. FLANIGAN, J.A. METCALF, R.E. WALKER, J. FALLOON, AND M. BASELER. HIV infection induces changes in CD4+ T-cell phenotype and depletions within the CD4+ T-cell repertoire that are not immediately restored by antiviral or immunebased therapies. Nature medicine, 3(5):533-540, 1997.
- [265] J C GEA-BANACLOCHE, L MARTINO, J M MICAN, C W HALLA-HAN, M BASELER, R STEVENS, L LAMBERT, M POLIS, H C LANE, AND M CONNORS. Longitudinal changes in CD4+ T cell antigen receptor diversity and naive/memory cell phenotype during 9 to 26 months of antiretroviral therapy of HIV-infected patients. AIDS research and human retroviruses, 16(17):1877-86, November 2000.
- [266] MONICA KHARBANDA, THOMAS W MCCLOSKEY, RAJENDRA PAHWA, MEI SUN, AND SAVITA PAHWA. Alterations in T-Cell Receptor Vbeta Repertoire of CD4 and CD8 T Lymphocytes in Human Immunodeficiency Virus-Infected Children. Clinical and diagnostic laboratory immunology, 10(1):53-58, 2003.
- [267] S KOSTENSE, F M RAAPHORST, D W NOTERMANS, J JOLING, B HOOIBRINK, N G PAKKER, S A DANNER, J M TEALE, AND F MIEDEMA. Diversity of the T-cell receptor BV repertoire in HIV-1-infected patients reflects the biphasic CD4+ T-cell repopulation kinetics during highly active antiretroviral therapy. AIDS, 12(18):F235-40, December 1998.
- [268] JOSEPH A CONRAD, RAMESH K RAMALINGAM, COLEY B DUN-CAN, RITA M SMITH, JIE WEI, LOUISE BARNETT, BRENNA C SI-MONS, SHELLY L LOREY, AND SPYROS A KALAMS. Antiretroviral therapy reduces the magnitude and T cell receptor repertoire diversity of HIV-specific T cell responses without changing T cell clonotype dominance. Journal of virology, 86(8):4213-21, April 2012.
- [269] PAUL D BAUM, JENNIFER J YOUNG, DIANE SCHMIDT, QIAN-JUN ZHANG, REBECCA HOH, MICHAEL BUSCH, JEFFREY MAR-TIN, STEVEN DEEKS, AND JOSEPH M MCCUNE. Blood T cell receptor diversity decreases during the course of HIV infection but the potential for a diverse repertoire persists. Blood, 119:3469–3477, February 2012.
- [270] SERGIO SERRANO-VILLAR, TALIA SAINZ, SULGGI A LEE, PE-TER W HUNT, ELIZABETH SINCLAIR, BARBARA L SHACKLETT, APRIL L FERRE, TIMOTHY L HAYES, MA SOMSOUK, PRISCILA Y HSUE, MARK L VAN NATTA, CURTIS L MEINERT, MICHAEL M LE-DERMAN, HIROYU HATANO, VIVEK JAIN, YONG HUANG, FREDER-ICK M HECHT, JEFFREY N MARTIN, JOSEPH M MCCUNE, SAN-TIAGO MORENO, AND STEVEN G DEEKS. HIV-infected individuals with low CD4/CD8 ratio despite effective antiretroviral therapy exhibit altered T cell subsets, heightened CD8+ T cell activation, and increased risk of non-AIDS morbidity and mortality. PLoS pathogens, 10(5):e1004078, May 2014.
- [271] W. NDIFON, H. GAL, E. SHIFRUT, R. AHARONI, N. YIS-SACHAR, N. WAYSBORT, S. REICH-ZELIGER, R. ARNON, AND N. FRIEDMAN. Chromatin conformation governs Tcell receptor J gene segment usage. Proceedings of

the National Academy of Sciences, **109**(39):15865–15870, September 2012.

- [272] STEPHEN C. THRELKELD, PEGGY A. WENTWORTH, SPYROS A. KALAMS, BARBARA M. WILKES, DEBBIE J. RUHL, ELISSA KEOGH, JOHN SIDNEY, SCOTT SOUTHWOOD, BRUCE D. WALKER, AND ALESSANDRO SETTE. Degenerate and promiscuous recognition by CTL of peptides presented by the MHC class I A3-like superfamily. Journal of immunology, 159(4):1648-1657, 1997.
- [273] T. UENO, H. TOMIYAMA, AND M. TAKIGUCHI. Single T Cell Receptor-Mediated Recognition of an Identical HIV-Derived Peptide Presented by Multiple HLA Class I Molecules. The Journal of Immunology, 169(9):4961-4969, November 2002.
- [274] GERALDINE M A GILLESPIE, GUILLAUME STEWART-JONES, JAYA RENGASAMY, TARA BEATTIE, JOB J BWAYO, FRANCIS A PLUM-MER, RUPERT KAUL, ANDREW J MCMICHAEL, PHILIPPA EAST-ERBROOK, TAO DONG, E YVONNE JONES, AND SARAH L ROWLAND-JONES. Strong TCR conservation and altered T cell cross-reactivity characterize a B\*57restricted immune response in HIV-1 infection. Journal of Immunology, 177(6):3893-3902, September 2006.
- [275] VANESSA VENTURI, HUI YEE CHIN, TEDI E ASHER, KRISTIN LADELL, PHILLIP SCHEINBERG, ETHAN BORNSTEIN, DAVID VAN BOCKEL, ANTHONY D KELLEHER, DANIEL C DOUEK, DAVID A PRICE, AND MILES P DAVENPORT. TCR beta-chain sharing in human CD8+ T cell responses to cytomegalovirus and EBV. Journal of immunology (Baltimore, Md. : 1950), 181(11):7853-7862, December 2008.
- [276] CHRISTOPH T BERGER, NICOLE FRAHM, DAVID A PRICE, BEAT-RIZ MOTHE, MUSIE GHEBREMICHAEL, KARI L HARTMAN, LEAH M HENRY, JASON M BRENCHLEY, LAURA E RUFF, VANESSA VENTURI, FLORENCIA PEREYRA, JOHN SIDNEY, ALESSANDRO SETTE, DANIEL C DOUEK, BRUCE D WALKER, DANIEL E KAUFMANN, AND CHRISTIAN BRANDER. High-functionalavidity cytotoxic T lymphocyte responses to HLA-B-restricted Gag-derived epitopes associated with relative HIV control. Journal of virology, 85(18):9334-45, September 2011.
- [277] DANIEL MENDOZA, CASSANDRA ROYCE, LAURA E RUFF, DAVID R AMBROZAK, MÁIRE F QUIGLEY, THURSTON DANG, VANESSA VENTURI, DAVID A PRICE, DANIEL C DOUEK, STEPHEN A MIGUELES, AND MARK CONNORS. HLA B\*5701-positive long-term nonprogressors/elite controllers are not distinguished from progressors by the clonal composition of HIV-specific CD8+ T cells. Journal of virology, 86(7):4014-8, April 2012.
- [278] CHIHIRO MOTOZONO, NOZOMI KUSE, XIAOMING SUN, PIERRE J RIZKALLAH, ANNA FULLER, SHINICHI OKA, DAVID K COLE, AN-DREW K SEWELL, AND MASAFUMI TAKIGUCHI. Molecular Basis of a Dominant T Cell Response to an HIV Reverse Transcriptase 8-mer Epitope Presented by the Protective Allele HLA-B\*51:01. Journal of immunology (Baltimore, Md. : 1950), 192(7):3428-34, April 2014.
- [279] N. KHAN, N. SHARIFF, M. COBBOLD, R. BRUTON, J. A. AINSWORTH, A. J. SINCLAIR, L. NAYAK, AND P. A. H. MOSS. Cytomegalovirus Seropositivity Drives the CD8 T Cell Repertoire Toward Greater Clonality in Healthy Elderly Individuals. The Journal of Immunology, 169(4):1984-1992, August 2002.
- [280] QIN OUYANG, WOLFGANG M WAGNER, STEFFEN WALTER, CLAU-DIA A MÜLLER, ANDERS WIKBY, GERALDINE AUBERT, TATJANA KLATT, STEFAN STEVANOVIC, TONY DODI, AND GRAHAM PAW-ELEC. An age-related increase in the number of

CD8+ T cells carrying receptors for an immunodominant EpsteinBarr virus (EBV) epitope is counteracted by a decreased frequency of their antigen-specific responsiveness. *Mechanisms of Ageing and Development*, **124**(4):477–485, April 2003.

- [281] RANGSIMA REANTRAGOON, ALEXANDRA J CORBETT, ISAAC G SAKALA, NICHOLAS A GHERANDIN, JOHN B FURNESS, ZHENJUN CHEN, SIDONIA B G ECKLE, ADAM P ULDRICH, RICHARD W BIRKINSHAW, ONISHA PATEL, LYUDMILA KOSTENKO, BRONWYN MEEHAN, KATHERINE KEDZIERSKA, LIGONG LU, DAVID P FAIR-LIE, TED H HANSEN, DALE I GODFREY, JAMIE ROSSJOHN, JAMES MCCLUSKEY, AND LARS KJER-NIELSEN. Antigen-loaded MR1 tetramers define T cell receptor heterogeneity in mucosal-associated invariant T cells. The Journal of experimental medicine, 210(11):2305-20, October 2013.
- [282] HUI YEE GREENAWAY, BENEDICT NG, DAVID A PRICE, DANIEL C DOUEK, MILES P DAVENPORT, AND VANESSA VENTURI. NKT and MAIT invariant TCRα sequences can be produced efficiently by VJ gene recombination. Immunobiology, 218(2):213-24, February 2013.
- [283] DEMIN LI AND XIAO-NING XU. NKT cells in HIV-1 infection. Cell research, 18(8):817-22, August 2008.
- [284] H. J. J. VAN DER VLIET, B. M. E. VON BLOMBERG, M. D. HAZEN-BERG, N. NISHI, S. A. OTTO, B. H. VAN BENTHEM, M. PRINS, F. A. CLAESSEN, A. J. M. VAN DEN EERTWECH, G. GIACCONE, F. MIEDEMA, R. J. SCHEPER, AND H. M. PINEDO. Selective Decrease in Circulating V 24+V 11+ NKT Cells During HIV Type 1 Infection. The Journal of Immunology, 168(3):1490-1495, February 2002.
- [285] CAROLINE S FERNANDEZ, ANTHONY D KELLEHER, ROBERT FIN-LAYSON, DALE I GODFREY, AND STEPHEN J KENT. NKT cell depletion in humans during early HIV infection. Immunology and cell biology, (January):1-13, April 2014.
- [286] EVAN W NEWELL AND MARK M DAVIS. Beyond model antigens: high-dimensional methods for the analysis of antigen-specific T cells. Nature biotechnology, 32(2):149-57, February 2014.
- [287] BRINDA EMU, WALTER J MORETTO, REBECCA HOH, MELISSA KRONE, JEFFREY N MARTIN, DOUGLAS F NIXON, STEVEN G DEEKS, AND JOSEPH M MCCUNE. Composition and Function of T Cell Subpopulations Are Slow to Change Despite Effective Antiretroviral Treatment of HIV Disease. PloS one, 9(1):e85613, January 2014.
- [288] JOSEPH P CASAZZA, MICHAEL R BETTS, LOUIS J PICKER, AND RICHARD A KOUP. Decay Kinetics of Human Immunodeficiency Virus-Specific CD8 + T Cells in Peripheral Blood after Initiation of Highly Active Antiretroviral Therapy. Journal of virology, 75(14):6508-6516, 2001.
- [289] JANE DEAYTON, AMANDA MOCROFT, PAULINE WILSON, VIN-CENT C EMERY, MARGARET A JOHNSON, AND PAUL D GRIF-FITHS. Loss of cytomegalovirus (CMV) viraemia following highly active antiretroviral therapy in the absence of specific anti-CMV therapy. AIDS, 13:1203-1206, 1999.
- [290] S WORRELL, J DEAYTON, P HAYES, VC EMERY, F GOTCH, B GAZZARD, AND E-L LARSSON-SCIARD. Molecular correlates in AIDS patients following antiretroviral therapy: diversified Tcell receptor repertoires and in vivo control of cytomegalovirus replication. HIV Medicine, 2:11-19, 2001.
- [291] T. YOKOSUKA. Predominant Role of T Cell Receptor (TCR)-alpha Chain in Forming Preimmune

TCR Repertoire Revealed by Clonal TCR Reconstitution System. Journal of Experimental Medicine, 195(8):991–1001, April 2002.

- [292] XIAOLING LIANG, LUISE U WEIGAND, INGRID G SCHUSTER, ELFRIEDE EPPINGER, JUDITH C VAN DER GRIENDT, ANDREA SCHUB, MATTHIAS LEISEGANG, DANIEL SOMMERMEYER, FLO-RIAN ANDERL, YANYAN HAN, JOACHIM ELLWART, ANDREAS MOOSMANN, DIRK H BUSCH, WOLFGANG UCKERT, CHRISTIAN PESCHEL, AND ANGELA M KRACKHARDT. A single TCR alpha-chain with dominant peptide recognition in the allorestricted HER2/neu-specific T cell repertoire. Journal of immunology (Baltimore, Md. : 1950), 184(3):1617-29, February 2010.
- [293] MICHAEL R. BETTS, DAVID R. AMBROZAK, DANIEL C. DOUEK, SEBASTIAN BONHOEFFER, JASON M. BRENCHLEY, JOSEPH P. CASAZZA, RICHARD A. KOUP, AND LOUIS J. PICKER. Analysis of total human immunodeficiency virus (HIV)specific CD4+ and CD8+ T-cell responses: relationship to viral load in untreated HIV infection. Journal of virology, 75(24):11983-11991, 2001.
- [294] LAURA PAPAGNO, CELSA A SPINA, ARNAUD MARCHANT, MAR-IOLINA SALIO, NATHALIE RUFER, SUSAN LITTLE, TAO DONG, GILLIAN CHESNEY, ANELE WATERS, PHILIPPA EASTERBROOK, P ROD DUNBAR, DAWN SHEPHERD, VINCENZO CERUNDOLO, VINCENT EMERY, PAUL GRIFFITHS, CHRISTOPHER CONLON, ANDREW J MCMICHAEL, DOUGLAS D RICHMAN, SARAH L ROWLAND-JONES, AND VICTOR APPAY. Immune activation and CD8+ T-cell differentiation towards senescence in HIV-1 infection. PLoS biology, 2(2):E20, February 2004.
- [295] SONIA BASTIDAS, FREDERIK GRAW, MIRANDA Z SMITH, HER-BERT KUSTER, HULDRYCH F GÜNTHARD, AND ANNETTE OXE-NIUS. CD8+ T cells are activated in an antigenindependent manner in HIV-infected individuals. Journal of immunology (Baltimore, Md. : 1950), 192(4):1732-44, February 2014.
- [296] JASON M BRENCHLEY, DAVID A PRICE, TIMOTHY W SCHACKER, TEDI E ASHER, GUIDO SILVESTRI, SRINIVAS RAO, ZACHARY KAZZAZ, ETHAN BORNSTEIN, OLIVIER LAMBOTTE, DANIEL ALT-MANN, BRUCE R BLAZAR, BENIGNO RODRIGUEZ, LEIA TEIXEIRA-JOHNSON, ALAN LANDAY, JEFFREY N MARTIN, FREDERICK M HECHT, LOUIS J PICKER, MICHAEL M LEDERMAN, STEVEN G DEEKS, AND DANIEL C DOUEK. Microbial translocation is a cause of systemic immune activation in chronic HIV infection. Nature medicine, 12(12):1365– 71, December 2006.
- [297] SAURABH MEHANDRU, MICHAEL A POLES, KLARA TENNER-RACZ, PATRICK JEAN-PIERRE, VICTORIA MANUELLI, PETER LOPEZ, ANITA SHET, ANDREA LOW, HIROSHI MOHRI, DANIEL BODEN, PAUL RACZ, AND MARTIN MARKOWITZ. Lack of Mucosal Immune Reconstitution during Prolonged Treatment of Acute and Early HIV-1 Infection. PLoS Medicine, 3(12):2335–2348, 2006.
- [298] F MALDARELLI, X WU, L SU, F R SIMONETTI, W SHAO, S HILL, J SPINDLER, A L FERRIS, J W MELLORS, M F KEARNEY, J M COFFIN, AND S H HUGHES. HIV latency. Specific HIV integration sites are linked to clonal expansion and persistence of infected cells. *Science*, 345:179– 83, 2014.
- [299] LILLIANB. COHN, ISRAELT. SILVA, THIAGOY. OLIVEIRA, RAFAELA. ROSALES, ERICAH. PARRISH, GERALDH. LEARN, BEATRICEH. HAHN, JULIEL. CZARTOSKI, M.JULIANA MCEL-RATH, CLARA LEHMANN, FLORIAN KLEIN, MARINA CASKEY, BRUCED. WALKER, JANETD. SILICIANO, ROBERTF. SILICIANO, MILA JANKOVIC, AND MICHELC. NUSSENZWEIG. HIV-1 Integration Landscape during Latent and Active Infection. Cell, 160(3):420-432, 2015.

- [300] MATTHIAS EGGER, BERNARD HIRSCHEL, PATRICK FRANCIOLI, PHILIPPE SUDRE, MARC WIRZ, MARKUS FLEPP, MARTIN RICKEN-BACH, RAFFAELE MALINVERNI, PIETRO VERNAZZA, AND MANUEL BATTEGAY. Impact of new antiretroviral combination therapies in HIV infected patients in Switzerland: prospective multicentre study. BMJ, 315:1194-1199, 1997.
- [301] MATTHIAS EGGER, MARGARET MAY, GENEVIÈVE CHÊNE, AN-DREW N PHILLIPS, BRUNO LEDERGERBER, FRANÇOIS DABIS, DOMINIQUE COSTAGLIOLA, ANTONELLA D ARMINIO MONFORTE, FRANK DE WOLF, PETER REISS, JENS D LUNDGREN, AMY C JUSTICE, SCHLOMO STASZEWSKI, CATHERINE LEPORT, ROBERT S HOGG, CAROLINE A SABIN, M JOHN GILL, BERND SALZBERGER, AND JONATHAN A C STERNE. Prognosis of HIV-1-infected patients starting highly active antiretroviral therapy : a collaborative analysis of prospective studies. The Lancet, 360:119–129, 2002.
- [302] JOSEPH C MUDD AND MICHAEL M LEDERMAN. CD8 T cell persistence in treated HIV infection. Current opinion in HIV and AIDS, 9(5):500-5, September 2014.
- [303] VICTOR LEUNG, JENNIFER GILLS, JANET RABOUD, CURTIS COOPER, ROBERT S HOGG, MONA R LOUTFY, NIMA MACHOUF, JULIO S G MONTANER, SEAN B ROURKE, CHRIS TSOUKAS, AND MARINA B KLEIN. Predictors of CD4:CD8 ratio normalization and its effect on health outcomes in the era of combination antiretroviral therapy. *PloS one*, 8(10):e77665, January 2013.
- [304] EDWIN LEEANSYAH, ANUPAMA GANESH, F QUIGLEY, ANDERS SO, JAN ANDERSSON, PETER W HUNT, MA SOMSOUK, STEVEN G DEEKS, JEFFREY N MARTIN, MARKUS MOLL, BARBARA L SHACK-LETT, AND JOHAN K SANDBERG. Activation, exhaustion, and persistent decline of the antimicrobial MR1restricted MAIT-cell population in chronic HIV-1 infection. Blood, 121(7):1124-1135, 2013.
- [305] JOHANNA MARIA EBERHARD, PHILIP HARTJEN, SILKE KUMMER, REINHOLD E SCHMIDT, MAXIMILIAN BOCKHORN, CLARA LEHMANN, ASHWIN BALAGOPAL, JOACHIM HAUBER, JAN VAN LUNZEN, AND JULIAN SCHULZE ZUR WIESCH. CD161+ MAIT Cells Are Severely Reduced in Peripheral Blood and Lymph Nodes of HIV-Infected Individuals Independently of Disease Progression. PloS one, 9(11):e111323, January 2014.
- [306] O-JIN LEE, YOUNG-NAN CHO, SEUNG-JUNG KEE, MOON-JU KIM, HYE-MI JIN, SUNG-JI LEE, KI-JEONG PARK, TAE-JONG KIM, SHIN-SEOK LEE, YONG-SOO KWON, NACKSUNG KIM, MYUNG-GEUN SHIN, JONG-HEE SHIN, SOON-PAL SUH, DONG-WOOK RYANG, AND YONG-WOOK PARK. Circulating mucosalassociated invariant T cell levels and their cytokine levels in healthy adults. Experimental gerontology, 49:47-54, January 2014.
- [307] ALISON MOTSINGER, DAVID W HAAS, ALEKSANDAR K STANIC, LUC VAN KAER, SEBASTIAN JOYCE, AND DERYA UNUTMAZ. CD1d-restricted human natural killer T cells are highly susceptible to human immunodeficiency virus 1 infection. The Journal of experimental medicine, 195(7):869-79, April 2002.
- [308] NADINE G PAKKER, DAAN W NOTERMANS, ROB J DE BOER, MARIJKE T L ROOS, FRANK DE WOLF, ANDREW HILL, JOHN M LEONARD, SVEN A DANNER, FRANK MIEDEMA, AND PETER T A SCHELLEKENS. Biphasic kinetics of peripheral blood T cells after triple combination therapy in HIV-1 infection: a composite of redistribution and proliferation. Nature medicine, 4(2):208-214, 1998.
- [309] TAISHENG LI, NING WU, YI DAI, ZHIFENG QIU, YANG HAN, JING XIE, TING ZHU, AND YANLING LI. Reduced thymic output is a major mechanism of immune reconstitution failure in HIV-infected patients after long-term antiretroviral therapy. *Clinical infectious diseases :*

an official publication of the Infectious Diseases Society of America, **53**(9):944–51, November 2011.

- [310] INEKE DEN BRABER, TENDAI MUGWAGWA, NIENKE VRISEKOOP, LISET WESTERA, RAMONA MÖGLING, ANNE BREGJE DE BOER, NEELTJE WILLEMS, ELISE H R SCHRIJVER, GERRIT SPIEREN-BURG, KOOS GAISER, ERIK MUL, SIGRID A OTTO, AN F C RUITER, MARIETTE T ACKERMANS, FRANK MIEDEMA, JOSÉ A M BORGHANS, ROB J DE BOER, AND KIKI TESSELAAR. Maintenance of peripheral naive T cells is sustained by thymus output in mice but not humans. Immunity, 36(2):288-97, February 2012.
- [311] J.D. WATSON AND F.H.C. CRICK. Molecular structure of nucleic acids. Nature, 171(4356):709-756, 1953.
- [312] DORIS T. ZALLEN. Despite Franklin's work, Wilkins earned his Nobel. Nature, 425(September):15, 2003.
- [313] F. SANGER, G.G. BROWNLEE, AND B.G. BARRELL. A two-dimensional fractionation procedure for radioactive nucleotides. Journal of Molecular Biology, 13(2):373–IN4, September 1965.
- [314] GG BROWNLEE AND F SANGER. Nucleotide sequences from the low molecular weight ribosomal RNA of Escherichia coli. Journal of molecular biology, 23:337– 353, 1967.
- [315] JM ADAMS, PGN JEPPESEN, F SANGER, AND BG BAR-RELL. Nucleotide sequence from the coat protein cistron of R17 bacteriophage RNA. Nature, 228(September):1009-1014, 1969.
- [316] W MIN-JOU, G HAEGEMAN, M YSEBAERT, AND W FIERS. Nucleotide Sequence of the Gene Coding for the Bacteriophage MS2 Coat Protein. Nature, 237(May):82-88, 1972.
- [317] W FIERS, R CONTRERAS, F DUERINCK, G HAEGEMAN, D IS-ERENTANT, J MERREGAERT, W MIN-JOU, F MOLEMANS, A RAEYMAEKERS, A VAN DEN BERGHE, G VOLCKAERT, AND M YSEBAERT. Complete nucleotide sequence of bacteriophage MS2 RNA: primary and secondary structure of the replicase gene. Nature, 260:500– 507, 1976.
- [318] RAY WU. Nucleotide sequence analysis of DNA. Nature, 51:501-521, 1972.
- [319] R PADMANABHAN AND RAY WU. Nucleotide Sequence Analysis of DNA. Biochemical and biophysical research communications, 48(5):1295–1302, 1972.
- [320] F SANGER, J E DONELSON, A R COULSON, H KOSSEL, AND D FIS-CHER. Use of DNA polymerase I primed by a synthetic oligonucleotide to determine a nucleotide sequence in phage f1 DNA. Proceedings of the National Academy of Sciences USA, 70(4):1209-1213, 1973.
- [321] R PADMANABHAN, ERNEST JAY, AND R WU. Chemical Synthesis of a Primer and Its Use in the Sequence Analysis of the Lysozyme Gene of Bacteriophage T4. Proceedings of the National Academy of Sciences USA, 71(6):2510-2514, 1974.
- [322] F SANGER AND AR COULSON. A rapid method for determining sequences in DNA by primed synthesis with DNA polymerase. Journal of molecular biology, 94:441-448, 1975.
- [323] ALLAN M MAXAM AND WALTER GILBERT. A new method for sequencing DNA. Proceedings of the National Academy of Sciences USA, 74(2):560–564, 1977.
- [324] A. R. COULSON FRED SANGER, S NICKLEN. DNA sequencing with chain-terminating. Proceedings of the National Academy of Sciences, 74(12):5463–5467, 1977.

- [325] F. SANGER, G.M. AIR, B.G. BARRELL, N.L. BROWN, A.R. COULSON, J.C. FIDDES, C.A. HUTCHISON, P.M. SLOCOMBE, AND M. SMITH. Nucleotide sequence of bacteriophage phi X174 DNA. Nature, 265:687–695, 1977.
- [326] ZG CHIDGEAVADZE AND R. SH. BEABEALASHVILLI. 2', 3'-Dideoxy-3'aminonucleoside 5'-triphosphates are the terminators of DNA synthesis catalyzed by DNA polymerases. Nucleic acids research, 12(3):1671-1686, 1984.
- [327] LLOYD M SMITH, STEVEN FUNG, MICHAEL W HUNKAPILLER, TIM J HUNKAPILLER, AND LEROY E HOOD. The synthesis of oligonucleotides containing an aliphatic amino group at the 5' terminus synthesis of fluorescent DNA primers for use in DNA sequence analysis. Nucleic acids research, 13(7):2399-2412, 1985.
- [328] WILHELM ANSORGE, BRIAN S SPROAT, JOSEF STEGEMANN, AND CHRISTIAN SCHWAGER. A non-radioactive automated method for DNA sequence determination. Journal of Biochemical and Biophysical Methods, 13:315-323, 1986.
- [329] WILHELM ANSORGE, BRIAN SPROAT, JOSEF STEGEMANN, CHRIS-TIAN SCHWAGER, AND MARTIN ZENKE. Automated DNA sequencing: ultrasensitive detection of fluorescent bands during electrophoresis. Nucleic acids research, 15(11):4593-4602, 1987.
- [330] JAMES M PROBER, GEORGE L TRAINOR, RUDY J DAM, FRANK W HOBBS, CHARLES W ROBERTSON, ROBERT J ZAGURSKY, AN-THONY J COCUZZA, MARK A JENSEN, AND KIRK BAUMEISTER. DNA Sequencing with Rapid for System Fluorescent Chain-Terminating Dideoxynuclcotides. Science, 238(4825):336-341, 1987.
- [331] HIDEKI KAMBARA, TETSUO NISHIKAWA, YOSHIKO KATAYAMA, AND TOMOAKI YAMAGUCHI. Optimization of parameters in a DNA sequenator using fluorescence detection. Nature biotechnology, 6:816–820, 1988.
- [332] HAROLD SWERDLOW AND RAYMOND GESTELAND. Capillary gel electrophoresis for rapid, high resolution DNA sequencing. Nucleic acids research, 18(6):1415– 1419, 1990.
- [333] JA LUCKEY AND HOWARD DROSSMAN. High speed DNA sequencing by capillary electrophoresis. Nucleic acids research, 18(15):4417–4421, 1990.
- [334] T HUNKAPILLER, RJ KAISER, BF KOOP, AND L HOOD. Largescale and automated DNA sequence determination. Science, 254(5028):59–67, 1991.
- [335] R STADEN. A strategy of DNA sequencing employing computer programs. Nucleic acids research, 6(7):2601-2610, 1979.
- [336] STEPHEN ANDERSON. Shotgun DNA sequencing using cloned DNase I-generated fragments. Nucleic Acids Research, 9(13):3015–3027, 1981.
- [337] DAVID A JACKSON, ROBERT H SYMONST, AND PAUL BERG. Biochemical Method for Inserting New Genetic Information into DNA of Simian Virus 40: Circular SV40 DNA Molecules Containing Lambda Phage Genes and the Galactose Operon of Escherichia coli. Proceedings of the National Academy of Sciencess of the USA, 69(10):2904-2909, 1972.
- [338] STANLEY N COHEN, ANNIE C Y CHANG, HERBERT W BOYERT, AND ROBERT B HELLINGT. Construction of Biologically Functional Bacterial Plasmids In Vitro. Proceedings of the National Academy of Sciencess of the USA, 70(11):3240-3244, 1973.

- [339] H KLENOW AND I HENNINGSEN. Selective Elimination of the Exonuclease Activity of the Deoxyribonucleic Acid Polymerase from Escherichia coli B by Limited Proteolysis. Proceedings of the National Academy of Sciencess of the USA, 65(2):168-175, 1970.
- [340] CHENG-YAO CHEN. DNA polymerases drive DNA sequencing-by-synthesis technologies: both past and present. Frontiers in microbiology, 5(June):305, January 2014.
- [341] LLOYD M SMITH, JANE Z SANDERS, ROBERT J KAISER, PETER HUGHES, CHRISS DODD, CHARLES R CONNEL, CHERYL HEINER, STEPHEN B H KENT, AND LEROY E HOOD. Fluorescence detection in automated DNA sequence analysis. Nature, 321(June):674-679, 1986.
- [342] WILHELM J ANSORGE. Next-generation DNA sequencing techniques. New biotechnology, 25(4):195-203, April 2009.
- [343] ERIC S. LANDER, LAUREN M. LINTON, BRUCE BIRREN, CHAD NUSBAUM, MICHAEL C. ZODY, JENNIFER BALDWIN, KERI DE-VON, KEN DEWAR, MICHAEL DOYLE, WILLIAM FITZHUGH, ROEL FUNKE, DIANE GAGE, KATRINA HARRIS, ANDREW HEAFORD, JOHN HOWLAND, LISA KANN, JESSICA LEHOCZKY, ROSIE LEVINE, PAUL MCEWAN, KEVIN MCKERNAN, JAMES MELDRIM, JILL P. MESIROV, CHER MIRANDA, WILLIAM MORRIS, JEROME NAY-LOR, CHRISTINA RAYMOND, MARK ROSETTI, RALPH SANTOS, AN-DREW SHERIDAN, CARRIE SOUGNEZ, NICOLE STANGE-THOMANN. NIKOLA STOJANOVIC, ARAVIND SUBRAMANIAN, DUDLEY WYMAN, JANE ROGERS, JOHN SULSTON, RACHAEL AINSCOUGH, STEPHAN BECK, DAVID BENTLEY, JOHN BURTON, CHRISTOPHER CLEE, NIGEL CARTER, ALAN COULSON, REBECCA DEADMAN, PANOS Deloukas, Andrew Dunham, Ian Dunham, Richard Durbin, LISA FRENCH, DARREN GRAFHAM, SIMON GREGORY, TIM HUB-BARD, SEAN HUMPHRAY, ADRIENNE HUNT, MATTHEW JONES, CHRISTINE LLOYD, AMANDA MCMURRAY, LUCY MATTHEWS, SI-MON MERCER, SARAH MILNE, JAMES C. MULLIKIN, ANDREW MUNGALL, ROBERT PLUMB, MARK ROSS, RATNA SHOWNKEEN, SARAH SIMS, ROBERT H. WATERSTON, RICHARD K. WILSON, LADEANA W. HILLIER, JOHN D. MCPHERSON, MARCO A. MARRA, ELAINE R. MARDIS, LUCINDA A. FULTON, ASIF T. CHIN-WALLA, KYMBERLIE H. PEPIN, WARREN R. GISH, STEPHANIE L. CHISSOE, MICHAEL C. WENDL, KIM D. DELEHAUNTY, TRA-CIE L. MINER, ANDREW DELEHAUNTY, JASON B. KRAMER, LISA L. COOK, ROBERT S. FULTON, DOUGLAS L. JOHNSON, SAN-DRA W. CLIFTON, PATRICK J. MINX, TREVOR HAWKINS, EL-BERT BRANSCOMB, PAUL PREDKI, PAUL RICHARDSON, SARAH WENNING, TOM SLEZAK, NORMAN DOGGETT, JAN-FANG CHENG, ANNE OLSEN, SUSAN LUCAS, MARVIN FRAZIER, CHRISTOPHER Elkin, Edward Uberbacher, Richard A. Gibbs, Donna M. MUZNY, STEVEN E. SCHERER, JOHN B. BOUCK, ERICA J. SODER-GREN, KIM C. WORLEY, CATHERINE M. RIVES, JAMES H. GOR-RELL, MICHAEL L. METZKER, SUSAN L. NAYLOR, RAJU S. Kucherlapati, David L. Nelson, George M. Weinstock, Yoshiyuki Sakaki, Asao Fujiyama, Masahira Hattori, Tetsushi Yada, Todd Taylor, Atsushi Toyoda, Takehiko ITOH, CHIHARU KAWAGOE, HIDEMI WATANABE, YASUSHI TO-TOKI, JEAN WEISSENBACH, ROLAND HEILIG, WILLIAM SAURIN, PATRICK WINCKER, FRANCOIS ARTIGUENAVE, PHILIPPE BROT-TIER, THOMAS BRULS, ERIC PELLETIER, ANDREAS RUMP, CATHERINE ROBERT, ANDRÉ ROSENTHAL, MATTHIAS PLATZER, GERALD NYAKATURA, STEFAN TAUDIEN, JOANN DUBOIS, DOUglas R. Smith, Jun Gu, Lynn Doucette-Stamm, Marc RUBENFIELD, KEITH WEINSTOCK, SHIZEN QIN, HONG MEI LEE, HUANMING YANG, JUN YU, JIAN WANG, GUYANG HUANG, LEROY HOOD, LEE ROWEN, ANUP MADAN, THE INSTITUTE FOR SYSTEMS BIOLOGY:, RONALD W. DAVIS, NANCY A. FED-ERSPIEL, A. PIA ABOLA, MICHAEL J PROCTOR, BRUCE A. ROE, FENG CHEN, HUAQIN PAN, JULIANE RAMSER, HANS LEHRACH, RICHARD REINHARDT, W. RICHARD MCCOMBIE, MELISSA DE LA BASTIDE, GABRIELE NORDSIEK, HELMUT BLÖCKER, KLAUS HOR-NISCHER, RICHA AGARWALA, L. ARAVIND, JEFFREY A. BAIley, Alex Bateman, Serafim Batzoglou, Ewan Birney, PEER BORK, DANIEL G. BROWN, CHRISTOPHER B. BURGE, LORENZO CERUTTI, HSIU-CHUAN CHEN, DEANNA CHURCH,

MICHELE CLAMP, RICHARD R. COPLEY, TOBIAS DOERKS, SEAN R. EDDY, EVAN E. EICHLER, TERRENCE S. FUREY, JAMES GALAGAN, JAMES G. R. GILBERT, CYRUS HARMON, YOSHIHIDE HAYASHIZAKI, DAVID HAUSSLER, HENNING HERM-JAKOB, KARSTEN HOKAMP, WONHEE JANG, L. STEVEN JOHN-SON, THOMAS A. JONES, SIMON KASIF, AREK KASPRYZK, SCOT KENNEDY, W. JAMES KENT, PAUL KITTS, EUGENE V. KOONIN, IAN KORF, DAVID KULP, DORON LANCET, TODD M. LOWE, AOIFE MCLYSAGHT, TARJEI MIKKELSEN, JOHN V. MORAN, NICOLA MULDER, VICTOR J. POLLARA, CHRIS P. PONTING, GREG SCHULER, JÖRG SCHULTZ, GUY SLATER, ARIAN F. A. SMIT, ELIA STUPKA, JOSEPH SZUSTAKOWKI, DANIELLE THIERRY-MIEG, JEAN THIERRY-MIEG, LUKAS WAGNER, JOHN WALLIS, RAYMOND WHEELER, ALAN WILLIAMS, YURI I. WOLF, KEN-NETH H. WOLFE, SHIAW-PYNG YANG, RU-FANG YEH, FRANCIS Collins, Mark S. Guyer, Jane Peterson, Adam Felsen-FELD, KRIS A. WETTERSTRAND, RICHARD M. MYERS, JEREMY SCHMUTZ, MARK DICKSON, JANE GRIMWOOD, DAVID R. COX, MAYNARD V. OLSON, RAJINDER KAUL, CHRISTOPHER RAYMOND, NOBUYOSHI SHIMIZU, KAZUHIKO KAWASAKI, SHINSEI MINOSHIMA, GLEN A. EVANS, MARIA ATHANASIOU, ROGER SCHULTZ, AND ARISTIDES PATRINOS. Initial sequencing and analysis of the human genome. Nature, 409(February):860-921, 2001.

[344] J.C. VENTER, M.D. ADAMS, E.W. MYERS, P.W.LI, B. J. MURAL, G G SUTTON, H O SMITH, M YANDELL, C A EVANS, R A HOLT, J D GOCAYNE, P AMANATIDES, R M BALLEW, D H HUSON, J R WORTMAN, O ZHANG, C D KODIRA, X H ZHENG, L CHEN, M SKUPSKI, G SUBRAMANIAN, P D THOMAS, J ZHANG, G L GA-BOR MIKLOS, C NELSON, S BRODER, A G CLARK, J NADEAU, V A MCKUSICK, N ZINDER, A J LEVINE, R J ROBERTS, M SI-MON, C SLAYMAN, M HUNKAPILLER, R BOLANOS, A DELCHER, I DEW. D FASULO, M FLANIGAN, L FLOREA, A HALPERN, S HANNENHALLI, S KRAVITZ, S LEVY, C MOBARRY, K REIN-ERT, K REMINGTON, J ABU-THREIDEH, E BEASLEY, K BIDDICK, V BONAZZI, R BRANDON, M CARGILL, I CHANDRAMOULISWARAN, R CHARLAB, K CHATURVEDI, Z DENG, V DI FRANCESCO, P DUNN, K EILBECK, C EVANGELISTA, A E GABRIELIAN, W GAN, W GE, F GONG, Z GU, P GUAN, T J HEIMAN, M E HIG-GINS, R R JI, Z KE, K A KETCHUM, Z LAI, Y LEI, Z LI, J LI, Y LIANG, X LIN, F LU, G V MERKULOV, N MIL-SHINA, H M MOORE, A K NAIK, V A NARAYAN, B NEELAM, D NUSSKERN, D B RUSCH, S SALZBERG, W SHAO, B SHUE, J SUN, Z WANG, A WANG, X WANG, J WANG, M WEI, R WIDES, C XIAO, C YAN, A YAO, J YE, M ZHAN, W ZHANG, H ZHANG, Q ZHAO, L ZHENG, F ZHONG, W ZHONG, S ZHU, S ZHAO, D GILBERT, S BAUMHUETER, G SPIER, C CARTER, A CRAVCHIK, T WOODAGE, F ALI, H AN, A AWE, D BALDWIN, H BADEN, M BARNSTEAD, I BARROW, K BEESON, D BUSAM, A CARVER, A CENTER, M L CHENG, L CURRY, S DANAHER, L DAVENPORT, R DESILETS, S DIETZ, K DODSON, L DOUP, S FERRIERA, N GARG, A GLUECKSMANN, B HART, J HAYNES, C HAYNES, C HEINER, S HLADUN, D HOSTIN, J HOUCK, T HOW-LAND, C IBEGWAM, J JOHNSON, F KALUSH, L KLINE, S KO-DURU, A LOVE, F MANN, D MAY, S MCCAWLEY, T MCIN-TOSH, I MCMULLEN, M MOY, L MOY, B MURPHY, K NEL-SON, C PFANNKOCH, E PRATTS, V PURI, H QURESHI, M REAR-DON, R RODRIGUEZ, Y H ROGERS, D ROMBLAD, B RUHFEL, R Scott, C Sitter, M Smallwood, E Stewart, R Strong, E Suh, R Thomas, N N Tint, S Tse, C Vech, G Wang, J WETTER, S WILLIAMS, M WILLIAMS, S WINDSOR, E WINN-DEEN, K WOLFE, J ZAVERI, K ZAVERI, J F ABRIL, R GUIGÓ, M J CAMPBELL, K V SJOLANDER, B KARLAK, A KEJARIWAL, H MI, B LAZAREVA, T HATTON, A NARECHANIA, K DIEMER, A Muruganujan, N Guo, S Sato, V Bafna, S Istrail, R Lippert, R Schwartz, B Walenz, S Yooseph, D Allen, A BASU, J BAXENDALE, L BLICK, M CAMINHA, J CARNES-STINE, P CAULK, Y H CHIANG, M COYNE, C DAHLKE, A MAYS, M Dombroski, M Donnelly, D Ely, S Esparham, C Fosler, H GIRE, S GLANOWSKI, K GLASSER, A GLODEK, M GOROKHOV, K GRAHAM, B GROPMAN, M HARRIS, J HEIL, S HENDERSON, J HOOVER, D JENNINGS, C JORDAN, J JORDAN, J KASHA, L KAGAN, C KRAFT, A LEVITSKY, M LEWIS, X LIU, J LOPEZ, D MA, W MAJOROS, J MCDANIEL, S MURPHY, M NEWMAN, T NGUYEN, N NGUYEN, M NODELL, S PAN, J PECK, M PETER-SON, W ROWE, R SANDERS, J SCOTT, M SIMPSON, T SMITH,

A SPRAGUE, T STOCKWELL, R TURNER, E VENTER, M WANG, M WEN, D WU, M WU, A XIA, A ZANDIEH, AND X ZHU. **The sequence of the human genome.** Science, **291**(5507):1304-51, February 2001.

- [345] PÅ L NYRÉN AND ARNE LUNDIN. Enzymatic method for continuous monitoring of inorganic pyrophosphate synthesis. Analytical biochemistry, 509:504–509, 1985.
- [346] EDWARD DAVID HYMAN. A New Method for Sequencing DNA. Analytical biochemistry, 174:423-436, 1988.
- [347] PÅ L NYRÉN. Enzymatic method for continuous monitoring of DNA polymerase activity. Analytical biochemistry, 238:235-238, 1987.
- [348] MOSTAFA RONAGHI, SAMER KARAMOHAMED, BERTIL PETTERS-SON, MATHIAS UHLEN, AND PÅ L NYRÉN. Real-Time DNA Sequencing Using Detection of Pyrophosphate Release. Analytical biochemistry, 242:84–89, 1996.
- [349] MOSTAFA RONAGHI, MATHIAS UHLEN, AND PÅ L NYRÉN. A Sequencing Method Based on Real-Time Pyrophosphate. Science, 281(5375):363-365, July 1998.
- [350] MARCEL MARGULIES, MICHAEL EGHOLM, WE ALTMAN, AND SAID ATTIYA. Genome sequencing in microfabricated high-density picolitre reactors. Nature, 437(7057):376-380, 2005.
- [351] DAN S TAWFIK AND ANDREW D GRIFFITHS. Man-made celllike compartments for molecular evolution. Nature biotechnology, 16:652–656, 1998.
- [352] SAMUEL LEVY, GRANGER SUTTON, PAULINE C NG, LARS FEUK, AARON L HALPERN, BRIAN P WALENZ, NELSON AXELROD, JIAQI HUANG, EWEN F KIRKNESS, GENNADY DENISOV, YUAN LIN, JEF-FREY R MACDONALD, ANDY WING CHUN PANG, MARY SHAGO, TIMOTHY B STOCKWELL, ALEXIA TSIAMOURI, VINEET BAFNA, VIKAS BANSAL, SAUL A KRAVITZ, DANA A BUSAM, KAREN Y BEESON, TINA C MCINTOSH, KARIN A REMINGTON, JOSEP F ABRIL, JOHN GILL, JON BORMAN, YU-HUI ROGERS, MARVIN E FRAZIER, STEPHEN W SCHERER, ROBERT L STRAUSBERG, AND J CRAIG VENTER. The diploid genome sequence of an individual human. PLoS biology, 5(10):e254, September 2007.
- [353] DAVID A WHEELER, MAITHREYAN SRINIVASAN, MICHAEL EGHOLM, YUFENG SHEN, LEI CHEN, AMY MCGUIRE, WEN HE, YI-JU CHEN, VINOD MAKHIJANI, G THOMAS ROTH, XAVIER GOMES, KARRIE TARTARO, FAHEEM NIAZI, CYNTHIA L TUR-COTTE, GERARD P IRZYK, JAMES R LUPSKI, CRAIG CHINAULT, XING-ZHI SONG, YUE LIU, YE YUAN, LYNNE NAZARETH, XIANG QIN, DONNA M MUZNY, MARCEL MARGULIES, GEORGE M WEIN-STOCK, RICHARD A GIBBS, AND JONATHAN M ROTHERG. The complete genome of an individual by massively parallel DNA sequencing. Nature, 452(7189):872-6, April 2008.
- [354] KARL V VOELKERDING, SHALE A DAMES, AND JACOB D DURTSCHI. Next-generation sequencing: from basic research to diagnostics. *Clinical chemistry*, 55(4):641-58, April 2009.
- [355] JAY SHENDURE AND HANLEE JI. Next-generation DNA sequencing. Nature biotechnology, 26(10):1135-45, October 2008.
- [356] MILAN FEDURCO, ANTHONY ROMEU, SCOTT WILLIAMS, IS-ABELLE LAWRENCE, AND GERARDO TURCATTI. BTA, a novel reagent for DNA attachment on glass and efficient generation of solid-phase amplified DNA colonies. Nucleic acids research, 34(3):e22, January 2006.
- [357] DAVID R. BENTLEY, SHANKAR BALASUBRAMANIAN, HAROLD P. SWERDLOW, GEOFFREY P. SMITH, JOHN MILTON, CLIVE G. BROWN, KEVIN P. HALL, DIRK J. EVERS, COLIN L. BARNES, HELEN R. BIGNELL, JONATHAN M. BOUTELL, JASON BRYANT,

RICHARD J. CARTER, R. KEIRA CHEETHAM, ANTHONY J. COX, DARREN J. ELLIS, MICHAEL R. FLATBUSH, NIALL A. GORM-LEY, SEAN J. HUMPHRAY, LESLIE J. IRVING, MIRIAN S. KAR-BELASHVILI, SCOTTM. KIRK, HENG LI, XIAOHAI LIU, KLAUS S. MAISINGER, LISA J. MURRAY, BOJAN OBRADOVIC, TOBIAS OST, MICHAEL L. PARKINSON, MARK R. PRATT, ISABELLE M. J. RASOLONJATOVO, MARK T. REED, ROBERTO RIGATTI, CHIARA RODIGHIERO, MARK T. ROSS, ANDREA SABOT, SUB-RAMANIAN V. SANKAR, AYLWYN SCALLY, GARY P. SCHROTH, MARK E. SMITH, VINCENT P. SMITH, ANASTASSIA SPIRIDOU, PETA E. TORRANCE, SVILEN S. TZONEV, ERIC H. VERMAAS, KLAUDIA WALTER, XIAOLIN WU, LU ZHANG, MOHAMMED D. Alam, Carole Anastasi, Ify C. Aniebo, David M. D. Bai-LEY, IAIN R. BANCARZ, SAIBAL BANERJEE, SELENA G. BAR-BOUR, PRIMO A. BAYBAYAN, VINCENT A. BENOIT, KEVIN F. BENSON, CLAIRE BEVIS, PHILLIP J. BLACK, ASHA BOODHUN, JOE S. BRENNAN, JOHN A. BRIDGHAM, ROB C. BROWN, AN-DREWA. BROWN, DALE H. BUERMANN, ABASS A. BUNDU, JAMES C. BURROWS, NIGEL P. CARTER, NESTOR CASTILLO, MARIA CHIARA E. CATENAZZI, SIMON CHANG, R. NEIL COOLEY, NATASHA R. CRAKE, OLUBUNMI O. DADA, KONSTANTINOS D. DI-AKOUMAKOS, BELEN DOMINGUEZ-FERNANDEZ, DAVID J. EARN-SHAW, UGONNA C. EGBUJOR, DAVID W. ELMORE, SERGEY S. ETCHIN, MARK R. EWAN, MILAN FEDURCO, LOUISE J. FRASER, KARIN V. FUENTES FAJARDO, W. SCOTT FUREY, DAVID GEORGE, KIMBERLEY J. GIETZEN, COLIN P. GODDARD, GEORGE S. GOLDA, PHILIP A. GRANIERI, DAVID E. GREEN, DAVID L. GUSTAFSON, NANCY F. HANSEN, KEVIN HARNISH, CHRISTIAN D. HAUDENSCHILD, NARINDER I. HEYER, MATTHEW M. HIMS, JOHNNY T. HO, ADRIAN M. HORGAN, KATYA HOSCHLER, STEVE Hurwitz, Denis V. Ivanov, Maria Q. Johnson, Terena JAMES, T. A. HUW JONES, GYOUNG-DONG KANG, TZVETANA H. KERELSKA, ALAN D. KERSEY, IRINA KHREBTUKOVA, ALEX P. KINDWALL, ZOYA KINGSBURY, PAULA I. KOKKO-GONZALES, ANIL KUMAR, MARC A. LAURENT, CYNTHIA T. LAWLEY, SARAH E. LEE, XAVIER LEE, ARNOLD K. LIAO, JENNIFER A. LOCH, MITCH Lok, Shujun Luo, Radhika M, Mammen, John W, Mar-TIN, PATRICK G. MCCAULEY, PAUL MCNITT, PARUL MEHTA. KEITH W. MOON, JOE W. MULLENS, TAKSINA NEWINGTON, ZEMIN NING, BEE LING NG, SONIA M. NOVO, MICHAEL J. ONEILL, MARK A. OSBORNE, ANDREW OSNOWSKI, OMEAD OS-TADAN, LAMBROS L. PARASCHOS, LEA PICKERING, ANDREW C. PIKE, ALGER C. PIKE, D. CHRIS PINKARD, DANIEL P. PLISKIN, JOE PODHASKY, VICTOR J. QUIJANO, COMERACZY, VICKI H. RAE, STEPHEN R. RAWLINGS, ANA CHIVA RODRIGUEZ, PHYLL-IDA M. ROE, JOHN ROGERS, MARIA C. ROGERT BACIGALUPO, NIKOLAI ROMANOV, ANTHONY ROMIEU, RITHY K. ROTH, NA-TALIE J. ROURKE, SILKE T. RUEDIGER, ELI RUSMAN, RAQUEL M. SANCHES-KUIPER, MARTIN R. SCHENKER, JOSEFINA M.SEOANE, RICHARD J. SHAW, MITCH K. SHIVER, STEVENW.SHORT, NING L. SIZTO, JOHANNES P. SLUIS, MELANIE A. SMITH, JEAN Ernest Sohna Sohna, Eric J. Spence, Kim Stevens, Neil SUTTON, LUKASZ SZAJKOWSKI, CAROLYN L. TREGIDGO, GER-ARDO TURCATTI, STEPHANIE VANDEVONDELE, YULI VERHOVSKY, Selene M. Virk, Suzanne Wakelin, Gregory C. Walcott, JINGWENWANG, GRAHAM J. WORSLEY, JUYING YAN, LING YAU, MIKE ZUERLEIN, AND ROGERS{, JANE AND MULLIKIN, JAMES C. and Hurles, Matthew E. and McCooke{, Nick J. and West, John S. and Oaks, Frank L. and Lundberg, Peter L. AND KLENERMAN, DAVID AND DURBIN, RICHARD AND SMITH, AN-THONY J. Accurate whole human genome sequencing using reversible terminator chemistry. Nature, 456(6):53-9, November 2008.

- [358] GERARDO TURCATTI, ANTHONY ROMIEU, MILAN FEDURCO, AND ANA-PAULA TAIRI. A new class of cleavable fluorescent nucleotides: synthesis and optimization as reversible terminators for DNA sequencing by synthesis. Nucleic acids research, 36(4):e25, March 2008.
- [359] SHANKAR BALASUBRAMANIAN. Sequencing nucleic acids: from chemistry to medicine. Chemical Communications, 47(26):7281-7286, 2011.
- [360] MICHAEL A QUAIL, MIRIAM SMITH, PAUL COUPLAND, THOMAS D OTTO, SIMON R HARRIS, THOMAS R CONNOR, ANNA BERTONI,

HAROLD P SWERDLOW, AND YONG GU. A tale of three next generation sequencing platforms: comparison of Ion Torrent, Pacific Biosciences and Illumina MiSeq sequencers. *BMC genomics*, **13**(1):341, January 2012.

- [361] CHENGWEI LUO, DESPINA TSEMENTZI, NIKOS KYRPIDES, TIM-OTHY READ, AND KONSTANTINOS T KONSTANTINDIS. Direct comparisons of Illumina vs. Roche 454 sequencing technologies on the same microbial community DNA sample. *PloS one*, 7(2):e30087, January 2012.
- [362]KEVIN JUDD MCKERNAN, HEATHER E PECKHAM, GINA L COSTA, F MCLAUCHLIN YUTAO FU ERIC F TSUNG CHRISTOPHER B CLOUSER, CISYLA DUNCAN, JEFFREY K ICHIKAWA, CLARENCE C LEE, ZHENG ZHANG, SWATI S RANADE, T DIMALANTA, FIONA C Hyland, Tanya D Sokolsky, Lei Zhang, Andrew Sheri-DAN. HAONING FU. CYNTHIA L HENDRICKSON, BIN LI, LEV KOTLER, JEREMY R STUART, A MALEK, JONATHAN M MAN-NING, ALENA A ANTIPOVA, DAMON S PEREZ, P MOORE, KATH-LEEN C HAYASHIBARA, MICHAEL R LYONS, ROBERT E BEAUDOIN, E COLEMAN, MICHAEL W LAPTEWICZ, ADAM E SANNICANDRO, MICHAEL D RHODES, RAJESH K GOTTIMUKKALA, SHAN YANG, VINEET BAFNA, ALI BASHIR, ANDREW MACBRIDE, CAN ALKAN, JEFFREY M KIDD, EVAN E EICHLER, MARTIN G REESE, M FRAN-CISCO, DE LA VEGA, AND ALAN P BLANCHARD. Sequence and structural variation in a human genome uncovered by short-read, massively parallel ligation sequencing using two-base encoding. Genome Research, 19:1527-1541, 2009.
- [363] JAY SHENDURE, GREGORY J PORRECA, NIKOS B REPPAS, XIAOXIA LIN, JOHN P MCCUTCHEON, ABRAHAM M ROSEN-BAUM, MICHAEL D WANG, KUN ZHANG, ROBI D MITRA, AND GEORGE M CHURCH. Accurate multiplex polony sequencing of an evolved bacterial genome. Science, 309(5741):1728-32, September 2005.
- [364] H P J BUERMANS AND J T DEN DUNNEN. Next generation sequencing technology: Advances and applications. Biochimica et biophysica acta, 1842(10):1932– 1941, July 2014.
- [365] TRAVIS C GLENN. Field guide to next-generation DNA sequencers. Molecular ecology resources, 11(5):759-69, September 2011.
- [366] RADOJE DRMANAC, ANDREW B SPARKS, MATTHEW J CALLOW, AARON L HALPERN, NORMAN L BURNS, BAHRAM G KERMANI, PAOLO CARNEVALI, IGOR NAZARENKO, GEOFFREY B NILSEN, George Yeung, Fredrik Dahl, Andres Fernandez, Bryan STAKER, KRISHNA P PANT, JONATHAN BACCASH, ADAM P Borcherding, Anushka Brownley, Ryan Cedeno, Linsu CHEN, DAN CHERNIKOFF, ALEX CHEUNG, RAZVAN CHIRITA, BENJAMIN CURSON, JESSICA C EBERT, COLEEN R HACKER, ROBERT HARTLAGE, BRIAN HAUSER, STEVE HUANG, YUAN JIANG, VITALI KARPINCHYK, MARK KOENIG, CALVIN KONG, Tom Landers, Catherine Le, Jia Liu, Celeste E McBride, MATT MORENZONI, ROBERT E MOREY, KARL MUTCH, HE-LENA PERAZICH, KIMBERLY PERRY, BROCK A PETERS, JOE PE-TERSON, CHARIT L PETHIYAGODA, KALIPRASAD POTHURAJU, CLAUDIA RICHTER, ABRAHAM M ROSENBAUM, SHAUNAK ROY, JAY SHAFTO, ULADZISLAU SHARANHOVICH, KAREN W SHANNON, Conrad G Sheppy, Michel Sun, Joseph V Thakuria, Anne TRAN, DYLAN VU, ALEXANDER WAIT ZARANEK, XIAODI WU, SNEZANA DRMANAC, ARNOLD R OLIPHANT, WILLIAM C BANYAI, BRUCE MARTIN, DENNIS G BALLINGER, GEORGE M CHURCH, AND CLIFFORD A REID. Human genome sequencing using unchained base reads on self-assembling DNA nanoarrays. Science (New York, N.Y.), 327(5961):78-81, January 2010.
- [367] JONATHAN M ROTHBERG, WOLFGANG HINZ, TODD M REARICK, JONATHAN SCHULTZ, WILLIAM MILESKI, MEL DAVEY, JOHN H LEAMON, KIM JOHNSON, MARK J MILGREW, MATTHEW ED-WARDS, JEREMY HOON, JAN F SIMONS, DAVID MARRAN, JA-SON W MYERS, JOHN F DAVIDSON, ANNIKA BRANTING, JOHN R

NOBILE, BERNARD P PUC, DAVID LIGHT, TRAVIS A CLARK, MAR-TIN HUBER, JEFFREY T BRANCIFORTE, ISAAC B STONER, SI-MON E CAWLEY, MICHAEL LYONS, YUTAO FU, NILS HOMER, MARINA SEDOVA, XIN MIAO, BRIAN REED, JEFFREY SABINA, ERIKA FEIERSTEIN, MICHELLE SCHORN, MOHAMMAD ALANJARY, EILEEN DIMALANTA, DEVIN DRESSMAN, RACHEL KASINSKAS, TANYA SOKOLSKY, JACQUELINE A FIDANZA, EUGENI NAMSARAEV, KEVIN J MCKERNAN, ALAN WILLIAMS, G THOMAS ROTH, AND JAMES BUSTILLO. An integrated semiconductor device enabling non-optical genome sequencing. Nature, 475(7356):348–52, July 2011.

- [368] NICHOLAS J LOMAN, RAJU V MISRA, TIMOTHY J DALLMAN, CHRYSTALA CONSTANTINIDOU, SAHEER E GHARBIA, JOHN WAIN, AND MARK J PALLEN. Performance comparison of benchtop high-throughput sequencing platforms. Nature biotechnology, 30(5):434-9, May 2012.
- [369] LD STEIN. The case for cloud computing in genome informatics. Genome Biology, 11:207, 2010.
- [370] WILLIAM J GREENLEAF AND AREND SIDOW. The future of sequencing : convergence of intelligent design and market Darwinism. Genome biology, 15:303, 2014.
- [371] ILLUMINA. Illuminas Board Unanimously Rejects Roches Unsolicited Tender Offer as Inadequate. http://investor.illumina.com/phoenix.zhtml?c= 121127&p=irol-newsArticle&ID=1658034, 2012. Accessed: 2014-10-19.
- [372] ERIC E SCHADT, STEVE TURNER, AND ANDREW KASARSKIS. A window into third-generation sequencing. Human molecular genetics, 19(R2):R227-40, October 2010.
- [373] THOMAS P NIEDRINGHAUS, DENITSA MILANOVA, MATTHEW B KERBY, MICHAEL P SNYDER, AND ANNELISE E BARRON. Landscape of next-generation sequencing technologies. *Analytical chemistry*, 83(12):4327–4341, 2011.
- [374] CHANDRA SHEKHAR PAREEK, RAFAL SMOCZYNSKI, AND ANDRZEJ TRETYN. Sequencing technologies and genome sequencing. Journal of applied genetics, 52(4):413-35, November 2011.
- [375] IVO GLYNNE GUT. New sequencing technologies. Clinical & translational oncology, 15(11):879-81, November 2013.
- [376] IDO BRASLAVSKY, BENEDICT HEBERT, EMIL KARTALOV, AND STEPHEN R QUAKE. Sequence information can be obtained from single DNA molecules. Proceedings of the National Academy of Sciencess of the USA, 100(7):3960-3964, 2003.
- [377] TIMOTHY D HARRIS, PHILLIP R BUZBY, HAZEN BABCOCK, ERIC BEER, JAYSON BOWERS, IDO BRASLAVSKY, MARIE CAUSEY, JEN-NIFER COLONELL, JAMES DIMEO, J WILLIAM EFCAVITCH, EL-DAR GILADI, JAIME GILL, JOHN HEALY, MIRNA JAROSZ, DAN LAPEN, KEITH MOULTON, STEPHEN R QUAKE, KATHLEEN STEIN-MANN, EDWARD THAYER, ANASTASIA TYURINA, REBECCA WARD, HOWARD WEISS, AND ZHENG XIE. Single-molecule DNA sequencing of a viral genome. Science (New York, N.Y.), 320(5872):106-9, April 2008.
- [378] JAYSON BOWERS, JUDITH MITCHELL, ERIC BEER, PHILIP R BUZBY, MARIE CAUSEY, J WILLIAM EFCAVITCH, MIRNA JAROSZ, EDYTA KRZYMANSKA-OLEJNIK, LI KUNG, DORON LIPSON, GEOF-FREY M LOWMAN, SUBRAMANIAN MARAPPAN, PETER MCINER-NEY, ADAM PLATT, ATANU ROY, SUHAIB M SIDDIQI, KATHLEEN STEINMANN, AND JOHN F THOMPSON. Virtual terminator nucleotides for next-generation DNA sequencing. Nature methods, 6(8):593-5, August 2009.
- [379] GENOMEWEB. Helicos BioSciences Files for Chapter 11 Bankruptcy Protection, 2012. Accessed: 2014-10-20.

- [380] ERWIN L. VAN DIJK, HÉLÈNE AUGER, YAN JASZCZYSZYN, AND CLAUDE THERMES. Ten years of next-generation sequencing technology. Trends in Genetics, 30(9), August 2014.
- [381] M J LEVENE, J KORLACH, S W TURNER, M FOQUET, H G CRAIGHEAD, AND W W WEBB. Zero-mode waveguides for single-molecule analysis at high concentrations. Science, 299(5607):682-6, January 2003.
- [382] JOHN EID, ADRIAN FEHR, JEREMY GRAY, KHAI LUONG, JOHN LYLE, AND GEOFF OTTO. Real-time DNA sequencing from single polymerase molecules. *Science*, 323(January):133-138, 2009.
- [383] BENJAMIN A FLUSBERG, DALE R WEBSTER, JESSICA H LEE, KEVIN J TRAVERS, ERIC C OLIVARES, TYSON A CLARK, JONAS KORLACH, AND STEPHEN W TURNER. Direct detection of DNA methylation during single-molecule, realtime sequencing. Nature methods, 7(6):461-5, June 2010.
- [384] FARZIN HAQUE, JINGHONG LI, HAI-CHEN WU, XING-JIE LIANG, AND PEIZUAN GUO. Solid-State and Biological Nanopore for Real-Time Sensing of Single Chemical and Sequencing of DNA. Nano today, 8(1):56-74, 2013.
- [385] JOHN J KASIANOWICZ, ERIC BRANDIN, DANIEL BRANTON, AND DAVID W DEAMER. Characterization of individual polynucleotide molecules using a membrane channel. Proceedings of the National Academy of Sciencess of the USA, 93(November):13770-13773, 1996.
- [386] JIALI LI, DEREK STEIN, CIARAN MCMULLAN, AND DANIEL BRAN-TON. Ion-beam sculpting at nanometre length scales. Nature, 412(July):2-5, 2001.
- [387] C DEKKER. Solid-state nanopores. Nature nanotechnology, 2:209-215, 2007.
- [388] JAMES CLARKE, HC WU, LAKMAL JAYASINGHE, AND ALPESH PATEL. Continuous base identification for singlemolecule nanopore DNA sequencing. Nature nanotechnology, 4(April):265-270, 2009.
- [389] MICHAEL EISENSTEIN. Oxford Nanopore announcement sets sequencing sector abuzz. Nature biotechnology, 30(4):295-6, April 2012.
- [390] N. J. LOMAN AND A. R. QUINLAN. Poretools: a toolkit for analyzing nanopore sequence data. *Bioinformatics*, pages 1–3, August 2014.
- [391] DANIEL BRANTON, DAVID W DEAMER, ANDRE MARZIALI, HA-GAN BAYLEY, STEVEN A BENNER, THOMAS BUTLER, MASSIMIL-IANO DI VENTRA, SLAVEN GARAJ, ANDREW HIBBS, STEVAN B JOVANOVICH, PREDRAG S KRSTIC, STUART LINDSAY, XINSHENG SEAN, ROBERT RIEHN, GAUTAM V SONI, VINCENT TABARD-COSSA, AND MENI WANUNU. The potential and challenges of nanopore sequencing. Nature biotechnology, 26(10):1146-1153, 2008.
- [392] F PAILLARD, G STERKERS, AND C VAQUERO. Transcriptional and post-transcriptional regulation of TcR, CD4 and CD8 gene expression during activation of normal human T lymphocytes. The EMBO journal, 9(6):1867-72, June 1990.
- [393] FLORENCE PAILLARD, GHYSLAINE STERKERS, AND CATHERINE VAQUERO. Correlation between up-regulation of lymphokine mRNA dn down-regulation of TCR, CD4, CD8 and lack mRNA as shwon by the effect of CsA on activated T lymphocytes. Biochemical and biophysical research communications, 186(2):603-611, 1992.

- [394] MICHAEL A FROHMAN, MICHAEL K DUSH, AND GAIL R MARTIN. Rapid production of full-length cDNAs from rare transcripts: Amplification using a single genespecific oligonucleotide primer. Proceedings of the National Academy of Sciences of the United States of America, 85(December):8998-9002, 1988.
- [395] MT SUZUKI AND SJ GIOVANNONI. Bias caused by template annealing in the amplification of mixtures of 16S rRNA genes by PCR. Applied and environmental microbiology, 62(2):625-630, 1996.
- [396] MARTIN F POLZ AND COLLEEN M CAVANAUGH. Bias in Template-to-Product Ratios in Multitemplate PCR Bias in Template-to-Product Ratios in Multitemplate PCR. Applied and environmental microbiology, 64(10):3724-3730, 1998.
- [397] LISA FÖHSE, JANINE SUFFNER, KARSTEN SUHRE, BENJAMIN WAHL, CORNELIA LINDNER, CHUN-WEI LEE, SUSANNE SCHMITZ, JAN D HAAS, STELLA LAMPRECHT, CHRISTIAN KOENECKE, ANDRÉ BLEICH, GÜNTER J HÄMMERLING, BERNARD MALISSEN, SEBASTIAN SUERBAUM, REINHOLD FÖRSTER, AND IMMO PRINZ. High TCR Diversity ensures optimal Function and Homeostasis of FoxP3(+) Regulatory T cells. European journal of immunology, 41(11):1-37, September 2011.
- [398] HARLAN S ROBINS, CINDY DESMARAIS, JESSICA MATTHIS, ROBERT LIVINGSTON, JESSICA ANDRIESEN, HELENA REIJONEN, CHRISTOPHER CARLSON, GEROLD NEPOM, CASSIAN YEE, AND KAREN CEROSALETTI. Ultra-sensitive detection of rare T cell clones. Journal of Immunological Methods, 375(1-2):14-19, January 2012.
- [399] CHRISTOPHER S CARLSON, RYAN O EMERSON, ANNA M SHER-WOOD, CINDY DESMARAIS, MOON-WOOK CHUNG, JOSEPH M PARSONS, MICHELLE S STEEN, MARISSA A LAMADRID-HERRMANNSFELDT, DAVID W WILLIAMSON, ROBERT J LIV-INGSTON, DAVID WU, BRENT L WOOD, MARK J RIEDER, AND HARLAN ROBINS. Using synthetic templates to design an unbiased multiplex PCR assay. Nature communications, 4:2680, October 2013.
- [400] SHUO LI, MARIE-PAULE LEFRANC, JOHN J MILES, ELTAF ALAM-YAR, VÉRONIQUE GIUDICELLI, PATRICE DUROUX, J DOUGLAS FREEMAN, VINCENT D A CORBIN, JEAN-PIERRE SCHEERLINCK, MICHAEL A FROHMAN, PAUL U CAMERON, MAGDALENA PLEBAN-SKI, BRUCE LOVELAND, SCOTT R BURROWS, ANTHONY T PA-PENFUSS, AND ERIC J GOWANS. IMGT/HighV QUEST paradigm for T cell receptor IMGT clonotype diversity and next generation repertoire immunoprofiling. Nature communications, 4(May):2333, September 2013.
- [401] W M SCHMIDT AND M W MUELLER. CapSelect: a highly sensitive method for 5' CAP-dependent enrichment of full-length cDNA in PCR-mediated analysis of mRNAs. Nucleic acids research, 27(21):e31, November 1999.
- [402] PAWEL ZAJAC, SAIFUL ISLAM, HANNAH HOCHGERNER, PETER LÖNNERBERG, AND STEN LINNARSSON. Base preferences in non-templated nucleotide incorporation by MMLV-derived reverse transcriptases. *PloS one*, 8(12):e85270, January 2013.
- [403] ELI GILBOA, SUDHA W MITRA, STEPHEN GOFF, AND DAVID BALTIMORE. A Detailed Model of Reverse Transcription of Crucial Aspects and Tests. Cell, 18(September):93-100, 1979.
- [404] D G PETERS, A B KASSAM, H YONAS, E H O'HARE, R E FER-RELL, AND A M BRUFSKY. Comprehensive transcript analysis in small quantities of mRNA by SAGElite. Nucleic acids research, 27(24):e39, December 1999.

- [405] PHUONG NGUYEN, JING MA, DEQING PEI, CAROLINE OBERT, CHENG CHENG, AND TERRENCE L GEIGER. Identification of errors introduced during high throughput sequencing of the T cell receptor repertoire. BMC genomics, 12:106, January 2011.
- [406] B. EWING, L. HILLIER, M. C. WENDL, AND P. GREEN. Base-Calling of Automated Sequencer Traces Using Phred. I. Accuracy Assessment. Genome Research, 8(3):175-185, March 1998.
- [407] BRENT EWING AND PHIL GREEN. Base-calling of Automated Sequencer Traces Using Phred. II. error probabilities. Genome research, 8(3):186, 1998.
- [408] T H RABBITTS, M P LEFRANC, M A STINSON, J E SIMS, J SCHRODER, M STEINMETZ, N L SPURR, E SOLOMON, AND P N GOODFELLOW. The chromosomal location of Tcell receptor genes and a T cell rearranging gene: possible correlation with specific translocations in human T cell leukaemia. The EMBO journal, 4(6):1461-5, June 1985.
- [409] J. E. SIMS, A. TUNNACLIFFE, W. J. SMITH, AND T. H. RAB-BITTS. Complexity of human T-cell antigen receptor beta-chain constant- and variable-region genes. *Nature*, 312:515–545, 1984.
- [410] U SCHNEIDER, H U SCHWENK, AND G BORNKAMM. Characterization of EBV-genome negative "null" and "T" cell lines derived from children with acute lymphoblastic leukemia and leukemic transformed non-Hodgkin lymphoma. International journal of cancer. Journal international du cancer, 19(5):621-6, May 1977.
- [411] ROBERT T ABRAHAM AND ARTHUR WEISS. Jurkat T cells and development of the T-cell receptor signalling paradigm. Nature reviews. Immunology, 4(4):301-8, April 2004.
- [412] H ABTS, M EMMERICH, S MILTENYI, A RADBRUCH, AND H TESCH. CD20 positive human B lymphocytes separated with the magnetic cell sorter (MACS) can be induced to proliferation and antibody secretion in vitro. Journal of immunological methods, 125(1-2):19-28, December 1989.
- [413] S MILTENYI, W MÜLLER, W WEICHEL, AND A RADBRUCH. High gradient magnetic cell separation with MACS. Cytometry, 11(2):231–238, 1990.
- [414] XI YANG, DI LIU, FEI LIU, JUN WU, JING ZOU, XUE XIAO, FANGQING ZHAO, AND BAOLI ZHU. HTQC: a fast quality control toolkit for Illumina sequencing data. BMC bioinformatics, 14(1):33, January 2013.
- [415] YANIV ERLICH, PARTHA P MITRA, W RICHARD MCCOMBIE, AND GREGORY J HANNON. Alta-Cyclic : a self- optimizing base caller for next-generation sequencing. Nature Methods, 5(8):679-682, 2008.
- [416] CHRISTIAN LEDERGERBER AND CHRISTOPHE DESSIMOZ. Basecalling for next-generation sequencing platforms. Briefings in bioinformatics, 12(5):489-97, September 2011.
- [417] N R LANDAU, T P ST JOHN, I L WEISSMAN, S C WOLF, A E SILVERSTONE, AND D BALTIMORE. Cloning of terminal transferase cDNA by antibody screening. Proceedings of the National Academy of Sciences of the United States of America, 81(18):5836-40, September 1984.
- [418] YAN WANG, DENNIS E PROSEN, LI MEI, JOHN C SULLIVAN, MICHAEL FINNEY, AND PETER B VANDER HORN. A novel strategy to engineer DNA polymerases for enhanced processivity and improved performance in vitro. Nucleic acids research, 32(3):1197–207, January 2004.

- [419] PETER MCINERNEY, PAUL ADAMS, AND MASOOD Z HADI. Error Rate Comparison during Polymerase Chain Reaction by DNA Polymerase. Molecular biology international, 2014:287430, January 2014.
- [420] ARON FAZEKAS, ROYCE STEEVES, AND STEVEN NEWMASTER. Improving sequencing quality from PCR products containing long mononucleotide repeats. *BioTechniques*, 48(4):277–85, April 2010.
- [421] E VIGUERA, D CANCEILL, AND S D EHRLICH. Replication slippage involves DNA polymerase pausing and dissociation. The EMBO journal, 20(10):2587-95, May 2001.
- [422] ROCHE. Sequencing Solutions Technical Note. http://roche-biochem.jp/prima/pdf/TechNote\_ WhatisFlowPatternB\_August2013.pdf, 2013.
- [423] SUSAN M HUSE, JULIE A HUBER, HILARY G MORRISON, MITCHELL L SOGIN, AND DAVID MARK WELCH. Accuracy and quality of massively parallel DNA pyrosequencing. Genome biology, 8(7):R143, January 2007.
- [424] ILLUMINA. Illumina Announces MiSeq(TM) Personal Sequencing System, 2011.
- [425] ILLUMINA. Off-Line Basecaller Software v1.6 User Guide Rev, 2009.
- [426] FELIX KRUEGER, SIMON R ANDREWS, AND CAMERON S OS-BORNE. Large scale loss of data in low-diversity illumina sequencing libraries can be recovered by deferred cluster calling. *PloS one*, 6(1):e16607, January 2011.
- [427] S P BALK, E C EBERT, R L BLUMENTHAL, F V MCDERMOTT, K W WUCHERPFENNIG, S B LANDAU, AND R S BLUMBERG. Oligoclonal expansion and CD1 recognition by human intestinal intraepithelial lymphocytes. Science (New York, N.Y.), 253(5026):1411-5, September 1991.
- [428] SÉBASTIEN WÄLCHLI, GEIR Å GE LØ SET, SHRADDHA KUMARI, JORUNN NERGÅ RD JOHANSEN, WEIWEN YANG, INGER SANDLIE, AND JOHANNA OLWEUS. A practical approach to Tcell receptor cloning and expression. PloS one, 6(11):e27930, January 2011.
- [429] GENOME WEB. Roche Shutting Down 454 Sequencing Business. http://www.genomeweb.com/sequencing/ roche-shutting-down-454-sequencing-business, 2013. Accessed: 2014-09-19.
- [430] BRIAN C. SCHAEFER. Revolutions in rapid amplification of cDNA ends: new strategies for polymerase chain reaction cloning of full-length cDNA ends. Analytical biochemistry, 277:255–273, 1995.
- [431] K MARUYAMA AND S SUGANO. Oligo-capping: a simple method to replace the cap structure of eukaryotic mRNAs with oligoribonucleotides. Gene, 138(1-2):171-4, January 1994.
- [432] J B EDWARDS, J DELORT, AND J MALLET. Oligodeoxyribonucleotide ligation to single-stranded cDNAs: a new tool for cloning 5' ends of mRNAs and for constructing cDNA libraries by in vitro amplification. Nucleic acids research, 19(19):5227-32, October 1991.
- [433] ANTHONY B TROUTT, MICHAEL G MCHEYZER-WILLIAMS, BALI PULENDRAN, AND G. J. V. NOSSAL. Ligation-anchored PCR: a simple amplification technique with single-sided specificity. Proceedings of the National Academy of Sciencess of the USA, 89(October):9823-9825, 1992.

- [434] ILLUMINA. Low-Diversity Sequencing on the Illumina. http://www.illumina.com/documents/products/ technotes/technote\_low\_diversity\_rta.pdf, 2013.
- [435] GLENN K FU, JING HU, PEI-HUA WANG, AND STEPHEN P A FODOR. Counting individual DNA molecules by the stochastic attachment of diverse labels. Proceedings of the National Academy of Sciences of the United States of America, 108(22):9026-31, May 2011.
- [436] JAMES A CASBON, ROBERT J OSBORNE, SYDNEY BRENNER, AND CORRAD P LICHTENSTEIN. A method for counting PCR template molecules with application to next-generation sequencing. Nucleic acids research, 39(12):e81, July 2011.
- [437] TEEMU KIVIOJA, ANNA VÄHÄRAUTIO, KASPER KARLSSON, MAR-TIN BONKE, MARTIN ENGE, STEN LINNARSSON, AND JUSSI TAIPALE. Counting absolute numbers of molecules using unique molecular identifier. Nature Methods, 9(1):1-5, 2012.
- [438] JOSHUA A. WEINSTEIN, NING JIANG, RICHARD A. WHITE, DANIEL S. FISHER, AND STEPHEN R. QUAKE. Highthroughput sequencing of the zebrafish antibody repertoire. Science (New York, N.Y.), 324(5928):807– 10, May 2009.
- [439] AP BIRD. DNA methylation and the frequency of CpG in animal DNA. Nucleic Acids Research, 8(7):1499-1504, 1980.
- [440] BRADLEY HALL, JOHN M MICHELETTI, POOJA SATYA, KRYSTAL OGLE, JACK POLLARD, AND ANDREW D ELLINGTON. Design, synthesis, and amplification of DNA pools for in vitro selection. Current protocols in molecular biology / edited by Frederick M. Ausubel ... [et al.], Chapter 24(October):Unit 24.2, October 2009.
- [441] KASPER D HANSEN, STEVEN E BRENNER, AND SANDRINE DU-DOIT. Biases in Illumina transcriptome sequencing caused by random hexamer priming. Nucleic acids research, 38(12):e131, July 2010.
- [442] PETER C FITZGERALD, ANDREY SHLYAKHTENKO, ALAIN A MIR, AND CHARLES VINSON. Clustering of DNA sequences in human promoters. Genome research, 14:1562–1574, 2004.
- [443] VLADIMIR B BAJIC, SIN LAM TAN, ALAN CHRISTOFFELS, CHRISTIAN SCHÖNBACH, LEONARD LIPOVICH, LIANG YANG, OLIVER HOFMANN, ADELE KRUGER, WINSTON HIDE, CHIKATOSHI KAI, JUN KAWAI, DAVID A HUME, PIERO CARNINCI, AND YOSHIHIDE HAYASHIZAKI. Mice and men: their promoter properties. *PLoS genetics*, 2(4):e54, April 2006.
- [444] SIMON J VAN HEERINGEN, WASEEM AKHTAR, ULRIKE G JACOBI, ROBERT C AKKERS, YUTAKA SUZUKI, AND GERT JAN C VEEN-STRA. Nucleotide composition-linked divergence of vertebrate core promoter architecture. Genome research, 21:410-421, 2011.
- [445] FANGLEI ZHUANG, RYAN T FUCHS, ZHIYI SUN, YU ZHENG, AND G BRETT ROBB. Structural bias in T4 RNA ligasemediated 3'-adapter ligation. Nucleic acids research, 40(7):e54, April 2012.
- [446] M P LEFRANC, V GIUDICELLI, C BUSIN, J BODMER, W MÜLLER, R BONTROP, M LEMAITRE, A MALIK, AND D CHAUME. IMGT, the International ImMunoGeneTics database. Nucleic acids research, 26(1):297–303, January 1998.
- [447] ALTAF ALAMYAR, VÉRONIQUE GIUDICELLI, SHUO LI, PATRICE DUROUX, AND MARIE-PAULE LEFRANC. IMGT/HIGHV-QUEST: THE IMGT WEB PORTAL FOR IM-MUNOGLOBULIN (IG) OR ANTIBODY AND T CELL RECEPTOR (TR) ANALYSIS FROM NGS HIGH THROUGHPUT AND DEEP SE-QUENCING. Immunome Research, 8(1):2, May 2012.

- [448] W. JAMES KENT. BLAT The BLAST-Like Alignment Tool. Genome Research, 12(4):656–664, March 2002.
- [449] STEPHEN F ALTSCHUP, WARREN GISH, WEBB MILLER, EU-GENE W. MYERS, AND DAVID J. LIPMAN. Basic Local Alignment Search Tool. Journal of Molecular Biology, 215:403-410, 1990.
- [450] M.S. WATERMAN, T.F. SMITH, AND W.A. BEYER. Some Biological Sequence Metrics \*. Advances in Mathematics, 20:367–387, 1976.
- [451] ALFRED V. AHO AND MARGARET J. CORASICK. Efficient string matching: an aid to bibliographic search. Communications of the ACM, 18(6):333-340, June 1975.
- [452] SAIMA HASIB, MAHAK MOTWANI, AND AMIT SAXENA. Importance of Aho-Corasick String Matching Algorithm in Real World Applications. International Journal of Computer Science and Information Technoloaies, 4(3):467-469, 2013.
- [453] VÉRONIQUE GIUDICELLI AND MARIE-PAULE LEFRANC. Ontology for immunogenetics: the IMGT-ONTOLOGY. Bioinformatics, 15(12):1047–1054, 1999.
- [454] DMITRIY A BOLOTIN, MIKHAIL SHUGAY, ILGAR Z MAMEDOV, EKATERINA V PUTINTSEVA, MARIA A TURCHANINOVA, IVAN V ZVYAGIN, OLGA V BRITANOVA, AND DMITRIY M CHUDAKOV. MITCR: software for T-cell receptor sequencing data analysis. Nature methods, (8), July 2013.
- [455] STEPHANIE GRAS, ZHENJUN CHEN, JOHN J MILES, YU CHIH LIU, MELISSA J BELL, LUCY C SULLIVAN, LARS KJER-NIELSEN, REBEKAH M BRENNAN, JACQUELINE M BURROWS, MICHELLE A NELLER, RAJIV KHANNA, ANTHONY W PURCELL, ANDREW G BROOKS, JAMES MCCLUSKEY, JAMIE ROSSJOHN, AND SCOTT R BURROWS. Allelic polymorphism in the T cell receptor and its impact on immune responses. The Journal of experimental medicine, 207(7):1555-67, July 2010.
- [456] REBEKAH M BRENNAN, JACQUELINE M BURROWS, THOMAS EL-LIOTT, MICHELLE A NELLER, STEPHANIE GRAS, JAMIE ROSSJOHN, JOHN J MILES, AND SCOTT R BURROWS. Missense single nucleotide polymorphisms in the human T cell receptor loci control variable gene usage in the T cell repertoire. British journal of haematology, pages 1-4, March 2014.
- [457] WENJING PAN, MIRANDA BYRNE-STEELE, CHUNLIN WANG, STAN-LEY LU, SCOTT CLEMMONS, ROBERT J ZAHORCHAK, AND JIAN HAN. DNA polymerase preference determines PCR priming efficiency. BMC biotechnology, 14(1):10, January 2014.
- [458] KATSUYUKI SHIROGUCHI, TONY Z JIA, PETER A SIMS, AND X SUNNEY XIE. Digital RNA sequencing minimizes sequence-dependent bias and amplification noise with optimized single-molecule barcodes. Proceedings of the National Academy of Sciences of the United States of America, 109(4):1347-52, January 2012.
- [459] ILGAR Z. MAMEDOV, OLGA V. BRITANOVA, IVAN V. ZVYAGIN, MARIA A. TURCHANINOVA, DMITRIY A. BOLOTIN, EKATERINA V. PUTINTSEVA, YURIY B. LEBEDEV, AND DMITRIY M. CHUDAKOV. Preparing Unbiased T-Cell Receptor and Antibody cDNA Libraries for the Deep Next Generation Sequencing Profiling. Frontiers in Immunology, 4(December):1-10, 2013.
- [460] MIKHAIL SHUGAY, OLGA V BRITANOVA, EKATERINA M MER-ZLYAK, MARIA A TURCHANINOVA, ILGAR Z MAMEDOV, TIMUR R TUGANBAEV, DMITRIY A BOLOTIN, DMITRY B STAROVEROV,

EKATERINA V PUTINTSEVA, KARLA PLEVOVA, CARSTEN LIN-NEMANN, DMITRIY SHAGIN, SARKA POSPISILOVA, SERGEY LUKYANOV, TON N SCHUMACHER, AND DMITRIY M CHUDAKOV. **Towards error-free profiling of immune repertoires**. *Nature methods*, (April):1–5, May 2014.

- [461] TAKAHIRO KANAGAWA. Bias and artifacts in multitemplate polymerase chain reactions (PCR). Journal of Bioscience and Bioengineering, 96(4):317–323, 2003.
- [462] XIAOYUN QIU, LIYOU WU, HESHU HUANG, PATRICK E MC-DONEL, ANTHONY V PALUMBO, JAMES M TIEDJE, AND JIZHONG ZHOU. Evaluation of PCR-generated chimeras, mutations, and heteroduplexes with 16S rRNA genebased cloning. Applied and environmental microbiology, 67(2):880-887, 2001.
- [463] E PIENAAR, M THERON, M NELSON, AND HJ VILJOEN. A quantitative model of error accumulation during PCR amplification. Computational biology and chemistry, 30(2):102-111, 2006.
- [464] immunoSEQ. http://www.immunoseq.com/. Accessed: 2014-10-16.
- [465] iRepertoire. http://www.irepertoire.com. Accessed: 2014-10-16.
- [466] CHARLES A JANEWAY. Approaching the Asymptote ? Evolution and Revolution in Immunology. Cold Spring Harbor Symposia on Quantitative Biology, 45:1– 13, 1989.
- [467] MARIELLE C GOLD, STEFANIA CERRI, SUSAN SMYK-PEARSON, MEGHAN E CANSLER, TODD M VOGT, JACOB DELEPINE, ERV-INA WINATA, GWENDOLYN M SWARBRICK, WEI-JEN CHUA, YIK Y L YU, OLIVIER LANTZ, MATTHEW S COOK, MEGAN D NULL, DAVID B JACOBY, MELANIE J HARRIFF, DEBORAH A LEWIN-SOHN, TED H HANSEN, AND DAVID M LEWINSOHN. Human mucosal associated invariant T cells detect bacterially infected cells. *PLoS biology*, 8(6):e1000407, January 2010.
- [468] BY MARK EXLEY, JORGE GARCIA, STEVEN P BALK, AND STEVEN PORCELLI. Requirements for CD1d Recognition by Human Invariant Valpha24+ CD4-CD8- T Cells. Journal of experimental medicine, 186(1):109–120, 1997.
- [469] M HAN, L HARRISON, P KEHN, K STEVENSON, J CURRIER, AND M A ROBINSON. Invariant or highly conserved TCR alpha are expressed on double-negative (CD3+CD4-CD8-) and CD8+ T cells. Journal of immunology (Baltimore, Md. : 1950), 163(1):301-11, July 1999.
- [470] JOSEPH P SANDERSON, KATHRIN WALDBURGER-HAURI, DIANA GARZÓN, GEDIMINAS MATULIS, SALAH MANSOUR, NICHOLAS J PUMPHREY, NIKOLAI LISSIN, PETER M VILLIGER, BENT JAKOB-SEN, JOSÉ D FARALDO-GÓMEZ, AND STEPHAN D GADOLA. Natural variations at position 93 of the invariant  $V\alpha 24$ -J $\alpha 18 \alpha$  chain of human iNKT-cell TCRs strongly impact on CD1d binding. European journal of immunology, 42(1):248–55, January 2012.
- [471] GIUSEPPE PANTALEO, JAMES F. DEMAREST, HUGO SOUDEYNS, CECILIA GRAZIOSI, FRANCOIS DENIS, JOSEPH W. ADELSBERGER, PERSEPHONE BORROW, MICHAEL S. SAAG, GEORGE M. SHAW, RAFICK P. SEKALY, AND ANTHONY S. FAUCI. Major expansion of CD8+ T cells with a predominant Vbeta usage during the primary immune response to HIV. Nature, 370:463-467, 1994.
- [472] W KOLOWOS, M SCHMITT, M HERRMAN, E HARRER, P LÖW, J R KALDEN, AND T HARRER. Biased TCR repertoire in HIV-1-infected patients due to clonal expansion of HIV-1-reverse transcriptase-specific CTL clones. Journal of Immunology, 162(12):7525-33, June 1999.

- [473] Z C KOU, J S PUHR, M ROJAS, W T MCCORMACK, M M GOODENOW, AND J W SLEASMAN. T-Cell receptor Vbeta repertoire CDR3 length diversity differs within CD45RA and CD45RO T-cell subsets in healthy and human immunodeficiency virus-infected children. Clinical and diagnostic laboratory immunology, 7(6):953-9, November 2000.
- [474] SANG KYUNG LEE, ZHAN XU, JUDY LIEBERMAN, AND PREMLATA SHANKAR. The functional CD8 T cell response to HIV becomes type-specific in progressive disease. The Journal of Clinical Investigation, 110(9):1339–1347, 2002.
- [475] MATHIAS LICHTERFELD, KATIE L WILLIAMS, STANLEY K MUI, SHIVANI S SHAH, BIANCA R MOTHE, ALESSANDRO SETTE, ARTHUR KIM, MARY N JOHNSTON, NICOLE BURGETT, NICOLE FRAHM, DANIEL COHEN, CHRISTIAN BRANDER, ERIC S ROSEN-BERG, BRUCE D WALKER, MARCUS ALTFELD, AND XU G YU. T cell receptor cross-recognition of an HIV-1 CD8+ T cell epitope presented by closely related alleles from the HLA-A3 superfamily. International immunology, 18(7):1179-88, July 2006.
- [476] DIRK MEYER-OLSON, KRISTEN W BRADY, MELISSA T BART-MAN, KRISTIN M O'SULLIVAN, BRENNA C SIMONS, JOSEPH A CONRAD, COLEY B DUNCAN, SHELLY LOREY, ATIF SIDDIQUE, RIKA DRAENERT, MARYLYN ADDO, MARCUS ALTFELD, ERIC ROSENBERG, TODD M ALLEN, BRUCE D WALKER, AND SPY-ROS A KALAMS. Fluctuations of functionally distinct CD8+ T-cell clonotypes demonstrate flexibility of the HIV-specific TCR repertoire. Blood, 107(6):2373-83, March 2006.
- [477] MATHIAS LICHTERFELD, XU G YU, STANLEY K MUI, KATIE L WILLIAMS, ALICJA TROCHA, MARK A BROCKMAN, RACHEL L ALLGAIER, MICHAEL T WARING, TOMOHIKO KOIBUCHI, MARY N JOHNSTON, DANIEL COHEN, TODD M ALLEN, ERIC S ROSEN-BERG, BRUCE D WALKER, AND MARCUS ALTFELD. Selective depletion of high-avidity human immunodeficiency virus type 1 (HIV-1)-specific CD8+ T cells after early HIV-1 infection. Journal of virology, 81(8):4199-214, April 2007.
- [478] XU G YU, MATHIAS LICHTERFELD, KATIE L WILLIAMS, JAVIER MARTINEZ-PICADO, AND BRUCE D WALKER. Random Tcell receptor recruitment in human immunodeficiency virus type 1 (HIV-1)-specific CD8+ T cells from genetically identical twins infected with the same HIV-1 strain. Journal of virology, 81(22):12666-9, November 2007.
- [479] BRENNA C. SIMONS, SCOTT E. VANCOMPERNOLLE, RITA M. SMITH, JIE WEI, LOUISE BARNETT, SHELLY L. LOREY, DIRK MEYER-OLSON, AND SPYROS A. KALAMS. Despite Biased TRBV Gene Usage against a Dominant HLA Recognition of Circulating Epitope Variants. The Journal of Immunology, 181:5137–5146, 2008.
- [480] MARIA CANDELA IGLESIAS, JORGE R ALMEIDA, SOLÈNE FAS-TENACKELS, DAVID J VAN BOCKEL, MASAO HASHIMOTO, VANESSA VENTURI, EMMA GOSTICK, ALEJANDRA URRUTIA, LINDA WOOLDRIDGE, MATHEW CLEMENT, STÉPHANIE GRAS, PASCAL G WILMANN, BRIGITTE AUTRAN, ARNAUD MORIS, JAME

Rossjohn, Miles P Davenport, Masafumi Takiguchi, Christian Brander, Daniel C Douek, Anthony D Kelleher, David a Price, and Victor Appay. Escape from highly effective public CD8+ T-cell clonotypes by HIV. Blood, 118(8):2138–49, August 2011.

- [481] CRYSTAL Y CHEN, DAN HUANG, SHUYU YAO, LISA HALLIDAY, GUCHENG ZENG, RICHARD C WANG, AND ZHENG W CHEN. IL-2 simultaneously expands Foxp3+ T regulatory and T effector cells and confers resistance to severe tuberculosis (TB): implicative Treg-T effector cooperation in immunity to TB. Journal of immunology (Baltimore, Md. : 1950), 188(9):4278-88, May 2012.
- [482] JOSEPH A. CONRAD, RAMESH K. RAMALINGAM, RITA M. SMITH, LOUISE BARNETT, SHELLY L. LOREY, JIE WEI, BRENNA C. SIMONS, SHANMUGALAKSHMI SADAGOPAL, DIRK MEYER-OLSON, AND SPYROS A. KALAMS. Dominant clonotypes within HIV-specific T cell responses are PD-1hi and CD127low reactivity and display reduced variant cross-reactivity. Journal of Immunology, 186(12):6871-6885, 2012.
- [483] G. B. STEWART-JONES, P. SIMPSON, P. ANTON VAN DER MERWE, P. EASTERBROOK, A. J. MCMICHAEL, S. L. ROWLAND-JONES, E. Y. JONES, AND G. M. GILLESPIE. Structural features underlying T-cell receptor sensitivity to concealed MHC class I micropolymorphisms. Proceedings of the National Academy of Sciences of the United States of America, pages 1–10, November 2012.
- [484] I. V. ZVYAGIN, M. V. POGORELYY, M. E. IVANOVA, E. A. KOMECH, M. SHUGAY, D. A. BOLOTIN, A. A. SHELENKOV, A. A. KURNOSOV, D. B. STAROVEROV, D. M. CHUDAKOV, Y. B. LEBEDEV, AND I. Z. MAMEDOV. Distinctive properties of identical twins' TCR repertoires revealed by high-throughput sequencing. Proceedings of the National Academy of Sciences, 111(16):5980-5985, April 2014.
- [485] JAMES M. HEATHER. Known specificity human TCR CDR3 sequences. http://dx.doi.org/10.6084/m9. figshare.1153817, 2014.
- [486] JIAN HAN, DAVID C SWAN, SHARON J SMITH, SHANJUAN H LUM, SUSAN E SEFERS, ELIZABETH R UNGER, AND YI-WEI TANG. Simultaneous Amplification and Identification of 25 Human Papillomavirus Types with Templex Technology Simultaneous Amplification and Identification of 25 Human Papillomavirus Types with Templex Technology. Journal of clinical microbiology, 44(11):4157-62, November 2006.
- [487] JIAN HAN. Molecular differential diagnoses of infectious diseases: is the future now? In YIWEI TANG AND CHARLES W STRATTON, editors, Advanced Technologies in Diagnostic Microbiology, chapter Molecular. Springer Publishing Company, New York, 2006.
- [488] J HAN. Amplicon rescue multiplex polymerase chain reaction for amplification of multiple targets (Patent US7999092). http://www.google.com/ patents/US7999092, 2011.